

Gyimah, Daniel (2016) *Empirical studies on share issuance and repurchase decisions*. PhD thesis.

http://theses.gla.ac.uk/7483/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given



Gyimah, Daniel (2016) Empirical studies on share issuance and repurchase decisions. PhD thesis.

EMPIRICAL STUDIES ON SHARE ISSUANCE AND REPURCHASE DECISIONS

Daniel Gyimah B. Com, MSc

Submitted in Fulfilment of the Requirements for the Degree of Doctor of Philosophy in Accounting and Finance

Adam Smith Business School College of Social Sciences University of Glasgow

July 2016

ABSTRACT

Financial constraints influence corporate policies of firms, including both investment decisions and external financing policies. The relevance of this phenomenon has become more pronounced during and after the recent financial crisis in 2007/2008. In addition to raising costs of external financing, the effects of financial crisis limited the availability of external financing which had implications for employment, investment, sale of assets, and tech spending. This thesis provides a comprehensive analysis of the effects of financial constraints on share issuance and repurchases decisions. Financial constraints comprise both internal constraints reflecting the demand for external financing and external financial constraints that relate to the supply of external financing. The study also examines both operating performance and stock market reactions associated with equity issuance methods.

The first empirical chapter explores the simultaneous effects of financial constraints and market timing on share issuance decisions. Internal financing constraints limit firms' ability to issue overvalued equity. On the other hand, financial crisis and low market liquidity (external financial constraints) restrict availability of equity financing and consequently increase the costs of external financing. Therefore, the study explores the extent to which internal and external financing constraints limit market timing of equity issues. This study finds that financial constraints play a significant role in whether firms time their equity issues when the shares are overvalued. The conclusion is that financially constrained firms issue overvalued equity when the external equity market or the general economic conditions are favourable. During recessionary periods, costs of external finance increase such that financially constrained firms are less likely to issue overvalued equity. Only unconstrained firms are more likely to issue overvalued equity even during crisis. Similarly, small firms that need cash flows to finance growth projects are less likely to access external equity financing during period of significant economic recessions. Moreover, constrained firms have low average stock returns compared to unconstrained firms, especially when they issue overvalued equity.

The second chapter examines the operating performance and stock returns associated with equity issuance methods. Firms in the UK can issue equity through rights issues, open offers, and private placement. This study argues that alternative equity issuance methods are associated with a different level of operating performance and long-term stock returns. Firms using private placement are associated with poor operating performance. However, rights issues are found empirically to be associated with higher operating performance and less negative long-term stock returns after issuance in comparison to counterpart firms that issue private placements and open offers. Thus, rights issuing firms perform better than open offers and private placement because the favourable operating performance at the time of issuance generates subsequent positive long-run stock price response. Right issuing firms are of better quality and outperform firms that adopt open offers and private placement.

In the third empirical chapter, the study explores the levered share repurchase of internally financially unconstrained firms. Unconstrained firms are expected to repurchase their shares using internal funds rather than through external borrowings. However, evidence shows that levered share repurchases are common among unconstrained firms. These firms display this repurchase behaviour when they have bond ratings or investment grade ratings that allow them to obtain cheap external debt financing. It is found that internally financially unconstrained firms borrow to finance their share repurchase when they invest more. Levered repurchase firms are associated with less positive abnormal returns than unlevered repurchase firms. For the levered repurchase sample, high investing firms are associated with more positive long-run abnormal stock returns than low investing firms. It appears the market underreact to the levered repurchase in the short-run regardless of the level of investments. These findings indicate that market reactions reflect both undervaluation and signaling hypotheses of positive information associated with share repurchase. As the firms undertake capital investments, they generate future cash flows, limit the effects of leverage on financial distress and ultimately reduce the risk of the equity capital.

TABLE OF CONTENTS

ABSTRACT	<u> </u>
TABLE OF CONTENTS	III
LIST OF TABLES	IX
ACKNOWLEDGEMENTS	XII
AUTHOR'S DECLARATION	XIII
CHAPTER 1	1
INTRODUCTION	1
1.1. Introduction	1
1.2. BACKGROUND AND MOTIVATION	2
1.3. THE CHOICE OF DATASETS	8
1.4. RESEARCH QUESTIONS AND KEY FINDINGS	10
1.5. CONTRIBUTIONS OF THE STUDY	14
1.6. ORGANISATION OF THE STUDY	18
CHAPTER 2	21
LITERATURE REVIEW: SHARE ISSUANCE DECISIONS	21
2.1. Introduction	21
2.2. REVIEW OF THEORIES OF SHARE ISSUANCE	21
2.2.1. Pecking Order and Asymmetric Information Hypotheses	22
2.2.2. Market Timing	25
2.2.3. Trade off Theory	27
2.2.4. Agency Theory	28
2.2.5. Managerial Ownership	29
2.2.6. Managerial Entrenchment	30
2.2.7. Earnings Management	30
2.3. MISPRICING AND THE EFFECTS OF INTERNAL FINANCIAL CONSTRAINTS	32
2.3.1. Valuation and Financing Decisions	32
2.3.2. Internal Financial Constraints and Corporate Financial Policy	34

2.3.3. Share Issuance and Internal Financial Constraints	37
2.3.4. Share Issuance and External Financial Constraints	39
2.4. SHARE ISSUANCE AND STOCK RETURNS	41
2.4.1. Underreaction Hypothesis	42
2.4.2. Rational Discount Rate	43
2.4.3. Investment-Based Explanation	43
2.4.4. Economic Perspective	44
2.5. TESTABLE HYPOTHESES	45
2.5.1. Share Issuance and Internal Financial Constraints	45
2.5.2. Share Issuance and External Financial Constraints	47
2.5.3. Share Issuance and Stock Returns	49
2.6. Conclusion	50
CHAPTER 3	52
LITERATURE REVIEW: SHARE ISSUANCE METHODS	52
3.1. Introduction	52
3.2. OVERVIEW OF SHARE ISSUANCE METHODS	52
3.2.1. Flotation Methods	53
3.2.2. UK issues	54
3.2.3. Costs of Issue Approach	54
3.2.4. Demise of Rights Issues	55
3.3. THEORIES OF SHARE ISSUANCE CHOICE	57
3.3.1. Quality Signalling	57
3.3.2. Monitoring and Certification Hypothesis	58
3.3.3. Managerial Entrenchment	59
3.3.4. Private Placement and Investor Relationship	61
3.4. STOCK PRICE REACTIONS TO SHARE ISSUANCE CHOICE	62
3.4.1. Initial Abnormal Stock Returns	62
3.4.2. Long Run Stock Returns	64
3.4.3. Operating Performance	69
3.5. TESTABLE HYPOTHESES	72
3.5.1. Equity Issues and Operating Performance	72
3.5.2. Long-run Stock Returns and Equity Issuance Methods	74

3.6. CONCLUSION 74

CHAPTER 4	<u>76</u>
LITERATURE REVIEW: SHARE REPURCHASE PROGRAMS	76
4.1. Introduction	76
4.2. OVERVIEW OF SHARE REPURCHASE	76
4.2.1. Methods of Share Repurchase	77
4.2.2. US Share Repurchases Regulations	79
4.3. THEORIES OF SHARE REPURCHASE	81
4.3.1. Undervaluation Hypothesis	81
4.3.2. Information Signalling Theory	83
4.3.3. Free Cash Flow Hypothesis	86
4.3.4. Market Timing	88
4.4. Empirical Evidence of Share Repurchases	89
4.4.1. Levered Share Repurchase	89
4.4.2. Share Repurchase and Investment Expenditure	91
4.4.3. Share Repurchase and Stock Returns	93
4.5. TESTABLE HYPOTHESES	95
4.5.1. Internal Financial Constraints and Levered Share Repurchase	95
4.5.2. Internal Financial Constraints, Levered Share Repurchase and Sto	ck
Returns	97
4.6. CONCLUSION	98
CHAPTER 5	100
DATA AND METHODOLOGY	100
5.1. Introduction	100
5.2. Data	100
5.2.1. Sample Period	100
5.2.2. Data for Equity Issues	102
5.2.4. Selection of matching firms/control firms	104
5.2.3. Data for Share Issuance Methods	105
5.2.4. Data Share Repurchase	107
.3. VARIABLE DEFINITIONS AND MODEL SPECIFICATION	109

5.3.1. The Effects of Mispricing and Financial Constraints on Equity Issua	nce
Decisions	110
5.3.1.1. Dependent Variables	110
5.3.1.2. Explanatory Variables	112
5.3.1.3. Control Variables	116
5.3.2. Share Issuance Methods: Operating Performance and Stock Returns	123
5.3.2.1. Dependent Variable:	123
5.3.2.2. Explanatory Variable	127
5.3.3. Share Repurchase and Internal Financial Constraints	133
5.3.3.1. Dependent Variable	133
5.3.3.2. Explanatory Variables	136
5.3.3.3. Control Variables	138
5.3.3.4. Model Specification	140
CHAPTER 6	144
SHARE ISSUANCE AND FINANCIAL CONSTRAINTS	144
6.1. Introduction	144
6.2. SAMPLE STATISTICS	146
6.2.1. Descriptive Statistics of Variables	146
6.2.1.1. Descriptive Statistics of Sample	146
6.2.1.2. Yearly Average of Equity Issues	149
6.2.1.3. Pearson's Correlation Matrix of Variables	150
6.3. TESTING THE EFFECTS OF FINANCIAL CONSTRAINTS ON SHARE ISSUANCE	Έ
	152
6.3.1. Internal Financial Constraints and Share Issuance	152
6.3.2. The Effects of External Financial Constraints	155
6.3.3. Share Issuance, Financial Constraints and Stock Returns	160
6.3.3.1. Univariate Results for Stock Returns	161
6.3.3.2. Cross-Sectional Regression Results for Stock Returns	164
6.4. ROBUSTNESS CHECKS AND ADDITIONAL RESULTS	166
6.4.3. Tobit Regression Estimation of Equity Issues	169
6.4.4. Random and Fixed Effects Logit Regression Estimation	170
6.5. Conclusion	174

CHAPTER 7	<u>176</u>
SHARE ISSUANCE METHODS AND OPERATING PERFORMANCE	176
7.1. Introduction	176
7.2. SAMPLE STATISTICS AND DIAGNOSTICS	179
7.2.1. Descriptive Statistics of Independent and Control Variables	179
7.2.1.1. Equity Issues by Issuance Method during the Sample Period	179
7.2.1.2. Descriptive Statistics of Variables	180
7.2.1.3. Pearson's Correlation Matrix of Equity Issue Characteristics	183
7.3. TESTING THE EFFECTS OF OPERATING PERFORMANCE ON THE CHOICE OF	F
EQUITY ISSUANCE METHODS	187
7.3.1. Multinomial Logit Regression Analysis of Equity Issuance Methods	187
7.3.2. Long-term Stock returns	192
7.3.3. Long-term Operating Performance	197
7.4. ROBUSTNESS AND SENSITIVITY ANALYSIS	198
7.4.1. Logit Regression Results for Abnormal Returns on Assets (aROA) a	nd
Returns on Equity (ROE)	198
7.4.2. Multinomial Logit Regression Results for Multiple Equity Issuance	
Method during the Period	203
7.4.3. Tobit Regression Estimation of Main Results	206
7.5. CONCLUSION	208
CHAPTER 8	210
LEVERED SHARE REPURCHASE AND INTERNAL FINANCIAL	
CONSTRAINTS	210
8.1. Introduction	210
8.2. SAMPLE STATISTICS	212
8.2.1. Descriptive Statistics of Independent and Control Variables	212
8.2.1.1. Share Repurchase during the Sample Period	212
8.2.1.2. Descriptive Statistics of Variables	214
8.2.1.3. Pearson's Correlation Matrix of Variables	217
8.3. TESTING THE EFFECTS OF INTERNAL FINANCIAL CONSTRAINTS ON LEVE	RED
Share Repurchase	219

REFERENCES	263
APPRENDICES	261
9.5. Suggestions for Future Research	258
9.4. LIMITATIONS OF THE STUDY	257
9.3.2. Implications for Practitioners and Policymakers	255
9.3.1. Implications for Academics	254
9.3. RESEARCH IMPLICATIONS AND RECOMMENDATIONS	254
9.2.3. Share Repurchase and Internal Financial Constraints	251
9.2.2. Share Issuance Methods and Operating Performance	249
9.2.1. Share Issuance and Financial Constraints	247
9.2. Main Findings of the Thesis	247
9.1. Introduction	247
CONCLUSION	247
CHAPTER 9	247
8.5. CONCLUSION	245
	243
8.4.6. Random and Fixed Effects Logit Regression Estimation of Main Re	esults
8.4.5. Tobit Regression Estimation of Main Results	241
8.4.4. Levered Share Repurchase and Financial Distress	239
8.4.3. Levered Share Repurchase and Bond Ratings	236237
8.4.2. Levered Share Repurchase and WW index Internal Financial Const	
8.4.1. Unconstrained firms defined using alternative cut-off point of KZ	234
8.4. ROBUSTNESS AND SENSITIVITY ANALYSIS	234
8.3.2.2. Share Repurchase and Long Run Abnormal Returns	231
8.3.2.1. Share Repurchase and Announcement Date Abnormal Returns	227
8.3.2. Share Repurchase and Stock Returns	226
Unconstrained Firms	219
8.3.1. Logistic Regression Results for Levered Share Repurchase of	

LIST OF TABLES

Table 3.1: Summary of Empirical Evidence of US Long Run Abnormal	
Returns Post Equity Issues	67
Table 3.2: Summary of Empirical Evidence of UK Long Run Abnormal	
Returns Post Equity Issues	68
Table 4.1: Summary of Empirical Evidence of US Announcement Date	
Abnormal Returns following Share Repurchase	95
Table 4.2: Summary of Empirical Evidence of Long Run Abnormal Return	ns
following Share Repurchases	95
Table 5.1: Annual Equit Issues	104
Table 5.2: Construction of Equity Issues by Method and the Distribution a	cross
the Sample Period	106
Table 5.3: Construction of Final Sample of Share Repurchase Announcement	ents
	108
Table 5.4: Number of Annual Share Repurchase Announcements	109
Table 6.1: Comparison of Issuing and Non-issuing Firms	147
Table 6.2: Yearly Average Amounts and Ratio of Equity Issues	149
Table 6.3: Pearson's Correlation Matrix of Variables	151
Table 6.4: Logit Regression Estimates of Internal Financial Constraints (K	\mathbf{Z}
index) and Share Issuance	154
Table 6.5: Logit Regression Estimates (Odds ratio) of Share Issuance	155
Table 6.6: The Effects of External Financial Constraints (Financial Crisis)	on
Share Issuance	157
Table 6.7: The Effects of External Financial Constraints (Market Liquidity	y) on
Share Issuance	159
Table 6.8: Univariate Results of Abnormal Stock Returns	163
Table 6.9: Cross-sectional Regression Analysis of Share Issuance and Sho	rt
Run Stock Returns	165
Table 6.10: Logit Regression Estimates of Internal Financial Constraints (WW
index) and Share Issuance	167
Table 6.11: The Effects of Financial Distress on Share Issuance	169
Table 6.12: Tobit Regression Analysis of Internal Financial Constraints ar	ıd
Share Issuance	170

Table 6.13: Random and Fixed Effects Logit Regression Estimation (Finar	iciai
Constraints and Share Issuance)	172
Table 6.14: Random and Fixed Effects Logit Regression Estimation Finance	cial
Constraints and Share Issuance (Crisis and Non-crisis)	173
Table 7.1: Yearly Average Amounts and Ratio of Equity Issues	180
Table 7.2: Summary Statistics of Firm Characteristics	182
Table 7.3: Pearson's Correlation Matrix of Equity Issue Characteristics (Ri	ights
Issues)	184
Table 7.4: Pearson's Correlation Matrix of Equity Issue Characteristics (O	pen
Offers)	185
Table 7.5: Pearson's Correlation Matrix of Equity Issue Characteristics (Pr	ivate
Placement)	186
Table 7.6: Multinomial Logit Regression Estimates of Choice of Equity	
Issuance Method	189
Table 7.7: Long Run Buy and Hold Abnormal Returns and Equity Issuance	3
Method	193
Table 7.8: Long Run Stock Returns and Operating Performance (Positive a	ınd
Negative)	194
Table 7.9: Long Run Stock Returns and Operating Performance (High and	
Low)	195
Table 7.10: Cross-sectional Regression Analysis of Long Run Stock Return	ns
	196
Table 7.11: Post-issue Changes in Operating Performance	197
Table 7.12: Multinomial Logit Regression Estimates of Equity Issuance	
Method (abnormal ROA)	199
Table 7.13: Multinomial Logit Regression Estimates of Equity Issuance	
Method (ROE)	201
Table 7.14: Multinomial Logit Regression Estimates of Changes in Firm's	
Issuance Method During the Sample Period	204
Table 7.15: Tobit Regression Analysis of Equity Issuance Method	207
Table 8.1: Yearly Average Proceeds and Ratio of Share Repurchase Proceeds	eds
	213

Table 8.2: Summary Statistics: Comparison of Unconstrained and Constrained	ned
Firms	215
Table 8.3: Summary Statistics: Comparison of Unlevered and Levered	
Repurchase	216
Table 8.4: Pearson's Correlation Matrix of Variables	218
Table 8.5: Logit Regression Estimates (Raw Coefficient Estimates) of Lev	ered
Repurchase and Internal Financial Constraints	221
Table 8.6: Logit Regression Estimates (Raw Coefficient Estimates) of Lev	ered
Repurchase and Internal Financial Constraints	222
Table 8.7: Logit Regression Estimates (Odds Ratio) of Levered Repurchas	e
and Internal Financial Constraints	223
Table 8.8: Logit Regression Estimates (Odds Ratio) of Levered Repurchas	e
and Internal Financial Constraints	224
Table 8.9: Univariate Analysis of 3-day Announcement Date Abnormal	
Returns $(CAR_{(-I,I)})$ following Share Repurchase	228
Table 8.10: Cross-sectional Regression Analysis of Announcement Date	
Abnormal Returns	230
Table 8.11: Univariate Analysis of 36-month Buy and Hold Abnormal Ret	urns
(BHAR _{36-month}) following Share Repurchase	232
Table 8.12: Cross-sectional Regression Analysis of Long Run Buy and Ho	ld
Abnormal Returns	233
Table 8.13: Logit Regression Estimates of Main Results (alternative cut-of	f
points)	235
Table 8.14: Logit Regression Estimates of Levered Repurchase and International	ıl
Financial Constraints (WW index)	236
Table 8.15: Logit Regression Estimates of Levered Repurchase and Bond	
Ratings	238
Table 8.16: Logit Regression Estimates of Levered Repurchase and Finance	ial
Distress	240
Table 8.17: Tobit Regression Analysis of Levered Share Repurchase	242
Table 8.18: Random and Fixed Effects Logit Regression of Levered	
Repurchase and Internal Financial Constraints	244

ACKNOWLEDGEMENTS

I wish to extend my gratitude to persons who assisted me in diverse ways, either consciously or unconsciously, towards the completion of this thesis. Foremost, I thank God for His immense wisdom and guidance throughout the period of this work. My supervisors, Prof. Chris Veld and Dr. Antonios Siganos, deserve special mention for their patience, encouragement and tremendous assistance that culminated into the successful completion of this work. I especially thank them for the amount of time they spent in reading the various drafts of the thesis and for the comments and suggestions they offered to improve the standard of this work. I am also grateful to have Prof. Seth Armitage and Dr. Betty Wu as my PhD examiners and for their useful comments.

I also wish to recognize the support of the entire faculty at the University of Glasgow for their critiques and valuable comments during the annual doctoral review. I enjoyed tremendous warmth during the course of this doctoral studies and I owe to all the doctoral students I have come across over the years. The spiritual and emotional support I received from my church, The Church of Pentecost-Glasgow, cannot be ignored and I say God richly bless all the brethren.

Finally, I thank my family for their constant love and affection that has seen me thus far in my academic pursuit.

AUTHOR'S DECLARATION

"I declare that, except where explicit reference is made to the contribution of others, that this thesis is the result of my own work and has not been submitted for any other degree at the University of Glasgow or any other institution.
Signature
Printed name: Daniel Gyimah"

Chapter 1 INTRODUCTION

1.1. Introduction

Corporate financing policy is an integral part of the operations of companies. Its importance underpins several decisions and affects both investing and financing decisions of the firm. Policies relating to investments in capital projects including research and development, cash distribution to shareholders in the form of dividend payments, and share repurchase decisions among others, have overarching effects on both the short term and long-term prospects of the company. On the basis of these decisions, investors determine whether or not to provide money to the company. The recent financial crisis of 2007/2008 highlighted the significant effects of financial constraints on corporate financial decisions including employment, tech spending, and capital spending (Campello et al. 2010). This thesis explores the effects of financial constraints on share issuance decisions and share repurchase programmes of firms. Specifically, the thesis explores the choice between share issuance and non-issuance when firms are financially constrained. Further analysis of share issuance examines the differential long-term stock returns associated with share issuance methods.

The first empirical chapter explores the effects of financial constraints (both internal and external) on share issuance decisions. Specifically, this chapter examines the simultaneous effects of mispricing and financial constraints on share issuance. Thus, it describes whether firms time their share issues even when they face extreme internal and external financial constraints. Further, it focuses on the type of firms, in terms of size and the time period when market timing of share issues is more plausible to limit the adverse stock price reactions. In the second empirical chapter, the thesis investigates whether the different long-term stock returns associated with equity issuance methods is driven by the level of firm operating performance. The literature on equity issuance methods asserts that stock returns differ according to the issuance method used. This chapter uses long-term operating performance to explain the

choice of equity issuance methods and the associated long-term stock returns. The third empirical chapter explores the motivation for unconstrained firms to borrow to repurchase their shares.

The aim of this chapter is to introduce the thesis. Section 1.2 presents the background and motivation for the study. Section 1.3 discusses the study dataset and the reasons for the choice of both UK and US sample firms. Section 1.4 sets out the research questions and provides the key findings of the thesis. Section 1.5 discusses the contribution of the study. Finally, section 1.6 provides the organisation of the thesis and briefly presents the content within each chapter.

1.2. Background and Motivation

There is an extant literature on securities issuance and repurchase and the effects on other corporate decisions have been extensively explored (for example Baker and Wurgler, 2002; Campello et al, 2010; Dong et al, 2012). In particular, corporate share issues are related to investments where firms with significant growth opportunities need external financing to undertake them. Moreover, share issues and repurchase alter the capital structure of the firm. The effects of corporate financing on capital structure began with the seminal paper by Modigliani and Miller (1958) where they argued, among others, that whether or not firm is made up of equity or debt does not matter, under the assumptions of perfect information, homogenous firms, no discriminating taxes and transaction costs of financing. However, several subsequent studies have challenged these assumptions and proved that in fact the source of financing has significant effects on capital structure, including the value of the firm and the costs of financing. Again, the effects on the firm of share issues and repurchase are reflected in the stock prices of the issuing or repurchasing firms. These stock price reactions are consistent across several jurisdictions and have informed corporate financing and investing policies.

Several theories have been postulated to explain equity issues and the stock price effects including information asymmetry (Akerlof, 1970; Myers and Majluf, 1984; Bayless and Chaplinsky, 1991), pecking order (Helwege and

Liang, 1996; Frank and Goyal, 2003), market timing (Baker and Wurgler, 2002; DeAngelo, DeAngelo, and Stulz, 2010; Dong, Loncarski, ter Horst, and Veld, 2012), agency (Jensen and Meckling, 1976; Jensen, 1986), and static trade-off (Shyam-Sunder and Myers, 1999). Equity issues are generally associated with negative abnormal stock returns that has commonly attributed to market underreaction to the information (Spiess and Affleck-Graves, 1995; Loughran and Ritter, 1997). Firms also repurchase their shares in response to undervaluation (Ikenberry and Vermaelen, 1996; Stephens and Weisbach, 1998), information signalling effects (Vermaelen, 1981; Comment and Jarrell, 1991), and free cash flow (Jensen, 1986; Dittmar, 2000; Grullon and Michaely, 2004), among others. These motivations are consistent with divulging positive information about the prospects of the firms and hence stock returns associated with share repurchase announcements are positive.

The first empirical chapter explores the effects of financial constraints on share issuance decisions. Since the work by Modigliani and Miller, corporate financing decisions have evolved over the years, even though the theoretical explanations for stock returns still remain consistent. An important factor related to equity issues is transactions costs in the form of investment banking fees that depends on both the characteristics of the issuing firms and the overall capital market conditions. For example, Butler et al. (2005) find that firms reduce the costs of equity financing by improving the market liquidity of their stock. In other words, more liquid stocks are associated with lower investment banking fees. This cost of raising equity is important because it impacts market timing tendencies. Stated differently, internal (related to available cash and the friction to obtain external cash) and external financial constraints (mostly related to macroeconomic conditions and the overall capital market conditions) play a vital role in equity issuance decisions. Hence, this study explores the influence of financial constraints on share issuance decisions. Financial constraints involve both internal cash availability and the speed at which firms obtain equity financing in the equity capital market. It also captures market liquidity and the effects of financial crisis at the time of issuing equity to highlight the external financial constraints associated with equity issues.

This chapter, therefore, extends the literature on market timing by examining the effects of financial constraints on share issuance decisions. If firms time their equity issues and also incur costs of issuing equity, then I argue that internally financially unconstrained firms should be more likely to issue overvalued equity than constrained firms. Less internally financially constrained firms are more likely to time their equity issues (Dong et al, 2012) to coincide with periods of favourable macroeconomic conditions (Korajczyk and Levy, 2003). Taken together, firms that are internally financially constrained face high finance costs when there are simultaneously unfavourable market conditions. During financial crisis, with low market liquidity, issuance costs increase. Therefore, consistent with the financial constraints hypothesis, prospects of equity market timing by issuing firms are limited. These firms generate more free cash flow that would allow them to take advantage of issuing overpriced equity.

The second empirical chapter examines the different method of equity issuance in the UK. Firms in the UK issue equity through three dominant methods namely rights issues, open offers, and private placement. Since the deregulation in 1986, the preference for rights issue method has reduced and the recent evidence shows that firms use more private placements than rights issues (Armitage, 2007). The issuance methods are also associated with different stock price returns (Slovin et al. 2000; Barnes and Walker, 2006). Therefore, theories such as the information asymmetry (Myers and Majluf, 1984), window of opportunity hypothesis (Choe et al. 1993; Bayless and Chaplinsky, 1996), investor over-optimism (Loughran and Ritter, 1997; Hertzel et al. 2002) explain the choice of issuance method and the stock price reactions. For example, Hertzel et al. (2002) find that public firms that place their shares privately are associated with positive announcement returns suggesting that investors are over-optimistic about the future prospects of the issuing firms. Thus, firms deliberately choose the method based on the firm characteristics and the market responds according to the information that the issue provides to the market.

Furthermore, the research also explains the long-term stock returns associated with equity issuance methods. Prior studies attribute the different stock returns associated with each equity issuance method on the basis of both firm and issues characteristics that relate to market-to-book ratio, prior stock returns, firm size, and issue discount. Overall, these characteristics measure the degree of information asymmetry associated with the issuance method. For example, since rights issues are made to existing shareholders, it is less likely that firm will exploit any asymmetric information effects. Thus, issues involving external investors, such as private placement, allow the firm to exploit asymmetric information effects. This study asserts that operating performance determined by firms' operations have more long-term effects on long-term stock returns than alternative measures of information asymmetry. Therefore, the research explores the effects of long-term operating performance on the choice of issuance method and the subsequent long-term stock returns. Unlike Levis (2005), Abhyankar and Ho (2007), Capstaff and Fletcher (2011), who provide evidence about long term stock returns, this study focuses on how operating performance determines the choice of equity issuance method and explain the long term stock returns associated with equity issuance methods.

The third empirical chapter focuses on share repurchases policies of firms. Share repurchase programs form an integral part of firm payout policy. US firms continue to repurchase their shares and actually repurchase transactions have increased beyond the pre-crisis levels in 2007. According to an article in The Economist, companies in the S&P 500 index repurchased a record \$500 billion in 2013 financed by 33 cents of every dollar of cash flow. Prior research and empirical evidence is consistent with firm cash distribution in the form of share repurchase or in excess of dividend payments to avoid investment in value-destroying projects (Jensen, 1986; Bhargava, 2010). It follows that share repurchase are more likely to be conducted by firms with substantial amount of free cash and which have limited growth opportunities. Moreover, the undervaluation hypothesis contends that firms repurchase in order to buy their own cheap shares and also signal to the market about the favourable future prospects (Ikenberry and Vermaelen, 1996; Stephens and

⁻

¹ Share buy-backs: The repurchase revolution, The Economist, September 13, 2014.

Weisbach, 1998; Dittmar, 2000). In sum, firms that are less internally financially constrained and have cash slack will repurchase their shares. Chen and Wang (2012) find evidence that financially constrained firms repurchase their shares due to managerial hubris. Other studies find that firms borrow at the time of share repurchase announcements (Minnick and Zhao, 2007; de Jong, Dutordoir, and Verwijmeren, 2011).

Recent corporate financing policies reveal that it has become a commonplace for firms to simultaneously issue new convertible bond and repurchase their shares (see de Jong et al., 2011). These firms utilise the cash from debt issuance to finance their share repurchase transactions. During the last decades, share repurchases have become very popular as a means to return cash to shareholders. In 2007, share repurchases peaked at more than \$700 billion near the market top.² A large number of these repurchases were funded with debt issues. In 2009 alone, 37 companies announced plans to spend \$39 billion on these levered repurchases.³ It may be understandable for internally financially constrained firms to repurchase their shares by borrowing due to limited financial resources available, but it seems puzzling for internally financially unconstrained firms to borrow money in order to repurchase shares. A recent example is Apple that announced in April 2013 that they would return \$100 billion to shareholders in the form of a repurchase by the end of 2013. Apple borrowed part of the money for this repurchase, despite having a huge cash stockpile.⁴ This repurchase trend evokes the following question: why would internally financially unconstrained firms borrow to finance share repurchase? This debt-financed share repurchase could be determined by the extent of investment undertaken by these firms. The study further explores the stock price implications for these debt-financed repurchase firms.

²See Michael Milken: "Why capital structure matters", The Wall Street Journal, April 21,

³Source: research by JP Morgan cited by Herb Greenbert in: "Debt to buy back stock", CNBC, November 8, 2011. See: http://www.cnbc.com/id/45209702 (retrieved on March 30, 2015).

⁴See Katy Burne and Mike Cherney: "Apple's record plunge into debt pool", The Wall Street Journal, April 30, 2013. Apple's case may be different from other companies, because much of its money is overseas; raising cash in the bond markets helps the company to reduce its large tax bill that would hit the company if it would bring back the cash to the U.S.

Unlike Chen and Wang (2012), this chapter of the thesis examines the share repurchase financing of unconstrained firms. For constrained firms, the incentive of buying back cheap or undervalued shares will motivate them to borrow to repurchase. However, Chen and Wang (2012) explicitly assert that managers of constrained firms repurchase due to overconfidence about the future prospects of the firms and managerial hubris. In respect of debt-financed repurchase, studies, such as Minnick and Zhao (2007) find that firms borrow to finance their share repurchase programs. Due to the effect of share repurchase on distress risk, these debt-financed repurchase firms report less positive announcement date abnormal returns. Bond returns associated with debtfinanced repurchases are also negative or less than an otherwise cash-financed share repurchase. Thus, share repurchase generates wealth transfer from bondholders to stockholders (Maxwell and Stephens, 2003). Also, another wealth transfer occurs between debt-financed share repurchase and cashfinanced share repurchase (Minnick and Zhao, 2007). However, de Jong et al. (2011) do not find that convertible debt issues are necessarily used to finance share repurchase. They contend that convertible arbitrage strategies explain the size and speed of share repurchase transactions. These previous studies highlight the share repurchase programs of constrained firms and the general evidence that firms systematically issue new debt during share repurchase announcements. Therefore, the present research revisits this phenomenon and examines the share repurchase financing of internally financially unconstrained firms.

The third empirical chapter, therefore, investigates the factors that drive unconstrained firms to conduct levered share repurchase. The intuition for this phenomenon, as argued in current study, is that even though unconstrained firms can use internal cash flows to buy-back their shares, these firms can take advantage of low interest rates to raise debt financing and together with the internal cash flows undertake investments when they also repurchase their shares. Internally financially unconstrained firms are more likely to conduct levered share repurchase when they also invest in capital projects. In addition, it examines the initial stock returns subsequent to share repurchase announcements. It also highlights the stock returns for debt-financed and cash-

financed share repurchase of both internally financially unconstrained and constrained firms.

1.3. The choice of datasets

The thesis uses dataset from both UK and US for important reasons. The first empirical chapter focuses on the effects of financial constraints on share issuance of UK listed firms and the second chapter notes the marked stock returns differences associated with different equity issuance methods. There is a limited empirical literature on UK equity market timing and especially the study about the effects of financial constraints on UK equity issues is unexplored. Indeed, most of the internal financial constraints studies use US dataset (For example, Fazzari, Hubbard, and Petersen, 1988; Kaplan and Zingales, 1997; Whited and Wu, 2006; Korajczyk and Levy, 2003). Dong et al. (2012) investigate financial constraints effects on equity issues for Canadian listed firms. Equity mispricing studies that focus on the UK include Marsh (1982), and Mahajan and Tartaroglu (2008). Unlike the previous studies, I explore the simultaneous effects of internal financial constraints and the financial crisis on share issuance decisions of UK firms. Therefore, UK presents an ideal setting to contribute to the general empirical literature on equity issuance decisions and delineates how financial constraints impact such corporate financing policies.

Another important reason to study UK equity issues is the presence of different equity issuance method as explored in the Chapter two of the thesis. UK deregulation of 1986 allowed firms to issue equity through means other than rights offerings. Since the deregulation, several changes have occurred from gradual recession of the rights issues dominance to near demise of rights issues. Armitage (2007) finds that the demise of rights issues is more pronounced that reflects the notion that placing better facilitates block selling than rights issues (Armitage, 1998; 2010). Again, placement saves time and money compared with rights issue and open offers and cost less than rights

⁵ Placing is an equity issuance method by which firms place new share issues by private negotiation or through accelerated bookbuilding in which investing institutions are publicly invited to bid for shares. Extensive discussion of methods of equity issuance is provided in Section 3.1.1 of Chapter 3.

issues (Wu, 2004). Using UK data, the study then captures the periodic changes in the choice of issuance method and the implications for stock return reactions. Stated differently, UK provides an ideal setting to explore how investor, firm, and regulatory changes have altered preference for private placement over rights offerings. The findings shed more light on equity issuance methods and provide new insights into equity issuance decisions.

Note that this unique feature of equity issues is uncommon in international markets. Only one or two dominant issuance approaches exist such as firm commitment in the US (Eckbo and Masulis, 1992)⁶, rights issues in the Netherlands (Kabir and Roosenboom, 2003), rights offering by family controlled businesses in Sweden (Crongvist and Nilsson, 2005), rights offerings and private placement in Australia (Brown and Chan, 2004) and rights issues in most European countries (Tsangarakis, 1996; Bohren, Eckbo, and Michalsen, 1997; Gajewski and Ginglinger, 2002; Pastor-Llorca and Martin-Ugedo, 2004). In an environment where choice of equity issuance prevails, firms are able to tailor their issues to specific motivations. For example, in an attempt to restrict ownership dilution, firms will prefer to issue shares to existing shareholders through rights offerings (Cronqvist and Nilsson, 2005). On the other hand, private placements, while diluting ownership by allowing new investors to subscribe to the shares, encourage monitoring (Wruck, 1989) and promote credible certification of the issue (Hertzel and Smith, 1993).

With regard to the third empirical chapter, the dataset on U.S. share repurchase is used. Despite the extensive research on share repurchase announcements in the US, the evolution of corporate actions and events reflects the changing dynamics in corporate financing policy. During the last few decades, share repurchases have become popular as a means to return cash to shareholders. In 2007 alone, US share repurchases peaked at more than \$700 billion near the market top (Milken, 2009). A large number of these repurchases were funded

-

⁶ Eckbo and Masulis (1995) report that U.S. companies have switched from uninsured rights to standbys and to firm commitment underwriting method. The firm commitment flotation method accounts for 99% of all issues. Under firm commitment, an underwriter commits to buy the entire equity issue and tries to resell it to investors.

with debt issues. In 2009 for instance, 37 US companies announced plans to spend \$39 billion on these levered repurchases. 7 Unlike most European countries, share repurchase programs in the US are authorized by the board of directors and hence do not need shareholder approval. The apparent flexibility associated with share repurchase differs markedly from those of other jurisdictions. Until 1998, share repurchases were illegal in some European countries such as France and Germany (Rau and Vermaelen, 2002; Andriosopoulos and Lasfer, 2015). Also in the UK, share repurchase programs are limited due to regulatory restrictions and undermine companies' ability to take advantage of undervaluation (Rau and Vermaelen, 2002; Oswald and Young, 2004). For example, UK companies are not allowed to buy back their shares in periods when managers are most likely to have superior information about future earnings. Whereas the regulatory limitations provide sufficient grounds to study UK share repurchase, the new trend of debt-financed repurchase motivates the use of US data. The regulatory environment in the US allows firms to exploit undervaluation and pursue other motivations that apparently benefit the firm. The study therefore uses US data to explore the reasons why US firms that are less financially constrained issue debt during share repurchase programs.

1.4. Research questions and key findings

The main research questions of the thesis are as follows:

1. How do internal and external financial constraints affect equity issuance decisions?

The extant literature provides evidence for equity market timing and posits that subsequent stock returns associated with equity issues are significantly negative. This study extends the market timing theory by incorporating the effects of both internal and external financial constraints on equity issuance decisions. The study uses a sample of 1257 equity issues by UK listed firms to test the simultaneous of effects of mispricing and internal financial constraints on share issuance decisions. The total sample spans over a period of 17 years

-

⁷ see footnote 3 in Section 1.2

between 1994 and 2010. The study uses both univariate and multivariate analysis to establish the effects of financial constraints on share issuance decision. It uses logit regression analysis to show the probability of equity issuance based on both internal and external financial constraints. Moreover, it highlights the stock returns associated with equity issues conditional on financial constraints.

Consistent with the literature (Baker and Wurgler, 2002; DeAngelo et al., 2010; Dong et al., 2012), the results provide strong support for the market timing theory, showing that UK firms time their equity issues. The second proposition of the thesis contends that internally financially constrained firms could still issue overvalued equity provided the equity market and/or the general economic conditions are favourable. Similar to related evidence, the findings show that internally financially constrained firms are more likely to issue overvalued equity similar to unconstrained firms during non-crisis period. However, as costs of external financing become high during economic downturn, internally financially constrained firms are less likely to issue equity even if the equity is highly overvalued. The study further finds that during period of financial crisis or when there is low market liquidity, small firms are less likely to issue overvalued equity. It also shows that large firms with limited growth opportunities and substantial amount of free cash flow can afford transaction costs of issuing overvalued equity even during significant financial crisis.

2. Does the long-term operating performance reflect the choice of equity issuance methods?

The deregulation in 1986⁸ introduced alternative issuance methods in the UK rights issues, open offers and private placements – and subsequent equity issues reflect certain firm characteristics. Firms are expected to select issuance method for which they have more prospects for timing their overvalued issues. Rights issuing firms have less incentive to sell equity with poor prospects,

11

-

⁸ In October 1986, the event known as the "Big Bang" took place, marking significant changes to the structure of UK financial markets. Among other consequences, an electronic trading system was introduced.

since existing shareholders sell shares to themselves in comparison to counterpart private placements and open offers. Equity issuance methods account for different short-term stock returns (Slovin et al., 2000; Barnes and Walker, 2006) and long-term stock returns (Ho, 2005; Andrikopoulos, 2009; Capstaff and Fletcher, 2011). However, most of these studies attribute the different stock returns to firm and issue characteristics that have short-term effects on stock returns. This study argues that differential information asymmetry allows firms to use the inside information to select the issuance method. ⁹ Since operating performance may vary after the issuance of alternative methods, the long-term stock returns would vary accordingly. In testing the hypothesis, the research uses UK equity issuance data from 1996 to 2010 to explore stock returns and contemporaneous operating firm performance for each of the subsequent three years after equity issuance.

Consistent with the developed hypothesis, the empirical results show that firms with better operating performance are more likely to conduct rights issues and are associated with the less negative long-term stock returns after rights issues than private placements and open offers. These results hold within a number of robustness tests, indicating that managers select an issuance method in line with their expectations of a firm's future operating performance. Moreover, firms are more likely to select private placement because less favourable contemporaneous operating performance information is not available to the market at the time of the equity issues. Stated differently, firms are less likely to issue overvalued equity to existing shareholders through rights issues. Long-term stock returns associated with equity issuance method are consistent with the long-term operating performance. These results suggest that over the long term, operating performance explains long-term stock returns beyond firm and issue characteristics such as market-to-book ratio, firm size, issue discount, prior stock returns, and institutional share ownership, among others.

3. Why do internally financially unconstrained firms borrow to repurchase their shares?

_

⁹ Cronqvist and Nilsson (2005) theoretically and empirically support the hypothesis that information asymmetry influences managers' decisions of whether to select rights issues or private placements within Swedish firms.

Since the effects of internal financial constraints transcend all corporate decisions, the study also focuses on the share repurchase implications. Specifically, the study investigates the factors that drive unconstrained firms to conduct levered share repurchase. The intuition for this phenomenon is that unconstrained firms, which have sufficient internal cash flows, would not need external financing to repurchase their shares. However, these firms can take advantage of low interest rates to raise debt financing and together with the internal cash flows undertake investments when they also repurchase their shares. Denis and Sibilkov (2010) assert that low cash holdings of internally financially constrained firms does not allow them to build cash reserves when they spend all available cash on investments. The study, therefore, hypothesises that unconstrained firms with bond ratings obtain debt financing at relatively low transaction costs in terms of interest payments. Consequently, internally financially unconstrained firms are more likely to conduct levered share repurchase when they also invest in capital projects.

Share repurchases are defined as levered or unlevered on whether there is an increase in debt capital a year prior to and after the date of repurchase. Unconstrained firms are defined using the *KZ* index. ¹⁰ First analysis estimates logit regressions to investigate the probability of levered share repurchase based on internal financial constraints and the interaction between internal financial constraints and investments. The results are consistent with the hypothesis that internally financially unconstrained firms borrow to finance their share repurchase. But more importantly, they are more likely to conduct levered share repurchase when they also undertake investments. These findings imply that unconstrained firms even though they can repurchase with internal cash flows, the additional cash from the debt issuance allows the firms to undertake more investments. Using bond ratings as a proxy for internal financial constraints, the study finds that levered share repurchase is related to firms with bond ratings and/or investment-grade firms.

¹⁰ Kaplan and Zingales (1997) constructed the KZ index in their study of the financial constraints characteristics of 49 low-dividend paying manufacturing firms. Using observable characteristics, Kaplan Zingales run an ordered logit to rank the firms. A firm with high KZ index is considered more financially constrained.

Additional analysis shows that the initial stock returns associated with share repurchase depend on the level of internal financial constraints and the financing of the share repurchase. Levered repurchase firms are associated with less positive abnormal returns than unlevered repurchase firms. In relation to unconstrained firms, high investing firms are associated with more positive abnormal stock returns than low investing firms. The implications of these findings are that market reactions reflect both undervaluation and signaling hypotheses of positive information associated with share repurchase. As the firms undertake capital investments, they generate future cash flows, limit the effects of leverage on financial distress and ultimately reduce the risk of the equity capital.

1.5. Contributions of the study

This thesis extends and complements the extant literature on share issuance and share repurchase announcements.

First, this study complements the existing literature on share issuance decisions. Studies such as Kaplan and Zingales (1997), Almeida et al. (2004) and Whited and Wu (2006) provide extensive explanations for the effects of financial constraints on corporate decisions, enumerating important characteristics that distinguish financially constrained firms from unconstrained ones. This study emphasises the extent to which the level of financial constraints affects the decision to time the issuance of shares. Financial constraints have significant impact on corporate decisions relating to employment, cuts in tech spending and capital spending (Campello, et al., 2010) and the costs of equity issues (Korajczyk and Levy, 2003; Dong et al., 2012; McLean and Zhao, 2014). This thesis argues that both internal and external financial constraints determine the equity market timing of firms. This financial constraints variable relates both to the internal cash flow limitations and the costs of external equity issues when there is low market liquidity and/or during periods of financial crisis. In other words, the issue of overvalued equity in the equity marketplace reflects both the degree of internal cash flows and external costs associated with such issues.

For example, during the financial crisis or periods of low market liquidity, financially unconstrained firms could issue overvalued equity due to both the ability to finance the issues and lower adverse selection costs associated with unconstrained firms compared to constrained firms. Given low market liquidity and/or during financial crisis - high costs of issuing equity - financially constrained firms are less likely to issue equity even if they are significantly overvalued (Korajczyk and Levy, 2003; Dong et al., 2012). Financially constrained firms, not only are find equity issues costly compared to unconstrained firms, but they also potentially are associated with higher adverse selection costs. Only unconstrained firms would have the financial resources to conduct equity issues and are therefore more likely to issue overvalued equity. Unlike previous studies, this thesis performs analysis of equity issues for simultaneous effects of internal and external financial constraints. Finally, stock price underperformance subsequent to equity issues is more severe for constrained than unconstrained firms, especially for issuance during financial crisis or low market liquidity. Overall, this study demonstrates that the effects of internal financial constraints on equity market timing is more pronounced when there is a concurrent external financing constraints, especially during periods of the financial crisis and/or low market liquidity. Simultaneous effects of internal and external financial constraints also determine subsequent stock price reactions following equity issues.

Second, the thesis provides evidence about the effects of operating performance on the choice of equity issuance method and the subsequent long run stock returns associated with equity issues in the United Kingdom. For the UK in particular, several equity issuance methods are available to firms and each method is associated with different level of asymmetric information effects. Different equity issuance methods induce different types and degree of market reactions. Theoretical models developed in the literature attribute these different market reactions to quality signalling that are associated with each issuance method (Eckbo and Masulis, 1992; Balachandran et al., 2008). Factors that signal firm quality include the issue price discount, volatility of returns and earnings management (Dissanaike et al., 2014). The extant literature shows that stock returns, especially short-term returns, are a function of both firm and

issue characteristics (Slovin et al., 2000; Barnes and Walker, 2006; Armitage, 2010). While these characteristics are transient and relate to the short term, operating performance will have more explanatory power in the long run than other firm characteristics that existing research used to explain the stock price performance following equity issues. Since operating performance is a long term performance metric it is better aligned to the changes in stock price performance than alternative factors that explain stock returns following equity issues. This thesis asserts that long run operating performance provide more explanatory power for the long run stock price performance following equity issues. Moreover, long run price performance also differs markedly among equity issues through rights issues, open offers and private placement. Operating performance explain firms inside information about their performance beyond the equity issuance period.

The study departs from most previous studies that look at the choice of equity issuance method on the basis of the post-issue stock returns. This study asserts that operating performance not only determines the choice of issuance methods but it also explains the post-issue long run stock returns associated with each equity issuance method. Operating performance explains better the firm quality theory over the long term. Subsequent to equity issues, firms conducting rights issues report better operating performance than open offers and private placement. Thus, firms associated with better operating performance seem more likely to conduct rights issues rather than open offers and private placements. Consistent with Capstaff and Fletcher (2011), the study further finds that rights issuing firms outperform open offers and private placement in the long run. Unlike Slovin et al. (2000) and Barnes and Walker (2006), rights issuing firms are associated with quality information signalling since it is unlikely that firms would issue overvalued equity to exploit existing shareholders. The study also finds that long run operating performance explains stock price underperformance associated with equity issues (Loughran and Ritter, 1997; Kabir and Roosenboom, 2003). This research argues that operating performance better explains the choice of equity issuance than firm and issue characteristics such as market-to-book ratio, stock returns volatility, issue discount, among others. In effect, the study asserts and finds strong

support for the effects of long-term operating performance on both the choice of equity issuance and long-term stock returns associated with UK equity issues.

Third, this study provides empirical evidence about levered share repurchase. According to conventional wisdom, firms repurchase their shares when they are undervalued (Ikenberry and Vermaelen, 1996; Stephens and Weisbach, 1998), have free cash flows (Jensen, 1986; Grullon and Michaely, 2004; Bozanic, 2010) and produce quality information signaling (Vermaelen, 1981, 1984; Grullon and Michaely, 2004; Baker et al. (2003)). Consistent with the free cash flow theory, it is less likely for internally financially constrained firms to conduct share repurchases unless they obtain external financing to support this program (Chen and Wang, 2012). It is equally puzzling for firms with high free cash flows or excess cash flows to borrow during share repurchase programs. The study provides evidence that internally financially unconstrained firms borrow to finance share repurchase programs. Payout policies make the implicit assumption that firms with significant financial resources do not necessarily require external financing in order to distribute cash to existing shareholders. Share repurchasing firms invest less in capital projects as more cash is distributed to shareholders (Jensen, 1986; Hahn and Lee, 2009). However, the study argues that the additional external financing acts as cash buffer for investments. Given that payout limits subsequent investments, I explore whether unconstrained firms issue debt capital at the time of share repurchase announcements and find evidence that unconstrained firms issue debt at the time of share repurchase announcements. The findings also support the hypothesis that internally financially unconstrained firms undertake more capital investments.

The study contributes to the share repurchase literature and in particular the financing of share repurchases by internally financially unconstrained firms. Firms do not only utilise internal cash flows to finance share repurchase but also a significant number of firms borrow to finance share repurchase (Chen and Wang, 2012; Lei and Zhang, 2015; Farre-Mensa et al., 2015). Whereas Chen and Wang (2012) assert that internally financially constrained firms

conduct share repurchases due to managerial overconfidence and hubris, this study highlights that internally financially unconstrained firms also borrow during share repurchase programs. For instance, unconstrained firms are more likely to be rated firms with investment grade ratings which allows them to obtain debt financing at low interest rates compared to an otherwise financially constrained firms. Stated differently, the degree of internal financial constraints influences the costs of debt financing and the ensuing effects of additional leverage on the costs of financial distress magnifies the overall costs of equity capital for constrained firms. Finally, Farre-Mensa et al. (2015) report that firms finance their payouts through debt and equity to both manage their capital structure and cash holdings. Their evidence shows that these firms could not have funded the payout without external financing. This result implies some of the firms are potentially financially constrained such that they could only finance their payouts using external financing. I find evidence similar to Farre-Mensa et al. (2015). Firms undertaking share repurchases obtain external debt-financing. In particular, I find that unconstrained firms borrow during share repurchase announcements and potentially utilize the additional cash buffers to finance investment expenditure. Overall, this study provides empirical evidence about levered share repurchases and show that investment expenditure explains levered share repurchase programs.

1.6. Organisation of the study

The thesis is organised into nine chapters.

Chapter 2 reviews the literature related to the effects of financial constraints on equity market timing. The chapter begins with the review of market timing theory and the other alternative theories that explain equity issuance decisions. It discusses the stock returns associated with equity issues and presents the theories that explain such stock price effects. Further, the chapter presents literature on internal and external financial constraints and discusses the impact on share issuance decisions. Finally, the chapter presents the testable hypotheses based on the theoretical and empirical evidence.

Chapter 3 provides literature review of equity issuance methods in UK. The chapter presents an overview of equity issuance methods in the UK and discusses the three main issuance methods available to UK firms. It then reviews the literature on short and long-term stock returns associated with equity issues and examines the theories that explain the returns. Finally, the chapter presents the testable hypotheses based on the literature.

Chapter 4 reviews the literature on share repurchase programs in the US. The chapter discusses the share repurchase regulations in the US and presents the types of share repurchasing programs. It further reviews the factors that drive share repurchase and the theories that explain the stock returns associated with share repurchase announcements. The chapter discusses the empirical evidence of internal financial constraints effects on share repurchase and highlights debt-financed share repurchase programs. Finally, the chapter presents the testable hypotheses based on the theoretical and empirical evidence.

Chapter 5 discusses the data and methodology used to test the hypotheses and answer the research questions. It, specifically, presents the dataset and how it was collected, the econometric estimations and the variables for the empirical chapters. In addition, it discusses the justifications for the proxies used in the models.

Chapter 6 investigates the simultaneous effects of market timing and financial constraints on share issuance decisions. The chapter presents the descriptive statistics of the variables and the regression analysis of the models explained in chapter 5. It also discusses the robustness tests conducted to corroborate the results of the main model.

Chapter 7 provides the empirical testing of the asymmetric information effects on long-term stock returns associated with equity issuance methods. It provides the descriptive statistics of the variables, the univariate analysis and the multivariate analysis of the choice of equity issuance methods. In addition, the chapter presents the regression analysis of the long-run stock returns associated with equity issuance methods and presents the robustness tests of the main model.

Chapter 8 examines the share repurchase financing of unconstrained firms and explores the factors that drive unconstrained firms to borrow during the share repurchase announcements date. This chapter presents the descriptive statistics of the repurchase sample, the regression analysis of the models explained in chapter 5, and outlines the robustness tests for the results of the main model.

Chapter 9 concludes the thesis by summarising the key findings, discussing the limitations of the study and proposing recommendations for future research.

Chapter 2

LITERATURE REVIEW: SHARE ISSUANCE DECISIONS

2.1. Introduction

Share issuance decisions have implications for changes in capital structure and firm value. These decisions also determine availability of financing for investment projects and can affect the overall risk exposure of the firm. However, the level of internal financial constraints could affect issue of external equity. The effect of the financial crisis in 2007/2008 highlights the importance of external financial constraints in equity issues. This chapter reviews the theories that underlie share issuance decisions. It further examines the literature on the effects of both internal and external financial constraints on shares issuance and discusses the stock returns associated with equity issues. The chapter also develops testable hypotheses that set out the focus of the empirical chapter 6 of the thesis.

The chapter has four sections. Section 2.2 reviews theories of share issuance. It discusses theories such as asymmetric information/pecking order theory, market timing, agency theory, and trade-off theory, among others. Section 2.3 focuses on the effects of mispricing and internal financial constraints on equity issues. The section examines the effects valuation and financing decisions, share issuance and internal as well external financial constraints. Section 2.4 explains the empirical evidence of stock returns associated with equity issues and the theories that explain the stock returns' behaviour. Section 2.5 develops the testable hypotheses and section 2.6 concludes the chapter.

2.2. Review of Theories of Share Issuance

Through corporate financing policies, firms obtain funds in the external capital market for varied reasons. For example, firms need financing to undertake new investments, to pay dividends, to maintain capital structure, among others. The most popular financing sources include debt and equity, as well as convertible debt. Several theories have been developed to explain in part the reasoning behind financing decisions and the subsequent security issuance. Several

theories explain the motivations for share issuance decisions. The study provides here a brief theoretical framework of some of the theories documented in the finance text. In the empirical analysis, various proxies are used to control for the effects of these competing theories that determine the issuance of equity.

2.2.1. Pecking Order and Asymmetric Information Hypotheses

Two related theories that explain security issuance decisions are the pecking order and the asymmetric information hypothesis. According to the pecking order theory, firms follow a hierarchical order of financing preferences; internal financing is preferred to external financing. Myers (1984) and Myers and Majluf (1984) explain that high costs of adverse selection associated with external financing means firms will only use retained earnings unless external financing is needed in which case debt is preferred to equity issues (debt attracts low adverse selection costs than equity financing). In a rational setting, undertaking valuable investment opportunities requires firms to have readily available internal cash flows without necessarily going to the external capital market. It is only after exhausting internal funds firms are encouraged to obtain financing from outside in order of debt followed by equity (Myers and Majluf, 1984). In an asymmetric information environment, financing choices that reveal positive information to the market are more likely to be preferred by managers. Firms that have expected high cash flows are in a better position to enter the capital market but high asymmetric information effects hugely affect such prospects. Thus, high valued firms are more likely to issue debt capital to signal substantially positive financial position than low valued firms, despite significantly high asymmetric information. The overriding intuition is that debt signals firm quality.

Unlike equity issues, debt financing is associated with low asymmetric information effects. As debt issues are made with the supposition of providing positive information about the firm's quality, price reactions should be positive. Thus, leverage increasing transactions such as debt issue and equity repurchase are highly correlated with positive stock price returns. Equity issue reduces any positive impact of the positive financial and operating effects due

to its associated adverse selection costs. The conjecture is that at high (low) asymmetric information, firms issue debt (equity). While equity issue costs much in terms of issue costs and asymmetric information effects, debt can be issued cheap and with relative rapidity; despite the subsequent increase in costs of financial distress. Myers (2001) explains that large debt financing is a major support for the pecking order theory. The empirical support is mixed but generally pecking order has some predictive power of financing decisions.

Studies such as Shyam-Sunder and Myers (1999) and Lemmon and Zender (2010) test the pecking order theory. For example, Lemmon and Zender (2010) use dataset from 1971 to 2001 and confirm how the pecking order influences financing choices. Firms that enter the debt market have debt capacity to support additional leverage, stable cash flows with large pool of collateral. Information transparency enables firms to borrow at the equilibrium interest rate and increases their ability to access capital from the external debt market. With debt capacity, additional debt financing will have limited impact on costs of financial distress, while indirectly rebalancing the capital structure. Such debt-issuing firms are supported by high levels of tangible assets and relatively stable profitability to guarantee stable cash flows.

However, it is possible for firms to obtain financing without necessarily going to the external market. Transactions such as employee options are unrelated to adverse selection costs; therefore, equity issues are not constrained by asymmetric information effects. If firms consistently pursue these issues, then the pecking order breaks down and is unable to determine which security-financing firms adopt. Again, studies show that financial deficits have limited predictive control of financing choice. According to the pecking order theory, equity issues are made as last resort (due to adverse selection costs) in a state of significant debt capacity concerns that restricts access to the debt market. Thus, consequent financial distress costs discourage additional debt financing. This theoretical explanation has been challenged in several studies citing different motivations for equity issues contrary to predictions of the pecking order hypothesis.

In the spirit of asymmetric information effects, firms with substantial cash stockpile should avoid the external capital market and finance investment using internal cash flows. Helwege and Liang (1996) confirm this theory that firms with greater cash surpluses avoid external funding, but their results do not imply that cash deficit forces firms into the external finance market. Interestingly, asymmetric information also does little to explain the pecking order theory as significant asymmetric information fails to trigger equity finance. Equity issuers dominate external security financing that, as emphasised by Fama and French (2005), is not due to duress or cash deficit as theory posits. Again, firms that have previously obtained external financing are highly probable of obtaining additional funding subsequently. Again, Frank and Goyal (2003) use broad cross sections of 768 publicly traded American firms from 1971 to 1998, similar to Shyam-Sunder and Myers (1999), and find that equity dominates debt issues unrelated to financial distress.¹¹

From the above discourse, firms willing to make equity offering should be prepared to manipulate the market prior to the financing decisions in order to reduce the information asymmetry. But Korajczyk et al. (1991) assert that information releases precede equity issues. Asymmetric information effects represent immense bane on equity offerings due to the tendency for rational investors to discount the stock prices as a result of a lack of symmetry in information between managers and investors. Arguing for dominating equity issues, they offer evidence that dispels the general notion of unattractive equity issuance decisions. They find a clustering of equity issues around information release dates. To obtain favourable response to security issuance, especially equity, firms are more inclined to, as a matter of timing philosophy, release 'good' information prior to equity offerings. The effect is to discount the tendency for stock price depression on issuing the security. Instructively, equity issues preceding positive information releases are associated with significant price premium. However, other systematic events can trigger price reaction in either direction.

_

¹¹ Also Fama and French (2005) highlight the limited impact financial distress and asymmetric information has on financing decisions.

2.2.2. Market Timing

A relatively new theory that builds on the predictions of the information asymmetry hypothesis is the market timing theory. Market timing is one of the financing theories that have been tested to influence corporate financing decisions. It emphasises managers' efforts at issuing securities, especially equity, at a time in the market where they perceive benefits in the form of high stock prices and/or leave the market at low stock prices. Baker and Wurgler (2002) define market timing as the tendency for firms to issue equity at high prices (when stock is overvalued) and repurchase at low prices (when stock is undervalued). Temporary fluctuations in the cost of equity relative to other sources of capital guarantee the prospects of market timing. Not only does market timing persists in theory but it has also received much attention in empirical studies. Nonetheless, the evidence is mixed and fails to provide a convincing argument about its dominance in security issuance decisions. Managers are optimistic when market conditions favour their financing decisions and that drives the opportunistic attitudes of firms that form the foundation of market timing. Survey evidence provides converging position of CFOs across countries and different economic environments. For example, Graham and Harvey (2001) survey 392 US CFOs and find that securities issues reflect the timing potentials of firms. 12

Furthermore, the theory of market timing relates to the effects of adverse selection costs associated with security issues with variations across both time and firms. This version of market timing reflects the asymmetric information theory of capital structure and explains that equity market timing depends on the degree of asymmetric information. According to Myers and Majluf (1984), the high costs of adverse selection due to information asymmetry constrain firms to issue equity. Several studies explore time-varying adverse selection costs (Lucas and McDonald, 1990; Korajczyk et al.1992) and adverse selection that varies across firms (Choe et al.1993). Information releases prior to equity

_

¹² Bancel and Mittoo (2004) studies 87 CFOs from 16 European countries while Brounen et al. (2006) provides findings on 313 CFOs from 4 European countries. Both evidence including Graham and Harvey (2001) suggests that managers actually time the market before security issues are made.

issues reduce the effects of information asymmetry and increase investor optimism (Korajczyk et al. 1991; Rajan and Servaes, 1997). By extension, the changing adverse selection costs should also reflect the variations in market-to-book ratio (Baker and Wurgler, 2002). Generally, firms prefer debt capital because it conveys information about firm quality and it avoids the asymmetric information effects associated with equity financing. However, beyond the tax advantages and the power of debt to reduce the agency costs of free cash flows, debt issues increase costs of financial distress and other agency costs, the prospects of debt overhang crowd (Myers and Majluf, 1984). Thus, in the presence of asymmetric information, and given the excessive restrictions debt places on undertaking growth options, equity issues can only be made to support investment for overvalued stock.

Another explanation to the market timing theory concerns the presence of irrational investors and/or managers and time-varying mispricing. An interpretation of mispricing is that investors are irrational and therefore make systematic errors in forming expectations about stock prices and hence the value of the firm. These perceptions mean that stocks can become significantly over or under-valued at particular points in time. Mispricing relates to overvaluation of equity and undervaluation of repurchase where overvalued equity are issued to profit from high stock prices and undervalued equity are repurchased because they are cheap. Survey evidence by Graham and Harvey (2001) and capital budgeting model by Stein (1996) both support managers' attempt to time their equity issues. Moreover, Baker and Wurgler (2002) assert that past changes in market-to-book ratio actually explain equity issues and subsequent changes in capital structure. The above findings suggest that firms systematically time their equity issues when their shares are overvalued.

Taken together, mispricing should coincide with low asymmetric information. Conversely, the high mispricing can neutralise the adverse selection costs of asymmetric information to motivate firms to issue overvalued equity. Therefore, mispricing has substantial predictive power for equity market timing because the total gains exceed the adverse announcement effects (Baker and Wurgler, 2000). The empirical analysis offers new dimensions to the market

timing theory by looking at how internal financial constraints affect the issue of overvalued equity and the subsequent stock price reactions.

2.2.3. Trade off Theory

Since security issues can affect capital structure and firm value, theory of static trade off holds that there is a target leverage ratio that triggers security issuance decisions when there is a deviation of the current ratio. Debt and equity issuance are dependent on the conditions of the capital mix at a point in time such that debt is issued when leverage is low and equity at high leverage ratio. By maintaining target leverage firms balance the benefits of debt capital from tax advantages and control of free cash flow problems and costs emanating from bankruptcy costs and other agency costs. Moreover, managers with high levels of free cash flow under their control can engage in inefficient investments. Leverage restricts managerial hubris through fixed interest commitments and limit managerial discretion (Jung et al. 1996).

Hovakimian et al. (2001) test the trade-off theory using debt-equity model and contend that results on trade off are statistically significant. Firms possess both assets in place evidenced by the value of tangible assets in the business and growth options represented by the level of intangible assets available. The interactions between these two alternatives inform firms about the choice of financing to access. Whereas tangible assets expressly provide collateral, they support debt financing more than equity and growth opportunities track equity financing. Hovakimian et al. (2001) find that more profitable firms have a relatively low leverage ratio that eventually triggers more debt financing to move the leverage ratio to target. On the other hand, since valuable growth opportunities reflect positively in the stock prices, firms are motivated to issue equity rather than debt to rebalance their leverage to target ratio. Specifically, they highlight the impact of repurchase on rebalancing the leverage.

It is noteworthy to understand the short run impact of market timing in the context of trade off theory. Timing security issues result in temporary distortion of the capital structure. Equity creates under leverage while debt and repurchase are leverage increasing transactions. Firms that consistently adjust

their capital mix to reflect the target ratio rebalance their leverage subsequent to timing activities. Thus, leverage adjustment curtails the effects of M/B on capital structure. Kayhan and Titman (2007) as well as Marsh (1982) emphasise that when all market timing opportunities disappear financing choices move the leverage towards optimum level. The studies by Shyam-Sunder and Myers (1999), using the partial adjustment technique, summarises the degree of explanatory power between trade-off and pecking order theories. Relying on the effects of tax advantages and financial deficits they find that pecking order can be rejected when firms follow static trade off but fail to reject static trade off in the event of pecking order. Despite the apparent bias in the sample towards larger firms, it is still intuitive to generate further investigation. A model which incorporates the effects of stock valuation and asymmetric information will be interesting to appreciate how this theory accommodates other factors other than tax advantages and costs of financial distress.

2.2.4. Agency Theory

The agency theory emphasises that organisations are characterised by principal-agency relation where shareholders (principal) entrust to managers (agents) the control of the business affairs. This separation of ownership from control creates conflict of interest due to the tendency for managers to pursue acts that benefit them at the expense of the shareholders. Jensen (1986) asserts that financing decisions only reflect the interest of the agents not the principal. Agency costs consideration forces firms to depart from pecking order in their financing decisions. High agency costs of debt reflect increased equity financing to support investment projects and managerial discretion forces firms to issue equity when there is no viable investment opportunity to pursue.

Firms with valuable investment opportunities that seek to grow issue equity or those without investment opportunities but are constrained to issue debt due to debt capacity concerns issue equity. The agency model confirms that such equity financing only benefits management at the expense of shareholders. Excessive leverage reduces managerial discretion. Management losses control over free cash flows due to fixed commitments of interest payment to creditors

and monitoring mechanisms inserted in debt covenants to limit potential managerial hubris. Increasing leverage imposes financial discipline on management. This in part works to curtail any opportunistic activities management would have otherwise had the leeway to undertake at the expense of shareholders. The agency theory underlies both the information asymmetry and market timing theories in such a way as to define the issuance of equity. Invariably, the empirical analysis undertaken in this study captures the effects of agency theory on the issuance of overvalued equity.

2.2.5. Managerial Ownership

Security issuance decisions to a large extent depend on the level of managerial interest attached to the firm. The impact of asset substitution can be allayed to a significant extent when managers have interest in the debt structure of the firm. Increased leverage exposes the firm to costly financial distress but managers in acting in the interest of shareholders can pursue risky investments that can yield high returns. However, in the event of failed projects much of the fixed payments to debt holders could be greatly affected. Managers with pension benefits and compensation packages have incentive to act in ways that protect the business from uncertain bankruptcy. Therefore, with a meaningful managerial investment in the debt capital of the firm, they seek to act in the interest of other outside debt holders and invariably fail to advance wealthenhancing programmes that benefit shareholders (Cassell et al. Managers tend to forfeit risky investments to preserve the financial health of the company. This reduces the agency costs of debt between managers and debt holders. For example, Jensen and Meckling (1976) emphasise that CEOs with inside debt holdings make investments and pursue financial policies that mitigate agency costs of debt. Since debt reveals positive information about the earnings of a company, it follows that CEO debt holdings should offer stock price premium. The above explanations indicate that managerial ownership can be a source of information asymmetry and hence affect the issuance of equity.

2.2.6. Managerial Entrenchment

In the context of Jensen's (1986) free cash flow theory, debt constrains managers from undertaking suboptimal investment decisions with the available cash that fosters managerial empire building objectives. To limit these tendencies, Jensen advocates for high levels of debt capital to disgorge cash to outsiders in the form of increased fixed interest payments. The real disadvantage associated with the agency costs of free cash flow theory is that it equally works to undermine investments in good projects, as do bad ones. However, as a corollary to this explanation, Zwiebel (1996) offers a new dimension to the power of debt in restraining managerial hubris.

While managers typically act in manner that tend to promote their personal interest, excessive debt levels presents a credible avenue for managers to restrict bad investment projects. Under managerial entrenchment theory, debt financing provides a moderating influence on managerial actions that invariably protects firm value. Takeover threats coupled with bankruptcy costs and loss of managerial entrenchment encourages managers to pay out dividends from increased debt financing. This, in effect, deactivates potential investments in bad projects. Thus, the power of debt in protecting against undertaking bad projects should be accompanied by a contemporaneous dividend payment of the available cash flow.

2.2.7. Earnings Management

As an accounting anomaly, firms manage earnings using revenues and/or expenses. Accounting earnings comprise both cash component that represents transactions for which corresponding cash payments/receipts have been made and accrual component where payments or receipts are deferred beyond the current accounting period. In order to portray a healthy earnings structure, firms can deliberately defer expenses and recognize immediately revenues in the books to shore up the earnings for the period. Thus, earnings management occurs when the discretionary accruals component of the earnings has been unjustifiably inflated. The effect is that subsequent earnings must reflect this

short term trend leading to further abnormal accruals in the reported financial statements.

The extant empirical evidence suggests that firms engaged in earnings management also undertake significant corporate financing decisions. For example, initial public offerings (IPO) (Teoh et al., 1998) and seasoned equity offerings (SEO) (Rangan, 1998; Teoh et al. 1998) follow periods of earnings management. While Teoh et al. (1998) propose that investors act naively to incidence of earnings management, Shivakumar (2000) emphasises the concept of 'managerial response' to investors anticipation of managers earnings management practices. The managers in turn rationally overstate earnings prior to offering announcements at least to the extent anticipated by the market. This perceived earnings manipulation causes the share prices to rise above the market expected valuation; the market response by discounting the share price to normal levels. Jensen (2005) provides theoretical underpinnings as the antecedent to the earnings management phenomenon.

As in the case of market timing, overvaluation can be linked to earnings management. The tendencies for information to filter through the market and investors opinions about the value of firms to converge over time stifle sustaining overvaluation in the long run. Since overvaluation is a temporary phenomenon, firms and managers readily attempt to benefit from this anomaly in the short run with dire consequences for shareholder value in the long run. Chi and Gupta (2009) test the hypothesis that relates overvaluation to earnings management previously theorised by Jensen (2005). Consistent with their predictions, overvalued equity engages in earnings management. The motivation is to justify that overvalued firms also report high operating performance. However, the effect of managing earnings in support of overvaluation excites market reactions. Chi and Gupta (2009) find negative stock returns and poor operating performance associated with overvalued firms, especially those that simultaneously manage their earnings. Information asymmetry relates to earnings management and the addition of appropriate proxy controls for the effects of earnings management on share issuance decisions.

2.3. Mispricing and the Effects of Internal Financial Constraints

This section reviews relevant literature about market timing and share issuance decisions. It also discusses the effects of financial constraints and establishes how the level of financial constraints can potentially undermine prospects of equity market timing. The section is divided into three subsections. Subsection 2.2.1 discusses the effects of valuation on financing decisions. Subsection 2.2.2 examines the influence of internal financial constraints on equity issues while subsection 2.2.3 reviews the effects of external financial constraints on share issuance decisions.

2.3.1. Valuation and Financing Decisions

Extant empirical evidence such as Akerlof (1970), Myers and Majluf (1984), Bayless and Chaplinsky (1991), (Helwege and Liang (1996), Frank and Goyal (2003), Baker and Wurgler (2002) DeAngelo et al. (2010) assert that capital structure changes have significant impact on firm value, contrary to the capital structure irrelevance proposition by Modigliani and Miller (1958). This notion implies that it is necessary that managers take financing decisions with the view to improve the overall shareholder value. Thus, the degree of mispricing enables firms to issue securities that enhance this value. Not only does mispricing encourage equity offering but other evidence also suggest that mispricing is the result of available growth opportunities. Firms with substantial investment options have the edge to source financing from external capital market. In the spirit of market timing, firms issue equity at high valuation (high M/B) and repurchase equity/issue debt at low valuation (low M/B). With increasing levels of equity (at high M/B), the capital mix tilts more towards low leverage and corresponding leverage increasing transactions (repurchase/debt issue) result in high leverage. Conclusively, firms that time the market in their issues have low leverage for equity issues and improve their leverage levels at low valuations.

Consequently, changes in valuation should trigger specific financing decisions in line with equity market timing proposition. If analysts are optimistic about the earnings of the company, there is the tendency for the market value to

deviate markedly from the fundamental value. This encourages opportunistic managers to issue equity to profit from the temporary price divergence. Baker and Wurgler (2002) emphasise that companies do not only make equity financing at high valuation and repurchase at low valuation but also capitalise on analyst favourable opinion about the company. In testing the persistence of market timing on capital structure, they study a sample of firms in IPO time and develop subsamples over a ten-year period to determine the changing patterns of capital structure as market valuation changes. Their results reveal that there is a persistent impact of market valuation changes on the capital structure on a long run basis. Capital structure reflects the past valuation movements. The net effects of high market-to-book values are to lower leverage independent of increased retained earnings and decreased debt.

Moreover, the degree of responsiveness to valuation effects relates to the level of available cash flow to conduct the appropriate security issuance. Internally financially sound firm will be better able to react to mispricing than an otherwise internally financially constrained entity. The intuition is that high transaction costs can impede the ability and the speed with which firms make issues in accordance with equity market timing hypothesis. Dong et al. (2012) highlight the influence of internal financial constraints on valuation effects. They study securities issue on the Toronto Stock Exchange and find that companies that have substantial timing opportunities will be inclined to issue equity only when they are internally financially unconstrained. Any issue of additional debt increases the leverage that is associated with costly financial distress. Similarly, the deduction from their findings is not too remote from the pecking order hypothesis. However, as they emphasise, equity issue is highly valuation dependent even when there are internal financial constraints.

However, other evidence suggests that M/B is related to growth opportunities rather than mispricing. High M/B reflects available growth options that firms can undertake. Investor optimism about future earnings underscores the potential for firms to turn the valuable investments into assets in place that is positive information to the market. This effect, coupled with other information released to the market, reinforces favourable market valuation of the company

stocks. Hovakimian (2006), for example, documents result consistent with the hypothesis that historical M/B tracks current expenditure and R&D such that high M/B drives higher current capital expenditure and research and expenditure, a proxy for growth opportunities. The effects of market timing on leverage disappear and growth opportunities dominate financing decisions. These findings cast doubt on the market timing persistent effects on leverage. The study explores the effects of internal financial constraints on the market timing of equity issues. Unlike previous studies, this study seeks to provide new understanding of the factors that determine the issuance of overvalued equity. Thus, in addition to the traditional idea of market timing dependent on mispricing or overvaluation, the focus of this research is an attempt to offer new insights into equity market timing phenomenon.

2.3.2. Internal Financial Constraints and Corporate Financial Policy

Firms are internally financially constrained if they face funding constraints that may restrict their ability to finance new or on-going projects. Internal financial constraints also mean the frictions or restrictions that impede access to external financing either through borrowings or equity issues (Fazzari et al. 1988; Kaplan and Zingales, 1997; Lamont et al. 2001; Almeida et al. 2004; Whited and Wu, 2006; Chen and Chen, 2012). Fazzari et al (1988) assert that in a perfect market where firms have unlimited access to capital market, investment decisions are independent of a firm's financial position. However, in an imperfect market and with limited internal financing firms that rely on external financing incur substantial cost to access both equity and debt capital. Low cost external financing implies firms are able to smooth their investments when there is insufficient internal financing. Firms are classified as internally financially constrained using a priori conditions such as size, dividend payments and leverage. Investment is driven by cash flow fluctuations when firms face significant costs of external financing. Firms are internally financially constrained if they are restricted in accessing external financing either through borrowings or equity issues. In other words, there is a wedge between internal and external costs of financing. By this definition, Kaplan and Zingales (1997) – hereafter KZ index - conclude that all firms are likely to be

classified as internally financially constrained when they incur transaction costs in securing external financing.

Firms are internally financially constrained if they are restricted in accessing external financing either through borrowings or equity issues. In other words, there is a wedge between internal and external costs of financing. By this definition, Kaplan and Zingales (1997) conclude that all firms are likely to be classified as internally financially constrained when they incur transaction costs in securing external financing. The appropriate differentiation of firms in terms of internal financial constraints depends on the degree of the costs between internal and external financing. Their findings contrast that of Fazzari et al. (1988) and raises concern about the link between investment-cash flow sensitivity and financing constraints. Using sample of 49 low dividend paying firms (identified by Fazzari et al. (1988) as constrained firms), they find that less financially constrained firms exhibit greater sensitivity than more constrained firms. The underlying assumption of monotonicity of investmentcash flow sensitivity with the degree of financial constraints is challenged. However, data snooping issues might be raised against the sample size and period.

Lamont et al. (2001) define financial constraints as frictions (in an imperfect market) that prevent firms from obtaining external financing to undertake investments. Financial constraints imply credit constraints or inability to borrow, dependence on bank loans, inability to issue equity and illiquidity of assets. However, Alti (2003) argues that financing constraints is not unique to only markets with frictions because the information signalling effects of cash flows will still hold in a model without market imperfections (Fazzari et al., 1988; Gomes, 2001). For instance, Campello et al. (2010) use survey measure of financial constraints where CFOs determine whether their firms are facing financial constraints due to unavailability and cost of credit. Consistent with the literature, they find that constrained firms planned cut in capital spending, employment and tech spending. In events where they anticipate restrictions in borrowings they would sell assets. Despite the apparent limitation on financial policy for constrained firms, Chen and Wang (2012) provide evidence to

suggest that a number of constrained firms repurchase equity. Firms rely on financial resources to undertake investment projects, make dividend payments, decide on security issuance or repurchase transactions, among other issues. Thus, the capacity to meet opportunities in terms of investments and other corporate financial policies is defined by whether or not the firm is financially unconstrained.

Moyen (2004) offers an explanation to why some inconsistencies exist between the different financial constraints measures, especially between FHP index and KZ index. Debt financing in a frictionless market and for firms less financially constrained affect internal cash flow and the level of investment likely to be undertaken. Unconstrained firms with access to the capital market can easily obtain debt financing to supplement the internal cash flows and consequently increase their investment spending when there are more favourable investment opportunities. Moreover, whereas unconstrained firms with more cash flow can afford to increase investment and dividends simultaneously, constrained firms can only choose to invest or pay dividends. Thus, in support of Kaplan and Zingales (1997), constrained firms have lower cash flow sensitivity than unconstrained firms. Again, using low dividends to proxy for financial constraints and consistent with the predictions of Fazzari et al (1988), she concludes that the unconstrained firms rather pay low dividends and have higher cash flow sensitivity. The ability of these firms to issue more debt means that equity holders of unconstrained firms receive lower dividends.

Whether or not firms anticipate and face financing constraints has implications for liquidity management, especially cash flow decisions. Almeida et al. (2004) develop a different measure of financial constraints which underlies the relevance of liquidity management. Kaplan and Zingales (1997), Almeida et al. (2004) and Chen and Chen (2012) all argue that investment-cash flow sensitivity is flawed as a measure of financial constraints. They assert that the ability to predict investment demand influences the explanatory power of the index. Liquidity is an important factor that affects the financial policies of firms. Chen and Chen (2012) contend that there has been a significant decline in investment-cash flow sensitivity over the years, including the period of the

credit crunch. For example, Almeida et al (2004) predict that firms with uncertain cash flows will save more to avert situation of forgoing anticipated investment projects in future or cut dividends, among others. The cash flow sensitivity of cash defines the propensity for firms to save out of cash flows. Similar to investment sensitivity to cash flow index, they find that constrained firms with uncertain cash flows have positive cash flow sensitivity of cash. Unconstrained firms do not rely solely on savings to undertake financial policies; their savings are not related to cash flows.

2.3.3. Share Issuance and Internal Financial Constraints

This study demonstrates the effects of financial constraints on equity issues beyond overvaluation and mispricing. A supplementary factor to mispricing is how the level of financial constraints could influence firms' equity issuing decisions. It should be noted that financing decisions incur substantial costs. Comparatively, transactions costs incidental to equity issues are normally higher than that of debt capital, consistent with the adverse selection costs theory (Myers, 1984 and Myers and Majluf, 1984). Therefore, firms that are financially sound stand a better chance to issue equity when market conditions are favourable. Internal financial constraints determine the firm's demand for external financing (McLean and Zhao, 2014). Ceteris paribus, constrained firms cannot issue equity at high costs given the supply of external funding provided by investors. At high valuations, low asymmetric information with financial deficit and over leverage, firms prefer to issue equity capital than debt financing. Extant empirical evidence supports this trend yet most studies fail to account for the effect of financial constraints on the latitude of firms to pursue their financing objectives. Thus, financial flexibility relates to the ability to enter the external capital market. Firms that are substantially financially constrained rarely enter the capital market. However, studies show disparate results.

Theoretically, security issuance decisions depend on the need to achieve target leverage ratio, level of debt capacity and asymmetric information effects and mispricing. The speed of adjusting to optimum leverage, especially using equity, increases with financial flexibility. Korajczyk and Levy (2003) stress

that financially constrained firms find it impossible and expensive to issue securities that quickly restore the optimum debt ratio.¹³ In the study of security issuance decisions of Canadian firms, Dong et al (2012) emphasise the influence of financial constraints on the choice and speed of issuance in relation to high valuation. They contend that less financially constrained firms have high probability to issue equity when they are overvalued. This is suggestive that market timing is enhanced with cash availability.

However, if firms issue equity at a time when there is the need for adequate cash flow to support investment then this weakens equity market timing opportunities. DeAngelo et al. (2010) find that market timing and life cycle play significant roles in financing decisions. Yet, at high valuation the need for "near term" cash for future investments drives equity financing. That is, not all firms take advantage of timing opportunities. With substantial financial slack or high cash balance, there is limited motivation to obtain additional external financing regardless of the level of stock price. It is only at high financial constraints will firms be forced to access outside financing. They find that close to 62.6% and 81.1% of equity issuers would have run out of cash or would have had subnormal cash balances respectively in the year after the issue.

Firms that are financially constrained rarely enter the external capital market. But these firms are more likely to absorb any cash flow shocks internally. Adequate internal funds provide buffer for investments expenditure with fewer tendencies to contact outsiders for financing. Financial deficit/surplus is the overriding factor in deciding to enter the external market. Firms are likely to relegate cost of financial distress from excessive debt capital to increase/decrease debt with substantial deficit/surplus. They adjust capital structure either in the face of above target debt or below target debt ratio. Byoun (2008) finds, in contrast to Kayhan and Titman (2007), that capital structure adjustment is more above-target ratio dependent than it is for below-target ratio. Financial surplus is used to retire debt capital for excessive debt

-

¹³ Faulkender, Flannery, Hankins and Smith (2012) find that financial constraints have, in terms of magnitude, larger effect on leverage rebalancing than market timing considerations.

but the tendency to issue more debt/repurchase equity in financial deficit is less pronounced.

2.3.4. Share Issuance and External Financial Constraints

Whereas internal financial constraints occur within the firm due to policy changes and decisions, external financial constraints is an economy-wide effect on the overall capital market. This study synthesises the effects of internal and external financial constraints in explaining equity market timing. The financial crisis of 2008 and the European debt crisis of 2011 demonstrate how external financing conditions impact corporate policies. Recent studies highlight changes in firms' investment and payout decisions during the financial crisis. For example, Campello et al. (2010) and Campello et al. (2011) and McLean and Zhao (2014) document the effects of financial crisis on firm investment, payout polices, employment and assets sales to finance investments. Ivashina and Scharfstein (2010) also show constraints on bank lending during the crisis. These findings are consistent with external financing constraints effects on corporate decisions. The financing conditions have direct impact on the costs of issuing equity as reflected in investment banking fees. For example, Lee et al. (1996) report that investment banking fees represent about 7% of the total proceeds from equity financing. 14 Thus, there is a direct correlation between low market liquidity and high investment banking fees, emanating from unfavourable external financing conditions (Corwin, 2003; Butler et al. 2005). Butler et al. (2005) show that difference in investment banking fees between most liquid and least liquid stock is about 21%.

That security issuance decisions occur contemporaneously with favourable market conditions is fundamental underlying factor of the market timing theory. If firms do not issue equity to undertake investments now then it is expected that the firms will stockpile cash subsequent to issuance. The apparent demand for cash to undertake future investments impel firms to obtain external financing with the aim to building cash flow as precaution, especially

¹⁴ Market liquidity depends on how ease firms can sell their shares and as the liquidity becomes low due to unfavourable conditions, investment banking fees increase to compensate for the extra effort and time to sell the shares.

during good times. Akin to market timing, this period represents high stock prices when firms can cash in on mispricing. A combination of low transactions costs, availability of valuable future investments, cash flow volatility and high liquidity typifies periods of favourable market and firm level conditions. Bolton et al. (2013) assert that improved financing conditions raises stock prices. Conversely, higher probability of a crisis and the adverse effects on stock prices and costs of issuance motivate firms to invest conservatively and build cash buffers now through significant cash savings. Firms are also more likely to issue equity now prior to the downturn in financing conditions. However, firms are unable to anticipate when external financing conditions will deteriorate which means firms can only respond to the ex post financing conditions.

Thus, equity market timing activities should be expressive during good market conditions compared to a recessionary period. Alti (2006) offers a new dimension to timing opportunities by looking at the market conditions under which equity issues are made, especially IPOs. He notes that since market conditions are orthogonal (statistically independent) of other factors affecting financing decisions, those conditions better isolate the timing behaviour. Particularly, financial distress and growth characteristics do not influence external financing. "Hot market" firms effectively time the market and issue more equity than cold market firms. The short run impact of market timing on capital structure explains the subsequent attempts to rebalance leverage levels after the equity issues. The less equity issues after IPO could be due to long run stock underperformance that makes equity significantly undesirable giving preference to debt financing.

With limited adverse selection costs, the overall market reactions to financing choice should reflect moderate effects on stock prices. Alti and Sulaeman (2012) draw distinctive effects of both high and low institutional investor demand for firms' equity and find a strong support for the increased likelihood for overvalued firms to issue equity when there is perceived institutional demand for the stock. The need for financing is wide ranging and could enforce issuance decisions that are suboptimal to the purposes for which they are

sought in the capital market. Market conditions are subject to the vagaries in the general economic environment with both good times and bad times. A key incentive for stock market transactions is liquidity that greatly fluctuates overtime. The extant literature on market timing provides evidence that firms could time their equity issues during favourable market conditions even when there is no immediate need for external equity financing (Baker and Wurgler, 2002; Huang and Ritter, 2009; Fama and French, 2005; DeAngelo et al. 2010)

By inference, firms that systematically time their security issues should, regardless of investment options, possess substantial financial slack, especially for equity financing. Market conditions that enhance equity offerings improve the cash position when there are no potentially viable investments to undertake. McLean (2011) studies these features as drivers of equity issuance to meet firm's precautionary motives. Discounting market timing as reflecting the cash savings following equity financing, he reports that issuers' do not record substantial low returns. A significant fall in cash flow over the years has made firms resort to equity issuance as source of cash to meet future investments commitments. Specifically, he finds that about 60% of recent decade share issuance proceeds are saved. There is increasingly less dependence on internal cash flow as firms have resorted to external financing as source for cash flow (Pontiff and Irvine, 2009; Bolton et al. 2013). As precautionary motive, firms maintain target cash-to-assets ratio that they adjust through share issuance decisions. In essence, the study focuses on how external financial constraints interact with internal financial constraints to determine the issuance of overvalued equity.

2.4. Share Issuance and Stock Returns

The empirical literature provides evidence of stock underperformance following the issue of equity (Loughran and Ritter, 1995; Spiess and Affleck-Graves, 1995). Stocks that trade in efficient markets respond instantaneously to new information that the market receives. Such information includes, among others, earnings announcements, corporate takeover and merger news, change in management. Security issuance likewise has signalling effects and rational market participants react accordingly. Post issue stock returns fall for equity

and rise with equity repurchase/debt issues. This section discusses the underreaction hypothesis of stock returns in subsection 2.3.1. Subsection 2.3.2 explains rational discount rate theory of stock returns whereas subsection 2.3.3 examines the investment-based explanation. Finally, subsection 2.3.4 explores the economic perspective underlying stock returns response to equity issues.

2.4.1. Underreaction Hypothesis

Equity issues convey "bad" news that triggers rational market response. Repurchase and debt financing signal good information about the prospects of the business and highlight earnings quality. Given the perceptions of the market concerning the financing choice, equity financing attracts low returns whereas repurchase/debt earns high returns. The extant empirical findings attribute this trend to market timing determined solely by the level of mispricing regardless of investment opportunities. Decisions to issue securities reflect managers' opportunistic behaviour to profit from the lack of symmetry between stock market value and the intrinsic or fundamental value.

Stock market reactions to security issuance are not immediate which suggests that investors are irrational and are conservative in receiving new information in the market. Loughran and Ritter (1995) posit that long run stock performance is explained by the underreaction hypothesis. Other studies that also allude to this explanation include Bilinski and Strong (2009), Spiess and Affleck-Graves (1995) and Womack (1996). All of this evidence suggests that long run performance reflects investor partial processing of information signaled by managers at announcement date with partial effect on prices. The remaining effects of the information accounts for the long run lower returns. Full signaling effects take time to be totally absorbed in the security prices. ¹⁵

-

¹⁵ Bilinski and Strong (2012) hold that the correction last for 17 months as opposed to 5 year period reported by Loughran and Ritter (1995) and Spiess and Affleck-Graves (1995) and only 6 months by Womack (1996). Ikenberry et al. (1995) find substantial post-repurchase abnormal four year buy and hold returns of 12.2%

2.4.2. Rational Discount Rate

An alternative reason for the behaviour of post-issue stock performance is the rational discount rate. The level of investor rationality is evident in the long run since conservative attitude to new information preclude immediate stock price reaction. After the effects of underreaction are subsided the market reverts back to the position prior to the issue. M/B and size impact diminishes with the correction period so that stock returns reflect discount rate. Generally, reduced leverage is associated with low risk that translates into low stock returns. Eckbo et al (2000) find that reduced risk associated with equity issue and reduced leverage compared to size and valuation effects accounts for the underperformance. Discount rate indicates a lower post issue risk that leads to adverse post issue performance.

2.4.3. Investment-Based Explanation

Long run returns subsequent to equity offering, as documented widely in the literature, exhibit price discounts. According to the market timing explanation, investors bid down the price in response to the issue to neutralise the initial missvaluation. Investment-based theory offers similar argument but presents a different reason to the price discount that follows equity offering. In this alternative orientation, the response of the stock prices relates to the change in risk profile that follows equity issues. As more equity financing is made to the investing public managers convert growth options into real assets. Butler et al. (2005) argue that in a time series study the amount of net financing (real investment-based), not the composition of the capital (market timing), has a more predictive power of long run abnormal stock performance. Thus, the level of capital inflows from both equity and debt issues decreases firm's required rate of returns due to conversion of risky growth options into less risky real assets. Lyandres et al. (2008) contend that equity issue converts risky growth options into fixed assets that reduce risks.

2.4.4. Economic Perspective

The economic argument to stock returns states that in the event of increased equity issues or massive buybacks/debt issues, there is the interplay of demand and supply forces, all other things being equal. Leverage increasing transactions reduce the supply of equity whereas transactions that generate additional equity into the market increases supply. That is, post issue price is affected by the size of the offering at a given market. Given the demand curve, any equity issue at high valuations causes a downward pressure on prices (Scholes, 1972). This is because the increased supply creates surplus that forces the price down. Similarly, in a situation of constant demand for the shares of the firm, repurchase/debt generate shortage from the reduced supply that compels a consequent rise in the stock prices. However, in an efficient market change in the size of the offering should rule out any possibility of change in price in response to the financing decisions.

Studies that propagate post-issuance underperformance ascribe the effects to only equity offerings. Thus, firms issue equity at high price ex ante and record low returns ex post. This is fundamentally the underlying phenomenon upon which market timing hinges. But, it is difficult to concede that only equity issuance takes place at a particular security issuance period. Firms that issue equity may as well issue other kinds of security concurrently, for example bank loans, private equity, and public debt among others. Billet et al. (2011) reveal a serious omitted variable bias that characterise prior studies. Using US data between 1980 and 2005 on issues of IPO, SEO, public debt, public equity and bank loans, they highlight the potential impact of multiple issues on stock price performance. The abnormal returns from Fama-Macbeth, Fama and French with augmented Carhart as well as BHAR all yield similar results.

Consistent with prior studies such as Baker and Wurgler (2002), stock price returns fall subsequent to equity issuance. Specifically, they find that firms perform worse when they engage in multiple security issues. Issue of three securities earns -4.9% annual returns compared with issuance of four security types return of -16.9% per year. However, the reason for this trend is not well articulated. Whether that is a new dimension to market timing, pecking order as

well as the impact of specific characteristics is not clear. In effect, multiple security type issues are rife and this study provides a new insight into the stock price performance following financing decisions.

2.5. Testable Hypotheses

This section develops the testable hypotheses from the literature reviewed in the preceding sections. This section consists of four subsections related to each developed hypothesis that are tested in Chapter 6 of the thesis. Subsection 2.5.1 discusses the hypothesis about share issuance and internal financial constraints while subsection 2.5.2 explains the hypothesis about share issuance and external financial constraints. In subsection 2.5.3, the hypothesis that share issuance and financial constraints depends on firm size is developed. Finally, subsection 2.5.4 explains the hypothesis that share issuance decisions affect stock returns.

2.5.1. Share Issuance and Internal Financial Constraints

The extant literature provides strong support for equity issues that precede periods of stock overvaluation or prior stock price run-ups (Graham and Harvey, 2001; Baker and Wurgler, 2002). As explained in Section 2.2, equity issues depend on factors such as costs of adverse selection associated with equity issues (Myers and Majluf, 1984; Lucas and McDonald, 1990; Korajczyk et al., 1992). Again, the empirical evidence reveals that equity issues are associated with negative asymmetric information effects when they need external equity financing. Therefore, consistent with a pecking order theory, firms avoid this cost by utilising internal cash or issue debt when they require additional financing (Myers and Majluf, 1984; Shyam-Sunder and Myers, 1999; Lemmon and Zender, 2010). Baker and Wurgler (2002) explain that the effects of asymmetric information correlate with overvalued equity or mispricing. Thus, the issue of overvalued equity coincides with high asymmetric information effects. This implies even when firms face significantly high asymmetric information effects associated with equity issues, the motivation to issue overvalued equity for market price exploitation

overrules the convention that firms issue equity during periods of low information asymmetry.

According to the trade-off theory, firms attempt to maintain a balanced capital structure based on a target debt-equity ratio (Shyam-Sunder and Myers, 1999; Hovakimian et al., 2001). Financial deficit is crucial in determining when firms issue securities. Given the high costs of adverse selection equity issues only enable firms to balance its excessive leverage position. Thus, at high leverage any additional debt issues will potentially drag the firm into costly financial distress and bankruptcy, even though debt issues produce tax advantage. This means the combination of potential discounting of the stock price and relatively high transaction costs does not encourage firms to undertake equity offerings. At high stock prices, firms are motivated to make equity offerings, but they must also possess the financial capacity to pay the high ancillary costs of issue. Opportunistic managers that issue/repurchase securities on the basis of the stock price must have the financial strength to also deal with costs (especially for equity issues) incidental to accessing the external capital market.

The direct costs such as gross fees paid to investment banks in conducting seasoned equity offering could be considerably high (Butler, Grullon, and Weston, 2005). For example, Lee et al. (1996) find that the average firm pays around 7% of the total proceeds to raise capital through an SEO. If firms only time their issues, then it is expected that they have substantial financial slack, especially for equity issuers. Without adequate investment opportunities cash-strapped entities are less inclined to access equity financing. Growth options dominate the demand for cash, and hence the need to obtain external financing when there is internal deficit. DeAngelo et al. (2010) stress that it is only the need for "near term" cash for future investments that motivates equity issue at high valuations. Firms that require cash for identified projects contact the outside investing public for financing without regard for market timing activities. It presupposes that rate of investment should increase with equity issues.

Moreover, Fama and French (2005) and Dong et al. (2012) contend that equity issuing firms are actually cash-rich. These firms do not necessarily require external financing. So following the argument by Baker and Wurgler (2002), and the evidence from Fama and French (2005) and Dong et al. (2012) and the high costs of equity issuance, one can conclude that timing the issue of overvalued equity should be more pronounced for cash-rich firms than it would be for cash-poor ones. Thus, for market timing to hold, equity issues must track financial surplus. This means managers that time the market should not be financially constrained. The only motivation for any equity offering should be occasioned by high stock prices with substantial financial slack such that high transaction costs do not impair the ability to conduct the issue. Firms are typically opportunistic and will issue equity/repurchase at high/low valuations. Fama and French (2005) highlight this proposition and find that equity issuers are not constrained with cash rather these entities hold significant cash balances prior to the issue. While this lends credence to market timing theory, it weakens the position of the pecking order phenomenon. Issue of overvalued equity under the market timing theory undermines the effects of pecking order theory. Stated differently, financial constraints do not curtail equity issue when firms with timing prospects issue equity. Therefore, equity issuers are less financially constrained and M/B suppresses factors that support pecking order (Dong et al, 2012).

The hypothesis following the above discussions is stated as follows:

 H_1 : There is high probability for less internally financially constrained firms to issue overvalued equity

2.5.2. Share Issuance and External Financial Constraints

According to McLean and Zhao (2014), internal financing constraints affect the demand for external financing. However, the supply of the funding will depend on the overall external financing conditions. Investors buy shares when they are more liquid such that they can freely trade them on the secondary market. Thus, market liquidity influences the supply or the availability of external funding, regardless of whether firms are internally financially

unconstrained to pay the transaction costs. External financing constraints – the degree of market liquidity or economic conditions (e.g. financial crisis) - not only affect the supply of funding but it also increases the transactions costs associated with equity issuance. Speculative securities such as equity attract significant sentiments that affect the supply and costs of equity issuance (Stein, 1996; Baker and Wurgler, 2006, 2007; Baker, 2009). Therefore, firms' market timing of equity issues is dependent on the external financing constraints as well. The degree of internal financial constraints should also affect the likelihood that firm issues overvalued equity during periods of high external financial constraints.

The financial crisis of 2008 and the European debt crisis of 2011 have both emphasised the effects of external financing constraints on corporate decisions. During crisis periods, market liquidity dampens and increases the costs of raising external equity financing (Corwin, 2003; Butler et al., 2005; McLean and Zhao, 2014). For instance, Naes et al. (2011) show that U.S. stock market liquidity is countercyclical, and argues that during recessions there is a "flight to quality" in equities. This is because during recessions investors sell stocks, especially riskier stocks, in favour of safer securities. Therefore, external financial constraints should limit equity market timing. Studies such as Korajczyk and Levy (2003) and Alti (2006) find equity-issuing effects of market conditions. Specifically, firms find it cheap to issue equity when there are favourable overall market conditions. Favourable market conditions enhance supply of funding and also reduce the costs of obtaining the external financing. Thus, firms are more likely to issue overvalued equity during periods of favourable market conditions. Less external financial constraints and high market liquidity should increase the probability of issuing overvalued equity. The likelihood of a firm issuing overvalued equity during periods financial crisis should be higher for firms with low internal financial constraints than for those with high internal financial constraints.

 H_2 : Firms are more likely to issue overvalued equity when external financial constraints are low (non-financial crisis period or periods of high market liquidity)

 H_3 : The probability of issuing overvalued equity during periods of high external financial constraints increases for firms with low internal financially constraints than those with high internal financial constraints

2.5.3. Share Issuance and Stock Returns

In markets where security issues depend on prevailing conditions, investors are expected to act in response to issuers' decisions. Generally, capital market transactions reflect favourable information either to the seller or the buyer. There is always the lack of symmetry between the two parties in terms of the information that encourage the party to act in the market. Therefore, rational investors discount stock prices to indicate situations of asymmetric information. Market timing theory asserts that managers issue equity at high stock prices and repurchase at low valuations. If investors are rational, then the immediate response following announcement of equity (at high prices) and repurchase/debt issue (at low prices) is to bid down prices or up respectively. Equity issues are associated with significant stock underperformance (Fama and French, 1993; Loughran and Ritter, 1995, 1997; Baker and Wurgler, 2002; Dong et al. 2012).

Several theories underlie the gradual stock price reaction to security issuance decisions. Loughran and Ritter (1995) propose that investors' reaction take time to reflect in the stock price. Largely, irrational attitude of investors ensure that stock prices do not immediately fall following equity and rise with repurchase/debt issues. Opportunistic managers, therefore, take advantage of high and low valuations to make equity offering and repurchase/debt issues respectively. The combination of high adverse selection costs and transaction costs suggest that managers only exploit investors when they issue equity. According to the underreaction theory, overvalued/undervalued firms that issue equity/repurchase record low/high long run returns. The degree of financial constraints has significant impact on the level of stock returns. Lamont et al. (2001) find evidence that the stock returns of financially constrained firms move together because they are subject to common shocks. But more

importantly, their findings are consistent with low stock returns associated with constrained firms.

Financial constraints and the consequence on investment decisions and other signalling financial policies such as dividend payments should reflect in the stock returns. Firms that are limited in their ability to undertake investment opportunities and pay dividends will have negative stock market reactions compared to those that undertake investments and pay dividends. Using KZ index as internal financial constraints proxy, Lamont et al (2001) and Chan et al (2010) find that stock returns of internally financially constrained firms move together indicating that they have common shocks. On the other hand, they find that unconstrained firms earn significantly higher stock returns than constrained firms. The degree to which issuing firms underperform depends on the level of internal financial constraints. Consistent with Lamont et al. (2001), internally financially unconstrained firms should report less negative stock returns than more constrained firms. Also, faced with high external financial constraints and high costs of issuing equity, the level of stock price underperformance should also be less negative for unconstrained firms than constrained firms. The hypotheses are stated as follows:"

H₄: Issuance of overvalued equity is associated with more negative short-term abnormal returns, especially for firms with high internal financial constraints

 H_5 : Issuance of overvalued equity is associated with more negative short-term abnormal returns, especially for firms with high internal financial constraints and during periods of high external financial constraints

2.6. Conclusion

This chapter has reviewed the theoretical as well as the empirical evidence of share issuance decisions. The aim of the chapter was to establish that not only mispricing but also the level of both internal and external financial constraints affects the share issuance decisions of firms. The chapter also discussed the

theories that explain stock returns associated with equity issues. Empirical evidence suggests that equity issues are associated with negative announcement date abnormal returns. This finding fails to account for the level of financial constraints. Hence, firms' decisions relating to equity issuance are likely to be influenced by mispricing as well as the level of financial constraints.

Chapter 3

LITERATURE REVIEW: SHARE ISSUANCE METHODS

3.1. Introduction

Firms in the UK are allowed to issue equity through rights issues, open offers, and private placement. This apparent choice of issuance method was affected by the deregulation in 1986. In deciding on the method, firms take into account several factors including the level of information asymmetry, the ownership structure of the firm, and whether equity issue can be timed. In view of that, the market also responds differently to the issuance methods. This chapter provides overview of the equity issuance method in the UK. It also reviews the theories that underlie the choice of each method and provides empirical evidence of the stock returns associated with each method. The chapter also develops testable hypotheses that set out the focus of the empirical chapter 7 of the thesis.

The chapter has five sections. Section 3.2 presents the general overview of share issuance methods in the UK. It discusses the flotation methods, UK issues, the costs and demise of rights issues. Section 3.3 explains the theories that underlie the choice of issuance method. This section examines firm characteristics that determine whether a firm selects rights issues, open offers, or private placement in issuing equity. Section 3.4 provides empirical evidence of stock returns as well as the operating performance associated with each equity issuance method. Section 3.5 develops the testable hypotheses and section 3.6 concludes the chapter.

3.2. Overview of Share Issuance Methods

Equity issues can be undertaken through a number of methods. The application of a particular method is influenced by a number of firm factors and managerial motivations for the transaction. For instance, a small firm that prefers a more concentrated ownership structure would opt for rights issues approach to avoid ownership dilution. Without issuing shares to new investors, the existing shareholders are able to control the affairs of the company without much external influence and monitoring. Similarly, the cost of conducting

equity issues affects the choice, all other things being equal. Thus, shareholder characteristics, direct cost of flotation and asymmetric information, among other factors, determine the equity issuance approach adopted by firms. This section discusses the flotation methods in subsection 3.2.1, UK issues in 3.2.2, the cost of rights issues in 3.2.3, and the demise of rights issues in 3.2.4.

3.2.1. Flotation Methods

Rights issues: Issues through rights issues, otherwise known as uninsured rights issues, ensure that new shares are offered to existing shareholders for subscription on pro rata basis. Shares offered to current shareholders attract price discount to the market price to encourage take-up. This implies that the shareholders to whom the shares are offered have the right to forfeit or sell their rights subsequent to taking up the offer. Rights issues are organised on pro rata basis to assure that all shareholders retain the degree of control they enjoyed prior to the offer. Therefore, shareholders who opt to reject the offer tend to be worse off after the transaction because they lose percentage of control and subsequent dividends payments. Again, any forfeited shares cannot be issued to external investors with firm losing on the amount that would accrue had they been issued to outsiders. Given that firms act in the interest of current shareholders, rights issues are more likely offered when the shares are undervalued (Capstaff and Fletcher, 2011).

Open offers: Open offers: Open offer is an offer of new shares usually at a discount to existing shareholders on a pre-emptive basis. The usual practice is to invite shareholders to apply for any number of shares. The terms of an open offer are similar to rights issues except that offers that are not taken up cannot be sold by the shareholders (Armitage, 2000). Unlike a rights issue, application forms are used (instead of *provisional allotment letters*) which cannot be traded *nil paid* and no arrangements are made for the sale of shares not taken up by shareholders. Thus, entitlements from an open offer are not tradable and therefore an open offer is only available to existing shareholders. The principle of *pre-emption* is usually achieved by providing for a guaranteed "minimum entitlement". If any shareholder declines his minimal entitlement, the excess shares are allocated to those applying for shares in excess of their minimum

entitlements. If excess applications cannot be met in full, applications are scaled down.

Private Placement (Placings): An issue where the entire offer is placed with one or more investors without issuing pro rata to existing shareholders is private placement. Firms normally target these few investors. The only distinguishing feature of private placement from both rights issues and open offers is that private placement is not made on pro rata basis. However, it is made under similar terms and conditions as the rights issues.

3.2.2. UK issues

Factors that motivate a choice of flotation method are varied and wide. Rights issues, until deregulation in 1986, were the only method used in the UK. Subsequent to the deregulation, firms had the option to issue using rights issues, placement, and open offers among others. Korteweg and Renneboog (2002) and Barnes and Walker (2006) both find open offer and placing firms are smaller with higher asymmetric information than rights issuers. The high asymmetric information is attributed to the high proportion of director ownership in such firms. However, placings that involve new and external investors allow the firm to divulge information to the market. Thus, placings are associated with information releases that reduce information asymmetry. This information release improves liquidity of the shares and reduces issue costs for the shares. Quality firms would also choose placement over rights issues to attract new investors and explore other financing avenues. The above features, coupled with the certification credibility, contribute to a positive price reaction to placement with negative price effects to rights issues (Slovin et al, 2000; Armitage and Snell, 2004; and Barnes and Walker, 2006; Levis, 1995).

3.2.3. Costs of Issue Approach

An important element in the choice of issuance method is the cost associated with each method. Earlier empirical evidence in the US documents that uninsured right issues are the cheapest flotation method. Smith (1977) finds that the average cost of conducting rights issues is 1.8% and 0.5% of gross

proceeds for industrial and utilities, respectively. Firm commitment costs 6.1% and 4.2% of gross proceeds respectively. These costs comprise underwriting fees and other direct expenses. The discount provided on the offer encourages subscription and the amount earmarked to raise from the issue. Firms that choose rights offerings maintain the number of shareholders and consequently the dividend. For anti-dilution reasons, rights offerings do not diminish earnings per share. Rights issues are much cheaper than placement in the UK yet firms prefer private placement to rights issues. Recent equity issues show that a surge in private placement and open offers at the expense of rights issues. Factors that account for this changing trend are unrelated to the costs of issuance.

3.2.4. Demise of Rights Issues

The demise of rights issues relates to the changing preference for placings and open offers since early 1990s following the deregulation of 1986. Changing firm characteristics explain in part the apparent lack of preference for rights issues. For instance, Eckbo and Masulis (1992) attribute the demise of rights issues in the US during 1980 to increased use of dividend reinvestment plans (DRIPs). DRIPs allow shareholders could choose to receive common stock in lieu of cash dividends. However, the introduction of DRIPs does not necessary explain the disappearance of rights issues for firm commitment. A number of factors explain the demise of rights issues or the increased preference for private placement in the UK.

Firstly, capital gains taxes affect the choice of rights issues (Smith, 1977). Since rights issues are conducted to benefit equally all the existing shareholders, those who do not subscribe to their rights become worse off. To forestall this anomaly, shareholders can elect to sell their rights or subscribe and sell them subsequently. This ensures that the value of the shareholding is not reduced and also takes advantage of the price discount. Smith finds that the incidence of capital gains tax on the discount discourages substantial discounts making rights offerings unpopular. Further, he asserts that investment banks effectively influence firms to adopt firm commitment regardless of the higher fees. Secondly, rights issues are made to only existing shareholders unless the

rights are subsequently sold to outside investors. Given this phenomenon, price effects and trading in shares are limited. The lack of external or new investors' involvement in the rights issues lowers the liquidity of the shares (Kothare, 1997).

Another reason for the demise is that shareholders that opt to sell their rights bear the responsibility and the cost of selling them. Unlike underwriting where the sale of shares rests with underwriters, shareholders can only sell on organised exchanges that require dealers and brokers. Hansen (1989) finds a high dealer and brokerage fees associated with the sale. Lastly, rights issues that are unsubscribed attract negative market reactions. Eckbo and Masulis (1992) find that low participation in the issue increases asymmetric information about the firm. Thus, market response to rights issues is more negative than firm commitment. It implies announcement price run-up is low for rights issues and high for firm commitment issues. Barnes and Walker (2006) assert that placements are preferred for firms with high asymmetric information.

Other empirical evidence attributes the demise of rights issues to the need for block selling of shares. Armitage (2007) finds that the price discount of between 15% and 20% of the market price is least attractive to maintain the preference for rights issues. Nevertheless, the demise of rights issues reflects the notion that placings better facilitate block selling than rights issues (Armitage, 1998; 2010). Again, placement saves time and money compared with rights issue and open offers. Placement also invites new investors and provides insurance as to the certainty of the offer taken up. Substantial resources are committed to placement to assure the right value of the firm that reduces adverse selection effects. Therefore, high quality firms prefer placement to rights issues.

Again, Eckbo and Masulis (1992) further describe adverse selection costs as a function of current shareholder take-up of rights issues. By inference, high adverse selection costs are associated with rights issues with low take-up whereas high take-up effectively offset any asymmetric information effects. Thus, in the definition of market reaction to rights issues, Eckbo and Masulis (1992) rely on the degree of current shareholder take-up of the offering.

Absolute take-up of the issue signifies quality information that eventually translates into positive market reaction. This intuition reflects the tendency for firms to advance shareholder value by not inviting new investors to participate in the distribution of the wealth.

However, the incidental low take-up by current shareholders could force managers to issue to new shareholders. Undervalued firms are successful with rights issues whereas firms that are relatively overvalued prefer firm commitment. The above suggests a more negative market reaction with firm commitment and a near zero stock returns associated with rights issues. On the average current shareholders are more informed about the firm than the outside/prospective investors thus by participating in new shares convey quality information to the market. Eckbo and Masulis (1992) assert that subscription pre-commitments reveal the success of the issues and the perception of the existing shareholders about the quality of the firm. This effect potentially attenuates negative reaction occasioned by stock issues.

3.3. Theories of Share Issuance Choice

In addition to the theories that affect equity issues as discussed in Chapter 2, this section discusses specific theories that explain the choice of issuance method. The section is made up of four subsections discussing different theories. Subsection 3.2.1 explains the quality signalling theory, 3.2.2 the monitoring and certification hypothesis, 3.2.3 he managerial entrenchment hypothesis, and 3.2.4 the investor relationship phenomenon.

3.3.1. Quality Signalling

In line with the asymmetric information hypothesis¹⁶, firms attempt to signal their quality through equity issuing transactions. Factors which define quality include, among others, lower discount, low information asymmetry, higher liquidity, undervaluation. In particular, the extant evidence suggests that price discount is an important gauge of firm quality (Hertzel and Smith, 1993; Slovin et al, 2000; Barclay et al, 2007; Balachandran et al. 2008). Quality firms are

_

¹⁶ Refer to Chapter 2 for discussion on information asymmetry hypothesis

more likely to preserve ownership control by issuing equity to existing shareholders (Balachandran et al. 2013). On the one hand, quality firms will issue equity to existing shareholders because they have no motivation to reduce shareholder wealth. It also suggests that it allows firms to retain control of the quality firm and not dilute ownership concentration. However, firms use private placement to signal undervaluation through investor certification of the issue (Hertzel and Smith, 1993). This evidence indicates that for low quality firms, private placement allows them to obtain approval for the issue in exchange for significantly large price discounts.

In the study of equity offering methods using a sample of 967 UK equity issues from 1996 to 2005, Balachandran et al. (2013) assert that quality firms are more likely to issue equity to existing shareholders. These firms have lower idiosyncratic risk, higher liquidity, and lower information asymmetry. Since private placements constitute few investors who can be targeted, low quality firms choose private placement. Larger price discount compensates for the costs of obtaining information about the firms and the issue. Moreover, the firms in turn obtain credible certification for the issue that in turn dampens adverse effects of perceived adverse selection associated with issuance to external investors. The above differences have significant implications for subsequent stock price reactions. Their study emphasises that, all else equal, firms will more likely signal their quality through private placement than rights issues when the price discount is low. However, for larger discount, shareholder wealth is more protected using rights issues.

3.3.2. Monitoring and Certification Hypothesis

Two related theories that underlie share issuance choice are monitoring and certification hypothesis. Monitoring means the ability and capacity with which investors assert their authority and control over corporate strategies and preserve firm resources. A well monitored firm pressures management to act in the overall interest of the shareholders. It ultimately reduces the agency costs associated with management of corporate entities. Wruck (1989) contends that active investors purchase private placements to increase managerial monitoring which ensures that corporate resources are utilised more efficiently and also

increase the probability of value-increasing takeovers. Wruck (1989) further asserts that the monitoring effects is different between private and public equity issues because public issues are made to the general public with purchasers not necessarily willing to actively involve themselves in the management of the firm. On the other hand, private placement mitigates the asymmetric information problem because private placement allows managers to negotiate with the purchasers to address any concerns about the firm.

Consistent with the information asymmetry hypothesis, private placement promotes credible investor certification of the equity issue. Hertzel and Smith (1993) contend that informed investors who purchase the equity in private placement certify the credibility of the market value of the issue. The implication of the certification is that it dispels market suspicion about firms' attempts to issue overvalued equity and also decreases the adverse selection associated with information asymmetry. Hertzel and Smith (1993) study the certification hypothesis using a sample of 106 private placement announcements from January 1, 1980 through May 31, 1987. The substantial discounts associated with private placement compensates for the costs incurred by investors in obtaining information about the equity issue and the firm. Thus, the discounts represent the price paid by the firm in exchange for the value certification. By certification, information asymmetry problem is resolved and it also highlights the equity undervaluation.

3.3.3. Managerial Entrenchment

The activities and interests of corporate managers could also influence the choice of issuance. Corporate managers are under huge pressure to meet shareholder objectives of maintaining shareholder wealth and creating value through investments. These managers are expected to manage the financial resources in a way consistent with the objectives of the firm. Through monitoring, labour market pressures, among other factors, managers are constrained from making value-destroying activities and paying substantial perquisites at the expense of the firm. Managerial entrenchment is the attempt by managers to make themselves valuable to shareholders and costly to replace (Shleifer and Vishny, 1989). Entrenched managers have control over corporate

resources and strategy. All else equal, entrenched managers are more likely to act in a way that is inconsistent with shareholder value creation. This effect manifests through excessive perquisite payments and the investments in value-destroying projects that undermines the long-term growth potential of the firm.

In the context of share issuance, private placement attracts new investors to encourage monitoring (Wruck, 1989). However, Barclay et al. (2007) assert that private placements are often made to passive investors. By placing equity with passive investors, managers undermine the effects of monitoring. Rather, the managers are able to strengthen their control over the firm. Barclay et al (2007) also stress that initial positive stock returns may suggest evidence of monitoring and certification. Over the long term, these purchasers of private placement do not actively involve themselves in the management of the firm. Thus, the passive investors do not become directors, CEOs, and do not acquire the firms or act in any capacity that demonstrates their active involvement in the management of the firm. The inactivity of the investors in the management of the firm solidifies managers' control of the firm. Managerial entrenchment hypothesis is consistent with managerial behaviour and the post-placement activities of the firm.

Barclay et al find significant price discount for private placements. Of the total 594 issues between 1979 and 1997, only 12% were active placements. The criterion for designating as active and passive is crude with serious limitations. For instance, the kind of interaction that would qualify placement as active is arbitrary. They further assert that the discount compensates for the subsequent stock price decline. Passive placements means management chooses uninformed investors who do not provide the monitoring (Wruck, 1989) and certification (Hertzel and Smith, 1993) for the issue. Thus, the lack of monitoring and active involvement of the new investors in the activities of the firm allows management to entrench itself. Therefore, management entrenchment motivation influences the choice of passive investors, accounts for the price discount and the long-term negative post-issue stock returns associated with private placement.

3.3.4. Private Placement and Investor Relationship

Another feature of private placement is the tendency to target few investors without selling the shares to the wider investor public. The absence of a wide public demand is beneficial to existing shareholders in a sense as to preserve their ownership interest without significant dilution. It is also consistent with targeting investors who provide certification (Hertzel and Smith, 1993) and adequate monitoring (Wruck, 1989) to justify the value of the share issues transaction. The effects of both certification and monitoring constitute positive information about the firm that enhance the credibility of the issue and increase firm value. However, a large number of private placements are associated with significant issue discount and negative post-issue stock returns (Armitage, 2007; Wruck and Wu, 2009; Armitage et al. 2014). In effect, this brings into focus the ability of the active investors that participate in the issue to determine the true value of the firm at the time of the issuance decision.

Wruck and Wu (2009) note that in the process of the private placements, issuing firms form relationships with investors. When new relationships are created it evokes positive signal to assure the market of adequate monitoring and credible certification that accompany the transaction. The positive announcement stock returns support this hypothesis. They study a total of 1976 private placements spanning years 1980 to 1999 of US public companies. Placements where new relationships are formed generate significant CAR (-3, 0) of 3.15% compared to insignificant returns of 0.63% for no relationships. They argue that subsequent improvement in operating performance and long run stock returns are the result of strong governance and increased monitoring occasioned by the new relationships. By extension, the subsequent performance of the firm depends on the degree and effectiveness of the governance invoked by the private placement transaction. Thus, private placement accompanied by good governance is anticipated to generate improved performance than an otherwise private placement that fails to generate improved governance. Also, governance should also address the extent of the new investors' involvement in the corporate affairs of the firm.

3.4. Stock Price Reactions to Share Issuance Choice

The stock price and operating performance associated with equity issues are discussed in this section. Subsection 3.4.1 provides evidence of short-run stock returns associated with each equity issuance method whereas 3.4.2 offers insights into the long-run stock performance of the different equity issuance methods. Finally, 3.4.3 focuses on the level of operating performance to each method.

3.4.1. Initial Abnormal Stock Returns

Consistent with theories of share issuance (Slovin et al., 2000; Barnes and Walker, 2006; Baker and Wurgler, 2002; Armitage, 2010; Dong et al. 2012), firms choose equity issuance to exploit either overvaluation or in response to low information asymmetry. For example, market timing of equity issues is associated with overvaluation. If different valuation can be attributed to each equity issuance method, then firms can time their issuance choice on the basis of the overvaluation. Stated differently, different level of information asymmetry should also explain the choice of issuance method. This approach follows that initial stock returns associated with equity issues should be different for each issuance method. Empirical literature provides significant evidence of the differential stock price reactions dependent on the equity issuance method. Thus, if firms choose issuance method on the basis of overvaluation, market reactions should be negative for the issuance method that is associated with overvaluation. Moreover, for method with significant information asymmetry, market reactions would be negative.

Seasoned equity offerings generally attract negative market reactions due to the inherent asymmetric information costs associated with them. Empirical evidence in the US is unanimous. For instance, Hansen (1988) finds AAR underwritten rights -2.61% for industrials and -1.21% for utilities whereas Eckbo and Masulis (1992) report AAR -1.39% for industrials and 0.23% for utilities. Similarly, for UK Armitage (2007) find that renounced rights issues are sold at significant discounts of about 8% but they argue this is more to do

62

¹⁷ See chapter 2 for overview of theories that explain share issuance decisions.

with issuer value uncertainty than overvaluation concerns. The extant literature asserts that rights issues attract low costs. However, as Armitage reports, when the rights issues are subsequently renounced and sold at discount to new investors the cost could potentially exceed that which would otherwise be incurred for open offers and private placement. On this evidence, take up of rights issues define the success of such transactions.

Firms wishing to signal quality, disperse ownership and reduce asymmetric information between insiders and outsiders adopted Placings. Moreover, placings effectively enhance ownership dispersion, enhance external monitoring and promote corporate control. Again, given that an underwriter buys the entire offerings and makes gains from bid-ask spread, thorough investigation about the value of the firm would be conducted before an underwriter accepts the placings. Slovin et al (2000) and Barnes and Walker (2006) contend, in consonance with Myers and Majluf's (1984) information asymmetry theory, that underwriting certification, ownership dispersion, increased external monitoring, and corporate control embodied in placings eliminate/reduce asymmetric information effects associated with equity financing.

Another explanation to the stock price reactions to equity issuance method is mispricing. Stock price underperformance associated with SEO in the US and UK has been attributed to the effects of market timing. All else equal, regardless of equity issuance method, post-issue abnormal stock returns is likely to be negative due to possible overvaluation; but the more firms are able to exploit the overvaluation through a specific issue approach, the more negative the market reactions to stock returns. The argument that managers aim to improve shareholder value supports the idea that they will be unwilling to issue overvalued equity to existing shareholders. Following this reasoning, firms would be more likely to issue overvalued stock to new investors due to information asymmetry and these investors can only obtain information by incurring substantial costs. This is similar to a case where ownership is concentrated at family level. With aversion to control dilution and increased monitoring following private placement, such firms would prefer rights

offerings (Cronqvist and Nilsson, 2005). Burch et al (2004) and Capstaff and Fletcher (2011) assert that firms are less likely to time their equity through rights issues than placings. Capstaff and Fletcher (2011) and Balachandran et al (2013) assert that rights issues firms are not of lower quality compared to placings and open offer firms.

3.4.2. Long Run Stock Returns

Theories such as asymmetric information, monitoring and certification, managerial entrenchment and mispricing underlie the long run price reactions associated with the choice of equity issuance methods. Following the asymmetric information hypothesis, long run performance of rights offerings and placings are akin to SEO. Several empirical studies confirm that firms that conduct both rights issues and placings record substantial negative abnormal returns. This evidence is consistent with findings in the US about market reactions subsequent to seasoned equity offerings. Prior to the deregulation, firms could exploit overvaluation, as they deemed profitable using rights offerings. With the introduction of different approaches to issuing equity, firms can distinguish themselves from others and communicate information to investors using a more appropriate technique. Ngatuni et al. (2007) review the differential long run performance of equity offering for rights issues and open offers at different time periods.

In the period between 1986 and 1995, rights issues were the dominant issue approach. Open offers and placings had been introduced but were yet to be more pronounced in the UK. Using a sample of 818 rights issues spanning a period from 1986 to 1995, Ngatuni, et al (2007) report a 5-year average BHAR of -41.80%. It is both statistically and economically significant. When disintegrated between different time periods, rights issues made between 1991 and 1995 recorded a reduction in the abnormal returns. Generally, the intuition is consistent with evidence from the US (e.g. Spiess and Affleck-Graves, 1995; Loughran and Ritter, 1995) that find negative abnormal returns following SEO. As open offers increases with rights issues gradually less preferred, the negative reaction to rights issues diminishes and open offers earn positive abnormal returns.

Also, using a sample of 670 UK rights issues and 391 placings from 1989 to 1997, Ho (2005) and Abhyankar and Ho (2007) conduct event study that focuses on the relative long run performance of equal and value-weighted portfolios. It would have been interesting to note the performance of both rights issues and placings given that prior studies have enumerated many different features that potentially yield different market reactions. They find significant negative abnormal returns over a three-year period. Specifically, rights issues earn equal-weighted BHAR and value-weighted BHAR which are respectively -19.50% and -8.53%. While consistent with other UK evidence, there is stark difference in terms of magnitude of the returns. For example, Levis (2005) find three year BHAR of -30.20% in UK.

The equity issuance decisions of public firms also determine the long-term stock performance. Public firms can issue equity securities to the public for subscription through firm commitment method of flotation. While this approach is commonly adopted in the US, a number of public firms also place equity private with investors. These issues are limited to specific investor or investor group without public involvement. If equity market timing hypothesis causes the market to penalize firms that conduct public issues due to perceived exploitation of overvaluation, then private placings of equity should at least cause limited negative market reactions. Hertzel et al. (2002) study 619 private placements over the period 1980 to 1996 for firms listed on the NYSE/AMEX/Nasdaq exchanges. They estimate both the stock returns and operating performance after the private placement to check for a consistent behavior of stock returns and firm operating performance.

Hertzel et al. (2002) generate positive stock price reactions to announcement of private equity of 2.4% in variant to the negative returns associated with public issues. The long run returns, however, compares with the negative returns recorded for public issues. For example, they find mean there-year buy-and-hold abnormal return for size-and-industry matched control firms of -38.18%. Spiess and Affleck-Graves (1995) report abnormal returns relative to size-and-industry-matched benchmark of about -22.84% for public issues of equity. This evidence contradicts the underreaction hypothesis that has been advanced in

explaining the stock market reactions following the announcement of equity issues. In effect, the long run returns lend support to investor over-optimism associated with 'windows of opportunity' at the announcement of both private and public equity issues. Again, and more strikingly, they find that private issues follow period of relatively poor operating performance that divorces the predictions of investor over-extrapolation of firm performance to the announcement of the issuance decisions.

According to the managerial entrenchment hypothesis, the long run stock returns associated with private placement differs from the initial stock returns. Barclay et al. (2007) emphasise that whereas the initial positive abnormal returns reflects the effects of monitoring and certification, over the long term the effects of managerial entrenchment become apparent. Firms make private placement to passive investors who do not intend to engage in the management of the firm. The inactivity of the new investors enhances the managerial control of the existing management on corporate resources, increases perquisite payments and ultimately decreases shareholder wealth. Thus, private placement promotes managerial entrenchment and hence results in unfavourable long-term stock price reactions.

Capstaff and Fletcher (2011) also contend that UK managers are less motivated to use rights offerings to exploit equity overvaluation. Rights issues of equity made to existing shareholders present limited scope for information asymmetry and mispricing and the potential for exploitation. Consistent with the quality-signalling hypothesis, rights issues are less overvalued and the market least anticipates firms to time their issues through rights offerings. This incidence of mispricing explains the long-term stock performance. Capstaff and Fletcher (2011) assert that rights offering firms perform better in the long run than other alternative issuance methods. The study of 772 equity issues from January 1996 to December 2007 computes cumulative abnormal returns (CAR), buyand-hold abnormal returns (BHAR) as well as calendar time portfolio returns for equity.

Rights issuing firms perform significantly better over the long term than placings. With similarly positive prior issuance performance for all methods, the subsequent underperformance is an indication of an attempt by firms to time their issues. The firm quality argument is consistent with rights issues because firms are less likely to time issues to existing shareholders with low information asymmetry. In line with the asymmetric information hypothesis, and consistent with equity market timing, placings are more likely to be timed even though recent evidence by Dionysiou (2015) suggests otherwise. In fact, she finds that placings are associated with high earnings quality and growth opportunities at the announcements. The effects of mispricing and information asymmetry manifests in the long term and define the stock price underperformance. This evidence supports the managerial entrenchment phenomenon (Barclay et al. 2007) and cast doubts on the findings by Slovin et al. (2000) and Barnes and Walker (2006). Again, these findings suggest that any positive stock returns associated with placings can only be transient; the monitoring and certification hypotheses explain these short-term returns (Wruck, 1989; Hertzel and Smith, 1993). Tables 3.1 and 3.2 summarise the empirical evidence of long run stock returns associated with equity issues.

Table 3.1: Summary of Empirical Evidence of US Long Run Abnormal Returns Post Equity Issues								
Authors	Period	Size	Post-issue Period	Returns				
Spiess and Affleck-Graves (1995)	1975-1989	1116	3-Year CAR	-18.67%				
Loghran and Ritter (1995)	1970-1990	3702	3-Year BHAR	-33.0%				
Brav et al. (2000)	1975-1992	3775	5-Year BHAR	-24.0%				
Eckbo et al. (2000)	1964-1995	3315	5-Year BHAR	-23.20%				

Table 3.2: Summary of Empirical Evidence of UK Long Run Abnormal Returns Post Equity Issues							
Authors	Period	Equity Method	Sample Size	Event Window	Returns		
Levis (1995)	1980-1988	Rights Issues	158	36-month BHAR	30.20%		
Ho (2005)	1989-1997	Rights Issues	562	36-month BHAR	-19.92%		
		Placings	252	36-month BHAR	-25.38%		
Abhyankar and Ho (2007)	1989-1997	Rights Issues	670	36-month BHAR	-19.92%		
Ngatuni et al. (2007)	1990-1995	Open Offers	132	36-month BHAR	15.60%		
	1991-1995	Rights Issues	421	36-month BHAR	-25.75%		
	1986-1990	Rights Issues	397	36-month BHAR	-34.45%		
Iqbal et al. (2009)	1991-1995	Open Offers	181	36-month BHAR	-21.65%		
Andrikopoulos (2009)	1988-1998	Rights Issues	1542	36-month BHAR	-24.39%		
Capstaff and Fletcher (2011)	1996-2007	Rights Offerings	168	36-month BHAR	-7.1% ^a		
		Placings	185	36-month BHAR	-15.21%		
		Open Offers	95	36-month BHAR	-2.62% ^a		
		Placings plus	76	36-month BHAR	-25.50%		
		Open Offers					
Iqbal et al. (2013)	1988-2008	Rights Issues	1146		-30.67%		

^a means the value is statistically insignificant

3.4.3. Operating Performance

Theoretically, the stock returns should correlate positively with operating performance. Poor performing firms are expected to have depressed stock prices. Thus, operating performance explains the stock performance of the firm. A number of studies have shown the stock market response to the issue of seasoned equity. The evidence is unanimous across several economic environments that there is a negative stock returns following equity issuance transactions. However, our understanding of the level of operating performance at the time and subsequent to equity issues is limited. Anecdotally, negative stock returns are expected to reflect deteriorating operating performance of the firms. Thus, from market timing standpoint, high operating performance should reflect the stock price run-up and high market-to-book ratio of equity issuing firms. Similarly, as stock price declines after the equity issuance transactions, operating performance is expected to equally deteriorate to give support to the underlying reasons for the adverse market reactions. In the context of issuance method, it can be argued that issuance method that mimics the market timing or information asymmetry theory should demonstrate similar reactions and effects as the equity offerings. Thus, high market-to-book ratio, stock price run-up and high operating performance should precede the announcement of the equity issues.

Empirical research covering this thesis include but not limited to Healy and Palepu (1990) who find no earnings effects following equity issues. They contend that SEO do not convey new information about earnings. However, Loughran and Ritter (1997) demonstrate that operating performance of equity issuing firms is significantly high prior to the issuance date. These firms subsequently report deteriorating performance that reflects the decline in stock returns. Teoh et al. (1998) as well as Rangan (1998) also assert that the activities of firms 'managing their earnings' in the lead up to the issuance decision underperform. Similar effects can be investigated when equity issues are disintegrated into the different issuance methods available to firms, especially in the UK. Kabir and Roosenboom (2003) study rights issues in the Netherlands and find that firms experience significant stock price decline at the

announcement of rights offerings. Rights offerings are the dominant flotation method in the Netherlands unlike UK where firms have choice among rights issues, open offers and private placement including variants of these methods. This compares well with the US where firm commitment is the only method adopted for equity issuing transactions. This study differs markedly from Kabir and Roosenboom (2003) since it considers three equity issuance methods that are available to UK firms. I explore the effects of differential operating performance on the choice of equity issuance methods among UK firms.

Their study covers the period from 1984 to 1995 for a total 67 rights issues (final sample of 58 issues) conducted by 62 different companies listed on the Amsterdam Stock Exchange. Rights issuing firms record announcement date abnormal return of -2.0% which decreases further to -2.8% in the next two days. Consistent with the stock price decline, operating performance subsequently deteriorates. This shows that negative stock return is not an anomaly but indicate market perception of the adverse operating performance after the issuance activity. Using return on assets measures such as net income to total assets, EBITDA to total assets as well as return on sales metrics like net income to sales and EBITDA to sales, they find consistent negative operating performance over 5-years after the equity issues. Information asymmetry between firms and shareholders explain this trend but they find no evidence of rights issues in response to favourable market conditions.

Also, differences in type of firms can account for the post-issue performance of firms conducting private placement. The equity issuance literature is replete with evidence that overvalued firms underperform the market subsequent to the issuance decisions. The work by Baker and Wurgler (2002) highlights the long run effects of mispricing on capital structure. Other studies such as Dong et al (2012), also find evidence in support of the overvaluation-performance relationship. As noted earlier Hertzel et al (2002), provide consistent relationship between stock returns and operating performance following equity issues. Chi and Gupta (2009) also assert that overvalued firms are more likely to engage in earnings management that explains the subsequent underperformance. However, differences such as the degree of overvaluation

and/or the level of growth opportunities can distinguish firms in terms of the level of underperformance.

Using a sample of 371 private placements between the period 1980 and 2000, Chou et al. (2009) explain the post-issue performance of private placements in relation to growth opportunities. High growth firms are more likely to manage earnings (Skinner and Sloan, 2002; McNichols, 2002). Thus, the perception for firms to engage in earnings management stimulates the adverse effects of private placements on stock returns. Market reactions reflect the notion that firms cannot maintain the high reported earnings prior to the equity issues. High growth opportunities evoke both management and investor optimism about the prospects of the firm (Hertzel et al, 2002; Marciukaityte et al, 2005). Overall, they find that high growth firms record negative 3-year abnormal stock returns between -15% and -37% that is consistent with the poor operating performance within the same time period. However, the cross-sectional regression of the abnormal returns is too simplistic since it does not capture the effects of factors such as issue discount, the level of demand and the other corporate governance factors.

Moreover, firms that manage earnings always struggle to maintain the same level of performance. Earnings that are backed by actual revenue generating activities are likely to be repeated in subsequent years. However, managed earnings are not consistent with the underlying business prospects. Chen et al. (2010) find evidence that firms that conduct private placement overstate their earnings prior to the announcement. Investors are unable to detect the excessive accruals in the earnings but subsequent earnings reversal causes the stock price to decline. These earnings reversals imply the firm is unable to sustain or justify the prior period earnings because the core business and revenue generating activities deviate from the reported figures. This phenomenon is consistent with the poor operating performance that occurs in the long run after the announcement of the private placement.

3.5. Testable Hypotheses

The evidence on financing decisions in the literature is varied and wide. Evidence in support of market timing asserts the overriding effects of valuation in determining the decision to issue equity or repurchase at specific times. For instance, Baker and Wurgler (2002) affirm that market timing enables companies to issue equity at high prices and repurchase at low prices. Hence, a coherent understanding of market timing prospects should occur at the point where at high (low) stock prices firms are also significantly overvalued (undervalued) to force equity issue (repurchase). Consequently, in a perfect and rational economic environment, market timing should generate immediate negative (positive) market reactions for equity issues (repurchase). Differences in the level of anticipated operating performance should explain the choice of equity issuance method.

3.5.1. Equity Issues and Operating Performance

As explained in Section 3.3, theories such as monitoring hypothesis (Wruck, 1989), certification hypothesis (Hertzel and Smith, 1993), and managerial entrenchment hypothesis (Barclay et al., 2007)) define the decision to conduct private placements. For instance, the monitoring hypothesis emphasise that private placements enhance firm value because active investors that purchase the shares monitor and ensure efficient use of corporate resources. Under the certification hypothesis (Hertzel and Smith, 1993), private placements create value since the shares are purchased by informed investors with considerable credibility for the market value. Finally, managerial entrenchment hypothesis assert that management places stock with friendly or passive investors who do not play active role in the managing the company thereby strengthening the position of existing management.

A number of characteristics define firms that choose rights issues compared with firms that conduct open offers and placings. Korteweg and Renneboog (2002) and Barnes and Walker (2006) find evidence smaller firms whose shares are less liquid and have a higher degree of information asymmetry are more likely to choose open offers and placings. A higher proportion of the

equity of these firms is also owned by directors. Given that placings are made to external investors it has the tendency to improve external monitoring of the firm and the liquidity of its shares (Slovin et al., 2000). Cronqvist and Nilsson (2005) explain that rights issues are used to retain family control (prevents dilution of ownership), whereas placements are used to issue shares to a strategic partner whose interests are aligned to that of the issuer. Placing process via negotiation with investors facilitates the purchase of large blocks better than the alternative method of selling rights on the market (Armitage, 2010). Unlike Slovin et al. (2000) and Barnes and Walker (2006), Capstaff and Fletcher (2011) argue that quality firms prefer rights issues to open offers and private placements.

The effect of operating performance subsequent to equity issues has been documented in the literature. Loughran and Ritter (1997) assert that firms that issue equity normally report substantial deterioration of operating performance. This means that the negative post-issue stock returns only reflects the poor operating outlook of the firm. Firms timing their equity issues when there is a priori improvement in operating performance are more likely to generate negative market reactions when the operating performance is not sustained. In studying private placements, Hertzel et al (2002) find that private issues of equity follow periods of poor operating performance. Kabir and Roosenboom (2003) also establish that Dutch rights issues also exhibit decline in operating performance. In the context of issuance choice, negative stock returns should correlate with operating performance. If overvaluation drives the issuance of equity using private placement and generates negative stock returns, one would expect deteriorating operating performance for private placement than rights issues and open offers. Factors that signal firm quality include the issue price discount, volatility of returns and earnings management (Dissanaike et al., 2014). However, operating performance should determine the choice of equity issuance method beyond these factors that have relatively short-term impact on subsequent stock performance of equity issuing firms.

 H_6 : Firms that anticipate positive operating performance are more likely to conduct rights issues than open offers or private placement

3.5.2. Long-run Stock Returns and Equity Issuance Methods

Market reactions to equity issues on the average are negative following market timing-informed issuance. Empirical evidence in the literature in relation to equity issues in the UK contends that different issuance methods attract different market effects. Slovin et al. (2000) and Barnes and Walker (2006) both find that rights issues attract more negative reactions that open offers and placement. This is the result of less asymmetric information effects associated with placement where firms increase external monitoring and dilute ownership. However, Capstaff and Fletcher (2011) in a recent study find, similar to Burch et al. (2004), that firms are less likely to time their equity through rights issues than placings. Thus, firms that conduct rights offerings do not potentially exploit overvaluation as do those that conduct equity issues through placings. If quality firms choose rights issues, then it follows that these firms are also more likely to outperform counterpart firms that choose open offers and private placement. In other words, the level of operating performance at the time of equity issuance has explanatory power for the subsequent stock returns associated with the equity issues. Firms that choose rights issues should report less negative stock returns than those that choose open offers and private placement."

H₇: Equity issues through private placement generate more negative abnormal returns than rights issues

3.6. Conclusion

This chapter provided a comprehensive review of equity issuance methods in the UK. It has also discussed the historical background to the equity issues and the changing trends in the choice of equity methods. The chapter also discussed the specific theories that explain the choice of each method and the stock returns associated with each issuance method. The discourse about the stock returns included both the short-run and the long-run price reactions to equity issues and also the level of operating performance associated with each issuance method. It appears the contemporaneous level of operating performance subsequent to the issuance decisions could influence the choice of

method. Based on the reviewed literature, the chapter also discussed the testable hypotheses that are analysed in Chapter 7 of the thesis.

Chapter 4

LITERATURE REVIEW: SHARE REPURCHASE PROGRAMS

4.1. Introduction

Share repurchase programs remain one of the main avenue by which firms distribute cash flows to existing shareholders. Corporate payout policies have changed over the world and in particular the US where different tax codes have shifted preference for dividend payments to share repurchase. In addition to tax, the incidence of stock options as part of corporate compensation programs as well as the prevalence of takeovers, has further allowed firms more preference for share repurchase. Moreover, the recent financial crisis and the debt financing of unconstrained firms present a new dimension to the share repurchase policy. The present study, therefore, explores the share repurchase behaviour of financially unconstrained firms. Apart from the general overview of share repurchase programs and the theories that underlie repurchase transactions, the focus of this chapter is to establish the empirical underpinnings of debt-financed repurchase of firms. In this light, the chapter develops testable hypotheses that set out the focus of the empirical chapter 8 of the thesis.

The chapter has four sections. Section 4.2 presents overview of share repurchase programs. It discusses the methods of and the regulations of share repurchase programs in the US. Section 4.3 focuses on the theories that explain repurchase transactions. Section 4.4 explains the empirical evidence of share repurchase. Section 4.5 discusses the testable hypotheses and section 4.5 concludes the chapter.

4.2. Overview of share repurchase

This chapter briefly lays out the general overview of share repurchase programs. The section consists of two subsections. Subsection 4.2.1 presents the different methods of conducting share repurchase as explained in the literature and adopted by firms, while 4.2.2 the share repurchases regulations in the US.

4.2.1. Methods of Share Repurchase

Open market share repurchase (OMR): OMR is the mode of share repurchase in which a company enters the open market, usually through a broker, to purchase its own shares. The average percentage of shares sought is approximately 7%. ¹⁸ OMR remains the most common method by which firms acquire their own shares. OMR accounts for a total of about 90% of total share repurchase transactions in the US (Stephens and Weisbach, 1998). In fact, between 1994-1999, OMRs accounted for between 95% and 98% of repurchase activity (Grullon and Ikenberry, 2000). Open market repurchases represent an option rather than an obligation for the firm to acquire shares Thus, firms are flexible to decide when and how much to repurchase. This flexibility implies announced shares can remain unpurchased over a period of time. In fact, Stephens and Weisbach (1998) show that between 74 and 82% of announced repurchases are completed whereas Kahle et al. 2005) show a completion rate of 90% among repurchase conducted by Fortune 500 firms. Also, Bhattacharya and Dittmar (2004) contend that open market repurchases afford managers the opportunity to engage in cheap talk. Thus, the mere announcement of a repurchase does not constitute the intention to actually repurchase the shares. This thesis analyses only OMRs given the high percentage of repurchase activity that constitute OMRs.

Tender offer: This method of repurchase involves the purchase of specified amount of shares at fixed price, normally above the market price. This transaction occurs until the expiration date of the offer. In a tender offer, the firm can set limit in terms of both the maximum or minimum amount that intends to buy from the shareholders. The premium associated with tender offer motivates shareholders to tender their shares and conveys quality information about the prospects of the firms. If a small number of shares is tendered the firm can withdraw the offer. However, for an excessively large tender, the firm can choose to buy pro rata or buy greater than the anticipated number of shares

_

¹⁸ Ikenberry et al. (1995) report that the average percentage of outstanding shares sought in all of the OMRs announced between January 1980 and December 1990 by firms listed on the American Stock Exchange, New York Stock Exchange, and NASDAQ was 6.6%.

initially specified. One peculiar feature of tender offer is that the shareholders know about the intentions of the firm to buy back the shares.

Dutch-auction tender offers: These repurchases are similar to fixed price tender offers in the way they are organised. Like fixed price tender offers, they also constitute management's attempt to buy back specified number of shares and with a real commitment to purchase such shares. However, firms set price range at which it is willing to purchase the shares from the shareholders. Comment and Jarrell (1991) report abnormal returns for fixed price tender offers, dutch-auction tender offers and open market repurchase of 11%, 8% and 2.3% respectively. The intuition is that whereas tender offers represent commitment to repurchase, OMR affords management flexibility without actual commitment to follow through with the announced repurchase.

Accelerated share repurchase or synthetic repurchases: These repurchases are combined with derivative securities such as put options, collars and forward contracts. The repurchase transaction is only consummated when the underlying derivative instrument is exercised. For example, the sale of a put option enables a firm to buy the underlying stock at a specified exercise price. The put option allows the firm to exercise the option and buy the share at or below the exercise price. Thus, the firm commits to buy specific number of shares on expiration of the put option. Firms with undervalued shares but have large growth opportunities and face high costs of financial distress prefer these contracts (Vermaelen, 2005). For example, in a forward contract, the company will receive cash when the stock price ends up above the forward price on maturity. In the event that stock price is below the forward price, the company has the option to issue shares to settle the price difference.

Privately negotiated or targeted share repurchase: Share repurchase organised through direct or private negotiations constitute efforts to buy back specified amount of shares from specific shareholders. This normally entails buying block of shares from a large holder. Either the shareholder(s) or the management of the firm can initiate the agreement for the transfer of shares. Sellers proactively make contacts with firms of their intentions to sell their shares back to the firms. This mode of repurchase is normally prevalent where

stocks are thinly traded such that sellers do not favour open market repurchase transactions (Peyer and Vermaelen, 2005). Large investors involved in target repurchase are well informed about the prospects of the firms. By initiating the sale of their shares to the company, it is unlikely that such repurchase will be motivated by undervaluation or signal private information about the firm. Moreover, reductions in cash flow might not reflect decrease in agency costs.

4.2.2. US Share Repurchases Regulations

This section discusses the US regulations as they apply to share repurchase. It lays out this regulatory framework to provide perspective since the thesis uses US data on share repurchase. A number of rules are instituted in the US Securities and Exchange Commission (SEC) regulations to control the share repurchase transactions of corporations. The initial motivation for the regulation of share repurchase is to ensure that corporations do not engage in excessive repurchase that will deplete the assets of the company. For instance, assets depletion affects debt repayment and accrued interests. Unlike open market repurchase, stringent reporting and disclosure requirements must be adhered to in respect of tender offer repurchases, insider trading and equity offerings.

Prior to 1982, open market repurchase in the US were regulated under the provisions of Sections 9(a) (2) and 10(b) and Rule 10b-5 of the Securities and Exchange Act of 1934. However, firms were liable to a charge for illegal price manipulation as result of the uncertainties inherent in the provisions. Price manipulation determines the price of future acquisitions and new issue of shares. However, firms can also have genuine business reasons for engaging in open market repurchase other than simply repurchasing shares to inflate the share prices. The SEC Rule 10(b)-18 of 1982 addresses this concern by giving firms safe harbour in respect of open market repurchase programs. A significant component of this rule is that compliance with the conditions of the rule and the disclosures are voluntary. The safe harbour rule explains the surge in US share repurchase transactions (Grullon and Michaely, 2002).

SEC Rule 10(b)-18 provides immunity from the anti-manipulation provisions of the Securities and Exchange Act of 1934. However, firms must meet four conditions to enjoy the immunity. First, repurchases are made through only one broker or dealer on a given day (Manner of purchase). Second, no repurchase is made as an opening transaction or during the last half hour of trading day (Timing condition). Third, no repurchase is made at a price exceeding the highest current independent bid price or the last independent sale price, whichever is higher (Price condition). Finally, non-block repurchase volume does not exceed the higher of (A) one round lot or (B) the number of round lots closest to 25% of the average daily trading volume of the preceding four calendar weeks.

Under the SEC Rule 10(b)-18 paragraph (d), a firm is not subject to legal liability for non-compliance if it does not abide by the conditions. The implication for non-compliance is that the firm ceases to enjoy protection or immunity under the safe harbour rule. Kim and Varaiya (2004) posit that there is likely conflict of interests between inside and outside shareholders due to the non-mandatory reporting and disclosure issues that characterise open market repurchases in the US. In the same way that firms are buying back shares insiders can also be selling their holdings to profit from the increased in the share price caused by the buyback activities. Amendments to the Rule 10(b)-18 were proposed on 26th January 2010 due to improved trading strategies and the automation of trading systems and technology that have contributed to trading speed. In the event of market-wide trading suspension, the issuer is allowed to purchase up to 100% of the average daily trading volume. Also, the time of purchase condition is relaxed at the reopening of trading on the day of the market-wide trading suspension or on the opening of trading on the day preceding the day of the market-wide trading suspension. The amendments also challenge the safe harbour rule in the event that issuers' insiders are involved in the sale of issuers' shares concurrent with the issuer buyback activity.

4.3. Theories of Share Repurchase

As with various corporate policies, several theories have been developed in the literature to explain the motivations for share repurchase programs. Firms conduct repurchase based on some firm characteristics that underlie various theories and hypotheses with empirical evidence. This section provides description of the theories that explain share repurchase programs. This section is divided into four subsections. Subsection 4.2.1 discusses the undervaluation hypothesis, 4.2.2 the information signalling theory, 4.2.3 the free cash flow hypothesis, and 4.2.4 the market timing theory.

4.3.1. Undervaluation Hypothesis

A firm is undervalued if the intrinsic or the fundamental value is higher than the market price. In other words, undervalued shares are cheaper in the market than their worth. The undervaluation hypothesis therefore posits that firms are more likely to repurchase their own shares in the market when they are cheap in comparison to their fundamental values. This phenomenon of share undervaluation stems from the information asymmetry between managersinsiders- and the external investors. Firms that are undervalued can inform the market about it through corporate transactions such as repurchase, acquisitions and dividend payments. Thus, firms or managers attempt to profit from the private information about the true value of the firm by buying back shares that are cheaper. Since this undervaluation contains information relevant for stock prices, investors are inclined to revise upwards their valuation upon the release of the private information. Moreover, in a share repurchase, managerial behavioural changes in terms to purchase or sale of ownership rights indicate the extent of the quality of the private information (Lee et al. 1992). This hypothesis underlines the financial management goal of maximising shareholder value; any attempt to buy back shares at a price higher than the undervalued price will destroy shareholder value. With significant manager share ownership in the firm, such repurchase transactions align the interests of managers with that of the shareholders.

Repurchase of undervalued shares at low stock prices stimulate high or improved stock price. This is because buyback of shares represents favourable information signal to the market about the future prospects of the firm. Stephens and Weisbach (1998) assert that repurchase is negatively related to the prior stock price performance. In other words, undervalued shares perform better subsequent to a share repurchase announcements. In a study of 450 open market repurchases from 1981 to 1990, Stephens and Weisbach find significant effects of low stock price on increasing the number of shares repurchased. Consistent with free cash flow hypothesis and liquidity argument, they further find evidence of levels of cash flows driving the repurchases (Jensen, 1986; Nohel and Tarhan, 1998; Dittmar, 2000; Grullon and Michaely, 2004). The timing of repurchase of undervalued shares is more effective in relation to open market operations that contain significant flexibility. Managers determine the when and how much of announced shares to repurchase which allows them to make actual repurchase when there is significant undervaluation (Ikenberry and Vermaelen, 1996; Stephens and Weisbach, 1998). According to Ikenberry and Vermaelen (1996), the flexibility feature allows managers to expand their investment opportunity set. This flexibility constitutes exchange option that affects the stock prices at the announcement date. In the long term, the option value increases the stock value of the long-term shareholders. In the empirical analysis discussed in Chapter 8, I control for prior stock stock returns.

Undervaluation can also be inferred from the level of analysts' forecasts of earnings that are related to the information asymmetry. However, since consensus analysts' estimate of earnings is public information, a more refined way of looking at undervaluation from information asymmetry perspective is to consider the anticipated future earnings prospects of firms rather than analyst estimates. D'Mello and Shroff (2000) compute an ex ante economic value from the manager's perspective. The difference between the estimated economic value and the market value constitutes undervaluation. However, the caveat is that the estimate of the economic value will be impacted by the assumptions about the future earnings and the cost of equity. They report that 74% of firms that repurchase via tender offers are undervalued. The positive information signal associated with share repurchase and managerial opportunism means

that managers/insiders are likely to buy more of the undervalued shares and profit from the subsequent price appreciation (Lee et al. 1992; D'Mello and Shroff, 2000). This affirms the expectations of private information associated with share repurchase.

4.3.2. Information Signalling Theory

A related theory to the undervaluation hypothesis is the information signalling that posits that share repurchase is a mechanism for information releases. The information-signalling hypothesis assumes asymmetric information between managers/insiders and investors. Therefore, share repurchase is one tool by which companies convey quality and credible information about the prospects of the firm to the market. Baker et al. (2003) survey managers and find that signalling is the most cited reason for share repurchase. Managers as insiders within the firm are likely to possess information that is yet to be released to the market. In a market where investors do not have complete information about a company, actions and decisions of managers have greater impact on the economic decisions of investors. Market reactions reflect the nature of the information and the magnitude of the 'positiveness' or the 'negativeness' of such information. Information that can be divulged to the market through corporate actions include among other things, the value of the firm, the future cash flow prospects of the firm including the earnings potentials from available growth opportunities. Undervalued shares reflect these positive characteristics such that the market attaches positive information to share repurchase announcements.

Several factors about the firm indicate information signalling associated with share repurchase. These conditions are likely to achieve the signalling effects that accompany repurchase and other forms of cash distributions. Firstly, since repurchase involve cash distribution, only cash-rich firms are more likely to conduct share repurchase programs. Repurchase allows the firm to disgorge excess cash flows to the shareholders in a way consistent with reducing the agency costs of free cash flow (Jensen, 1986). Moreover, undervalued shares signify highly probability that firms anticipate improved earnings in the long-run. Thus, the predictions of the signalling hypothesis resonate with perceived

improved earnings and cash flows subsequent to repurchase transactions. The firms, through repurchase releases information about both the current and future cash flows. This should be associated with improved operating performance and positive stock returns (Nohel and Tarhan, 1998; Grullon and Michaely; 2004). Firms signal quality information to the market and the outside investors through repurchase. If firms distribute free cash flows to reduce agency costs, that should be preclude the availability of growth opportunities.

Secondly, the decision of firms to acquire part of its outstanding shares by paying premium for them indicates anticipated future prospects in terms of enhanced operating performance. Investor positive reactions to repurchase are consistent with this anticipated improved performance. Vermaelen (1981; 1984) formalises the signalling theory of share repurchase to explain the stock price reactions associated with share repurchase. In this model, the effects of information hypothesis depend on the premium paid for the shares, the target fraction of shares acquired and the fraction of insider holdings in the company. The information signalling theory has more predictive explanation for the postrepurchase abnormal returns. Repurchase at a premium using tender offers signifies that the future prospects of the future are more likely to improve. This is because insiders will lose much of their stakes in the event of failure. Vermaelen (1981) finds that insiders do not tender their shares but rather increase the target fraction to signal to the market of quality information. The stock price reactions reflect the premium paid, the increase target fraction, and the high amount of insider shareholding in the firm (Vermaelen, 1984). He finds this to be true for also open market repurchase where signalling is the most plausible explanation for the abnormal returns.

Comment and Jarrell (1991) extend the information-signalling hypothesis of Vermaelen (1981; 1984) using three repurchase methods. They emphasise that, even though repurchase is associated with positive abnormal returns, the level of stock impact depends on how much information signals the repurchase sends to the market. Relative signalling effects of fixed price tender offers, dutch-auction tender offer and open market repurchase yield different stock

price reactions. The level of premium determines the risk of loss of managerial wealth; quality of signals increases with insider loss of false signals. Comment and Jarrell (1991) study the stock price effects of three repurchase methods using over the period between 1984 and 1988. They find that firms pay premium of 20.6% and 12.8% for fixed price tender offers and dutch auction respectively. Open market repurchase does not normally involve premium and as such have a limited information signal effects than the tender offers. However, open market repurchase constitute high volume of share repurchase to signal undervaluation. The average stock returns at announcement are 11% for fixed price tender offer and 8% for dutch-auction tender offer. Open market repurchase is associated with average stock returns of about 2%.

A third factor that explains the information signalling effects of share repurchase is the volume of repurchased shares. The level of information signalling is positively correlated with the volume of open market share repurchase. The number of shares bought in an open market repurchase determines the risk exposure of the insiders. Market interprets that as information signal that motivates insiders to assume the high risk of ownership. By using internal cash flow in a leverage increasing repurchase transaction, the riskiness of the firm increases due to reduced cash holdings and increased leverage. Also, the exposure to earnings volatility and other macroeconomic dynamics impact on the value of ownership. For an insider-owner, transactions that increase utility is preferred and pursued. Open market repurchase has signalling effects unrelated to premium payment such that stock price increases are related to the number of shares bought in open market repurchase (Comment and Jarrel, 1991; Ikenberry and Vermaelen, 1996). McNally (1999) posits that higher expected earnings compensates for the risk exposure associated with increased insider shareholdings. Insiders generate utility from open market repurchase due to signals of greater expected earnings. Like previous findings, the model predicts high market value (high stock prices) in relation to quantity of shares repurchased the number of insider holdings and the riskiness of the expected earnings.

All the above discussion assumes high costs for low value firms which signal false information about the quality of the firm. This high cost discourages mimicking of share repurchase programs (Massa et al, 2007). The inherent flexibility and lack of commitment in open market repurchase has the tendency for managers intending to shore up low stock prices to announce repurchase without significantly making actual purchase of the shares (Vermaelen, 1984; Stephens and Weisbach, 1998). If repurchase is inspired by the objective to send 'false' signals-where earnings improvement are not actually expected in the long run-the long run stock performance should reflect this anomaly. Thus, the market reactions after the initial announcement period are expected to reverse once investors process the actions of managers to mislead them. Once the repurchase is announcement more information is released to the market in a way as to reduce the level of asymmetric information. At this point the market gathers more information and obtains the actual value of the firm. Studies such as Fried (2004) and Chan et al. (2010) report that some managers announce open market repurchase with the view to mislead investors just to increase stock prices. For suspect cases, the immediate market reaction to a buyback announcement does not differ from that generally observed. However, over longer horizons, suspect firms do not experience improvement in their economic performance.

4.3.3. Free Cash Flow Hypothesis

Another factor that may define the share repurchase behaviour of firms is the free cash flow hypothesis. According to the free cash flow hypothesis, firms with excess cash flows distribute it to shareholders rather than investing in value-destroying projects. In other words, share repurchases preserve shareholder value more than sub-optimal investments for firms with excess cash flows. The separation of ownership creates two kinds of conflict: conflicts between management and shareholders on one hand, and conflicts between shareholders and bondholders on the other hand (Jensen and Meckling, 1976). The conflict between managers and owners creates agency problem because managers are more likely to pursue their interests at the expense of the owners. Therefore, managers may commit firm's resources into activities that benefit

such as extensive perquisites, empire building, and investment in valuedestroying activities. Repurchase limits available cash and restricts the overinvestment projects (Jensen, 1986).

Unlike the undervaluation and the information signalling models, the positive investor reactions to repurchase are unrelated to perceived improved performance that is associated with payout policies. Repurchase under free cash flow hypothesis does not predict improved operating performance but lead to reduced agency costs of free cash flow. Under the free cash flow model, repurchase and other corporate payouts should increase with low growth opportunities or negative NPV projects. Lang and Litzenberger (1989) support this conjecture and report that high positive market reactions are associated with low-q firms than high-q firms subsequent to dividends payments. The source of the cash flow expended in the payout could be generated from sources other than cash reserves. Firms can undertake asset sales and/or issue debt financing prior to the corporate payout decisions. For example, as part of restructuring, loss-making assets could be sold and cash proceeds spent in more efficient projects or paid out to shareholders.

Nohel and Tarhan (1998) also argue that asset sales-performance improvement is related to free cash flow hypothesis. Significant asset sales prior to repurchase transactions generate increase in operating performance. Performance increase reflects the deployment of proceeds from sales of poorly performing assets as part of restructuring programme. Again, the performance effects resonate with low-growth firms without any increases in investment opportunities. Market reactions occur at the announcement date and do not extend over the long run. These firms also record significant reduction in systematic risk. Thus, the positive stock returns associated with repurchase are explained by the downward revision of firm risk perception and corresponding rise in investor expectations about the prospects of the firms.

The predictions of the free cash flow hypothesis also assert that managers forgo capital expenditures and disgorge excess cash flows to shareholders. This effect discounts the conflicts of interest that persist between managers and shareholders. The agency theory demonstrates an inherent motivation for

managers to seek to enhance their benefits at the expense of shareholders. Firms with limited growth opportunities should reduce agency costs of free cash flows by paying out excess cash flows to shareholders (Grullon and Michaely, 2004; Bozanic, 2010). In this spirit, share repurchase should correlate negatively with capital expenditure and research and development expenditure. The reduction in capital expenditure, R&D expenses, and subsequent decline in cash reserves is a feature of mature firms. Growth firms on the hand, have unlimited growth opportunities and are more likely to conduct repurchase in line with the signalling model (Liang et al. 2013). Repurchasing firms record reduction in systematic risk and cost of capital (Nohel and Tarhan, 1998) and inform market about reduction in both the agency costs and risks that stimulates positive market reactions.

4.3.4. Market Timing

According to Baker and Wurgler (2002), firms issue overvalued equity and repurchase undervalued equity in keeping with market timing phenomenon. This theory suggests that managerial behaviour also sheds lights on the stock market performance of share repurchase. Consistent with the undervaluation hypothesis, managers with timing ability should only repurchase when the share price is lower. Empirical evidence suggests that few or no shares are actually bought following repurchase announcements (Ikenberry and Vermaelen, 1996; Stephens and Weisbach, 1998). If managers truly have timing ability they will increase actual purchase of shares already announced when the prices are low. Thus, Chan et al. (2007) relate mispricing with the actual buyback activity and long run stock performance. There is only pseudomarket timing if stock performance depends on repurchase announcement behaviour (Schultz, 2003). Chan et al. (2007) find that different portfolio technique yields similar returns and that past stock performance do not have a negative relationship with repurchase announcements. However, they find significant evidence for the increased actual buy back activity when the past stock prices fall. This is consistent with the managerial timing ability that open market repurchases are actually bought when managers perceive stocks are undervalued.

4.4. Empirical Evidence of Share Repurchases

Recent corporate decisions concerning share repurchase provides evidence about alternative financing for share repurchase programs. Firms do not necessarily utilise internal cash flows to buy back their shares. In fact, both constrained and unconstrained firms borrow to finance share repurchase. This section, therefore, provides insights into levered share repurchase. Subsection 4.4.1 discusses levered share repurchase whereas 4.4.2 examines repurchase and investment expenditure. Subsection 4.4.3 provides empirical evidence of stock returns associated with repurchase announcements.

4.4.1. Levered Share Repurchase

The argument from the free cash flow hypothesis is consistent with share repurchase financing through excess cash flows. In order to restrict investment in value-destroying projects and the payment of excessive perquisites, firms with free cash flows are expected to distribute same to the existing shareholders. Thus, cash-rich firms utilise their internal cash to pay for the purchase of their own shares. Stated differently, cash-rich firms are unlikely to borrow to finance their share repurchase transactions. The potential reduction in agency costs as result of distributing excess cash flow to shareholders through share repurchase transactions is positive news to the shareholders (Jensen, 1986). Following from the above, cash-poor firms with excess debt capacity will tend to use debt financed repurchases (Minnick and Zhao, 2007). However, cash-poor firms incur huge costs of financial distress from simultaneous debt issues and share repurchase, especially if they do not have excess debt capacity. Thus, the effects of increasing debt through borrowings might limit cash-poor firms from conducting debt-financed share repurchase.

Debt issues during periods of share repurchase has become commonplace in recent corporate events. In fact, significant amount of levered share repurchases occurred in the period prior to the financial crisis in 2008. The amount of such debt-financed repurchases peaked at more than \$700 billion in

2007 (Milken, 2009). ¹⁹ This trend has continued during and after the financial crisis. The incidence of debt-financed repurchase of unconstrained firms during the crisis period has been attributed to tax costs. For instance, Foley et al. (2007) assert that the consequence of tax costs associated with repatriation of foreign income to the US has resulted in huge amount of cash holdings by firms. This evidence suggests that cash-rich firms are multinationals with significant oversees presence. A significant amount of these cash holdings are domiciled in these oversees operations. Accordingly, it is unlikely that these firms will distribute excess cash flows to finance share repurchase. However, few large firms account for a significant amount of this foreign cash. Therefore, the incidence of levered share repurchase is not defined by only the cash-rich multinationals that do not want to repatriate their foreign earnings for tax purposes.

According to the precautionary motive for holding cash, future uncertainties in the capital market motivates firms to build cash buffers which will allow them to undertake future investments (Opler et al. 1999; Bates et al. 2006). This motive also holds for debt issues when interest rates are at a record low. More significant, it also relates to the asymmetric information effects of external financing. High information asymmetry causes an increase in the costs of external financing. Myers and Majluf (1984) posit that firms should obtain external financing during periods of low information asymmetry. It also predicts that debt issues are less information sensitive and as such involves low costs of issuing them. Moreover, the level of information asymmetry reflects the degree of firm internal financial constraints. All else equal, unconstrained firms are likely to have low information asymmetry compared to an otherwise constrained firm. Therefore, the costs of debt financing should be low for unconstrained firms.

¹⁹ See Michael Milken, "Why Capital Structure Matters: Companies that repurchased stock two years ago are in a world of hurt, "*The Wall Street Journal*, April 21, 2009. He states that companies such as General Electric, AIG, Motorola, and Home Depot all used new debt to finance stock buybacks prior to the year 2007.

The agency cost of debt²⁰ is an important factor that affects debt financing by firms. Due to agency costs of debt firms are unable to issue significant amount of debt capital. Highly leveraged firms face default and bankruptcy risks that prevent them from obtaining debt financing that will allow them to undertake investment projects. Thus, agency costs of debt results in underinvestment (Myers, 1977) and asset substitution (Jensen and Meckling, 1976). The prospect of agency costs of debt implies the degree of internal financial constraints will determine the financing of share repurchase. Constrained firms are more likely to face high agency costs of debt because of limited cash flow and the associated high costs of external financing. Conversely, internally financially unconstrained firms with excess debt capacity are able to increase their debt ratios without significant agency costs. Therefore, one would expect unconstrained firms with valuable investments opportunities to engage in debt-financed share repurchase.

4.4.2. Share Repurchase and Investment Expenditure

The predictions of the free cash flow hypothesis suggest that repurchase significantly reduces cash, cash flows and investment. It also assumes that firms with limited growth opportunities undertake repurchase in order to distribute excess cash flow to shareholders instead of spending on value-destroying activities. In that sense firms are able to reduce risk and costs of capital because repurchase is less risky than the assets in place (Grullon and Michaely, 2004). Conversely, the undervaluation and signalling hypotheses do not necessarily imply limited growth options for firms, even though undervalued firms are more likely to have low growth options. What these theories suggest is that repurchase signals quality information about the future prospects about the firm. By inference, therefore, undervaluation and signalling do not anticipate significant reduction in investment.

From the foregoing, it can be stated that firm lifecycle determines whether free cash flow or the information signalling motivates repurchase announcements.

_

²⁰ Agency costs of debt arise when the interests of shareholders differ from that of debtholders, and also among different classes of debtholders.

Hence, it is anticipated that growth firms would repurchase based on information signalling and mature firms would be motivated by the free cash flow hypothesis (Grullon and Michaely, 2004; Liang et al., 2013). Generally, repurchase reduce cash flow and liquidity in a way that would limit investment expenditure. Thus, share repurchasing firms should invest less in capital projects as more cash are distributed to shareholders (Jensen, 1986; Hahn and Lee, 2009). But, firms could still invest insofar as cash flow is not significantly depleted subsequent to the repurchase announcements. Moreover, through levered repurchase firms would be able to maintain sufficient cash flows to allow for investment activities. Stated differently, the benefits of investing in capital projects could explain the levered share repurchase transactions of unconstrained firms. Market to book ratio is a control variable in the empirical analysis since it captures the effects of growth opportunities.

The financial crisis of 2008 showed dramatic changes in firm's financing and investment behaviours. Internally financially constrained firms are more likely to be hugely affected by financial crisis in terms of investment activities. Campello et al. (2010) and Campello et al. (2011) show that constrained firms planned deeper cuts in investments during the financial crisis. Chen and Wang (2012) provide evidence about share repurchase effects on cash, cash flow, leverage, and investments. They find that constrained firms generally experience significant declines in cash, cash flow, investments, and significant increases in leverage after repurchase activity. Likewise, unconstrained firms also experience declines in cash and increases in leverage but no changes in cash flow and investment. This evidence suggests that leverage is more likely undertaken to finance repurchase program by constrained firms. Conversely, for unconstrained firms the additional leverage provides cash buffer to maintain investment expenditure. Overall, investment has significant influence on repurchase programs, especially for unconstrained. Given the effects of the financial crisis, I control for crisis in the empirical analysis as discussed in Chapter 8.

4.4.3. Share Repurchase and Stock Returns

Share repurchase, according to the undervaluation, free cash flow and information signalling theories, provides positive information to the market. This information includes favourable future financial prospects and growth about the firm, the reduction of the costs of free cash flows not invested in value-destroying activities. In response to the positive information, stock prices increase over the pre-announcement levels. Thus, share repurchase announcements are associated with positive initial abnormal returns. This returns characteristics indicate that repurchasing firms outperform non-repurchasing firms. Prior research attributes the positive returns to the undervaluation hypothesis (Stephens and Weisbach (1998); D'Mello and Shroff (2000); Comment and Jarrell (1991)), the free cash flow theory (Grullon et al., 2002; Grullon and Michaely, 2004; Bozanic, 2010), and the information-signalling hypothesis (Vermaelen, 1981, 1984; Comment and Jarrell, 1991; Ikenberry and Vermaelen, 1996).

The overall evidence find average 3-day abnormal returns of about 3% for US repurchase announcements. Table 4.1 summarises the evidence of abnormal stock returns associated with repurchase announcements. This figure compares well with returns for repurchase in Canada However, the returns are lower for countries with regulatory restrictions to repurchase announcements (Rau and Vermaelen, 2002; Andriosopoulos and Lasfer, 2011). For example, Rau and Vermaelen (2002) find 11-day average abnormal returns of 1.14% and Oswald and Young (2004) 3-day abnormal returns of 1.24%; Andriosopoulos and Lasfer (2015) find 1.68% all in UK, 2.32% in Germany, and 0.80% in France.

Moreover, the long run returns similarly reflect the positive information that accompany repurchase announcements. Market reactions seem to follow the initial returns. Table 4.2 presents empirical evidence of long run returns associated with repurchase announcements which shows that over the period between 2 to 4 years after repurchase, stock returns are significantly positive. This return pattern implies the market initial underreact to the repurchase announcements until sufficient information is released to the market. Grullon

and Michaely (2004) explain market underreacts to repurchase because they underestimate the decline in the cost of capital.

Three hypotheses provide reasons for the market underreaction to repurchase announcements and hence the positive returns associated with share repurchase. First, risk-change hypothesis emphasise that repurchase results in decline in firm risk. According to Grullon and Michaely (2004), firms with limited growth opportunities are more likely to conduct share repurchase. As these firms buyback their shares, they return cash which otherwise would have been wasted on value-destroying investments to reduce the costs of free cash flows. Thus, the repurchase is less risky than assets in place. Grullon and Michaely further explain that repurchasing firms experience significant reduction in systematic risk and cost of capital compared to non-repurchasing firms.

Second, the liquidity hypothesis asserts that the abnormal returns could be caused by an omitted priced factor. The evidence on the liquidity hypothesis is mixed. For example, Barclay and Smith (1988), Brockman and Chung (2001), and Pastor and Stambaugh (2003) find that repurchase reduces liquidity and the abnormal returns reflect price for the reduced liquidity. However, repurchase could increase depth of the sell side market by supporting market makes and adding downside liquidity to falling stock markets (Grullon and Michaely, 2002). These effects of repurchase improve liquidity.

Finally, the long run returns could be explained by the overreaction hypothesis. According to this hypothesis, management belief about market overreaction to recent publicly available information influences repurchases decisions. Thus, prior events determine the long run returns such that management buys back to correct market reaction to bad news. What this hypothesis predicts is consistent with the undervaluation hypothesis because under-valued stocks are priced cheap in the market. All three hypotheses provide better understanding to the positive stock returns associated with repurchase announcements.

Table 4.1: Summary of Empirical Evidence of US Announcement Date Abnormal Returns following Share Repurchase						
Authors	Sample Period	Sample Size	Event Window	Abnormal Returns		
Comment and Jarrell (1991)	1984-1989	1,197	CAR (-1,1)	2.3%		
Peyer and Vermaelen (2005)	1984-2001 6,470		CAR (-1,1)	2.39%		
Peyer and Vermaelen (2009)	1991-2001	3,481	CAR (-1,1)	2.39%		
Grullon and Micahely (2002)	1972-2000	15,843	CAR (-1,1)	2.57%		
Grullon and Micahely (2004)	1980-1997	4,443	CAR (-1,1)	2.71%		
Stephens and Weisbach (1998)	1981-1992	450 CAR (-1,1)		2.69%		
Vermaelen (1981)	1970-1978	243	CAR(-1,0)	3.37%		
Chen and Wang (2012)	1990-2007	4,710	BHAR(-2,0)	1.17%		
Chan et al. (2004)	1980-1996	5508	AR(-2,2)	2.18%		
Chan et al. (2010)	1980-2000	7628 AR(-2,2)		1.80%		
Ikenberry et al. (1995)	1980-1990	1239	CAR(-2,2)	3.54%		
Maxwell and Stephens (2003)	1980-1997	6541	CAR(-1,1)	1.49%		

Table 4.2: Summary of Empirical Evidence of Long Run Abnormal Returns following							
Share Repurchases							
Authors	Sample Period	Sample Size Holding		Abnormal Returns			
			Period				
Chan et al. (2010)	1980-2000	7628	2-year BHAR	8.6%			
Ikenberry et al. (1995)	1980-1990	1239	4-year HPR	12.14%			
Ikenberry et al. (2000)	1989-1997	1060	3-year CAR	21.40%			
Chan et al. (2004)	1980-1996	5508	4-year BHAR	23.56%			
Peyer and Vermaelen (2009)	1991-2001	3481	4-year F&F	24.25%			
			CAR				

4.5. Testable Hypotheses

This section discusses the testable hypotheses.

4.5.1. Internal Financial Constraints and Levered Share Repurchase

In the information signalling framework, firms repurchase to convey quality and credible information about the prospects of the firm to the market in terms of both the current and future cash flows (Baker et al., 2003; Grullon and

Michaely, 2004). The undervaluation hypothesis posits that firms repurchase when their stock prices are undervalued or cheap (Ikenberry and Vermaelen, 1996; Stephens and Weisbach, 1998). The free cash flow hypothesis asserts that firms repurchase in order to disgorge excess cash flows to shareholders (Jensen, 1986; Stephens and Weisbach, 1998). Therefore, firms with limited growth opportunities would rather distribute cash instead of investing in valuedestroying projects. This evidence is in the spirit of maximising shareholder value even though it undermines the pursuit of personal interests of the managers. Stock returns response reflects the decrease in agency costs associated with free cash flow, when managers engage in cash distribution, including share repurchase. Nohel and Tarhan (1998), Dittmar (2000), and Grullon and Michaely (2004), all find support for the free cash flow hypothesis. This theory assumes that share repurchase transactions reflect cash distribution of firms with considerable excess cash reserves. Anecdotal evidence suggests that internally financially unconstrained firms are more likely to repurchase their shares than firms with significant cash constraints.

This chapter addresses two broad arguments related to share repurchase transactions. Firstly, the study asserts that firms borrow to finance their share repurchase transactions. Secondly, Chen and Wang (2012) emphasise that internally financially constrained firms also conduct share repurchases. Grullon and Michaely (2004) as well as Denis and Sibilkov (2009) show that a firm may reduce their investments after share repurchase activities due to limited available cash flows. Moreover, Campello et al. (2010) and Campello et al. (2011) show that constrained firms planned deeper cuts in investments during the financial crisis. Thus, share repurchasing firms should invest less in capital projects as more cash is distributed to shareholders (Jensen, 1986; Hahn and Lee, 2009). However, borrowings should allow firms to build cash balances that support both investment expenditure and share repurchase. Firms that hold high cash balances are likely to preserve their cash balances and conduct debt issuance and use the proceeds to support investment expenditure and repurchase shares.

Therefore, the research further argues that unconstrained firms borrow to finance share repurchases insofar as they engage in more capital investments. The motivation for this conjecture is that unconstrained firms with debt rating and/or investment grade ratings are able to obtain cheap and available debt financing than constrained firms (Kaplan and Zingales, 1997; Almeida and Campello, 2007). This idea follows that while constrained firms need debt financing to repurchase their shares, the high costs of external debt limit the amount they can obtain to finance both the repurchase and investments. Consequently, unconstrained firms are likely to build cash reserves sufficient to undertake both share repurchase and investments. Again, the above discourse is consistent with the conjecture that the level of internal financial constraints determines the probability of levered share repurchase. More significantly, the financing of the repurchase transaction differs according to the level of cash flow available to the firm. Thus, the need for cash to finance investment expenditure at the time of repurchasing shares motivates firms to borrow in order to have sufficient cash reserves to invest while at the same returning cash to shareholders. The following hypotheses are thus tested:

 H_8 : Internally financially constrained firms are more likely to conduct debt-financed repurchase than unconstrained firms

 H_9 : Internally financially unconstrained firms borrow to repurchase especially when they also undertake investments

4.5.2. Internal Financial Constraints, Levered Share Repurchase and Stock Returns

In terms of stock returns, the study tests the hypothesis that the announcement date abnormal returns associated with share repurchase reflect the level of firm internal financial constraints and financing of the share repurchase programme. The extant literature find support for stock price outperformance following share repurchase programs both announcement date (Vermaelen, 1981; Stephens and Weisbach, 1998; Grullon and Micahely, 2004; Peyer and Vermaelen, 2009) and long run (Ikenberry et al., 1995; Chan et al., 2010). However, whether internal cash flow or borrowings are used to finance the

share repurchase program should produce different stock price reactions, both at the announcement date and over the long run. Cash-financed repurchase indicates a more likely positive outlook for the firm; this evidence is highlighted in the information-signalling model. Moreover, the free cash flow hypothesis shows that share repurchase using cash reserves decreases agency costs of free cash flow. Debt-financed repurchase increases leverage and financial distress more than cash-financed repurchase (Jensen and Meckling, 1976; Myers, 1977; Chen and Wang, 2012). The leverage increases cause a fall in the value of the equity capital.

Stated differently, unconstrained firms that borrow to finance share repurchase should have less increases in leverage and hence financial distress costs than constrained firms. To the extent that unconstrained firms borrow to both conduct share repurchase and invest more in capital projects (given that projects are value-increasing), these firms should be associated with more positive abnormal returns than low investing firms and /or constrained levered repurchase firms. In addition to the announcement date abnormal returns, the risk hypothesis and liquidity hypothesis explanations to the long run stock returns suggest that unconstrained firms which invest should be associated with more positive long run returns, especially for the levered repurchase firms."

 H_{10} : Cash-financed repurchase is associated with more positive abnormal returns than debt-financed repurchase, especially when the firm is internally financially unconstrained.

 H_{II} : All else equal, internally financially unconstrained firms which conduct levered share repurchase should be associated with more positive returns for high investing firms than low investing firms.

4.6. Conclusion

This chapter has presented the overview of share repurchase and provided empirical support for share repurchases announcements. The aim of the chapter was to establish that repurchase financing is changing with firms using borrowings to finance share repurchase. However, for internally financially unconstrained firms, which do not necessarily need external financing to conduct repurchase, the additional borrowings allow them to invest more. The chapter also discussed the stock returns associated with repurchase announcements. Empirical evidence suggests that share repurchase announcements are associated with positive announcement date abnormal returns. This finding fails to account for the level of financial constraints and the source of financing of the repurchase program that can affect the stock returns. Therefore, firms are likely to take into account these factors and the stock price implications when undertaking share repurchase programs. The study examines the repurchase behaviour of unconstrained firms, especially when they borrow to finance their repurchase transactions.

Chapter 5

DATA AND METHODOLOGY

5.1. Introduction

This chapter discusses the data and methodology applied to carry out the empirical analysis. The objective of this chapter is to explain the data selection procedure and criteria and also to present the empirical models used in testing the hypotheses. The rationale for the research design is discussed in this chapter. The chapter consists of three sections: Section 5.2 examines the sample selection criteria and the design of the study, Section 5.3 the empirical models and estimation techniques including the measurements of the variables used. Section 5.4 concludes.

5.2. Data

This section discusses the sample period, sources of the data used and how the final sample is derived. The section is divided into four sections. Section 5.2.1 presents the criteria for the sample period selection. Section 5.2.2 focuses on the data for the equity issues, whereas Section 5.2.3 the data for the equity issuance methods. Finally, Section 5.2.4 examines the data for share repurchase programs.

5.2.1. Sample Period

The thesis uses two datasets for the entire analysis and the testing of the hypotheses. Data on UK equity issues are collected from Securities Data Company (SDC) and accounting data from Datastream over the period 1st January 1994 to 31st December 2010. These equity issues are then divided into the type of equity issuance method that is the focus of the second empirical chapter (Chapter 7). Since the deregulation of 1986, UK listed firms are allowed to issue equity through rights issues, open offers, and private placement. Hence, the initial year of the sample period is chosen due to the preference for open offers and private placement. The final data comprise 1257

equity issuance announcements. This represents a comprehensive dataset compared to previous studies.²¹

The second dataset comprises open market share repurchases of US firms between January 1, 1990 and December 31, 2012 and is collected from the (SDC) US Mergers and Acquisitions database. Financial statement data are extracted from the Compustat database while market and stock returns data are collected from the Center for Research in Security Prices (CRSP) database. Share repurchases announcements data from SDC is relatively more complete after 1985 and from the 1990 it became readily available and more comprehensive. The final data consists of 5874 share repurchase announcements over the 23-year period. The choice of US data is important since US firms make significant debt-financed repurchase. 22 This dataset compares favourably with data used in prior studies both in terms of length of time period and the sample size. For example, Stephens and Weisbach (1998) used a sample of 450 repurchase announcements from 1981 to 1990, Grullon and Michaely (2004) used data of 4443 repurchase announcements from 1980 to 1997, and Chen and Wang (2012) used a sample of 4710 repurchase announcements between 1990 to 2007.

The study uses datasets for equity issues (including the equity issuance methods) and share repurchase programs. While the equity issues are for UK listed firms, the share repurchase sample is derived from US listed firms. The aim of the UK equity issues is to explore the effects of market timing and financial constraints on equity issuance decisions (H1, H2, H3, H4, and H5) as discussed in Section 2.5 of Chapter 2. Moreover, the UK equity issues are split into the issuance methods - rights issues, open offers, and private placement - to investigate the long run operating and stock performance associated with each issuance method (H7 and H8) as discussed in Section 3.5 of Chapter 3. For the purpose of examining why unconstrained firms conduct levered share

_

²¹ Prior studies such as Slovin et al., (2000) used 296 equity issues (made up of 220 rights offerings and 76 open offers), Armitage (2000) used a sample of 928 rights issues and 450 open offers from 1985 to 1996, Barnes and Walker (2006) used 868 equity issues consisting of 600 rights offers and 268 placings from 1989 to 1998, and Capstaff and Fletcher (2011) used a sample of 772 equity issues made up of 229 rights issues, 129 open offers, 299 placings, and 115 placings with open offers from January, 1996 to December, 2007.

²² The choice of datasets is discussed in Section 1.3 of Chapter 1.

repurchase (H_{9} , H_{10} , H_{11} , H_{12} , and H_{13}), a total of 5874 share repurchase announcements are used as discussed in Section 4.5 of Chapter 4. In the subsections that follow, the sample procedure for the equity issues and repurchase announcements as well as the sample statistics are discussed.

5.2.2. Data for Equity Issues

The data on the security issues obtained from the Securities Data Company (SDC) are matched using datastream codes (see subsection 5.2.3 below), with the corresponding accounting, stock market prices and valuation data from Datastream. Data on analyst forecast earnings are obtained from the Institutional Brokers Estimate System (IBES). Financial firms are excluded from the sample in line with most research due to volatility of data variables and in regard to regulatory factors that affect financial firms. Examples of studies that exclude financial firms include; Hovakimian, Opler and Titman (2001); Dittmar and Thakor (2007); Oswald and Young (2008); Cook and Tang (2010); DeAngelo et al. (2010). In addition to regulatory differences, financial firms also have a different capital structure from non-financial firms. For example, banks equity issuance decisions may be affected by the minimum capital requirements.

Data are sorted first using sedol number since that represents the code for listed companies on the London Stock Exchange. After applying this initial screening criterion, about 88% of the companies with sedol numbers are found which are then used together with the company ticker symbols to identify the corresponding datastream codes. In cases where a firm has both sedol and ticker symbols further check that the datastream codes provided by sedol and ticker are the same. Where sedol numbers are missing (about 12%), both the ticker and the company names are used to find the datastream codes. Also, further checks with the proceeds resulted in the exclusion of companies without figures for proceeds. This also resulted in the deletion of a further 0.18% of the data to arrive at the final sample for the equity issues.

Data restrictions are as follows:

- i. Issues from SDC must have corresponding accounting data from datastream
- ii. Issues must have relevant sedol codes, datastream codes and ticker symbols
- iii. Firms without any of the codes are deleted
- iv. If the codes do not match with the appropriate corresponding codes in datastream, the firm is eliminated. Thus, both the sedol and datastream codes must match with the codes extracted from datastream
- v. Where feasible, ticker and company names are used to reconcile the disparities. These are just few instances where three of the four criteria are inconsistent.
- vi. Only firms that made any security issues are considered. Therefore, firms without any issues are ignored.
- vii. Following Byoun (2011), only firms with positive market-to-book ratios are included in the final sample. Negative market to book ratios is difficult to interpret since they imply a sustained level of negative earnings by the firm.

Final sample comprises 1257 equity issues over the sample period 1994 to 2010 as reported in Table 5.1.

Table 5.1: Annual Equit Issues				
Year	No. of issues			
1994	63			
1995	63			
1996	83			
1997	51			
1998	49			
1999	58			
2000	126			
2001	104			
2002	51			
2003	77			
2004	65			
2005	44			
2006	47			
2007	83			
2008	54			
2009	159			
2010	80			
Total	1257			

Notes: This table shows the annual equity issues across the sample period.

5.2.4. Selection of matching firms/control firms

The procedure for selecting matching firms is similar to Capstaff and Fletcher (2011), Spiess and Affleck-Graves (1995), Loughran and Ritter (1997) and Hertzel et al (2002). A control firm for each sampled firm is extracted from the listed firms on the London Stock Exchange. At any year-end, firms that previously issued equity are excluded from the list of potential control firms to avoid confounding results. In each of the year-end prior to the security issuance date, one control firm for each sample firm that satisfies certain conditions is identified. A control firm must belong to the same industry group as the sample firm and must be of comparable size in terms of specified range of market value compared to the sample firm.

Once selected, a two-year time lag is imposed before a previously selected control firm is allowed to re-enter the pool of potential control firms. This allows to measure operating performance for the three years to compare with the performance of the sampled firms. Furthermore, a three-year stock return is

estimated to reflect the level of operating performance following the decision to issue equity through three issuance methods. In addition to the cumulative abnormal returns, buy and hold abnormal returns (BHAR) for sampled firms and control firms over three years post-issue are estimated. Following Hertzel et al (2002), the size-matched benchmark ensures that the control firm has market value larger than, but closest to, the sample firm. The matching is done 12 months prior to the announcement date of the issuing firms to better compare the prior issue abnormal returns.

5.2.3. Data for Share Issuance Methods

The final data collected for equity issues as described above are split into the equity issuance methods. Annual operating and monthly stock market returns are collected from Datastream. Since stock returns are estimated for each firm up to three years after the issuance of equity; therefore, stock returns are collected until December 2013. Again, the data is restricted to non-financial firms with at least one-year return data available after the issuance. Panel A of Table 5.2 outlines the procedure for constructing the final sample; Panel B shows the annual number of equity issuances during the sample period. The final sample consists of 328 rights issues, 321 open offers, and 608 private placements. The number of issues over the sample period shows the high frequency use of rights issues during the 1990s, and a significant increase in usage for alternative private placements and open offers within recent years. This decline in usage of rights issues and the increase in private placements and open offers have been reported in prior UK studies (Barnes and Walker, 2006; Armitage, 2010).

²³ The *TIME* datatype is used to explore whether a firm is delisted after the issuance (zero returns in the month of delisting), and firms are not required to have all three-year returns after issuance to avoid look-ahead biased.

Table 5.2: Construction of Equity Issues by Method and the Distribution across the Sample Period Panel A: Construction of Final Sample of Equity Issues Initial Thomson One Banker sample (follow-on equity issues of UK listed 5986 firms from 1994 to 2010) Exclude issues not on London Main Market (3785)Exclude issues which are not either ordinary and/or common shares (53)Exclude VCTs and investment trusts (891)1257 Final sample Made up of: Rights issues 328 Open offers 321 **Placings** 608 **Total** 1257

Year	Rights Issues (RI)	Open Offers (OO)	Private Placement PP	Total 63	
1994	42	0	21		
1995	39	0	24	63	
1996	48	0	35	83	
1997	29	10	12	51	
1998	17	17 27 5		49	
1999	16 26 16		16	58	
2000	18	47	61	126	
2001	19	44	41	104	
2002	13	13 29 9		51	
2003	11	41	25	77	
2004	11	29	25	65	
2005	19	4	21	44	
2006	11	4	32	47	
2007	2	10	71	83	
2008	7	8	39	54	
2009	24	29	106	159	
2010	2	13	65	80	
Total	328	321	608	1257	

Notes: Panel A shows the procedure for the construction of the equity issues by method whereas Panel B gives the annual equity issues by issuance method.

5.2.4. Data Share Repurchase

The second dataset consist of 5874 share repurchase announcements of US firms. Following Maxwell and Stephens (2003) and Minnick and Zhao (2007), this study identifies the announcements of open market repurchases program from the Securities Data Company (SDC). Fama and French (2001) and Grullon and Michaely (2002) calculate repurchase using the balance sheet which give similar results. According to Grullon and Michaely (2002) the correlation between the two measures is 0.97 and the dollar amounts are also similar. However, the announced repurchase from SDC gives the specific dates to help compute the returns prior to the announcement. Using Compustat measure of repurchase is also problematic. For example, Stephens and Weisbach (1998) indicate that since Compustat reports repurchase in value terms and the price at which shares are repurchased is assumed. Moreover, Compustat measure is likely to overstate repurchase since it is an aggregation of all security repurchases and retirements during the quarter or year.

Hence, data on open market share repurchases are collected of US firms between January 1, 1990 and December 31, 2012 from the Securities Data Company (SDC) US Mergers and Acquisitions database. Financial statement data are extracted from the Compustat database while market and stock returns data are collected from the CRSP database. Financial firms (SIC codes between 6000 and 6999) are excluded due to the stringent regulatory oversight under which they operate and their different capital structure (e.g. Denis and Sibilkov, 2010; Chen and Wang, 2012). Finally, following Stephens and Weisbach (1998), repurchase made by the same firm within three years of the previous announcements are excluded to eliminate repeated repurchase announcements (in line with Ikenberry and Vermaelen, 1996; and Lie, 2005). The final sample is made up of 5874 repurchase announcements as reported in Table 5.3.

_

²⁴ Jagannathan et al (2000); Fama and French (2001) and Grullon and Michaely (2002) define repurchase as the total expenditure on common and preferred stocks (Compustat #115) minus reduction in the number of preferred stocks outstanding (Compustat #56).

Table 5.3: Construction of Final Sample of Share Repurchase Announcements					
Initial Thomson One Banker sample	12296				
Exclude issues without corresponding Compustat code for matching	(5191)				
(permno, gvkey or 6cusip codes)					
Exclude announcements without share repurchase value/amount	(959)				
Exclude sample with negative MB or no stock returns data	(272)				
Final Sample	5874				

Notes: This table shows construction of the final sample of share repurchase. It indicates the initial sample and the restrictions and exclusions that resulted in the final sample.

Consistent with the research the third objective the sample is defined as levered or unlevered depending on the source of financing. The idea is to test the hypothesis that drives firms to either commit internally generated funds to conduct repurchases or borrow with the goal of using the proceeds to finance equity repurchase. Levered share repurchase occurs when debt issues finance the repurchase. Following Minnick and Zhao (2007), share repurchase is levered if there is a positive change in the debt capital either one year prior to or within two years after the repurchase announcement. The sample is further classified according to the level of internal financial constraints.²⁵ Table 5.4 Panel A reports the number of share repurchase for unconstrained firms during the sample period. Out of 5874 share repurchase announcements, a total of 4700 repurchases are made by unconstrained firms of which 3359 unlevered and 1341 levered repurchases. There seems no particular pattern at the total share repurchases and the distribution between unlevered and levered repurchases during the sample period. A significant number of share repurchases is therefore financed through borrowings, even though the particular firms are unconstrained.

_

²⁵ The measure of financial constraints is discussed in section 5.2.1.2

Table 5.4: Number of Annual Share Repurchase Announcements										
	Panel A: Full sample				Panel B:		Panel C:			
	•			Un	Unconstrained			Constrained		
Year	0	1	Total	0	1	Total	0	1	Total	
1990	195	38	233	159	32	191	36	6	42	
1991	94	19	113	86	18	104	8	1	9	
1992	162	32	194	150	24	174	12	8	20	
1993	124	46	170	114	39	153	10	7	17	
1994	179	74	253	147	57	204	32	17	49	
1995	222	86	308	181	65	246	41	21	62	
1996	289	153	442	253	113	366	36	40	76	
1997	271	145	416	230	103	333	41	42	83	
1998	453	189	642	346	123	469	107	66	173	
1999	341	127	468	254	76	330	87	51	138	
2000	164	92	256	126	60	186	38	32	70	
2001	126	29	155	111	19	130	15	10	25	
2002	90	33	123	80	21	101	10	12	22	
2003	73	27	100	61	18	79	12	9	21	
2004	107	46	153	89	31	120	18	15	33	
2005	130	99	229	112	77	189	18	22	40	
2006	92	101	193	80	86	166	12	15	27	
2007	157	141	298	132	100	232	25	41	66	
2008	302	92	394	258	62	320	44	30	74	
2009	95	40	135	82	27	109	13	13	26	
2010	116	72	188	99	58	157	17	14	31	
2011	135	129	264	117	103	220	18	26	44	
2012	103	44	147	92	29	121	11	15	26	
Total	4020	1854	5874	3359	1341	4700	661	513	1174	

Notes: This table shows the annual share repurchase announcements. Panel A provides the annual share repurchase announcements for the full sample; Panel B gives the annual repurchase announcements for unconstrained firms and Panel C shows the annual repurchase announcements for the constrained firms subsample.

.3. Variable Definitions and Model Specification

This section justifies the research approach used in the study. Variables used are defined including how they are measured. The model used is also specified. The section is further divided into three subsections. Subsection 5.3.1 defines the dependent, explanatory, and control variables for exploring the effects of mispricing and financial constraints on equity issues. It also discusses the model used to test the hypotheses related to equity issues. Subsection 5.3.2 presents the dependent, explanatory, and control variables used to examine the

long run operating and stock performance associated with equity issuance methods. Again, model for the analysis is explained in this subsection. Finally, subsection 5.3.3 explains the variables and models used to examine the levered share repurchase behaviour of internally financially unconstrained firms. The variables used are consistent with prior literature (see for example Grullon and Michaely, 2002; Chen and Wang, 2012) which has shown that these variables theoretically and empirically have significant effects on share issuance and repurchase decisions of firms as discussed further in the subsections that follow.

5.3.1. The Effects of Mispricing and Financial Constraints on Equity Issuance Decisions

The first part of the thesis examines the equity issuance decisions of firms given the degree of financial constraints. This section discusses the dependent variables, the explanatory, and control variables. It also specifies the model for the empirical testing of the hypotheses.

5.3.1.1. Dependent Variables

The dependent variables are discussed in the sections below.

i. Indicator Variable: Equity Issuance

The study explores the propensity for firms to issue equity given the degree of financial constraints. It also controls for other firm characteristics that have been explored in the empirical literature to determine equity issuance decisions. Given that the research examines probability of equity issues, the dependent variable therefore is a dummy variable; 1 for equity issues and 0 for non-equity issuance.

ii. Abnormal Stock Returns

The second dependent variable is the abnormal stock returns associated with equity issues, Event studies methodology is used to measure the stock price impact of corporate events. The early work by Fama et al. (1969) laid the foundation for other intriguing event studies in respect of corporate events.

Their study focused on stock split and market reactions but since then other corporate events such as seasoned equity offerings (SEO), initial public offerings (IPO), mergers and acquisitions (M&A), repurchase, among others have received considerable attention in event studies in the empirical corporate finance literature. Brown and Warner (1980) explore event studies using monthly data and subsequently use daily data when they became available (Brown and Warner, 1985).

Abnormal stock returns are estimated using the market model where daily return of a particular security is compared with the daily return of the benchmark to compute daily excess or abnormal returns.²⁶ For UK data, the benchmark is the FTSE ALL SHARE INDEX. The market model calculates abnormal returns as the difference between the actual returns and the OLS fitted returns. Thus, the residuals or regression errors represent the abnormal returns of the market model. The market model therefore adjusts for both market-wide factors and systematic risk of each security.

Abnormal returns are computed using the market model as follows:

$$AR_{it} = R_{it} - (\alpha_i + \beta_i R_{mt})$$

Where:

AR_{it} is the abnormal returns

 R_{ii} is the actual stock returns

 $\alpha_i + \beta_i R_{mt}$ is the fitted or predicted returns from the OLS regressions with α_i and β_i as regression coefficients

 R_{mt} is the market index returns.

Cumulative abnormal returns, *CAR* is estimated from the abnormal returns using the market model.

²⁶ Studies which have used the market model to estimate abnormal returns include Dong et al. (2012).

111

_

5.3.1.2. Explanatory Variables

The proxies used to measure the effects of mispricing and financial constraints on equity issuance decisions are discussed in the following subsections. In addition to their definitions, limitations of each proxy are also explained.

i. Mispricing: Market-to-book ratio (MB)

According to proponents of market timing theory, misvaluation of firm value represents mispricing that motivates managers to issue overvalued equity and repurchase undervalued. Market-to-book ratio (MB) by definition encompasses firm value from market perspective as well as the fundamental or intrinsic value as determined by the companies' fundamentals. Given this background, any difference between what the market perceives as firm value and the firm's valuation results in mispricing. Summarily, higher market value than the book value generates high MB whereas lower market value than the book value generates lower MB. Thus, MB defines stock mispricing (Fama and French, 1992; Rhodes-Kropf et al., 2005; Elliot et al., 2008). Consistent with the literature on equity issues, overvalued firms are more likely to issue equity whereas undervalued firms prefer to buy back their shares (Baker and Wurgler, 2002; Dong et al., 2012). The degree of mispricing defined as the ratio of market price to fundamental value also explains firm stock returns (Fama and French, 1992; Daniel et al. 2001; Barberis and Huang, 2001).²⁷

Consistent with prior studies, MB is the proxy for mispricing. Accordingly, high MB is expected to have a positive relationship with share issuance decisions. The effects of mispricing also indicate that MB has an expost negative relationship with stock returns. Mispricing or misvaluation is defined as MB and computed as follows:

$$MB_{it} = \frac{BVA_{it} - BVE_{it} + MVE_{it}}{BVA_{it}}$$

Where:

BVA_{it} is the book value of total assets

²⁷ Further evidence of the effects of mispricing on share issuance decisions and subsequent stock returns is discussed in chapter 2.

ii. Internal Financial Constraints (KZ index)

The main proxy for internal financial constraints is the Kaplan and Zingales' (1997) index (hereafter KZ index). ²⁸ Kaplan and Zingales (1997) constructed this index in their study of the financial constraints characteristics of 49 low-dividend paying manufacturing firms. The sample of 49 firms over 15-year period (from 1970-1984) replicates that of Fazzari et al. (1988). Low-dividend paying firms exhibit strong relation between investment and cash flow. Using observable firm characteristics, Kaplan and Zingales (1997) rank these firms on ordinal scale form least to most constrained and run ordered logit model. This is an objective measure that has gained substantial currency in the empirical literature as an indicator of financial constraints (Baker et al. 2003). It is hypothesised that internally financially constrained firms will be impeded in responding to high valuations to issue equity. This implies less internally financially constrained firms quickly take advantage of mispricing; therefore, there is a negative correlation between valuation and degree of internal financial constraints.

Following studies such as Baker et al., (2003) and Dong et al., (2012), the proxy for internal financial constraints used in this study is computed below.

KZ index is defined as follows:

$$KZ = 1.002(CF/TA) + 3.139LEV$$
 $39.368(DIV/TA)$ $1.315(CASH/TA)$

Where:

CF is the cash flow measured as net income plus depreciation

²⁸ Lemmon and Zender (2010); Almeida et al. (2004) use dividend payment and debt capacity as proxy for financial constraints which the KZ-index captures in addition to other factors that affect the level of financial flexibility. Again, Helwege and Liang (1996); Lemmon and Zender (2010); DeAngelo et al (2010) adopt Altman's Z-score which define financial distress as evidenced by excessive leverage rather than the lack of financial surplus.

TA is total assets LEV is leverage defined as the long term debt over lagged total assets

CASH is defined as cash and cash equivalents

A higher *KZ-index* represents highly constrained firms that have debt capacity concerns, have low cash balance (limited financial slack) and pay no dividends. According to the pecking order theory, firms with high *KZ-index* issue equity. Similarly, less financially constrained are better placed to respond to both valuation effects and the need to adjust the leverage to achieve target debt ratio.

iii. Internal Financial Constraints (WW index)

For robustness of results, an alternative proxy for financial constraints is used to analyse the effects of internal financial constraints on share issuance decisions. This proxy for financial constraints is the Whited and Wu (2006) index hereafter the WW index calculated as follows:

$$WW = -0.091(CF/TA) - 0.062(DIVDUM) + 0.021(LTD/TA)$$
$$-0.044\log TA + 0.102INDSG - 0.035SG$$

Where:

CF is cash flow; TA is total assets; LTD is long-term debt; INDSG is the firm's three-digit industry sales growth; SG is the firm's sales growth; and DIVDUM is a dummy variable that takes a value of 1 if a firm pays cash dividends and 0 otherwise. A high WW-index denotes that the firm is more internally financially constrained. In line with the KZ-measure, firms in the 5th quintile are classified as internally financially constrained and unconstrained otherwise.

iv. Financial Distress (Taffler's Z score)

Financial distress (used for UK data) measure is used as alternative proxy for financial constraints measure. The degree of financial distress supports the pecking order theory. In the context of this hypothesis, firms that are potentially liable to financial distress prefer equity financing to debt. Financial distress, as explained in theoretical finance, is the inability of company to meet its financial obligations, especially to creditors. Factors such as excessive debt giving rise to fixed interest payments, lack of profitability particularly in

situations where sales levels slump in response to economic downturns and the lack of liquid assets that could be sold to defray the obligations. Firms plagued with costly financial distress correlate with severe debt capacity concerns which limit external debt financing. Again, there is a lack of financial slack such that inadequate internal funds result in financial deficit.

For example, Helwege and Liang (1996), Lemmon and Zender (2010), DeAngelo et al. (2010) state that Z-score correlates with excessive leverage rather than the lack of financial surplus. Thus, financial distress explains financial constraints due to the high leverage associated with distressed firms. The effects of debt capacity concerns, lack of internal funds and profitability restrict any attempts to access the external debt market. Effectively, financially distressed firms are more likely to enter the external equity market for financing in order to undertake investments projects. The correlation with internal financial constraints also suggests that financially distressed firms are less likely to issue equity due to high costs of equity issuance.

Financial distress, though correlates with internal financial constraints, also have partial effects of financing decisions and in particular determining the leverage levels. The trade-off theory emphasises the costs associated with excessive leverage. Here financial distress, rather than internal financial constraints, is used to account for the effects of potential bankruptcy costs on equity financing. Therefore, Taffler's Z-score (a measure of business failure for UK firms) is used as a measure of financial distress. By using multiple discriminant analysis, Taffler's (1983) provides a Z-score measure that is computed as four weighted ratios. He used stepwise linear discriminant analysis to develop a model that was able to discriminate effectively between failed and healthy companies between the beginning of 1969 and the end of 1976. The model is described as follows:

$$Z = 3.20 + 12.18 * PROF + 2.50 * WCAP - 10.68 * FRISK + 0.0289 * LIQD$$

Where:

PROF is the ratio of profit before tax to average current liabilities WCAP is ratio of current assets to total liabilities

FRISK is current liabilities divided by total assets LIQD refers to number of credit interval that is defined as [(Current assets-Inventory-Current liabilities) / (Sales-Profit before tax +Depreciation)]

Negative *Z score* represents financially distressed companies that are in danger of failure; whereas those with positive *Z score* are solvent companies.

v. Financial Crisis (CRISIS) and Market Liquidity (LIQD)

The extant literature offers evidence of the effects of financial crisis and market liquidity on corporate decisions. ²⁹ Campello et al. (2010), Ivashina and Scharfstein (2010), Campello et al. (2011), and McLean and Zhao (2014) indicate that external financial constraints, evidenced by the financial crisis, affect investment, employment, and cash holdings of firms. Financial crisis (*CRISIS*) is a dummy variable that is 1 for the period after 2007 and 0 otherwise. Market Liquidity (*LIQD*), on the other hand, measures the ease of raising capital that is related to the costs of external financing. Internally financially constrained firms are less likely to issue overvalued equity when market liquidity is low. Market liquidity is computed as the difference between 12-month London Interbank Offer Rate (*LIBOR*) and the Bank of England Base Rate (*BOEBR*). This is as similar measure of market liquidity used by Harford (2005) for US equity market.

$$LIQD_{t} = LIBOR_{t}$$
 $BOEBR_{t}$

Where:

LIBOR_t is the 12-month London Interbank Offer Rate $BOEBR_t$ is the Bank of England Base Rate

5.3.1.3. Control Variables

In addition to the key parameters that have been defined as affecting financing decisions, the study incorporates a number of additional variables that have partial effects on security issuance decisions. Rajan and Zingales (1995) identify four firm characteristics that have significant impact on financing

²⁹ See Chapter 2 for discussion of the effects of external financial constraints on corporate financing policies.

decisions. Particularly, leverage levels correlate with each of these variables that in turn determine financing decisions as a result of the leverage levels. The four Rajan and Zingales (1995) leverage determining factors include firm size, profitability, and asset tangibility and growth opportunities. However, growth opportunities are already captured in the measure of mispricing given as *MB*.

The following control variables are included.

i. Profitability (PROF)

Profitability is defined in the model as the earnings before interest, taxes and depreciation (EBITDA) divided by total assets and expressed as a percentage (in other words as the return on assets). Jensen (1986) asserts that free cash flow problems associated with profitable firms force them to issue debt capital. Again, evidence attests that profitable firms have high levels of internal funds and as such are unlikely to be financially distressed. Moreover, profitability translates into growth opportunities in subsequent periods. That is, profitable firms are more likely to investment in research and development that produces opportunities for growth. Thus, with growth options firms issue equity more than debt financing and hence record low leverage levels. Following from that, and according to the pecking order theory, these firms are unlikely to access the capital debt market for financing. Therefore, profitable firms have a low leverage. It is defined accordingly as:

$$PROF_{it} = \frac{INC_{it}}{CE_{it}}$$

Where:

INC_{it} is net income before extraordinary items CE is common equity of the firm i at time t

ii. Firm size (SIZE)

Firm size is measured as the log of net sales/market capitalisation has a positive relationship with leverage. This is due to the ability of large firms to shield against financial distress and access capital debt market. Empirical

evidence asserts that small firms possess considerable growth opportunities, low leverage yet greater uncertainty due to information asymmetry. Large firms, on the other hand, are mostly matured entities that have exhausted growth options but are highly followed by analysts. The effects of analysts' coverage limit asymmetric information effects. Again, large firms have substantial tangible assets which provide collateral facilities to access debt financing. Intuitively, small firms are less likely to follow the pecking order, even though this argument is sternly challenged in the literature (Lemmon and Zender, 2010). In terms of leverage levels, the collateral facility from the high degree of asset tangibility ensures that large firms have high leverage than small entities. The conclusion from this analysis reflects the positive relationship between firm size and leverage.

Using Rajan and Zingales' (1995) definition, firm size is calculated as the natural logarithm of book value of total assets (also Barclay and Smith, 1995; Dong et al., 2012). Other studies such as Antoniou et al. (2006) use log of sales. The reasoning for scaling with natural logarithm is to curtail the bias associated with outliers and errors. Residuals get bigger with bigger values for the dependent variable. This apparent anomaly is inevitable because error or change in the value of an outcome variable is often a percent of the value rather than an absolute value. The natural logarithm of a variable neutralises the residuals for the bigger value. Moreover, the difference in size among small and large firms is controlled using logarithm rather than the "raw" values. This eliminates potential skewness associated with large values. There is neutrality with firm size and the effects on financing choice or leverage removes any bias.

$$SIZE_{it} = LogTA_{it}$$

Where:

LogMV_{it} is the natural logarithm of book value of total assets

iii. Earnings Deviation (ED)

Earnings deviation and accrual quality demonstrate information gap between managers and investors. This measure captures the effects of information asymmetry associated with equity issues as discussed in Chapter 2. Dittmar and Thakor (2007) use earnings deviation as an agreement parameter where high value represents higher agreement between managers and investors. Dong et al (2012), however, argue that absolute value of earnings deviation better defines the degree of disagreement between managers and investors. Lee and Masulis (2009) and Billet et al. (2013) assert that accrual quality is an alternative proxy for asymmetric information that is more related to earnings management. Whereas investors rely on financial statements to take economic decisions, insiders own the source of such information reported on the financial statements. A lower quality of such information therefore represents a source of asymmetric information.

To proxy for asymmetric information, in line with the pecking order theory, an estimate of the earnings deviation, showing the absolute deviation of actual earnings from analysts' consensus forecast is given as:

$$ED_{it} = \left| \frac{EPS_{actual} - EPS_{forecast}}{EPS_{actual}} \right|.$$

Where:

 EPS_{actual} is the earnings per share reported by the firm i at time t $EPS_{forecast}$ is the consensus analysts' average forecast of earnings per share for the firm i at time t

iv. Capital Expenditure (CAPX)

Investment is used to represent capital expenditure. Firms with valuable investments are more likely to obtain external financing. For example, firms issue additional debt to undertake investment (DeAngelo et al. 2011; Dudley, 2012). Also given debt capacity concerns, firms would obtain equity financing to provide cash flow for investment. Thus, mispricing drives investment. Firms

undertake investment when the stock is overpriced through the issuance of equity financing to support the capital or cash required for the investment project. This view relates closely with the market timing phenomenon. Polk and Sapienza (2009) assert that mispriced firms undertake abnormal investment and are associated with post-investment adverse returns. This implies mispriced firms are more likely to issue equity because of investment opportunities. Investment tracks mispricing more when the firms are dominated with high R&D and also possess short-term horizon investors.

It is defined by the capital expenditure over the prior period divided by total assets and is given as:

$$CAPX_{it} = \frac{CAPX_{it-1}}{TA_{it}}$$

Where:

 $CAPEX_{it-1}$ is capital expenditure for the prior period TA_{it} is total assets

5.3.1.4. Model Specification and Estimation Technique

This section discusses the econometric model for testing the effects of mispricing and financial constraints on equity issues and the associated stock returns. The first part tests the following hypotheses (H_1 , H_2 , and H_3) as discussed in Section 2.5 of Chapter 2:

 H_1 : There is high probability for less internally financially constrained firms to issue overvalued equity

H₂: Firms are more likely to issue overvalued equity when external financial constraints are low (non-financial crisis period or periods of high market liquidity)

H₃: The probability of issuing overvalued equity during periods of high external financial constraints increases for firms with low internal financially constraints than those with high internal financial constraints

The hypotheses specifications are related to a binary dependent variable. This is because the study tests the probability or the likelihood of equity issues in the presence of a set of firm characteristics. For the binary dependent variable, OLS estimation technique is not an appropriate model. For example, OLS estimation of the linear probability model (LPM) violates certain assumptions. LPM has the defect that the conditional probability is not constrained to lie between zero and one (Aldrich and Nelson, 1984; Cameron and Trivedi, 2005; Wooldridge, 2010). Thus, the predictions of the conditional probability can be greater than one or less than zero. Moreover, the errors are inherently heteroscedastic. Unlike LPM, logistic regression estimated under maximum likelihood estimation (MLE) restricts the predicted probability within the range 1 and 0 and also corrects for heteroscedastic errors. Studies such as Cox, (1970) Tennant (1977), Silvapulle (1981), Scott and Wild (1986 and 1991) are apply the MLE to various research. Therefore, logistic regression model is an appropriate model for binary dependent variable.

The logistic regression model is stated as follows:

$$\Pr(y_i = 1 | x_i) = \frac{e^{x_i \beta}}{1 + e^{x_i \beta}}....(1)$$

$$\Pr(y_i = 1 | x_i) = \beta_0 + \beta_1 K Z_{it} + \beta_2 M B_{it} + \beta_3 M B * K Z_{ti} + \sum_{i=1}^n \beta_i X_{it} + \varepsilon_{it} \dots (1)$$

Where:

 β_0 is the constant term

KZ, is the indicator variable for levered repurchase

MB_{it} is the market-to-book ratio for mispricing

 $MB*KZ_{it}$ is the interaction term between market-to-book ratio and internal financial constraints

 X_{ii} is the set of control variables: pre-announcement abnormal returns (AR), dummy for financial crisis (CRISIS) and market liquidity (LIQ), firm size (LMV), capital expenditure (CAPX), earnings deviation (ED), profitability (PROFIT), time dummies, and industry dummies

Logistic regression models have been applied extensively in the empirical finance and corporate finance literature. Martin (1977) explores early warning of bank failure, Martin (1996) the method of payment in corporate acquisitions, Fama and French (2001) firm characteristics and the propensity to pay dividends, Goyal and Park (2002) the board leadership structure and CEO turnover, Baker and Wurgler (2004) link the propensity to pay dividends to catering incentives, and Denis and Osobov (2008) he reasons for dividend payments. This study adopts a similar estimation model to study the propensity for equity issues given the level of internal financial constraints and a set of firm characteristics.

The following pooled OLS regression model is used to test the above hypothesis about the post-issue stock returns.

H₄: Issuance of overvalued equity is associated with more negative abnormal returns, especially for firms with high internal financially constraints

 H_5 : Issuance of overvalued equity is associated with more negative abnormal returns, especially for firms with high internal financially constraints and during periods of high external financial constraints

$$CAR_{i} = \beta_{0} + \beta_{1}KZ_{i} + \beta_{2}MB_{i} + \beta_{3}MB * KZ_{i} + \sum_{i=1}^{n} \beta_{i}X_{i} + \varepsilon_{i}.....(2)$$

Where:

 β_0 is the constant term

 KZ_{ii} is the indicator variable for internal financial constraints which is 1 for constrained and 0 for unconstrained

MB_{it} is the market-to-book ratio for mispricing

 $MB*KZ_{it}$ is the interaction term between market-to-book ratio and internal financial constraint

 X_{it} is the set of control variables: pre-announcement abnormal returns (AR), dummy for financial crisis (CRISIS) and market liquidity (LIQ),

firm size (LMV), capital expenditure (CAPX), earnings deviation (ED), profitability (PROFIT), time dummies, and industry dummies

All regression estimations throughout the thesis are carried out using the STATA software.

5.3.2. Share Issuance Methods: Operating Performance and Stock Returns

The second research objective of the study is to explore the effects of operating performance on the choice of equity issuance methods. It also uses the operating performance to examine the different long run stock performance associated with rights issues, open offers, and private placement. This section discusses the dependent variables, the explanatory, and control variables. It also specifies the model for the empirical testing of the hypotheses.

5.3.2.1. Dependent Variable:

The following sections explain the dependent variables used to carry out the empirical analysis.

i. Indicator Variable for Rights Issues, Open Offers, and Private Placement

The study compares rights issues with open offers on one hand, and rights issues and private placement on the other. Thus, for the first part the dependent variable is 1 for rights issues and 0 for open offers. In the second part, the dependent variable is 1 for rights issues and 0 for private placement.

ii. Long term abnormal returns

Long-term returns estimation presents important statistical issues that need to be considered in order to avoid biased estimates. Unlike short-run abnormal returns with limited stock price volatility over short estimation window, long run abnormal returns can be affected by the estimation method used and produces biased estimates and misspecified test statistics. Kothari and Warner (1997) and Barber and Lyon (1997) both analyse extensively the properties of long-run abnormal returns. Both studies find significant statistical problems

with long-run abnormal returns depending on the method of estimation. Cumulative abnormal returns, buy-and-hold abnormal returns and Fama and French (1993) three-factor models have been employed to estimate long-run abnormal returns in the empirical literature. Ritter (1991), for example, note that the use of *CAR* and *BHAR* depends on the specific question to be answered. Thus, it is more likely that *CAR* and *BHAR* will produce different estimates.

Moreover, Barber and Lyon (1997) assert that empirical power and specification of test statistics are affected by the method of estimation and the approach for developing benchmark. Long-run abnormal returns benchmarks used in the literature include reference portfolio (market index or size decile portfolios), control firms approach and the three-factor model of Fama and French (1993). Fama (1998) states that long-term return anomalies are sensitive to methodology. Further, Lyon et al. (1999) recognise the problems associated with analysis of long-run abnormal returns. Bias estimates and test statistics misspecification stem from new listing bias, rebalancing bias and skewness bias (Barber and Lyon, 1997). They argue that new listing bias occurs because new firms are allowed to the market index that is benchmark for the sample firms' long-run abnormal returns. Again, compound returns of market index are computed with implicit periodic rebalancing whereas returns of sample firms do not involve rebalancing. Finally, long-run abnormal returns are positively skewed.

Control firm approach to calculating abnormal returns is able to offset the sources of bias and misspecification depending on whether *CAR* or *BHAR* is the estimation method. Without the reference portfolio benchmark such as the market index, the bias from new listing, rebalancing and skewness are eliminated. Control firm and sample firm are identical in several respects and experience comparable effects when measured against the market index. *CAR* is also associated with measurement bias. Barber and Lyon (1997) contend that *BHAR* using the control firm approach produces unbiased and well-specified test statistics since it eliminates the bias from new listing, rebalancing,

skewness and measurement. The current study follows this approach that has been applied in a number of studies in the literature.³⁰

a. Cumulative Abnormal Returns (CAR)

To estimate cumulative abnormal returns for each cross section firm, the expected returns, using market model that regresses stock returns on market return, is computed. Market returns are defined as the total return on the FTSE ALL SHARE index. The relationship of the market model is thus stated as follows:

$$R_{it} = \alpha_i + \beta_i R_{mt} + \varepsilon_{it}$$

$$(1)$$

$$AR_{it} = R_{it} - E(R_{it})$$

$$(2)$$

Where:

$$E(R_{it}) = \stackrel{\wedge}{\alpha}_i + \stackrel{\wedge}{\beta}_i R_{mt}$$
(3)

The parameters in equation (1) are the stock returns for each firm, R_{it} ; the market return defined as R_{mt} , and ε_{it} is the idiosyncratic risk associated with the stock returns. α_i and β_i are the alpha and beta coefficients of the regression model to be estimated. Both stock returns and market returns are the daily returns associated with the stock and market index respectively. The estimation window of 250 days prior to the announcement date is used.

The cumulative abnormal returns (CAR) are estimated as the sum of the abnormal returns (AR) within a specified event window. This is stated as follows:

_

³⁰ Loughran and Ritter (1995), Spiess and Affleck-Graves (1995), Veld and Veld-Merkoulova (2004), Hertzel et al (2002) and Capstaff and Fletcher (2011) all use the control firm approach in calculating long-run abnormal returns.

$$CAR_{it} = \sum_{t=1}^{T} AR_{it}$$

Following Barber and Lyon (1997), the *t*-test is defined as the ratio of the cumulative abnormal returns to the standard deviation of the *CAR* of the firms divided by the root of number of firms.

$$t - test_{CAR_T} = \frac{CAR_T}{\sigma(CAR_{iT})/\sqrt{N}}$$

b. Buy-and-Hold Abnormal Returns (BHAR)

Buy and hold abnormal returns (*BHAR*) measure the difference between compounded actual return and the compounded predicted return. The compounding feature associated with *BHAR* better simulates the effect of an event on an investor's portfolio. Again, *BHAR* provides a good measure of the long run investor experience under the long run event studies (Loughran and Ritter, 1995). *CAR* and *BHAR* both complement each other due to their peculiar limitations. Whereas *CAR* fails to capture the compounding effects, *BHAR* can also yield incorrect statistically significant abnormal performance due to short-term return fluctuations. Therefore, the problems of extreme skewness occasioned by *BHAR* are curtailed when double-checked with *CAR*.

Similarly, the *BHAR* is computed after estimating the expected or predicted returns from the regression of stock returns and market returns. Buy and hold return is the daily compounded return on the equity security for each firm over a specified time period in days.

The buy and hold return (BHR) of each firm is given as

$$BHR_{iT} = \prod_{t=1}^{T} [1 + R_{it}] - 1.$$

In same fashion, the expected or predicted buy and hold return is calculated as

$$E(BHR_{iT}) = \prod_{t=1}^{T} [1 + E(R_{it})] - 1$$

Finally, BHAR, the difference between the buy and hold returns (BHR_{iT}) and the expected buy and hold returns $E(BHR_{iT})$, is computed using the equation below.

$$BHAR_{it} = \prod_{t=1}^{T} [1 + R_{it}] - \prod_{t=1}^{T} [1 + E(R_{it})]$$

Test of statistical significance of the mean returns is computed using the standard *t*-test as follows:

$$t - test_{BHAR_T} = \frac{BHAR_T}{\sigma(BHAR_{iT})/\sqrt{N}}$$

5.3.2.2. Explanatory Variable

This section discusses the proxies that are used to measure the effects of operating performance on share issuance method and stock returns. In addition to their definitions, limitations of each proxy are also explained.

i. Operating Performance (ROA)

The extant literature reports strong empirical evidence of stock underperformance following SEOs (Loughran and Ritter, 1995; Spiess and Affleck-Graves, 1995) that is related to the poor operating performance (McLaughlin et al. 1996; Loughran and Ritter, 1997; Levis, 1995).³¹ The study predicts that the level of operating performance influences equity issuance method and hence explains the long run stock performance differences among issuance methods. If high profitability is favourable market information, then rights issuing firms are more likely to be profitable than open offers and private placement (Capstaff and Fletcher, 2011). This should also explain long run stock outperformance for rights issues than open offers and private placement.

Profitability is defined using returns on assets (ROA) and their abnormal components over matched firms. The following relations explain each measure.

$$ROA_{it} = \frac{EBIT_{it}}{TA_{it}}$$

_

³¹ Chapter 3 provides comprehensive discussions of stock and operating performance associated with equity issuance methods

Where:

ROA is the return on assets. It can also be calculated as the ratio of EBIT, earnings before interest and taxes to TA,

5.3.2.3. Control variables

The control variables are used in previous studies to explain the stock price effects associated with equity issues and equity issuance methods. Most of these variables are based on theories that have been explored extensively in the literature. In addition to variables that explain equity issues in general as discussed in preceding sections, other variables specifically determine the choice among the equity issuance methods. Following the extant literature, the variables are defined in the same manner to reflect the same effects as indicated in the literature. The proxy for each variable is discussed as follows:

i. Mispricing (MB)

Mispricing affects the choice of equity issuance method due to its relation to high information asymmetry (Livingston et al., 2005). Moreover, Slovin et al. (2000) and Barnes and Walker (2006) find that overvalued equity are issued to existing shareholders rather than privately placed with external investors. However, Capstaff and Fletcher (2011) assert that rights issuing to existing shareholder are less likely to be overvalued. Thus, all else equal, high *MB* is negatively related to the probability of rights issues and positively related to private placement and open offers. *MB* is defined in the Section 5.3.1.2.

ii. Pre-issue Abnormal Returns (Prior AR)

AR is the cumulative abnormal stock returns (CAR) estimated using the market model over estimation window (-255,-21) relative to the announcement date. The pre-issue AR is the cumulative abnormal returns from day -30 through day -2. Thus, the AR is defined as CAR (-30,-2). It is expected that AR should have a positive relationship with mispricing and thus increase the probability of equity issues through private placement.

iii. Firm Size (SIZE)

Firm size is measured as the natural logarithm of book value of total assets and has a positive relationship with equity issues. This is because large firms to have low information asymmetry and are less likely to be mispriced. Large firms are mostly matured entities that have exhausted growth options but are highly followed by analysts. Small firms possess considerable growth opportunities, low leverage yet greater uncertainty due to information asymmetry.

iv. Discount (DISC)

Empirical evidence suggests that equity issues are associated with significant discounts between the offer price and the market price at the announcement date. Armitage et al. (2012) find that about 90% of UK open offers and placings are at a discount of about 26%. However, for large discounts existing shareholder value is more protected if rights offering are used to issue equity. This is because discount transfers wealth to the new investors that will subscribe to the equity issues, when the discounted equity is issued to external investors through private placement. While this constitutes costs to existing shareholders who do not participate the new shares, due to lack of pre-emptive rights, the discount could compensate placees for the cost of investigating the issuer (Hertzel and Smith, 1993; Balachandran et al., 2013). However, Barclay et al (2007) contend that the large size of the discount undermines such conjecture.

Following the information asymmetry hypothesis, overvalued firms should attract high discounts (Hertzel et al, 2002). Armitage et al. (2014) establish that discounts are related to illiquidity of issuers' shares, financial distress and inelastic demand for the shares. Slovin et al. (2000), Armitage (2002), and Balachandran et al. (2008) report that rights issue price reaction is significantly and negatively related to the issue discount. However, Barnes and Walker (2006) assert that high discounts are more likely to be associated with placings than rights issues. This is consistent with overvaluation effects that evoke equity issues through placings. High discount (more negative value) is more

likely to be associated with issues that reflect overvaluation. Thus, equity issues are timed should attract deep discount to compensate investors.

Issue price discount is calculated using the Barnes and walker (2006) and Armitage et al. (2014) formula that compares the pre-announcement date market price and the offer price. Thus, discount is given as follows:

$$DISC_{it} = \frac{P_{offer} - P_{market}}{P_{market}} ,$$

Where:

 $P_{\it offer}$ is the offer price stated in SDC at the announcement date of the equity issues

 P_{market} is the prior announcement date market price reported

v. Accruals Quality (AQ)

Consistent with the information asymmetry and earnings management hypotheses, firms with better accrual quality are more likely to issue equity to existing shareholders. Hertzel and Smith (1993), Wu (2004), Balachandran et al. (2013) argue that firms since private placement investors can obtain the true value of the firm at a cost; firms with high information asymmetry are more likely to issue equity through private placements. Accruals quality is the modified Dechow and Dichev (2002) measure of earnings quality defined in the preceding sections. This is consistent with studies such as Lee and Masulis (2009), Balachandran et al. (2013), and Armitage et al. (2014),

$$CA_{it} = \alpha_i + \gamma_1 CFO_{it-1} + \gamma_2 CFO_{it} + \gamma_3 CFO_{it+1} + \gamma_4 \Delta SALES_{it} + \gamma_5 PPE_{it} + \varepsilon_{it}$$

Where:

 CA_{it} =total current accruals = Δ current assets - Δ current liabilities - Δ cash + Δ debt in current liabilities; Δ =changes from year t to year t-1

CFO = cash flow from operations= net income before extraordinary itemstotal accruals and total accruals=current accruals-depreciation and amortization;

SALES =total sales revenue

PPE = property, plant and equipment. Each of the variables is scaled by the average of total assets between year t-1 and year t.

Accruals quality (AQ) is computed as the standard deviation of the regression residuals from t-4 through t where larger standard deviation at year t reflects a poor earnings quality and hence high asymmetric information. All variables are scaled by average total assets over year t-1 and t. Overall, four regressions using a total of four lagged values and one lead value of data over the sample period 1989-2011 is estimated. The firm-specific measure of earnings management is the standard deviation of the four regression residual values. Larger standard deviations suggest poor accruals quality and greater information asymmetry. This variable is denoted as ACC_QUAL .

vi. Idiosyncratic Risk (IDYRISK)

Idiosyncratic volatility is measured as the standard deviation of the excess market return relative to the date of equity issue announcements (Dierkens, 1991; Krishnaswami and Subramanian, 1999). Balachandran et al. (2013) find lower idiosyncratic risk for rights offerings than open offers and placement. This indicates that issue to existing shareholders do not produce adverse selection costs that are associated with external shareholders. Hence, firms with lower asymmetric information are more likely to conduct rights offerings. Therefore, lower idiosyncratic risk indicates that the firm is higher quality firm (Balachandran et al., 2008).

vii. Financial Crisis (CRISIS

Financial crisis is an indicator variable of 1 for the period 2008-2010 and 0 otherwise.

5.3.2.4. Model Specifications

This section discusses the econometric model for testing the effects of operating performance on the choice of equity issuance method and the long run stock performance as discussed in Section 3.4 of Chapter 3. The first part

tests the following hypotheses (H₆) using the multinomial logit regression model.

H_6 : Firms that anticipate positive operating performance are more likely to conduct rights issues than open offers or private placement

The multinomial logit regression model is stated as follows:

$$\Pr(y_i = 1 | x_i) = \beta_0 + \beta_1 ROA_{it} + \sum_{i=1}^n \beta_i X_{it} + \varepsilon_{it}$$
....(3)

Where:

 β_0 is the constant term

 ROA_{it} is the returns on assets for long run operating performance

 X_{it} is the set of control variables: market-to-book ratio (MB), ratio of issue proceeds over market value (P/MV), pre-announcement abnormal returns (AR), dummy for financial crisis (CRISIS),, firm size (LMV), economic growth rate (GDP), accruals quality (ACC_QUAL), idiosyncratic risk (IDY_RISK), issue price discount (DISC), time dummies, and industry dummies

Long run stock returns are estimated using the OLS estimation to test the following hypothesis.

H₇: Equity issues through private placement generate more negative abnormal returns than rights issues

The OLS model is stated as below:

$$BHAR_{i} = \beta_{0} + \beta_{1}ROA_{i} + \beta_{2}ISSUE_{i} + \beta_{3}ROA*ISSUE_{i} + \sum_{i=1}^{n} \beta_{i}X_{i} + \varepsilon_{i}.....(4)$$

Where:

 β_0 is the constant term

 ROA_i is the returns on assets for long run operating performance

*ISSUE*_i is the dummy variable for equity issuance method which is 1 for rights issues and 0 otherwise

 $ROA_i * ISSUE$ is the interaction between operating performance and equity issuance method

 X_i is the set of control variables: market-to-book ratio (MB), ratio of issue proceeds over market value (P/MV), pre-announcement abnormal returns (AR), dummy for financial crisis (CRISIS),, firm size (LMV), economic growth rate (GDP), accruals quality (ACC_QUAL), idiosyncratic risk (IDY_RISK), issue price discount (DISC), time dummies, and industry dummies

5.3.3. Share Repurchase and Internal Financial Constraints

The third research objective of the study is to examine the share repurchase behaviour of internally financially unconstrained firms. In other words, the study explores the effects of internal financial constraints on the financing of share repurchase programs through either internal cash flows or external borrowings. This section discusses the dependent variables, the explanatory, and control variables. It also specifies the model for the empirical testing of the hypotheses.

5.3.3.1. Dependent Variable

According to the developed hypotheses (H₈), internally financially constrained firms are more likely to borrow to finance repurchase programs. However, for unconstrained firms, the need to undertake investments are the time of repurchasing the shares motivates them to obtain debt financing (H₉). Subsequent stock returns should be more positive for unlevered repurchase firms than levered repurchase firms (H₁₀). For the levered repurchase firms, high investing firms are likely to be associated with more positive stock returns than low investing firms (H₁₁). These hypotheses are discussed in Section 4.5 of Chapter 4. Here the dependent variables that explore the hypotheses are discussed.

i. Levered and Unlevered repurchase (LEVD)

Repurchase is defined as levered/unlevered in relation to the level of borrowings that firms are predicted to have committed to finance the share repurchase. Thus, in this context changes in debt capital within the period of the repurchase represent borrowings with a view to financing the repurchase. According to Minninck and Zhao (2007), levered repurchases are those that are

financed by debt issues. The debt issues must be made within one year prior to two years post share repurchases. Thus, changes in long-term debt capital on the balance sheet are used to determine whether or not debt-financing influences repurchase. Firms that issued bonds or increased their borrowing within the repurchase time period specified are classed as levered repurchase (debt-financed); otherwise they are unlevered repurchase (cash-financed).

The sample is then classified in terms of levered or unlevered share repurchase based on the increase in the total debt. Levered share repurchase occurs when bond issues finance the repurchase. Share repurchase is defined as levered if there is a positive change in the debt capital one year prior to and/or two years after the repurchase announcement.³² Following this definition, the dependent variable is an indicator variable that is 1 for levered and 0 for unlevered repurchase. The use of indicator variable for levered repurchase measures the probability that a firm conducts levered share repurchase given other firm characteristics. These firm characteristics are discussed under the explanatory and control variables sections. Studies such as Minnick and Zhao (2007), Dong et al. (2012), and Balachandran et al. (2013) use indictor variable as dependent variable in their studies of either equity issues or share repurchase.

ii. Stock returns

Prior research finds positive abnormal stock returns around the announcements of share repurchase programs. Signalling, free cash and undervaluation hypotheses are all consistent with the stock returns subsequent to share repurchase. Positive information signals are transmitted to the market because these theories posit that share repurchase decisions indicate that firms have positive future prospect in terms of earnings. Also, the agency theory/free cash flow hypothesis asserts that excess cash flows are distributed to shareholders when there are limited investment opportunities in order to preserve shareholder value. Therefore, share repurchase announcements mean managers do not waste resources on negative NPV projects. Maxwell and Stephens

_

³² Share repurchase in the highest quintile- in either one year prior to or two years after repurchase announcements- are classified as levered repurchase whereas those in the other quintiles are unlevered repurchase.

(2003) find that not only does repurchase increase stock returns but it also transfers wealth from bondholders to shareholders. Minnick and Zhao (2007) report similar evidence in respect of levered repurchase and unlevered repurchase. Unlevered share repurchase is associated with high abnormal stock returns than levered share repurchase. Chen and Wang (2012) find that abnormal stock returns increases more for unconstrained firms than for constrained firms. Chan, Ikenberry and Lee (2007) also report evidence of abnormal stock performance following buyback announcements. dependent variables are the initial and long run abnormal returns defined below.

a. Initial Abnormal Returns

Following previous studies, the study uses 3-day event window to estimate the initial abnormal returns associated with share repurchase announcement. Studies such as Chen and Wang (2012) use the 3-day window for initial returns. Initial announcements date abnormal stock is computed as the threeday cumulative abnormal returns (CAR) over the period from day -1 through day +1. Thus, initial abnormal returns is defined as CAR(-1,1).

b. Long Run Abnormal Returns

Long-run abnormal stock returns are computed over 12 to 36 months following repurchase announcements, based on Barber and Lyon's (1997)methodology. 33 Buy-and-hold abnormal returns are estimated as the difference between buy and hold returns for the repurchase firm and buy and hold returns for the control firm as discussed in Section 5.3.2.1. Control firm must be within the same size decile, book-to-market (B/M) quintile, and KZ index quintile as the repurchasing firm (similar to Chen and Wang, 2012). In line with Lyon et al. (1999), t-statistics are bootstrapped skewness-adjusted.

³³ Refer to section 5.2.2.1 for discussions on cumulative abnormal returns and buy and hold abnormal returns.

5.3.3.2. Explanatory Variables

The following sections discuss the explanatory variables that are used to test the hypotheses about the levered share repurchase behaviour of internally financially unconstrained firms.

i. Internal Financial constraints (KZ index and WW index)

Corporate events are impacted by the degree of financial constraints. However, Chen and Wang (2012) find evidence which indicates that a sample of internally financially constrained firms repurchase their shares. Repurchase for undervalued firms are also associated with less financially constrained firms. Internally financially constrained firms would be more likely to borrow in order to repurchase their shares. However, the study predicts that unconstrained firms that undertake investment projects would also be likely to conduct levered repurchase. Consistent with the hypothesis, the study explores the extent to which internally financially unconstrained firms engage in levered repurchase. In line with Chen and Wang (2012), share repurchase in the highest quintile are classified as internally financially constrained whereas those in the other quintiles are internally financially unconstrained.³⁴

ii. Bond Ratings (RATINGS)

Share repurchase increases leverage and therefore results in debt ratings downgrades. It follows that the bond ratings at the time of share repurchase would determine the extent of debt-financed repurchase. Interest payments for firm borrowings will be influenced by the creditworthiness of the firms, including other factors. Investment grade firms will more likely obtain debt financing at a cheaper cost than speculative grade firms. Maxwell and Stephens (2003) and Minnick and Zhao (2006) both find that bond ratings are twice likely to be downgraded as upgraded after the announcement of repurchase programs. This is due to the relatively increased risk of excessive debt capital. These downgrades are inspired by the likely risk of financial distress following

³⁴ See subsection 5.2.1.2 for comprehensive discussion on the measurement of the KZ index and WW index. Chapter 4 examines the literature review about the effects of financial constraints on share repurchase programs.

debt issuance and repurchase. Following from this evidence, we expect levered repurchasing firms to report significant downgrades especially when they are internally financially constrained.

iii. Financial Distress (Z score)

Excessive leverage capital increases the risk of bankruptcy. Repurchase reduces the share capital and invariably increases the debt component of the capital of the firm. Even though repurchase does not necessary change firm value, it results in distress risk. Opler and Titman (1994) find a significant and positive relationship between leverage and financial distress. Internal financial constraints in relation to repurchase will have a far greater effect on distress risk than unconstrained firm. Chen and Wang (2012) find that constrained firms with high actual repurchase ratios report significantly greater distress risk than unconstrained firms. This is due to the lower cash levels and higher leverage ratios occasioned by the repurchase transaction. Moreover, increased debt levels increase the probability of default on the bonds and expose the firm to financial distress. The likelihood of downgrades is related to the increase distress risk (Maxwell and Stephens, 2003; Minnick and Zhao, 2006). Purnanandam (2008) asserts that following debt issuance firms immediately institute risk-management policies due to probable financial distress risk. It is predicted that financial distress risk will increase for repurchase especially for levered repurchase conducted by internally financially constrained firms.

Financial distress is measured using the Altman (1968) Z-score for US data.³⁵

This is given as

$$Z - score = 1.2X_1 + 1.4X_2 + 3.3X_3 + 0.6X_4 + 0.999X_5$$

 X_1 is working capital divided by book value of assets; X_2 is retained earnings over book value of assets; X_3 is the earnings before interest and taxes over book value of assets; X_4 is the market value of equity over book value of assets and X_5 is the net sales over book value of assets.

³⁵ Studies that use Altman (1968) Z-score to measure financial distress include Purnanandam (2008), Chen and Wang (2012); among others.

Firms with value less than 1.81 as distressed. Financial distress dummy is 1 for Z-score value less than 1.81 and 0 otherwise. It must be stressed that Taffler's Z-score is used as proxy for financial distress for UK data whereas Altman's Z-score is used for US data.

iv. Investment (INV)

Investment is the total capital expenditure plus research and development (R&D) over lagged total assets. Share repurchasing firms should invest less in capital projects as more cash are distributed to shareholders (Jensen, 1986; Hahn and Lee, 2009). However, debt-financed share repurchase should allow firms to also invest more. Therefore, firms investing more are likely to borrow to finance share repurchase.

5.3.3.3. Control Variables

The following variables are included in the regressions to control for the effects on share repurchase announcements.

i. Undervaluation (MB)

Market to book ratio is a proxy for undervaluation (Baker and Wurgler, 2002; Dong et al, 2012; Li and McNally, 2007; Grullon and Michaely, 2002). Market to book ratio relates to undervaluation hypothesis by which undervalued firms are more likely to repurchase their shares. It is cheaper to purchase because the suppressed market price of the shares means firms pay significantly low price for the same number of shares than they would for overvalued shares. Thus, low market to book ratio indicates undervaluation and also correlates positively with low prior stock returns.

According to the agency hypothesis, firms disgorge excess cash when there are limited positive net present value projects to invest the cash flow. Thus, available growth opportunities compete with cash paid to shareholders by way of dividends and repurchase. High growth opportunities reduce the probability of share repurchase (Boudry et al. 2013). We expect high stock returns following share repurchase for firms with low growth opportunities. Ikenberry

and Vermaelen (1996) assert that positive stock returns following repurchase explains that management commit resources in a way to benefit long term shareholders when there are no growth opportunities. Undervalued firms are more likely to repurchase their shares because the shares trade at relatively lower prices than their fundamental values (Stephens and Weisbach, 1998; D'Mello and Shroff, 2000). Thus, low *MB* firms are more likely to repurchase their shares than high *MB*.

ii. Firm size (SIZE)

Large firms are more likely to repurchase their shares due to lack of growth opportunities. Cash distributions are made in respect of investment financing including acquisitions, dividend payments and share repurchase, among others. Free cash flow hypothesis enjoins firms to disgorge excess cash flow to shareholders especially when positive net present value projects are limited. Firm size (SIZE) is given as natural logarithm of market value of equity. Firm size is computed as follows:

$$SIZE_{it} = LogTA_{it}$$

Where:

LogMV is natural logarithm of book value of total assets

All else equal, large firms that are also mature have limited growth opportunities and therefore are likely to distribute excess cash flow to shareholders. As firms increase in size and become mature, the limited investment opportunity creates free cash flow that can be distributed to shareholders. Thus, firm size positively correlates with share repurchase announcements and is consistent with the free cash flow hypothesis (Grullon and Michaely, 2004; Chen and Wang, 2012; Liang et al, 2013).

iii. Prior returns (Prior AR)

According to the undervaluation hypothesis, firms with prior poor stock performance have the incentive to repurchase their shares. The repurchase signals quality information and causes the share price to rise. Li and McNally (2007) Grullon and Michaely (2002) find evidence consistent with this

prediction. In the same view, prior stock returns should also motivate whether or not firms borrow to finance to repurchase. Thus, prior stock returns are positively related to announcement date stock returns. Following prior studies such as Maxwell and Stephens (2003), Dong et al (2012), the *prior AR* is the cumulative abnormal returns from day -30 through day -2. Thus, the *AR* is defined as *CAR* (-30,-2).

iv. Share repurchases ratio (REP_RATIO)

This is the ratio of repurchase amount to market value of equity. This variable estimates the repurchase value relative to firm size. Firm size correlates positively with amount of share repurchase since large firms have limited growth opportunities but huge amount of free cash flows. Chen and Wang (2012) report that high repurchase ratio is associated with positive initial abnormal returns. It is therefore expected that high repurchase value should be associated with levered share repurchase. All else equal, unconstrained firms that borrow should be able to repurchase more as a percentage of firm size. Thus, high repurchase ratio is consistent with unconstrained firms, debt-financed repurchase, and large firms.

5.3.3.4. Model Specification

As stated in Section 4.5 of Chapter 4, the study predicts that internal financial constraints and investments influence the share repurchase financing of firms. The testable hypotheses are as follow:

H₈: Internally financially constrained firms are more likely to conduct debt-financed repurchase than unconstrained firms

H₉: Internally financially unconstrained firms borrow to repurchase especially when they also undertake investments

Logistic regression estimation is used to empirically test the above hypotheses about the effects of internal financial constraints on levered share repurchase. The logistics regression model is stated as follows:

$$P = \beta_0 + \beta_1 K Z_{it} + \beta_2 INV_{it} + \beta_3 K Z * INV_{it} + \sum_{i=1}^n \beta_i X_{it} + \varepsilon_{it} \dots (5)$$

Where:

 β_0 is the constant term

 KZ_{it} is the proxy for internal financial constraints (KZ index)

 INV_{it} is the investment expenditure of the firm during the repurchase

period

 $KZ*INV_{it}$ is the interaction term between financial constraints (KZ index) and investment expenditure (INV)

 X_{ii} is the set of control variables: market-to-book ratio (MB), preannouncement abnormal returns (AR), dummy for financial crisis (CRISIS), firm size (LMV), share repurchase ratio (RPR), time dummies, and industry dummies

The second part of the research objective explores the stock returns associated with levered share repurchase programs. It focuses on both the initial and long run stock returns of firms that conduct levered repurchase and simultaneously invest in capital projects. The hypotheses are thus stated below:

 H_{10} : Cash-financed repurchase is associated with more positive abnormal returns than debt-financed repurchase, especially when the firm is internally financially unconstrained.

 H_{11} : All else equal, internally financially unconstrained firms which conduct levered share repurchase should be associated with more positive returns for high investing firms than low investing firms.

The following pooled OLS regression model is used to test the above hypothesis about the post-repurchase stock returns similar to Stephens and Weisbach (1998), Maxwell and Stephens (2003), Chen and Wang (2012), Dong et al. (2012),

$$CAR_{i} = \beta_{0} + \beta_{1}LEVD_{i} + \beta_{2}INV_{i} + \beta_{3}LEVD * INV_{i} + \sum_{i=1}^{n} \beta_{i}X_{i} + \varepsilon_{i}.....(6)$$

Where:

 β_0 is the constant term

 $LEVD_i$ is the indicator variable for levered repurchase

 INV_i is the investment expenditure of the firm during the repurchase period

 $LEVD*INV_i$ is the interaction term between indicator variable levered repurchase (LEVD) and investment expenditure (INV)

 X_i is the set of control variables: market-to-book ratio (MB), preannouncement abnormal returns (AR), dummy for financial crisis (CRISIS), firm size (LMV), share repurchase ratio (RPR), time dummies, and industry dummies.

5.4. Conclusion

This chapter discussed the research design including definition of variables and the methodology for carrying out the empirical analyses. It provided overview of the data and sources and justified the choice of the data. The study uses two datasets for equity issues on one hand and repurchase on the other hand. Data on equity issues are made up of a sample of UK listed firms and collected from SDC and Datastream. The final sample is 1257 equity issues from the period January 1, 1994 to December 31, 2010. For the analysis of operating performance and stock returns associated with equity issuance methods, the data is restricted to the period from January 1, 1996 to December 31, 2010 and divided into rights issues, open offers, and private placement. For the share repurchase sample, the data consist of US firms repurchase announcements made between January 1, 1990 and December 31, 2012. The final sample consists of 5874 share repurchase announcements and collected from SDC. For this dataset, the accounting data and stock prices are collected from Compustat and CRSP, respectively.

The chapter also presented and defined the variables that explain the hypotheses developed in Chapters 2, 3, and 4. Since the research objectives examine the probability of equity issues and share repurchase, the dependent variable is an indicator variable. This means the logistic regression model and

multinomial logit regression (for equity issuance methods) are used to test the first part of the research objectives as summarised in Chapter 1. The second part of the research objective examines the stock returns associated with equity issues and repurchase and for that the appropriate dependent variable is determined as abnormal stock returns. The analysis of stock returns is carried out using OLS regression analyses. The above methodology is discussed in line with previous studies.

Finally, the explanatory and control variables are discussed in this chapter. Also, the models for testing the hypotheses are specified. Logistic regression models are specified to test the effects of mispricing and internal financial constraints on equity issues including the effects of operating performance on equity issuance methods. Chapters 6 and 7 provide empirical results for the models proposed for equity issues. The models for testing the levered share repurchase behaviour of financially unconstrained firms are also defined. Chapter 8 provides empirical results for the model stated to analyse the share repurchase announcements.

Chapter 6

SHARE ISSUANCE AND FINANCIAL CONSTRAINTS

6.1. Introduction

This chapter discusses the empirical analysis of the effects of financial constraints on share issuance decisions of UK listed firms. A number of studies explore whether firms' share issuance decisions reflect mispricing. The market timing (e.g., Taggart, 1977; Marsh, 1982; Asquith and Mullins, 1986) is the prevalent developed theory, stating that managers issue equity when firms' share prices are overvalued. Several studies (e.g., Loughran and Ritter, 1995; Baker and Wurgler, 2002) empirically support the market timing theory, showing that overvalued equity is associated with significant negative abnormal announcement returns. For example, the empirical evidence by Baker and Wurgler (2002) highlights the long-term effects on capital structure of firms' market timing behaviour of equity issues. Their evidence indicates that not only do firms time their equity issues but also that the equity issues determine the long-term changes in the capital structure.

The recent financial crisis provides a new dimension to equity market timing phenomenon. The argument explored in this thesis is whether the market timing theory is incomplete, since it does not capture firms' financial constraints according to which managers time their equity issuance as long as their firms are in good enough financial condition to pay for the costs associated with external equity financing. In particular, the effects of the financial crisis not only reduced available financing in the market but it also increased the costs of such financing. For example, Korajczyk and Levy (2003) assert that firms are better placed to obtain external financing during periods of favourable market conditions. Thus, whether or not firms time their equity issues should be influenced by both the firm's own internal financial constraints and the external financial conditions.

Within a mature market, most listed firms have rather limited financial constraints, and this thesis focuses on the variation of distress during the financial crisis to test whether financial constraints influence firms' equity

issuance decisions. Ivashina and Scharfstein's (2010) highlight the significant drop of bank lending during the financial crisis showing that bank lending reduced by 47 percent in the fourth quarter of 2008 in relation to its prior quarter. Campello et al. (2010) also survey CFOs from US, European, and Asian firms and reported significant difficulties in accessing external funding during the recent financial crisis, even for funding profitable projects. More than half of the firms in their sample faced difficulties to access funding during the crisis.³⁶

Studies such as Frank and Goyal (2003), Fama and French (2005), and Dong et al. (2012) look at equity issues of listed firms relative to the level of financial constraints. In particular, Dong et al. (2012) use Canadian firms' issuance decisions between 1998 and 2007 and support that the issue of overvalued equity is conditional on the level of firms' financial constraints during the whole sample period. However, Fama and French (2005), using data between 1973 and 2002, and Frank and Goyal (2003), between 1971 and 1998, find no evidence that financial constraints affect the decision of US firms to issue equity. They argue that in anticipation of investment opportunities, firms could retain cash even if they are financially constrained. These mixed results cast doubt on whether firm financial constraints impact equity issuance decisions. Using UK firm sample, this analyses the extent to which financial constraints influence market timing's predictions as long as focusing on the recent financial crisis. This chapter presents both univariate and multivariate analyses to explain the effects of financial constraints on share issuance decisions.

The chapter consists of four parts. Section 6.2 describes the sample used to perform the empirical analysis and treats outliers. Section 6.3 presents the multivariate analysis of the effects of financial constraints on share issuance decisions. Section 6.4 discusses additional results, performs logit regression diagnostic tests, and undertakes robustness checks using different estimation techniques. Section 6.5 concludes the chapter.

-

³⁶ According to Howcroft et al. (2014) even the level of syndicated banks loans, which involves a group of banks borrowing to a single party, was reduced significantly during the financial crisis among European banks.

6.2. Sample Statistics

In this section, the descriptive statistics of the variables selected to undertake the empirical analysis are discussed. The variables presented here are used in prior studies to explain equity issuance decisions of firms. Thus, the variables allow the study to test the effects of market timing and financial constraints on share issuance among listed UK firms. Section 6.2.1 presents the descriptive statistics for the final sample after outliers are considered.

6.2.1. Descriptive Statistics of Variables

This section presents the descriptive statistics of independent and control variables used to undertake the empirical analysis. It focuses on the statistics of the firm characteristics of both equity issuing and non-issuing firms.

6.2.1.1. Descriptive Statistics of Sample

Table 6.1 presents the descriptive statistics of the variables selected to estimate the effects of financial constraints on share issuance decisions. In the presence of extreme values/outliers, the regression analysis will produce spurious results that will distort analysis. In other words, outliers are deviations from the normal value that would produce meaningful results, both in statistical terms and in theory. However, even when outliers are due to errors by removing them do not necessarily reduce estimation bias (Bollinger and Chandra, 2003). The method of treating outliers depends on the magnitude and the potential effects on the analysis of the regression estimations. Therefore, for this sample, most of the variables are winsorised. Whereas trimming and truncation of data reduces the sample size, winsorising variables only restricts the magnitude of the outliers while maintaining the sample size.

Table 6.1 provides descriptive statistics of some firm characteristics that distinguish equity issuing from non-issuing firms. It also performs independent sample t-test of the mean difference between issuing and non-issuing firms. This mean difference test provides preliminary differences in firm characteristics that define issuing and non-issuing firms. Panel A shows the descriptive statistics for the issuing firms whereas Panel B displays the descriptive statistics for the matched non-issuing firms. The mean difference

tests are reported in Panel C. Firms issue equity to coincide with periods of high stock price run-ups. Thus, market timing of equity issues occurs when firms have significantly high stock prices in the period leading up to the issuance decisions. These firms issue overvalued equity in an attempt to profit from the temporary high stock prices.

Table 6.1: Comparison of Issuing and Non-issuing Firms							
			f Issuing a	nd Non-	issuing F	<u>'irms</u>	
Panel A: Issuing firms							
Variables	N	Mean	Median	S.D	Min	Max	
MB	1144	4.32	2.08	6.07	0.00	28.36	
Prior AR	1252	0.07	0.04	0.28	-0.44	0.75	
ED	868	1.15	0.60	1.88	0.00	10.18	
SIZE	1144	5.12	5.06	0.73	3.13	8.08	
PROF	1257	-0.21	0.01	0.81	- 4.61	2.12	
CAPX	1137	0.05	0.02	0.11	-0.02	2.16	
KZ	815	3.31	1.22	6.30	-1.39	38.63	
Z	1154	-5.42	2.14	20.19	-87.00	19.25	
LIQD	1257	0.00	0.00	0.00	-0.00	0.01	
Panel B: N	on-issui	ng					
firms							
Variables	N	Mean	Median	S.D	Min	Max	
MB	779	2.04	1.69	1.34	0.57	3.89	
Prior AR	1248	0.01	0.00	0.32	-1.31	1.61	
ED	790	0.47	0.26	0.38	0.10	0.99	
SIZE	1257	4.29	4.34	1.90	-2.12	10.94	
PROF	1070	0.08	0.09	0.10	-0.06	0.20	
CAPX	1017	0.09	0.04	0.24	0.00	6.45	
KZ	992	0.06	-0.05	0.25	-0.05	1.32	
Z	1045	1.28	2.88	13.04	-53.55	34.68	
LIQD	1257	0.00	0.00	0.00	-0.00	0.01	
Variables	N	Mean	Median	S.D	Min	Max	
Panel C: T	est of M	ean Diff	erence of i	issuing a	nd non-i	ssuing	
Variables]	N	Mean di	ifference	1	-stat	
MB	19	23	2.2	8***	(1	0.32)	
AR	25	500	0.0	6***	(5.15)	
ED	16	558	0.6	8***	(1	0.01)	
SIZE	24	101	0.84^{-1}		(1	3.97)	
PROF	23	327	-0.28***		(-11.34)		
CAPX	21	54	-0 ()4***	(-	(-5.55)	
KZ	18	307	3.2	3 25***		(16.23)	
Z	21	99	-6.6	59***	,	(-9.13)	
LIQD	25	514	0.	00	(0.00)	

Notes: This table shows the descriptive statistics of the variables used in the empirical analysis for both issuing and non-issuing firms after winsoring extreme values or outliers. Variables included market to book ratio (MB), pre-announcement abnormal returns (Prior AR), earnings deviation measure of asymmetric information (ED), logarithm of book value of total assets (SIZE), the ratio of earnings before interest, taxes and depreciation to total assets (PROF), ratio of capital expenditure to total assets (CAPX), financial

constraints measure (KZ), Taffler's Z score measure of financial distress (Z), and market liquidity (LIQD). T-statistics are shown in parenthesis.

denote significance at the 1% level.

As can be seen from Table 6.1, the mean MB for issuing firms is 4.32 compared to 2.04 for non-issuing firms. AR is 0.07 or 7% for issuing firms and 0.01 or 1% for non-issuing firms. The mean difference for MB is 2.28 and statistically significant at the 1% level. For AR, the mean difference is 6% and significant at the 1% level. The table also shows mean difference of 0.68 for ED at 1% significance level. The difference in firm size is also positive (0.84) and significant at 1% level whereas PROF and CAPX differences are negative and significant at 1% level. Mean difference for KZ and Z score are significantly positive and negative respectively at the 1% level. Finally, the difference in LIQD is not statistically significant.

These results provide preliminary evidence consistent with the extant evidence on issuance of overpriced equity. Both MB and AR show that issuing firms are more overvalued and subsequently are more likely to time their issues than noissuing firms. Market timing of equity issues should be consistent with low information asymmetry (Lucas and McDonald, 1990; Korajczyk, et al., 1992; Baker and Wurgler, 2002). However, the results on ED indicate that issuing firms have high information asymmetry than non-issuing firms. This could possibly explain the post issue stock returns. It also appears that issuing firms are large firms, unprofitable and do not spend much on capital expenditure. Again, the low profitability could indicate that such firms are in need of external financing without which they would be unable to undertake investment projects. Thus, the need for "near term" cash flow is consistent with equity issuing (DeAngelo et al. 2010). It also appears that issuing firms are more internally financially constrained than non-issuing firms but not internally susceptible to financial distress. Thus, issuing firms have internal cash flow limitations and external financing is readily needed. The findings about equity issuing firms are in line with prior literature about market timing (Baker and Wurgler, 2002; DeAngelo et al., 2010; Dong et al., 2012).

6.2.1.2. Yearly Average of Equity Issues

Table 6.2 presents the number of yearly equity issues during the sample period, the average value of the equity proceeds, and the average ratio of the proceeds to market value. As can be seen from the table, the annual issues are generally stable during the early part of the sample period from 1994 to 1999. During years 2000 and 2001, the number of equity issues increases significantly and drops to the levels in the 1990s between years 2002 and 2008. For the equity issue proceeds, the table show that much of the proceeds were generated between the periods from 2006 to 2009. This result is rather interesting given that the same period experienced the effects of the financial crisis. However, the ratio of proceeds to market value (PMV) indicates a stable value of proceeds per unit of firm value. It appears the large firms issued more equity during the financial crisis compared to small firms. This evidence implies that the effects of financial crisis less constrained large firms in obtaining external equity financing than would for small firms.

Table 6.2: Yearly Average Amounts and							
Ratio of Equity Issues							
Year	N	Proceeds	PMV				
1994	63	67.24	0.30				
1995	63	122.70	0.35				
1996	83	57.97	0.35				
1997	51	44.57	0.20				
1998	49	57.00	0.23				
1999	58	61.67	0.26				
2000	126	76.12	0.29				
2001	104	136.19	1.80				
2002	51	153.04	0.69				
2003	77	25.92	0.11				
2004	65	31.38	0.33				
2005	44	76.41	0.34				
2006	47	100.67	0.36				
2007	83	129.40	0.30				
2008	54	113.38	0.45				
2009	159	250.36	1.95				
2010	80	59.37	0.23				
Total	1257	103.84	0.66				

Notes: This table shows the annual equity issues of sample firms. It also indicates the average yearly proceeds from equity issues (PROCEEDS) and the ratio of proceeds to market value of firms (PMV).

6.2.1.3. Pearson's Correlation Matrix of Variables

Logistic regression is used to estimate the model that tests the effects of internal financial constraints on share issuance (H_1 , H_2 , and H_3) as explained in Section 5.3.1.4. However, OLS is used to test hypothesis H_4 and H_5 about the stock returns following share issuance. 'No multicollinearity', which is an important assumption for the OLS, requires that the explanatory variables and control variables are not perfectly or highly intercorrelated with each other. In estimating regression coefficient using OLS, everything else being equal, it is better to have less correlation between independent variables (Wooldridge, 2009, p. 98). A correlation coefficient close to or equal to +1 and -1 suggests high collinearity or perfect multicollinearity among the variables. Table 6.3 provides the Pearson's correlation matrixes of independent and control variables for the sample of equity issues.

As the shown in Table 6.3, none of the variable indicates high collinearity with the highest being 0.61 for ED and KZ. In the multivariate analysis, separate results are run excluding the ED. Most of the variables are correlated at either the 1% or 5% level, with most of the correlations lying close or below 0.20. This results implies low multicollinearity and thus tolerable for OLS estimation. The high correlation between ED and KZ indicates perhaps high correlation between asymmetric information and internal financial constraints. Also, MB seems to indicate high internal financial constraints with correlation coefficient of 0.44. Overall, the results show that multicollinearity is not a problem for estimating the regression model using OLS.

	Table 6.3: Pearson's Correlation Matrix of Variables									
	KZ	MB	Prior AR	ED	SIZE	PROFIT	CAPX	PMV	CRISIS _{dum}	MKTLIQ
KZ	1.00									
MB	0.44^{***}	1.00								
Prior AR	0.10^{***}	0.14***	1.00							
ED	0.61***	-0.12***	0.03	1.00						
SIZE	-0.02	0.01	0.06^{**}	-0.01	1.00					
PROF	-0.28***	-0.23***	0.10^{***}	-0.01	0.23***	1.00				
CAPX	0.06^{*}	0.01	-0.06**	-0.02	-0.03	-0.13***	1.00			
PMV	0.04	-0.01	0.00	0.01	-0.06**	-0.01	-0.01	1.00		
CRISIS _{dum}	-0.05	-0.13***	0.01	0.12***	0.14***	-0.05*	0.01	0.04	1.00	
MKTLIQ	0.05	-0.08***	0.01	0.09^{***}	-0.04	0.00	-0.03	-0.02	-0.01	1.00

Notes: Variables are defined as KZ index measure of internal financial constraints (KZ), market to book ratio (MB), pre-announcement abnormal returns (Prior AR), earnings deviation as measure of asymmetric information (ED), logarithm of book value of total assets (SIZE), profitability (PROFIT), capital expenditure (CAPX), ratio of issue value to market value (PMV), financial crisis dummy (CRISIS_{dum}) and measure of market liquidity (MKTLIQ). *, ** and *** denote significance at the 10%, 5% and 1% level respectively.

6.3. Testing the Effects of Financial Constraints on Share Issuance

This section tests the hypotheses specified in Section 2.5.1 and 2.5.2 of Chapter 2 using the logit regression model (Model 1) stated in Chapter 5.

Model 1 discussed in Section 5.3.1.4 tests the effects of internal financial constraints on equity issues. The model is re-stated below:

$$\Pr(y_i = 1 | x_i) = \beta_0 + \beta_1 K Z_{it} + \beta_2 M B_{it} + \beta_3 M B * K Z_{ti} + \sum_{i=1}^n \beta_i X_{it} + \varepsilon_{it} \dots (2)$$

The model tests the probability of equity issuance given a set of firm characteristics for both the issuing and non-issuing firms. Since the model represent logit regression estimation, the dependent variable is 1 for issuing firm and 0 for non-issuing firm.

6.3.1. Internal Financial Constraints and Share Issuance

[Hypothesis H_1]

Hypothesis H_1 is re-stated below:

H_1 : There is high probability for less internally financially constrained firms to issue overvalued equity

Table 6.4 presents the logit regression results about the probability of equity issues when firms are both mispriced and internally financially constrained. Regression Model 1 shows the results for the effects of internal financial constraints on equity issues holding constant the firm level variables including market to book ratio. Model 2 gives the results of the logit regression when internal financial constraints and mispricing are interacted (MB*KZ). Models 3 and 4 uses prior AR to proxy for stock price run-ups as alternative measure of mispricing. Here the regression outputs also contain control variables as well as year and industry dummies to control for year and industry specific effects.

The coefficient of KZ is negative but statistically insignificant in Models 1 and 1. Consistent with the hypothesis, this negative coefficient indicates that internally financially constrained firms are less likely to issue equity. First, constrained firms might have significant asymmetric information effects and second they will be less likely to pay the costs of equity issuance. However, it

can be argued that constrained firm's even need external financing to enable them invest, especially when the investments are value-increasing. But the evidence seems to suggest that financial constraints limit the demand for external equity financing. Also, the coefficient of MB is positive and statistically significant at the 1% level. More importantly, MB*KZ is significantly positive. Overall, the results provide strong evidence for market timing of equity issues. For Models 3 and 4, where ED is excluded from the control variables (due to high correlation between KZ and ED), the coefficient of KZ significant at the 5% level in Model 3. The variable of interest is the interaction term MB*KZ which tests the probability of equity issues given simultaneous effects of mispricing and internal financial constraints. The coefficient of MB*KZ is positive and statistically significant at the 1% level. Thus, the results are similar with or without the ED variable.

Consistent with the developed hypothesis (H_I) , constrained firms are less likely to take advantage of issuing overvalued equity. This implies issuing overvalued equity dominates the presence of internal financial constraints. For example, whereas financially constrained firms will less likely issue equity, when this same equity is mispriced, the firms are more likely to issue even when they are financially constrained. These results do not support the findings by DeAngelo et al. (2010) and Dong et al. (2012) who assert that due to costs of equity issuance only unconstrained firms are likely to issue overvalued equity. In other words, financially unconstrained firms undertake market timing of equity issues. Constrained firms might issue overvalued equity when the overall market liquidity is high or general economic environment is favourable. These results could also mean that constrained firms are willing to trade short term advantage of issuing overvalued equity for subsequent stock price underperformance following the equity issues. The results of the odds ratios presented in Table 6.5 highlight the effects of internal financial constraints and mispricing on the probability to issue equity. For example, the odds ratio for MB*KZ of 5.257 in Model 2 implies constrained firms and overvalued firms are about 5 times more likely to issue equity than unconstrained and undervalued firms.

Table 6.4: Logit Regression Estimates of Internal Financial Constraints (KZ index) and Share Issuance

	Model 1	Model 2	Model 3	Model 4				
Dependent Variable is Issuance (1) vs Non-Issuance (0)								
KZ	-1.003***	-1.931***	-0.091	-0.451**				
	(0.00)	(0.00)	(0.60)	(0.03)				
MB	0.479***	0.433***	0.485***	0.453***				
	(0.00)	(0.00)	(0.00)	(0.00)				
MB*KZ		1.660***		0.937***				
		(0.00)		(0.01)				
SIZE	0.287^{***}	0.288^{***}	0.354^{***}	0.349***				
	(0.00)	(0.00)	(0.00)	(0.00)				
ED	1.160***	1.186***						
	(0.00)	(0.00)						
PROF	-4.314 ^{***}	-4.191 ^{***}	-5.377***	-5.203***				
	(0.00)	(0.00)	(0.00) -7.993***	(0.00)				
CAPX	-8.631***	-8.800		-8.208***				
	(0.00)	(0.00) -1.989***	(0.00) -1.825***	(0.00)				
Constant	-2.001	-1.989 ^{***}	-1.825***	-1.727***				
	(0.00)	(0.00)	(0.00)	(0.00)				
Year Dummies	Yes	Yes	Yes	Yes				
Industry Dummies	Yes	Yes	Yes	Yes				
N	1366	1366	1477	1477				
Pseudo R^2	0.315	0.322	0.254	0.258				
chi2	241.753	278.796	185.602	194.979				
p-value	0.000	0.000	0.000	0.000				

Notes: This table shows the logit regression results for equity issues. The dependent variable is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), market-to-book ratio (MB), the interaction term (MB*KZ), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROFIT), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.** and **** denote significance at the 5% and 1% levels respectively.

Table 6.5: Logit Regression Estimates (Odds ratio) of Share Issuance							
	Model 1	Model 2	Model 3	Model 4			
Dependent Variable is Issuance (1) vs Non-Issuance (0)							
KZ	0.367***	0.145***	0.913	0.637**			
	(0.00)	(0.00)	(0.60)	(0.03)			
MB	1.614***	1.542***	1.624***	1.574***			
	(0.00)	(0.00)	(0.00)	(0.00)			
MB*KZ		5.257***		2.552***			
		(0.00)		(0.01)			
SIZE	1.332***	1.334***	1.425***	(0.01) 1.418***			
	(0.00)	(0.00)	(0.00)	(0.00)			
ED	3.190***	3.275***					
	(0.00)	(0.00)					
PROF	0.013***	0.015***	0.005^{***}	0.005^{***}			
	(0.00)	(0.00)	(0.00)	(0.00)			
CAPX	$(0.00) \\ 0.000^{***}$	$(0.00) \\ 0.000^{***}$	0.000^{***}	0.000***			
	(0.00)	(0.00)	(0.00)	(0.00)			
Year Dummies	Yes	Yes	Yes	Yes			
Industry Dummies	Yes	Yes	Yes	Yes			
N	1366	1366	1477	1477			
pseudo R^2	0.315	0.322	0.254	0.258			
chi2	241.753	278.796	185.602	194.979			
p-value	0.000	0.000	0.000	0.000			

Notes: This table shows the exponentiated coefficients of the logit regression results for equity issues. The dependent variable is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), market-to-book ratio (MB), interaction term (MB*KZ), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.** and *** denote significance at the 5% and 1% levels respectively.

6.3.2. The Effects of External Financial Constraints

[Hypotheses H_2 and H_3]

The analysis and results produced in this section tests the following hypotheses about the effects of external financial constraints on share issuance decisions given the level of firm's internal financial constraints.

 H_2 : Firms are more likely to issue overvalued equity when external financial constraints are low (non-financial crisis period or periods of high market liquidity)

 H_3 : The probability of issuing overvalued equity during periods of high external financial constraints increases for firms with low

internal financially constraints than those with high internal financial constraints

The results in Table 6.6 represent the analysis of the effects of external financial constraints on the interaction between mispricing and internal financial constraints on share issuance decisions. Financial crisis and market liquidity measure external financial constraints. Pre-crisis period is defined as the period between 1994 and 2007 whereas crisis period is from 2008 to 2010. Thus, between 2008 and 2010 there are external financial constraints compared to the period prior to year 2008. Similarly, low market liquidity represents external financial constraints compared to period of high market liquidity (refer to Section 5.3.1.2 of Chapter 5. for discussions on market liquidity). The independent and control variables are the same as those used in Table 6.4. Again, Table 6.6 provides the results when financial crisis is used as proxy for external financial constraints whereas Table 6.7 gives the results for when market liquidity is a measure of external financial constraints. The results for the crisis period are shown in Models 1 and 2 of Table 6.6 respectively. Models 3 and 4 are the results for non-crisis period. Hypothesis H₂ asserts that firms issue overvalued equity during non-crisis or periods of high market liquidity. Conversely, hypothesis H₃ argues that unconstrained firms are able to issue overvalued equity during external financial constraints.

The coefficient for KZ is negative and statistically significant at the 1% level for Models 1 and 3. This implies, as already hypothesised, that internally financially constrained firms are less likely to issue equity. Since these effects are consistent in both crisis and non-crisis periods, it also implies that the equity issuance behaviour of internally financially constrained firms is independent of the level of external financial constraints. On the other hand, and in line with the market timing literature, the positive and statistically significant coefficient for MB in both Models 1 and 3 suggests that regardless of the level of external financial constraints, firms are more likely to issue overvalued equity. These results also show that large firms and firms with high asymmetric information are likely to issue equity regardless of external financial constraints. However, less profitable firms and those that undertake

less capital expenditure are more likely to issue equity. It appears profitable firms do not need external financing since they would have accumulated internal funds to undertake corporate expenditure. It also suggests that profitable firms follow the pecking order theory of external financing by which firms would issue equity only after exhausting both internal funds and debt capacity.

Table 6.6: The Effects of External Financial Constraints (Financial Crisis) on Share Issuance

Crisis) on Share Issuance								
Cri	sis	Non-	Crisis					
Model 1	Model 2	Model 3	Model 4					
Dependent Variable is Issuance (1) vs Non-Issuance (0)								
-3.233***	-1.059***	-1.254***	-0.251					
(0.00)	(0.00)	(0.00)	(0.38)					
0.248***	0.461***	0.458***	0.443***					
(0.01)	(0.00)	(0.00)	(0.00)					
-3.045***	-1.099	1.337***	(0.00) 1.239***					
(0.00)	(0.12)	(0.00)	(0.01) 0.393***					
0.325^{***}	0.377^{***}	0.315***	0.393***					
(0.01)	(0.00)	(0.00)	(0.00)					
2.255***		0.807^{***}						
(0.00)		(0.00)						
-7.120***	-11.726***	-3.337***	-3.958***					
(0.00)	(0.00)	(0.00)	(0.00)					
-10.353***	-8.098***	-8.691 ^{***}	-8.421***					
(0.00)	(0.00)	(0.00)	(0.00)					
-2.273**	-1.990 ^{***}	-2.358***	-2.430 ^{***}					
(0.01)	(0.00)	(0.00)	(0.00)					
Yes	Yes	Yes	Yes					
Yes	Yes	Yes	Yes					
368	394	998	1083					
0.502	0.342	0.291	0.263					
130.816	72.249	175.377	172.479					
0.000	0.000	0.000	0.000					
	Cri Model 1 Variable is Iss -3.233*** (0.00) 0.248*** (0.01) -3.045** (0.00) 0.325*** (0.01) 2.255* (0.00) -7.120*** (0.00) -10.353*** (0.00) -2.273* (0.01) Yes Yes 368 0.502 130.816	Crisis Model 1 Model 2 Variable is Issuance (1) vs Notes -3.233*** -1.059*** (0.00) (0.00) 0.248*** 0.461*** (0.01) (0.00) -3.045*** -1.099 (0.00) (0.12) 0.325*** 0.377*** (0.01) (0.00) -7.120*** -11.726*** (0.00) -10.353** -8.098*** (0.00) -2.273 -1.990** (0.01) (0.00) Yes Yes Yes Yes 368 394 0.502 0.342 130.816 72.249	Crisis Non-Issuance (I) Wariable is Issuance (I) vs Non-Issuance (I) -3.233*** -1.059*** -1.254*** (0.00) (0.00) (0.00) 0.248*** 0.461*** 0.458**** (0.01) (0.00) (0.00) -3.045*** -1.099 1.337*** (0.00) (0.12) (0.00) 0.325*** 0.377*** 0.315*** (0.01) (0.00) (0.00) 2.255*** 0.807*** (0.00) (0.00) (0.00) -7.120*** -11.726*** -3.337** (0.00) (0.00) (0.00) -10.353*** -8.098*** -8.691*** (0.00) (0.00) (0.00) -2.273* -1.990*** -2.358** (0.01) (0.00) (0.00) Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes Yes					

Notes: This table displays the logit regression results for equity issues given internal and external financial constraints. The results for the crisis period are given in Model 1, noncrisis period subsample in Model 2, low market liquidity in Model 3 and high market liquidity in Model 4. The dependent variable for this logit regression is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), market-to-book ratio (MB), the interaction term (KZ*MB, logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. ** and **** denote significance at the 5% and 1% level, respectively.

Additional variable, MB* KZ, measures the simultaneous effects of mispricing and internal financial constraints for both low and high external financial constraints. For the crisis period in Model 1, the coefficient of MB* KZ is - 3.045 which is significant at the 1% level. However, the coefficient of MB*KZ for the non-crisis period is 1.337 and significant at the 1% level. In Model 4 where ED is excluded, the coefficients of MB*KZ is still positive and statistically significant at the 1% level. Table 6.7 produces the logit regression using market liquidity as a measure of external financial constraints. The coefficient of MB* KZ is again negative and significant at the 1% level for the low market liquidity subsample given in Model 2. This is different for the high market liquidity subsample where the coefficient of MB* KZ is positive and significant at the 5% level as given in Model 3. Thus, the two definitions of external financial constraints produce similar results for the effects of mispricing and internal financial constraints on share issuance decisions.

Consistent with the developed hypothesis, and prior study by McLean and Zhao (2014), frictions in the external equity market or the general economic conditions seem to have significant effects on the share issuance decisions of firms. Whereas internal financial constraints determine the demand for external financing, the level of external financial constraints influences the amount of capital that investors would be willing to provide to firms through equity offerings. Thus, market timing behaviour depends to a large extent on the overall economic conditions when firms will be better off to issue equity during periods of favourable market conditions (Korajczyk and Levy, 2003). According to the conjecture about the effects of external financial constraints, the benefits of issuing overvalued equity seem to overshadow the effects of external financial constraints for financially unconstrained firms. Similarly, for the constrained firms, the effects of internal financial constraints do not seem to impact the issue of overvalued equity when the overall external market conditions are favourable.

Overall, these results provide strong support for the hypothesis about the issuance of equity and the effects of mispricing and financial constraints. Therefore, the hypotheses H₁, H₂, and H₃ are accepted based on the results in

this section of the study. Unlike previous studies, this thesis performs analysis of equity issues for simultaneous effects of internal and external financial constraints. In other words, the issue of overvalued equity in the equity marketplace reflects both the degree of internal cash flows and external costs associated with such issues.

Table 6.7: The Effects of External Financial Constraints (Market Liquidity) on Share Issuance

Liquidity) on Share Issuance								
	Low Marke	et Liquidity_	High Marke	et Liquidity				
	Model 1	Model 2	Model 3	Model 4				
Dependent Variables is Issuance (1) vs Non-Issuance (0)								
KZ	-2.038***	-0.414	-1.910***	-0.542				
	(0.00)	(0.11)	(0.00)	(0.20)				
MB	0.426***	0.420***	0.527***	0.591***				
	(0.00)	(0.00)	(0.00)	(0.00)				
MB*KZ	-1.689***	-0.918**	1.667**	0.829				
	(0.00)	(0.02) 0.285***	(0.03)	(0.22)				
SIZE	0.224***	0.285***	0.452***	0.534***				
	(0.00)	(0.00)	(0.00)	(0.00)				
ED	1.298***		1.060***					
	(0.00)		(0.00)					
PROF	-3.448***	-4.680 ^{***}	-6.762***	-7.032***				
	(0.00)	(0.00)	(0.00)	(0.00)				
CAPX	-9.049 ^{***}	-8.509 ^{***}	-6.856	-6.910				
	(0.00)	(0.00)	(0.21)	(0.14)				
Constant	-1.781***	-1.338***	-2.749**	-2.920***				
	(0.00)	(0.01)	(0.03)	(0.00)				
Year Dummies	Yes	Yes	Yes	Yes				
Industry Dummies	Yes	Yes	Yes	Yes				
N	962	1033	404	444				
Pseudo R^2	0.296	0.223	0.425	0.367				
chi2	196.177	130.704	120.903	113.487				
p-value	0.000	0.000	0.000	0.000				

Notes: This table displays the logit regression results for equity issues given internal and external financial constraints. The results for the crisis period are given in Model 1, noncrisis period subsample in Model 2, low market liquidity in Model 3 and high market liquidity in Model 4. The dependent variable for this logit regression is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), market-to-book ratio (MB), the interaction term (KZ*MB), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.** and *** denote significance at the 5% and 1% level, respectively.

These results are consistent with the evidence of prior studies about market timing of equity issues (e.g. Baker and Wurgler, 2002; Dong et al., 2012).

Thus, internally financially constrained firms are less likely to issue overvalued equity. The evidence in this section further demonstrates that unconstrained firms issue overvalued equity even during periods of significant external financial constraints. Also, for the constrained firms the probability of issuing overvalued equity is evidenced only during periods of favourable economic conditions. These findings are similar to studies such as Corwin (2003); Butler et al. (2005); McLean and Zhao (2014) which argue that during crisis periods, market liquidity dampens and increases the costs of raising external equity financing. For instance, Naes et al. (2011) show that U.S. stock market liquidity is countercyclical, and argue that during recessions there is a "flight to quality" in equities. Moreover, the results of the control variables seem to provide further evidence of market timing of equity issues among UK listed firms.

6.3.3. Share Issuance, Financial Constraints and Stock Returns

[Hypothesis H_4 and H_5]

The hypotheses about stock returns are discussed in Section 2.5.4 of Chapter 2. This section tests these hypotheses using the pooled OLS regression model (Model 2) stated in Section 5.3.1.4 of Chapter 5. The hypotheses are re-stated below:

H₄: Issuance of overvalued equity is associated with more negative abnormal returns, especially for firms with high internal financially constraints

 H_5 : Issuance of overvalued equity is associated with more negative abnormal returns, especially for firms with high internal financially constraints and during periods of high external financial constraints

The pooled OLS estimation model used to test the hypothesis about stock returns as discussed in Section 5.3.1.4 of Chapter 5 is stated below.

$$CAR_{i} = \beta_{0} + \beta_{1}KZ_{i} + \beta_{2}MB_{i} + \beta_{3}KZ * MB_{i} + \sum_{i=1}^{n} \beta_{i}X_{i} + \varepsilon_{i}.....(2)$$

From the model above, the dependent variable is the cumulative abnormal returns (CAR) over 6 months post equity issue. KZ and MB are the main

independent variables with set of control variables defined in Section 5.3.1.3 of Chapter 5.

6.3.3.1. Univariate Results for Stock Returns

The extant literature provides substantial evidence of stock price underperformance following equity issues. Various explanations have been advocated to justify the stock price performance of equity issuing firms as discussed in Section 2.4 of Chapter 2. Overall, evidence suggests that stock underperformance is consistent with market timing hypothesis and overvaluation for both initial public offerings and seasoned equity offerings (Loughran and Ritter, 1995; Spiess and Affleck-Graves, 1995; Baker and Wurgler, 2000, 2002; Dong et al. 2012). Thus, by issuing overvalued equity, firms suffer from adverse market perception that the value of the firm does not reflect the true value of the fundamentals. It seems to follow that overvalued equity persists when there is significant information asymmetry associated with the equity issuing firms.

This section provides results about the short-term stock returns associated with equity issues depending on both mispricing and internal financial constraints. It further looks at how the returns behave when there are external financial constraints, particularly during the financial crisis. According to the developed hypotheses, all else equal, constrained firms that issue overvalued equity should underperform counterpart unconstrained firms. Also, the returns should be more negative for financially constrained firms that issue overvalued equity during periods of financial crisis. The rationale for this expectation is that the firm suffers from both internal constraints and possible adverse effects of friction in the external equity market and the general unfavourable economic conditions (Ivashina and Scharfstein, 2010; and McLean and Zhao, 2014).

Table 6.8 shows the univariate results. Panel A examines the abnormal stock returns for the full sample partitioned into high/low MB/KZ in line with hypothesis H_5 . According to H_5 , overvalued firms are associated with more negative abnormal returns, especially when they are faced with high internal financial constraints. The total abnormal returns are -9.8% (significant at the

1% level) and insignificant -0.8% for high MB and low MB respectively. The mean difference returns between high and low MB is -9.5% which is statistically significant at the 1% level. For the KZ partition, high and low KZ subsamples record abnormal returns of respectively -6.0% and 3.4%, with both significant at the 1% level. The mean difference is -2.5% which is insignificant. For the high KZ subsample, the abnormal returns for high MB and low MB are -10.2% (significant at 1% level) and 0.2% (not significant) with a mean difference which is statistically significant the 1% level. This evidence is similar to Lamont et al. (2001) who argue that financially unconstrained firms report less negative stock returns than more constrained firms. However, these results contradict Whited and Wu (2006) who explain that the financial-constraints effects dominates the firm size effects. For the low KZ firms, the mean difference returns between high and low MB is negative and statistically significant at the 1% level. Thus, the abnormal returns indicate that internal financial constraints and mispricing determines the level of stock price underperformance following equity issues. In particular, it appears that financially constrained firms perform worse when they issue overvalued equity than when the equity overvaluation is low.

Panels B and C show the abnormal returns based on the period of financial crisis and pre-crisis, respectively. Thus, the sample is split into two subsamples for crisis and non-crisis period to investigate the effects of external financial constraints on the results. The mean abnormal stock returns difference between high KZ and low KZ, during crisis period as shown in Panel B, is -11.4% and is significant at 5% level. However, for non-crisis subsample in Panel C, the mean difference returns between high KZ and low KZ firms is insignificant. For the high KZ firms, the mean difference returns between high and low MB is negative and significant at 1% level for the non-crisis period. The same evidence is true for the low KZ firms. It appears the market is unable to differentiate between high and low KZ during the pre-crisis period. Therefore, the market seems to react unfavourably to both firms when equity issues are made.

PANEL A: Full Sample MB and KZ							
		High KZ	Low KZ	Total	Mean diff.		
	N	1	2	3	t-stat		
High MB	408	-0.102	-0.068	-0.098	-0.033		
		(-6.41)***	(-1.53)	(-8.21)***	(-0.72)		
Low MB	404	0.002	-0.024	-0.008	0.025		
		(0.09)	(-0.97)	(-1.64)	(0.86)		
Total	812	-0.060	-0.034	-0.053	-0.025		
		(-4.16)***	(-6.30)***	(-5.89)***	(-1.06)		
Mean diff		-0.104***	(-6.30)*** -0.087***	(-5.89)*** -0.095			
t-stat		(-5.09)	(-2.98)	(-5.16)			
PANEL B	: Durin	g Crisis MB a	nd KZ				
	N	High KZ	Low KZ	Total	Mean diff.		
High MB	62	-0.099	0.022	-0.083	-0.121		
		(-3.56)***	(0.30)	(-3.05)***	(-1.54)		
Low MB	168	-0.047	0.023	-0.006	-0.070		
		(-0.90)	(0.41)	(-0.29)	(-0.87)		
Total	230	-0.091	0.023	-0.063	-0.114		
		(-3.38)***	(0.05)	(-2.76)***	(-2.26)**		
Mean diff		-0.052	-0.011	-0.066			
t-stat		(-0.93)	(-0.15)	(-1.48)			
PANEL C	PANEL C: Pre-Crisis MB and KZ						
	N	High KZ	Low KZ	Total	Mean diff.		
XX: 1 X (D)	A 10	0.420	0.40.	0.400	0.00.		

Table 6.8: Univariate Results of Abnormal Stock Returns

IANEL	TANEL C. TIE-CISIS WID and KZ								
	N	High KZ	Low KZ	Total	Mean diff.				
High MB	240	-0.139	-0.105	-0.109	-0.035				
		$(-2.69)^{**}$	(-5.49)***	(-7.65)***	(-0.62)				
Low MB	342	-0.038	0.007	-0.008	-0.045				
		(-1.41)	(0.40)	(-0.49)	(-1.41)				
Total	582	-0.057	-0.048	-0.050	-0.009				
		(-3.84)	(-4.66)***	(-5.21)***	(-0.34)				
Mean diff		-0.109	-0.116	-0.108					
t-stat		$(-3.41)^{***}$	$(-5.16)^{***}$	$(-5.26)^{***}$					

Notes: The table shows the abnormal stock returns associated with share issues. Panel A displays the results for the short run returns based for the full sample partitioned into high/low KZ and MB. Panel B shows the stock returns for crisis period (2008-2010) and Panel C for the non-crisis period (1994-2007). T-statistics are shown in parenthesis. ** and *** denote significance at the 5% and 1% level, respectively.

Overall, the evidence is consistent with prior findings about post-issue stock returns. But importantly, internal financial constraints seem to explain the level of stock underperformance for overvalued firms. Given that the results in Section 6.5.1 suggest that financially constrained firms are less likely to issue overvalued equity, the more negative stock returns associated with such firms justify why these firms would rather not issue overvalued equity. Financially unconstrained firms are better off than constrained firms when they issue overvalued equity. These findings are consistent with Dong et al. (2012) who

indicate that due to issuance costs, unconstrained firms rather than constrained firms should issue overvalued equity. However, the univariate results on post-issue stock returns do not entirely support the findings by Korajczyk and Levy (2003) and McLean and Zhao (2014) about the effects of external financial constraints on equity issuance. This can be extended to imply that the market is unable to differentiate better performing firms from poor performing firms when there are adverse general economic conditions. In sum, the equity issuance decisions and the subsequent stock price performance is determined by the combined effects of both mispricing and internal/external financial constraints.

6.3.3.2. Cross-Sectional Regression Results for Stock Returns

Table 6.9 reports the results for pooled cross-sectional regressions estimation of the short run abnormal stock returns associated with equity issues. The dependent variable is short-run abnormal stock returns, CAR (2,120) using Model 2 discussed in Section 5.3.1.1 and Section 5.3.1.4 of Chapter 5 respectively. According to hypothesis H₄ financially constrained firms that issue overvalued equity are associated with more negative abnormal stock returns than unconstrained firms. Hypothesis H₅, on the other hand, explores the effects of external financial constraints on the stock returns of equity issuing firms given the level of internal financial constraints. Model 1 provides the results for the full sample, Model 2 gives the results for the sample period during the financial crisis and Model 3 restricts the sample period to the precrisis period. As usual, pre-crisis period is defined as the period between 1994 and 2007 whereas crisis period is from 2008 to 2010. The independent and control variables are the same as those used in Table 6.4.

From Model 1 for the full sample, the coefficient of KZ is not statistically significant. Also, MB has a coefficient of 0.008 that is statistically significant at the 1% level. The coefficients of SIZE and ED are not significant with PROF also showing a positive and statistically significant coefficient at the 1% level. CAPX shows a positive and marginally significant coefficient at the 10% level. These results imply that the level of internal financial constraints do not affect stock returns of equity issuing firms. MB does not show the expected

sign for negative effects of overvaluation on stock returns. It also shows that profitable firms are associated with positive stock price response whereas capital expenditure has marginal positive effects on stock returns. Finally, the coefficient of MB*KZ is not statistically significant for the full sample.

Table 6.9: Cross-sectional Regression Analysis of Share

Issuance and Short Run Stock Returns										
	Full	Crisis	Non-Crisis							
	Model 1	Model 2	Model 3							
Dependent Variable is the Post-announcement CAR _(2,120)										
KZ	0.068	0.322***	-0.048							
	(0.26)	(0.01)	(0.45)							
MB	0.008***	0.009	0.008^{***}							
	(0.00)	(0.14)	(0.01)							
MB*KZ	-0.065	-0.296**	0.030							
	(0.27)	(0.02)	(0.63)							
SIZE	0.027	0.026	0.036^{*}							
	(0.11)	(0.47)	(0.08)							
ED	-0.013	-0.042**	-0.005							
	(0.16)	(0.02)	(0.65)							
PROF	0.056^{***}	0.024	0.054***							

(0.00)

 0.240^*

(0.05)

-0.198^{*}

(0.05)

Yes

Yes

756

CAPX

N

Constant

Year Dummies

Industry Dummies

(0.59)

-0.040

(0.89)

-0.117

(0.54)

Yes

Yes

226

(0.00)

 0.302^*

-0.293

(0.02)

Yes

Yes

530

(0.02)

adj. R^2 0.125 0.150 0.132 Notes: This table shows the cross-sectional regression results for stock returns. The results for the full sample are given in Model 1, crisis period subsample in Model 2 and non-crisis period subsample in Model 3. The dependent variable CAR (2,120). Variables included in the regressions include KZ measure of internal financial constraints, market-to-book ratio (MB), the interaction term (KZ*MB), prior stock returns (Prior AR), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.*, ** and *** denote significance at the 10%, 5% and 1% level, respectively.

The results for the crisis and non-crisis subsamples shed more light on the stock returns impacts of equity issuance decisions. In Models 2 and 3, the coefficient of KZ is positive and significant at 1% level and negative and insignificant respectively. MB is positive in both models but only significant in Model 3. The coefficient of ED is negative and statistically significant at the

5% level in Model 2 indicating the adverse selection costs of equity issues. Profitable firms and capital expenditure are associated with positive post issue returns in the non-crisis period. More importantly, the coefficient of the interaction term MB*KZ is negative and statistically significant at the 5% level for the crisis period whereas it is not significant during the non-crisis period.

It appears the multivariate analysis suggests that the effects of internal financial constraints and mispricing on stock returns of equity issuing firms are influenced by the crisis period. Periods of financial crisis between 2008-2010 are characterised by significant slump in stock prices. Moreover, the supply of equity financing is low with high associated transaction costs. Taken together, these features of the stock market during financial crisis mean that firms issuing equity are likely to generate low post-issue stock returns. Given that overvaluation and internal financial constraints are associated with stock price underperformance of equity issuing firms, the hypothesis about stock price is accepted. Hence, the issue of overvalued equity by internally financially constrained firms is associated with stock price underperformance, especially during significant external financial constraints.

6.4. Robustness Checks and Additional Results

This section provides additional analysis to ensure the results are robustness to alternative definitions of internal financial constraints. It also presents logit regression diagnostic tests and the results using alternative estimation technique for the main results.

6.4.1. Internal Financial Constraints (WW index) and Share Issuance

The results in Table 6.10 analyses the effects of financial constraints on share issuance in testing hypothesis H_I . Here, the financial constraints measure is the Whited and Wu (2006) index. Using alternative measure of internal financial constraints enhances the robustness of the results when the KZ-index was used. Studies such as Chen and Wang (2012) use this measure together with the KZ-index to test the effects of financial constraints on share repurchase. Models 1 and 2 use market to book ratio as proxy for mispricing, whereas Models 3 and 4 use prior stock returns as measure of mispricing. This is similar to Table 6.4

and also includes the interaction term MB*WW to measure the effects of mispricing and internal financial constraints on share issuance as shown in Models 2 and 4.

Table 6.10: Log	it Regression ints (WW ind			nancial								
Constra	Model 1	Model 2	Model 3	Model 4								
Dan and Jant I	Dependent Variable is Issuance (1) vs Non-Issuance (0)											
Dependent variable is issuance (1) vs ivon-issuance (0)												
WW	0.171	0.651**	-0.047	0.391^{*}								
	(0.38) 0.433****	(0.01)	(0.78)	(0.06)								
MB	0.433***	(0.01) 0.479^{***}	(0.78) 0.414^{***}	(0.06) 0.449***								
	(0.00)	(0.00)	(0.00)	(0.00)								
MB*WW		(0.00) 1.049***		0.988***								
		(0.00)		(0.00)								
SIZE	0.277^{***}	0.286***	0.317***	(0.00) 0.332^{***}								
		(0.00)	(0.00)	(0.00)								
ED	(0.00) 1.003****	0.981***										
	(0.00) -3.858***	(0.00)										
PROF	-3.858***	-3.850***	-4.598 ^{***}	- 4.697***								
	(0.00)	(0.00)	(0.00) -5.679***	(0.00)								
CAPX	-9.552 ^{***}	-9.512***	-5.679 ^{***}	-5.655 ^{***}								
	(0.00) -1.399**	(0.00) -1.535***	(0.00)	(0.00)								
Constant	-1.399 ^{**}	-1.535 ^{***}	-0.897*	-1.083**								
	(0.01)	(0.01)	(0.06)	(0.02)								
Year Dummies	Yes	Yes	Yes	Yes								
Industry Dummies	Yes	Yes	Yes	Yes								
N	1176	1176	1517	1517								
Pseudo R^2	0.301	0.307	0.226	0.232								
chi2	203.286	208.807	117.392	127.065								

Notes: This table shows the logit regression results for equity issues. The dependent variable is I for share issuance and 0 for non-issuance. Variables included in the regressions include WW-index measure of internal financial constraints (WW), market-to-book ratio (MB), the interaction term (MB*WW), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROFIT), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.** and **** denote significance at the 5% and 1% levels respectively.

0.000

0.000

0.000

0.000

p-value

From the table, the coefficient of the interaction term MB*WW is positive and statistically significant at the 1% level. This result indicates that financially constrained firms issue overvalued equity. But perhaps, as argued in hypothesis H_2 the effects of external financial constraints might drive the issue of overvalued equity beyond the effects of internal financial constraints. What the results show is both KZ index and WW index provide similar outcome of the

effects of internal financial constraints on share issuance. The positive coefficient of MB*WW in Model 4 is similar to that in Model 2 and indicates that excluding ED due to high correlation with KZ does not affect the results.

6.4.2. Share Issuance and the Effects of Financial Distress

Additional tests of the probability of equity issues use financial distress to proxy for internal financial constraints. Firms that have high costs of financial distress are highly levered firms with limited debt capacity to issue debt financing and hence suffer from significant costs of bankruptcy. As explained earlier, financially distressed firms are more likely to require external equity financing unless high costs of external financing restricts their ability to issue overvalued equity. Also, given that equity issues are associated with high transactions costs, financially distressed firms may be unable to afford such issues. In the logit regression in Table 6.11, Z is the Taffler's Z score measure of financial distressed discussed in Section 5.3.1.2 of Chapter 5. The control variables are the same as used in the previous regressions.

In line with the hypothesis, the coefficient of the interaction term MB*Z is negative for the full sample (Model 1) and the crisis period subsample (Model 2) and statistically significant at the 1% level and 5% level respectively. In Model 3, which represents the non-crisis period subsample, the coefficient of Z*MB is positive and significant at the 1% level. Like previous results, this result indicates that financially distressed firms are less likely to issue overvalued equity, especially during financial crisis. However, in the non-crisis period these firms are more likely to issue overvalued equity apparently because the external market conditions favour stock price performance than during crisis period. Thus, costs of external financing play a key role in market timing. Costs of external financing reflect lack of available finance and low investor sentiments such that investors are less likely to purchase shares that are riskier during financial crisis and/or low market liquidity. Overall, the findings are consistent with financial constraints impact on market timing.

Table 6.11: The Eff	ects of Finan	cial Distress or	Share Issuance
	Full	Crisis	Non-crisis
	Model 1	Model 2	Model 3
Dependent Var	iable is Issua	nce (1) vs Non-I	ssuance (0)
Z	0.132	0.883**	-0.352
	(0.52) 0.550***	(0.04)	(0.20)
MB	0.550^{***}	(0.04) 0.386***	(0.20) 0.543****
	(0.00)	(0.00)	(0.00) 1.045***
MB*Z	-1.406 ^{***}	(0.00) -1.514**	1.045***
	(0.00) 0.293***	(0.03) 0.319^{**}	(0.00) 0.333****
SIZE	0.293***	0.319^{**}	0.333***
	$(0.00) \\ 0.977^{***}$	(0.02)	(0.00)
ED	0.977^{***}	1.912***	0.637***
	(0.00) -4.568***	(0.00) -6.808***	$(0.00)_{***}$
PROF	-4.568 ^{***}	-6.808***	-3.976 ^{***}
	(0.00)	(0.00) -10.147***	(0.00) -8.097***
CAPX	(0.00) -8.317***	-10.147***	-8.097***
	(0.00) -1.729***	(0.00)	$(0.00)_{***}$
Constant	-1.729 ^{***}	-2.340**	-2.048***
	(0.00)	(0.02)	(0.00)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	1299	354	945
Pseudo R^2	0.311	0.464	0.290
chi2	227.144	92.252	157.826
p-value	0.000	0.000	0.000

Notes: This table shows the logit regression results for equity issues given financial distress. The results for the full sample are given in Model 1, crisis period subsample in Model 2 and non-crisis period subsample in Model 3. The dependent variable for this logit regression is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include Taffler's Z score measure of financial distress (Z), market-to-book ratio (MB), the interaction term (Z*MB), prior stock returns (Prior AR), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. *, ** and **** denote significance at the 10%, 5% and 1% level, respectively.

6.4.3. Tobit Regression Estimation of Equity Issues

Table 6.12 provides the tobit regression results for equity issues given financial constraints and mispricing. The dependent variable is the dollar value of equity issues divided by the market value of equity. This variable explains the effects of financial constraints and mispricing on the amount of shares issued. Model 1 give the results of the effects of internal financial constraints (KZ) and market to book ratio (MB) on the value of equity issued whereas Model 2 examines

the simultaneously effects of mispricing and internal financial constraints (MB*KZ) on the value of equity issued. Models 3 and 4 exclude ED as a control variable due to high correlation with KZ. As can be seen from the table, the coefficient of MB*KZ is positive and statistically significant at the 1% level in Models 1 and 3, indicating that financially constrained firms issue overvalued equity.

Table 6.12: Tobit Regression Analysis of Internal Financial										
C	onstraints an	d Share Issu	ance							
	Model 1	Model 2	Model 3	Model 4						
Dependent Variable is Proceeds over Market Value (P/MV)										
KZ	1.697***	0.324	1.205****	-0.130						
	(0.00)	(0.27)	(0.00)	(0.54)						
MB	-0.000	-0.013	0.014	-0.011						
	(0.99)	(0.31)	(0.24)	(0.36)						
MB*KZ		1.665***		1.966***						
		(0.00)		(0.00)						
SIZE	-0.880***	-0.876***	-0.831***	-0.840 ^{***}						
	(0.00)	(0.00) -0.095**	(0.00)	(0.00)						
ED	-0.187***	-0.095**								
	(0.00)	(0.01)								
PROF	0.234***	0.254***	0.166^{**}	0.211***						
	(0.00)	(0.00)	(0.03)	(0.01)						
CAPX	-0.269	-0.407	-0.148	-0.396						
	(0.78) 8.530***	(0.67)	(0.88)	(0.68)						
Constant	8.530***	8.448***	7.956***	8.017***						
	(0.00)	(0.00)	(0.00)	(0.00)						
Year Dummies	Yes	Yes	Yes	Yes						
Industry Dummies	Yes	Yes	Yes	Yes						
N	752	752	763	763						
Pseudo R^2	0.006	0.006	0.006	0.006						
p-value	0.000	0.000	0.000	0.000						

Notes: This table shows the Tobit regression results for equity issues. The dependent variable is the dollar value of share repurchase divided by the prior year market value of equity. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), market-to-book ratio (MB), the interaction term (MB*KZ), prior stock returns (Prior AR), the interaction term (AR*KZ), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.** and **** denote significance at the 5% and 1% level respectively.

6.4.4. Random and Fixed Effects Logit Regression Estimation

The random and fixed effects regressions are estimated as robustness check. Whereas logit regression is the main estimation technique and estimates as cross-sectional regression, random and fixed effects accounts for the time series aspect of the data. Fixed effects estimation leads to significant loss in efficiency that is compensated by consistent coefficients. Thus, standard errors are larger for fixed effects estimation because of loss of observations by dropping all time-invariant regressors. Coefficients of time-varying independent variables may even be difficult to identify in nonlinear models with fixed effects (Cameron and Trivedi, 2005, p. 715). Fixed effects use only within variation of the regressors and are much less precise compared to random effects model. This contrasts with the random effects estimation that corrects for individual correlation over time.

The random effects logit regression model is stated as follows:

$$\Pr(y_{it} = 1 | x_{it}) = \beta_0 + \beta_1 K Z_{it} + \beta_2 M B_{it} + \beta_3 M B * K Z_{it} + \sum_{i=1}^n \beta_i X_{it} + \alpha_i + \varepsilon_{it} \dots (1)$$

And the fixed effects logit regression model is given below.

$$\Pr(y_{it} = 1 | x_{it}) = \beta_1 K Z_{it} + \beta_2 M B_{it} + \beta_3 M B * K Z_{it} + \sum_{i=1}^{n} \beta_i X_{it} + \alpha_i + \varepsilon_{it} \dots (2)$$

The results presented in Table 6.14 are the random and fixed effects estimation of the main results.

From the table, Models 1, 2, and 3 are for random effects whereas fixed effects estimation results are given in Models 4 to 6. Both the random effects and fixed effects results confirm the previous findings reported in the chapter about the effects of financial constraints and mispricing on share issuance decisions. Here again, the coefficient of KZ is significantly negative for all the models whereas MB is significantly positive. Also, the coefficient of the interaction term KZ*MB is significantly positive in all the models. These results provide further emphasis that internally financially constrained firms are less likely to issue equity. It also shows that firms are more likely to issue overvalued equity. However, given overvalued equity, internally financially constrained firms are more likely to issue equity indicating that internally financially constrained firms probably derive significant benefits from issuing overvalued equity. Thus, the results after using alternative estimation techniques still support the

hypotheses about the effects of internal financial constraints and mispricing on share issuance decisions.

Estimation (Financial Constraints and Share Issuance) Variables (1) (2) (3) (4) Dependent Variable is Issuance (1) vs Non-Issuance (0) KZ -0.943**** -1.831*** -0.867*** -2.161*** (0.00) (0.00) (0.00) (0.00) MB 0.434**** 0.392*** 0.217*** 0.182*** 0.182*** MB*KZ 1.556*** 0.288*** 0.400 0.000) (0.00) MB*KZ 0.288*** 0.289*** 0.412*** 0.397*** 0.397*** SIZE 0.288*** 0.289*** 0.412*** 0.397*** 0.397*** (0.00) (0.00) (0.00) (0.00) ED 1.169*** 1.179**** 0.811*** 0.860*** 0.860*** PROF -4.122** -4.008*** -1.483** -1.506* 0.00) CAPX -7.667** -7.839*** -8.463** -8.561** 0.00) CAPX -7.667** -7.839*** -8.463** -8.561** 0.00) Constant -2.268*** -2.184*** -2.184*** (0.00) (0.00) (0.00) (0.00) Year Dummies Yes Yes Yes Industry Dummies Yes Yes <th>Table 6.13: Ra</th> <th>ndom and</th> <th>Fixed Effect</th> <th>ts Logit Regr</th> <th>ession</th>	Table 6.13: Ra	ndom and	Fixed Effect	ts Logit Regr	ession
Variables (1) (2) (3) (4) Dependent Variable is Issuance (1) vs Non-Issuance (0) KZ -0.943*** -1.831*** -0.867*** -2.161*** (0.00) (0.00) (0.00) (0.00) MB 0.434*** 0.392*** 0.217*** 0.182*** 0.182*** (0.00) (0.00) (0.00) (0.00) MB*KZ 1.556*** 2.201*** 2.201*** (0.00) (0.00) (0.00) (0.00) SIZE 0.288*** 0.289*** 0.412*** 0.397*** 0.811*** 0.860*** ED 1.169** 1.179*** 0.811*** 0.860*** 0.860*** ED 1.169** 1.179**** 0.811*** 0.860*** 0.860*** ED (0.00) (0.00) (0.00) (0.00) PROF -4.122*** -4.008*** -1.483*** -1.506** -1.506*** (0.00) (0.00) (0.00) (0.00) CAPX -7.667*** -7.839*** -8.463*** -8.561** -8.561*** (0.00) (0.00) (0.00) (0.00) Constant -2.268*** -2.184*** -2.184*** (0.00) (0.00)	Estimation (Fi	nancial Co	nstraints an	d Share Issu	ance)
Dependent Variable is Issuance (1) vs Non-Issuance (0) KZ -0.943*** -1.831*** -0.867*** -2.161*** (0.00) (0.00) (0.00) (0.00) (0.00) MB 0.434*** 0.392*** 0.217*** 0.182*** (0.00) (0.00) (0.00) (0.00) MB*KZ 1.556*** 2.201*** (0.00) (0.00) (0.00) (0.00) SIZE 0.288*** 0.289*** 0.412*** 0.397*** ED 1.169*** 1.179**** 0.811*** 0.860**** ED 1.169*** 1.179**** 0.811** 0.860**** ED 4.122*** -4.008*** -1.483*** -1.506*** (0.00) (0.00) (0.00) (0.00) (0.00) CAPX -7.667**** -7.839*** -8.463*** -8.561** (0.00) (0.00) (0.00) (0.00) (0.00) Constant -2.268**** -2.184*** Yes Yes Yes <		Randon	n Effects	Fixed	Effects
KZ -0.943*** -1.831*** (0.00) (0.00) (0.00) (0.00) -2.161*** (0.00) (0.00) (0.00) MB 0.434*** 0.392*** (0.217*** 0.182*** 0.217*** 0.182*** (0.00) (0.00) 0.00) (0.00) (0.00) (0.00) MB*KZ 1.556** (0.00) (0.00) (0.00) 2.201*** (0.00) SIZE 0.288*** 0.289*** (0.00) (0.00) (0.00) (0.00) 0.397*** (0.00) (0.00) (0.00) (0.00) ED 1.169*** 1.179*** 0.811** 0.860*** (0.00) (0.00) (0.00) (0.00) 0.00) (0.00) (0.00) (0.00) (0.00) PROF -4.122*** -4.008*** -1.483*** -1.506*** (0.00) (0.00) (0.00) (0.00) 0.00) (0.00) (0.00) (0.00) (0.00) CAPX -7.667*** -7.839*** -8.463*** -8.561*** (0.00) (0.00) 0.00) (0.00) (0.00) (0.00) Constant -2.268*** -2.184*** (0.00) (0.00) 0.00) (0.00) (0.00) Year Dummies Yes Yes Yes Yes Industry Dummies Yes Yes Yes Yes N 1366 1366 1366 1366 chi-squared 232.996 235.243 449.709 471.257	Variables	(1)	(2)	(3)	(4)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Dependent Va	riable is Iss	suance (1) vs	Non-Issuanc	
MB 0.434*** 0.392*** 0.217*** 0.182*** (0.00) (0.00) (0.00) (0.00) MB*KZ 1.556*** 2.201*** (0.00) (0.00) (0.00) SIZE 0.288*** 0.289*** 0.412*** 0.397*** (0.00) (0.00) (0.00) (0.00) (0.00) ED 1.169*** 1.179*** 0.811*** 0.860*** (0.00) (0.00) (0.00) (0.00) (0.00) PROF -4.122*** -4.008*** -1.483*** -1.506** (0.00) (0.00) (0.00) (0.00) (0.00) CAPX -7.667*** -7.839*** -8.463*** -8.561*** (0.00) (0.00) (0.00) (0.00) (0.00) Constant -2.268**** -2.184*** (0.00) (0.00) (0.00) (0.00) Year Dummies Yes Yes Yes N 1366 1366 1366 1366 pseudo R²	KZ	-0.943***	-1.831***	-0.867***	-2.161***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.00)	(0.00)	(0.00)	(0.00)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	MB	0.434^{***}	0.392^{***}	0.217^{***}	0.182^{***}
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			(0.00)	(0.00)	
SIZE 0.288^{***} 0.289^{***} 0.412^{***} 0.397^{***} ED 1.169^{***} 1.179^{****} 0.811^{****} 0.860^{***} PROF -4.122^{****} -4.008^{****} -1.483^{****} -1.506^{****} CAPX -7.667^{***} -7.839^{***} -8.463^{***} -8.561^{***} Constant -2.268^{***} -2.184^{***} -2.184^{***} (0.00)(0.00)(0.00)(0.00)Year DummiesYesYesYesYesN1366136613661366pseudo R²1366136613661366chi-squared232.996235.243449.709471.257	MB*KZ				2.201***
SIZE 0.288^{***} 0.289^{***} 0.412^{***} 0.397^{***} ED 1.169^{***} 1.179^{****} 0.811^{****} 0.860^{***} PROF -4.122^{****} -4.008^{****} -1.483^{****} -1.506^{****} CAPX -7.667^{***} -7.839^{***} -8.463^{***} -8.561^{***} Constant -2.268^{***} -2.184^{***} -2.184^{***} (0.00)(0.00)(0.00)(0.00)Year DummiesYesYesYesYesN1366136613661366pseudo R²1366136613661366chi-squared232.996235.243449.709471.257			(0.00)		(0.00)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	SIZE	0.288^{***}	0.289^{***}	0.412^{***}	0.397^{***}
PROF			(0.00)	(0.00)	(0.00)
PROF -4.122*** -4.008*** -1.483*** -1.506*** (0.00)	ED	1.169***	1.179***	0.811***	
CAPX		(0.00)	(0.00)	(0.00)	(0.00)
CAPX	PROF	-4.122***	-4.008***	-1.483***	-1.506***
CAPX -7.667*** -7.839*** -8.463*** -8.561*** (0.00) (0.00) (0.00) (0.00) Constant -2.268*** -2.184*** (0.00) (0.00) Year Dummies Yes Yes Yes Yes Industry Dummies Yes Yes Yes Yes N 1366 1366 1366 1366 pseudo R² 0.255 0.267 chi-squared 232.996 235.243 449.709 471.257		(0.00)	(0.00)	(0.00)	(0.00)
Constant (0.00) (0.00) (0.00) (0.00) Year Dummies Yes Yes Yes Yes Industry Dummies Yes Yes Yes N 1366 1366 1366 1366 1366 pseudo R² (232.996 235.243 449.709 471.257	CAPX	-7.667 ^{***}	-7.839 ^{***}	-8.463***	-8.561***
Constant -2.268*** (0.00) (0.00) -2.184*** (0.00) Yes Y		(0.00)	(0.00)		(0.00)
Year Dummies Yes Yes <t< td=""><td>Constant</td><td>-2.268***</td><td></td><td></td><td></td></t<>	Constant	-2.268***			
Industry Dummies Yes Yes Yes Yes N 1366 1366 1366 1366 pseudo R² 0.255 0.267 chi-squared 232.996 235.243 449.709 471.257			(0.00)		
N 1366 1366 1366 1366 pseudo R ² 0.255 0.267 chi-squared 232.996 235.243 449.709 471.257	Year Dummies	Yes	Yes	Yes	Yes
pseudo R ² 0.255 0.267 chi-squared 232.996 235.243 449.709 471.257	Industry Dummies	Yes	Yes	Yes	Yes
chi-squared 232.996 235.243 449.709 471.257	N	1366	1366	1366	1366
1	pseudo R ²			0.255	0.267
p-value 0.000 0.000 0.000 0.000	chi-squared	232.996	235.243	449.709	471.257
	p-value	0.000	0.000	0.000	0.000

Notes: This table shows the random and fixed effects logit regression results for equity issues. The results for random effects logit regression are given in Models 1 to 3 whereas fixed effects logit regression results are shown in Models 4 to 6. The dependent variable for this logit regression is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include KZ measure of internal financial constraints (KZ), market-to-book ratio (MB), the interaction term (MB*KZ), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. *, * and * denote significance at the 10%, 5% and 1% level, respectively.

Similarly, further analysis of the effects of external financial constraints on equity issuance given the interaction of internal financial constraints and mispricing is robust to random and fixed effects estimations. Table 6.14 shows

that constrained firms are likely to issue overvalued equity only during periods of economic upturn. However, during crisis periods, as shown in both the random and effects models, only financially unconstrained firms are likely to issue overvalued equity.

Table 6.14: Random and Fixed Effects Logit Regression Estimation Financial Constraints and Share Issuance (Crisis and Non-crisis)

		n Effects	Fixed Effects									
	Crisis	Non-crisis	Crisis	Non-crisis								
Variables	1	2	3	4								
Dependent V	Dependent Variable is Issuance (1) vs Non-Issuance (0)											
KZ	-3.523***	-1.027**	-3.672***	-1.071***								
	(0.00)		(0.00)	(0.01)								
MB	(0.00) 0.268^{***}	(0.02) 0.398***	-0.036	0.301***								
	(0.01)	(0.00) 1.249**	(0.53) 4.260***	(0.00) 1.358***								
MB*KZ	3.173***	1.249**	4.260^{***}	1.358***								
	(0.00)	(0.02) 0.289***	(0.00)	(0.01)								
SIZE	0.326***	0.289^{***}	0.321***	0.341***								
	(0.01)	(0.00) 0.799***	(0.00)	(0.00)								
ED	2.149***	0.799***	1.171***	0.710***								
	(0.00)	(0.00)	(0.00)	(0.00)								
PROF	-7.300***	-3.020***	-2.921***	-1.752***								
	(0.00) -9.739***	(0.00) -7.853***	(0.00) -7.685***	(0.00) -8.221***								
CAPX			-7.685 ^{***}	-8.221***								
	(0.00) -2.424***	(0.00)	(0.00)	(0.00)								
Constant	-2.424***	-2.157***										
	(0.00)	(0.00)										
Year Dummies	Yes	Yes	Yes	Yes								
Industry Dummies	Yes	Yes	Yes	Yes								
N	368	998	368	998								
pseudo R ²			0.422	0.262								
chi-squared	80.140	150.627	200.488	337.112								
p	0.000	0.000	0.000	0.000								

Notes: This table displays the random and fixed effects logit regression results for equity issues given internal and external financial constraints. The results for the random effects estimation are given in model 1 (crisis) and 2 (non-crisis) and for fixed effects in model 3 (crisis) and 4 (non-crisis). The dependent variable for this logit regression is 1 for share issuance and 0 for non-issuance. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), market-to-book ratio (MB), and the interaction term (MB*KZ), logarithm of book value of total assets (SIZE), earnings deviation (ED) which measures asymmetric information effects, profitability (PROF), capital expenditure (CAPX) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.

*** and **** denote significance at the 5% and 1% level, respectively.

These results demonstrate the robustness of the earlier results using the logit regression estimation. Overall, the results in this chapter are robust to the alternative estimation technique and further strengthen the hypothesis that both internal and external financial constraints play important roles in the market timing of equity issues. It is only when there are external financial constraints will financially constrained firms be restricted in their attempts to time their equity issues.

A formal test of the appropriateness and comparison of random and fixed effects models is carried. Appendices 6A to 6C present the hausman test output for comparing random and fixed effects estimations. The starting point for the hausman test is that the two models do not produce significantly different coefficients. Thus, the null hypothesis is that the coefficient is not different between random and fixed effects in which case both estimations are appropriate otherwise the fixed effects is chosen because it gives consistent coefficients. As can be seen from the hausman test output in Appendix 6A, the chi-square is 185 and probability is 0.0000 indicating that the null hypothesis is rejected in favour of the alternative hypothesis. Therefore, the difference in the coefficients between random and fixed effects is systematic. Hence, the fixed effects output is more appropriate with consistent coefficient.

6.5. Conclusion

This chapter presented empirical results to explore the share issuance decisions of firms. It specifically investigated the simultaneous effects of financial constraints and mispricing on share issuance decisions. The main hypothesis tested in this chapter was that financial constraints limits market timing of equity issues. Thus, financially constraints firms are less likely to issue overvalued equity, especially when market conditions are unfavourable. This hypothesis emphasised that the costs of issuing equity increases substantially when there is economic crisis that also limits the availability of financing. The implication of this hypothesis was that financially unconstrained firms would be in a better position to issue overvalued equity even when the market conditions are less favourable. The results used the Kaplan and Zingales (1997) index and Whited and Wu (2006) index as well as Taffler's Z score as the proxies for internal financial constraints. Financial crisis and market liquidity are proxies for external financial constraints.

Consistent with prior studies, the results of this chapter support the developed hypotheses H₁, H₂, and H₃ about share issuance and financial constraints. The results presented in this chapter show that financially constrained firms are more likely to issue overvalued equity. Results further support the effects of external financial constraints on the interaction between internal financial constraints and mispricing. Specifically, the results showed in this chapter stress that firms are more likely to issue overvalued equity during periods of favourable market conditions or during non-crisis period, even when they are financially constrained. Thus, favourable market conditions allow firms to, regardless of their internal financing constraints, to exploit overvaluation advantages. However, only unconstrained firms are able to issue overvalued equity during periods of significant external financing constraints. Moreover, the chapter addresses issues with estimation assumptions and also presented results using alternative proxies for financial constraints. Finally, the chapter presented robust results for alternative estimation techniques.

The findings in this chapter highlight the market timing activities of firms. Studies such as Stein (1996), Graham and Harvey (2001), Baker and Wurgler (2002), Dong et al. (2012), all find evidence about equity market timing by which firms issue overvalued equity. Also, Korajczyk and Levy (2003), Dong et al. (2012) and (McLean and Zhao, 2014) assert that internal financial constraints restricts external financing. However, the results in this chapter differ markedly from prior studies on several fronts. For example, Baker and Wurgler (2002) do not explore the effects of financial constraints on equity market timing and Dong et al. (2012) only investigate the effects of internal financial constraints on equity issues. The results presented in this chapter, on the other hand, synthesise the effects of both internal and external financial constraints and demonstrate how these two determine the issuance of overvalued equity. This evidence has not previously been explored in the literature and therefore the results presented in this chapter extend the extant literature on equity issuance and market timing.

Chapter 7

SHARE ISSUANCE METHODS AND OPERATING PERFORMANCE

7.1. Introduction

This section presents the empirical results and discusses the effects of operating performance on the choice of share issuance methods and the subsequent long run stock returns associated with rights issues, open offers, and private placement. Firms that issue equity tend to be associated with poor long-term stock performance (Loughran and Ritter, 1995; Kabir and Roosenboom, 2003; Capstaff and Fletcher, 2011). Three main explanations have been suggested for this price pattern. According to the information asymmetry hypothesis (e.g., Myers and Majluf, 1984), managers have superior knowledge about their firms, compared to market participants, and use their private information to issue equity prior to public knowledge. Loughran and Ritter (1997) assert that investor over-optimism at the time of equity issue causes firms to overweigh recent good operating performance consistent with the over-optimism hypothesis. Finally, according to the window of opportunity hypothesis (Choe et al., 1993; Bayless and Chaplinsky, 1996), overvalued firms issue equity during favourable market conditions. Long-term stock returns tend to be poor after issuance, and are significantly in the magnitude based on the equity issuance method used.

This study sheds light on why long-term stock returns after issuance tend to be poor and, more importantly, why alternative issuance methods are associated with different levels of long-term stock returns. As argued in this thesis, firms that make rights issues have less incentive to sell equity with poor prospects, since existing shareholders sell shares to themselves in comparison to counterpart private placements and open offers. Based on the information asymmetry hypothesis, managers are therefore expected to use their inside information to select the issuance method. Since operating performance may vary after the issuance of alternative methods, the long-term stock returns would vary accordingly. In line with a number of studies in the literature (e.g.,

Cooper et al., 2008), operating performance is related positively with stock returns, and therefore rights issues are expected to result in better operating performance and long-term stock returns after issuance compared to counterpart private placements and open offers.

Several studies demonstrate that the method of equity issuance has implications for post-issue stock returns (Slovin et al., 2000; Barnes and Walker, 2006; Armitage, 2010; Capstaff and Fletcher, 2011) and operating performance (Loughran and Ritter, 1997; Hertzel et al., 2002; Kabir and Roosenboom, 2003). These findings suggest a direct relationship between the post-issue long run stock returns and the operating performance. Hence, as argued in the extant literature, not only does the issuance method choice affect the long run stock performance but this performance is largely dependent on the level of operating performance. It is argued that firms' expectations about future operating performance can be exploited in the choice of issuance method. If firms were motivated to maximise shareholder value, then they would more likely choose rights offerings when operating performance is anticipated to be good. The choice of rights issues transfers part of the benefits to the existing shareholders. Stated another way, for anticipated poor operating performance, issuance which involves outside investors will be preferred in order to exploit less informed investors, all other things being equal. Thus, this study tests the effects of long-term operating performance on the decision to choose one equity issuance method over other(s).

Unlike UK, where three issuance methods exist for firms, rights issues are the main equity issuance method in many European countries (e.g., Gajewski and Ginglinger, 2002; Bohren et al., 1997; Tsangarakis, 1996; Pastor-Llorca and Martin-Ugedo, 2004), but is uncommon in the US where firm commitments are the method most commonly used. Eckbo and Masulis (1992) and Kothare (1997) explain that US firms' non-selection of rights offerings may be due to relatively high transaction costs and capital gain taxes, and they estimate that more than 80% of US equity issuance is for non-rights offerings. Therefore, UK equity issues constitute three widely used equity issuance methods and offer an ideal environment to explore whether information asymmetry and

firms' operating performance after issuance is aligned with differential longterm stock returns.

The literature on the choice of equity issuance has been widely explored in the UK where firms are entitled to issue equity through rights issues, open offers, and private placement. Largely, such decisions have been attributed to the motivations of the market timing theory by which firms issue overvalued equity. However, the evidence is mixed even though most studies (for example Slovin et al. 2000; Barnes and Walker, 2006) find that rights issues are more likely to be timed. Information asymmetry is a related theory to market timing by which firms are deemed to have inside information about the issue more than the investors. Therefore, firms are likely to be influenced by such information to exploit potential investors through the issuance method. Operating performance is one variable which firms can have a better forecast than outside investors who will need more information, mostly at a cost, to determine the true value of an issuing firm. Thus, the hypotheses explained in Section 3.5 of Chapter 3 explore the influence of expected operating performance in determining whether a firm chooses rights issues, open offers, or private placement at the time of equity issues. Specifically, the study predicts that firms are more likely to choose rights issues than open offer or private placement when they anticipate high operating performance. It follows, therefore, that rights issuing firms are associated with less negative post-issue long run stock price performance compared to open offers and private placement.

The chapter consists of four parts. Section 7.2 describes the sample used to perform the empirical analysis. Section 7.3 presents the multivariate analysis of the effects of operating performance on the choice of equity issuance method and the subsequent stock price performance. Section 7.4 discusses additional results and performs robustness checks using alternative definitions of operating performance and estimation techniques. It also conducts logit regression diagnostic tests to determine the appropriateness of the models. Section 7.5 concludes the chapter.

7.2. Sample Statistics and Diagnostics

This section discusses the descriptive statistics of the variables used to undertake the empirical analysis. The study follows prior studies and uses variables that explain the choice of equity issuance method. Section 7.1.1 presents the descriptive statistics for the final sample.

7.2.1. Descriptive Statistics of Independent and Control Variables

This section presents the descriptive statistics of independent and control variables used to undertake the empirical analysis. It focuses on the statistics of the firm characteristics of equity issues and issuance methods. Thus, the equity sample is divided into rights issues (RI), open offers (OO), and private placement (PP).

7.2.1.1. Equity Issues by Issuance Method during the Sample Period

Table 7.1 presents the number of yearly equity issues that are made through rights issues, open offers, and private placement over the sample period. It also includes the anticipated proceeds from the equity issues, and the average ratio of the proceeds to market value for each issuance method. The results in the table show that rights issues were relatively more dominant in the early part of the sample period. However, private placement increased significantly in the year 2000 and 2001 and was relatively dominant from 2008 to 2010. Thus, more issues are made through private placement and open offers than rights issues, und relying the 'rights issues demise' phenomenon. Both Burton et al. (2005) and Armitage (2007) show that rights issue have significantly reduced as an equity issuance method at least in the period after the year 2000. Also, it can be seen from the table that the amount of proceeds from rights issues is higher than those generated from open offers and private placement. This is surprising given that greater number of private placement and open offers are made by equity issuing firms than rights issues. It appears open offers and private placement attract higher issue price discount to appeal to or compensate the new investors.

Finally, the ratio of proceeds to market value indicates a stable value of proceeds per unit of firm value. During the years 2001, 2002, 2004 2008, and

2009, the proceeds per market value for rights issues is greater than one indicating that the proceeds is more than 1 times the firm size. In fact, the total proceeds per market value for rights issues is 2.51 compared to 0.13 and 0.27 for open offers and private placement respectively. This implies small firms make rights issues or that the proceeds far exceed the value of the firm. It could also suggest that the stock price of rights issues is relatively higher than that of open offers and private placement.

-	Table 7.1: Yearly Average Amounts and Ratio of Equity Issues											
		Rights Issue	•		Open Offer			Private Placement				
Year	N	Proceeds	PMV	N	Proceeds	PMV	N	Proceeds	PMV			
1994	42	73.36	0.23	0	0.00	0.00	21	55.00	0.43			
1995	39	86.86	0.42	0	0.00	0.00	24	180.92	0.24			
1996	48	71.48	0.46	0	0.00	0.00	35	39.44	0.19			
1997	29	54.97	0.20	10	30.29	0.29	12	31.34	0.15			
1998	17	94.52	0.39	27	37.55	0.14	5	34.50	0.15			
1999	16	100.32	0.28	26	36.55	0.22	16	63.84	0.31			
2000	18	284.13	0.98	47	32.29	0.18	61	48.52	0.19			
2001	19	493.61	9.21	44	16.23	0.08	41	99.31	0.23			
2002	13	569.76	2.80	29	4.55	0.03	9	29.58	0.19			
2003	11	23.99	0.15	41	15.18	0.06	25	44.40	0.18			
2004	11	80.39	1.55	29	8.89	0.06	25	35.91	0.21			
2005	19	135.71	0.45	4	20.55	0.11	21	33.40	0.28			
2006	11	203.24	0.55	4	51.59	0.19	32	71.55	0.32			
2007	2	163.38	0.64	10	110.82	0.30	71	131.06	0.30			
2008	7	241.07	1.16	8	46.23	0.22	39	104.23	0.36			
2009	24	872.36	12.08	29	77.79	0.17	106	156.75	0.38			
2010	2	196.01	0.41	13	16.97	0.13	65	63.65	0.24			
Total	328	200.95	1.96	321	30.39	0.13	608	90.24	0.28			

This table shows the annual equity issues of sample firms for rights issues, open offers, and private placement. It also indicates the average yearly proceeds from equity issues (PROCEEDS) and the ratio of proceeds to market value of firms (PMV).

7.2.1.2. Descriptive Statistics of Variables

Table 7.2 provides the descriptive statistics for rights issues (Panel A), open offers (Panel B), and private placement (Panel C). It also shows the t-test of the mean difference between rights issues and open offers on one hand and rights issues and private placement on the other hand (as shown in Panel D). The variables include market to book ratio (MB), pre-issue announcement date abnormal returns (Prior AR), profitability (PROF), logarithm of market value (SIZE), issue price discount (DISC), accrual quality as a measure of earnings management (ACC QUAL), idiosyncratic risk (RISK), issue proceeds

(PROCEEDS) and the proceeds relative to market value (PMV). These variables measure both firm and issue quality and determines the choice of issuance method as discussed in Section 5.3.2.3 of Chapter 5.

As can be seen from the table, rights issues firms report highest MB of 4.49 compared to 4.02 and 4.31 for open offers and private placement, respectively. However, the mean difference tests show that the MB values are not different for rights issues, open offers and private placement. For the prior AR, rights also experience lower stock returns (Prior AR) prior to the issuance than open offers and private placement. In particular, the cumulative abnormal returns prior to each issuance are 2.0% for rights issues, 6.0% for open offers and 10.0% for private placements. It also shows that private placement experience statistically significant higher returns than rights issuing firms with mean difference of 8.2% significant at the 1% level. This result indicates that private placement is more likely to be timed when firms experience stock price runups. Moreover, PROF shows that rights issuing firms are better performing firms than those which conduct open offers and private placement. Overall, these results offer the first indication that managers that issue equity through rights issues have less of an incentive to time their equity issues using rights issues.

Furthermore, rights issuers are also bigger firms than open offers, as indicated by SIZE. Rights issues are associated with lower discount (DISC), accruals quality (ACC_QUAL) and idiosyncratic volatility RISK than private placements, and rights issuers generated more proceeds both in money terms and per unit of market value. These results are consistent with the hypothesis that rights issuing firms are less likely to exploit existing shareholders by issuing overvalued equity. On the contrary, it appears firms using private placement are of low quality seeking to exploit investors who might have insufficient about the firm. These results show that rights issues are perceived as a high quality issuance method, with lower information asymmetry.

Table 7.2: Summary Statistics of Firm Characteristics										
PANEL A: Rights Issues (RI)										
Variables	N	Mean	Median	SD	Min	Max				
MB	293	4.31	2.06	6.18	0.01	25.55				
Prior AR	327	0.02	0.01	0.24	-0.44	0.75				
PROF	313	0.08	0.09	0.04	0.02	0.12				
SIZE	307	11.24	11.43	1.89	6.58	18.15				
DISC	263	0.53	0.13	0.86	-0.34	1.64				
ACC QUAL	240	0.08	0.05	0.09	0.00	0.65				
RISK	327	0.03	0.02	0.02	0.00	0.08				
PROCEEDS	328	200.95	43.49	872.84	0.12	12277.61				
PMV	285	1.96	0.24	15.31	0.00	202.51				
PANEL B: O	pen Of	fers (OO)								
Variables	N	Mean	Median	SD	Min	Max				
MB	291	4.02	1.88	5.87	0.04	28.36				
Prior AR	321	0.06	0.03	0.30	-0.44	0.75				
PROF	309	0.02	0.04	0.07	-0.06	0.10				
SIZE	299	10.30	10.31	1.52	6.48	15.03				
DISC	274	1.44	0.79	1.65	-0.24	3.55				
ACC QUAL	213	0.11	0.07	0.13	0.00	0.91				
RISK	320	0.04	0.03	0.02	0.00	0.11				
PROCEEDS	321	30.39	7.39	69.35	0.07	648.49				
PMV	294	0.13	0.06	0.15	0.00	1.18				
PANEL C: Pi	rivate l	Placement	(PP)							
Variables	N	Mean	Median	SD	Min	Max				
MB	560	4.48	2.21	6.12	0.00	28.31				
Prior AR	604	0.10	0.06	0.29	-0.44	0.75				
PROF	584	0.04	0.05	0.07	-0.06	0.12				
SIZE	576	11.25	11.07	1.64	6.46	18.58				
DISC	518	0.53	0.21	1.00	-0.51	1.81				
ACC_QUAL	460	0.09	0.06	0.10	0.00	1.08				
RISK	603	0.03	0.03	0.02	0.00	0.10				
PROCEEDS	608	90.24	30.24	226.55	10.00	2815.26				
PMV	565	0.28	0.20	0.30	0.02	4.60				
PANEL D: To	ests of	Mean diffe	erence							
		ghts vs Ope	en offers		ghts vs Pla	cement				
Variables	N	diff.	t-stat	N	diff.	t-stat				
MB	584	0.287	(0.57)	853	-0.175	(-0.40)				
Prior AR	648	-0.032	(-1.51)	931	-0.074***	(-3.95)				
PROF	622	0.057***	(12.74)	897	0.042***	(9.19)				
SIZE	606	0.939	(6.73)	883	-0.015	(-0.12)				
DISC	537	-0.904***	(-7.92)	781	0.005	(0.07)				
ACC_QUAL	453	-0.031***	(-2.90)	700	-0.013*	(-1.67)				
RISK	647	-0.012***	(-7.08)	930	-0.008***	(-6.01)				
PROCEEDS	649	170.561	(3.49)	936	110.706	* (2.95)				
PMV	579	1.835**	(2.06)	850	1.685***	(2.62)				

Notes: This table shows the descriptive statistics of the variables used in the empirical analysis for the choice of equity issuance methods. Panel A provides results for rights issues, Panel B for open offers, Panel C for private placement, and Panel C tests the mean difference of variables between rights issues and open offers as well as rights

issues and private placement. Variables include market to book ratio (MB), preannouncement abnormal returns (Prior AR), profitability (PROF), logarithm of book value of total assets (SIZE), issue price discount (DISC), measure of earnings management (ACC_QUAL), measure of idiosyncratic risk (RISK), issue proceeds (PROCEEDS), and ratio of issue value to market value (PMV). T-statistics are shown in parenthesis in Panel D. ** and *** denote significance at the 5% and 1% level respectively.

7.2.1.3. Pearson's Correlation Matrix of Equity Issue Characteristics

Logistic regression is used to estimate the model that tests the effects of operating performance on the choice of equity issuance methods (H_7) as explained in Section 5.3.3.4. However, OLS is used to test hypothesis H_8 about the stock returns following equity issuance choice. As discussed in Chapter 6 Section 6.2.1.3, a correlation coefficient close to or equal to +1 and -1 suggests high collinearity or perfect multicollinearity among the variables. Tables 7.3, 7.4 and 7.5 provide the Pearson's correlation matrixes of independent and control variables for the sample of rights issues, open offers and private placement.

High correlations are recorded for PROCEEDS and PMV simply because PMV is derived from PROCEEDS. However, only PMV is used in the regression analysis. Most of the variables are correlated at either the 1% or 5% level, with most of the correlations lying close or below 0.20. This results implies low multicollinearity and thus tolerable for OLS estimation. Overall, the results show that multicollinearity is not a problem for estimating the regression model using OLS.

	Table 7.3: Pearson's Correlation Matrix of Equity Issue Characteristics (Rights Issues)											
	ROA	MB	Prior AR	SIZE	DISC	ACC_QUAL	RISK	PROCEEDS	PMV	$ m CRISIS_{dum}$		
ROA	1.000											
MB	-0.061	1.000										
Prior AR	-0.008	0.242^{***}	1.000									
SIZE	0.364***	-0.049	-0.035	1.000								
DISC	-0.058	-0.084	-0.028	-0.013	1.000							
ACC_QUAL	-0.405***	0.141^{*}	-0.079	-0.284***	0.177^{**}	1.000						
RISK	-0.164**	0.112^{*}	0.084	-0.174***	0.304***	0.314***	1.000					
PROCEEDS	0.160^{**}	-0.026	0.014	0.465^{***}	-0.007	-0.074	0.043	1.000				
PMV	0.121^{*}	-0.021	0.030	0.357^{***}	-0.036	-0.059	0.055	0.961***	1.000			
CRISIS _{dum}	0.095	-0.055	-0.042	0.280^{***}	0.274^{***}	-0.090	0.298^{***}	0.180^{***}	0.143^{**}	1.000		

Notes: Variables are defined as returns on assets (ROA), market to book ratio (MB), pre-announcement abnormal returns (Prior AR), logarithm of book value of total assets (SIZE), issue price discount (DISC), measure of earnings management (ACC_QUAL), measure of idiosyncratic risk (RISK), issue proceeds (PROCEEDS), ratio of issue value to market value (PMV), and financial crisis dummy (CRISIS_{dum}). *, ** and **** denote significance at the 10%, 5% and 1% level respectively.

	Table 7.4: Pearson's Correlation Matrix of Equity Issue Characteristics (Open Offers)										
	ROA	MB	Prior AR	SIZE	DISC	ACC_QUAL	RISK	PROCEEDS	PMV	$ m CRISIS_{dum}$	
ROA	1.000										
MB	-0.098*	1.000									
Prior AR	0.041	0.100^{*}	1.000								
SIZE	0.195***	-0.139**	-0.007	1.000							
DISC	-0.176 ^{***}	-0.034	-0.104*	-0.065	1.000						
ACC QUAL	-0.314***	0.066	0.038	-0.121*	0.089	1.000					
RISK	-0.323***	-0.031	0.015	-0.103 [*]	0.338^{***}	0.234^{***}	1.000				
PROCEEDS	0.141^{**}	-0.054	0.107^{*}	0.445^{***}	0.063	-0.077	0.068	1.000			
PMV	0.112^{*}	0.109^{*}	0.148^{**}	0.369^{***}	-0.006	-0.085	0.019	0.760^{***}	1.000		
CRISIS _{dum}	0.073	-0.153***	-0.058	0.033	0.230***	-0.079	0.357***	0.165***	0.117**	1.000	

Notes: Variables are defined as returns on assets (ROA), market to book ratio (MB), pre-announcement abnormal returns (Prior AR), logarithm of book value of total assets (SIZE), issue price discount (DISC), measure of earnings management (ACC_QUAL), measure of idiosyncratic risk (RISK), issue proceeds (PROCEEDS), ratio of issue value to market value (PMV), and financial crisis dummy (CRISIS_{dum}). *, ** and **** denote significance at the 10%, 5% and 1% level respectively.

	Table 7.5: Pearson's Correlation Matrix of Equity Issue Characteristics (Private Placement)											
	ROA	MB	Prior AR	SIZE	DISC	ACC_QUAL	RISK	PROCEEDS	PMV	CRISISdum		
ROA	1.000											
MB	0.018	1.000										
Prior AR	-0.082*	0.106^{**}	1.000									
SIZE	0.104^{**}	-0.005	-0.103**	1.000								
DISC	0.046	-0.067	-0.104**	-0.008	1.000							
ACC QUAL	-0.043	0.210^{***}	0.073	-0.240***	0.046	1.000						
RISK	-0.110**	0.004	0.074^{*}	-0.238***	0.226^{***}	0.205***	1.000					
PROCEEDS	0.065	-0.002	-0.018	0.522^{***}	0.062	-0.126***	-0.053	1.000				
PMV	0.026	-0.075*	0.011	0.238***	0.088^*	-0.077	0.043	0.633***	1.000			
CRISIS _{dum}	-0.049	-0.165***	-0.009	0.087^{**}	0.209^{***}	0.040	0.291^{***}	0.115^{***}	0.180^{***}	1.000		

Notes: Variables are defined as returns on assets (ROA), market to book ratio (MB), pre-announcement abnormal returns (Prior AR), logarithm of book value of total assets (SIZE), issue price discount (DISC), measure of earnings management (ACC_QUAL), measure of idiosyncratic risk (RISK), issue proceeds (PROCEEDS), ratio of issue value to market value (PMV), and financial crisis dummy (CRISIS_{dum}). *, ** and **** denote significance at the 10%, 5% and 1% level respectively.

7.3. Testing the Effects of Operating Performance on the Choice of Equity Issuance Methods

This section tests the hypotheses specified in Section 3.5 of Chapter 3 using the multinomial logit regression model (Model 3) stated in Chapter 5. The hypothesis is re-stated below:

H_6 : Firms that anticipate positive operating performance are more likely to conduct rights issues than open offers or private placement

Model 3 discussed in Section 5.2.2.4 tests the effects of operating performance on the choice of equity issuance method of UK listed firms. The model is restated below:

$$\Pr(y_i = 1 | x_i) = \beta_0 + \beta_1 ROA_{it} + \sum_{i=1}^n \beta_i X_{it} + \varepsilon_{it}$$
....(3)

The model tests the probability of equity issuance through rights issues depending on the level of anticipated operating performance and set of firm characteristics. Since the model represents multinomial logit regression estimation, the dependent variable is 1 for rights issues and 0 otherwise.

7.3.1. Multinomial Logit Regression Analysis of Equity Issuance Methods

[Hypothesis H_6]

This section presents the results of the multinomial logit regression analyses about the choice of equity issuance method given operating performance. The main variable of interest is ROA (measure of operating performance) that measures the level of information asymmetry. Following the extant literature, factors that are controlled for include measures of issue quality using DISC, PMV or firm characteristics as indicated by SIZE, MB, Prior AR, RISK (Barnes and Walker, 2006; Dissanaike et al. 2014) and ACC_QUAL, which is an accounting measure, to capture the information asymmetry hypothesis (Armitage et al. 2014) or earnings management (Teoh et al., 1998; Chi and Gupta, 2009). CRISIS dummy is used to control for macroeconomic conditions. The parameter coefficient of ROA proxies for operating performance, since if managers have superior knowledge they will raise equity through private placements and open offers when their firms are expected to

perform poorly after issuance. Moreover, and as earlier argued, firms that issue rights issues have less of an incentive to sell equity to existing shareholders when their next period operating performance is poor.

The MB and Prior AR also capture the over-optimism hypothesis (Loughran and Ritter, 1997), according to which investor over-optimism at the time of equity issue causes firms to overweigh conditions on the issuance day. Thus, high MB and Prior AR indicate a high level of optimism in a market and make investors optimistic about post-issuance stock performance. In addition, the model includes control for the recent financial crisis (CRISIS) by assigning the value of 1 if an issuance was made during the period 2008-2010 and 0 otherwise. This variable captures the window of opportunity hypothesis (Choe et al., 1993; Bayless and Chaplinsky, 1996), according to which overvalued firms issue equity during favourable market conditions. Finally, the estimations include year dummies to control for different levels of usage among the issuance methods during the sample period and industry dummies to control for industry characteristics.

Panel A of Table 7.6 presents the results for the multinomial logit regression for the choice between rights issues and open offers. Accordingly, the dependent variable is 1 for open offers (Panel A) and private placement (Panel B) and 0 for rights issues (the base outcome). This model explores whether managers select a rights issuance method based on their expectation of future operating performance. As can be seen from the table, the coefficient of ROA is significantly negative in year 1 (Models 1-3), year 2 (Model 4) and year 3 (Model 5). Unexpectedly, MB and SIZE are not significant but AR is significantly positive in year 1. Also, DISC is positive and significant at 1% level in Models 1-5 of Panel A whereas DISC is significantly negative in Panel B at 5% level in all the Models except Model 4. ACC_QUAL is significantly negative in all the Models of Panel A but not Panel B whereas RISK is not significant. PMV is positive and significant and CRISIS is significantly positive in all the models.

These results about ROA is consistent with the hypothesis that high operating performance increases the probability of rights issues at the expense of open

offers and private placement as shown in Panels A and B, respectively. Also, the evidence about AR means that high stock price run-ups increase the probability of open offers and high price discount is consistent with open offers but not private placement. These findings suggest that firms are less likely to time their rights issues. But the decision to issue equity through rights issues is negatively affected by the adverse market conditions. Thus, during periods of economic crisis, firms are more likely to make open offers and private placement than rights issues. As Capstaff and Fletcher (2011) argue, it seems rights issuing firms are quality firms compared to firms that undertake open offers and private placement. Studies such as Slovin et al. (2000) and Barnes and Walker (2006) find evidence that is contrary to what this study reports. Both studies use sample period that is characterised by the dominance of rights issues, especially during the 1980s and early 1990s. The study uses sample period that mirrors Capstaff and Fletcher (2011) and reinforces the trend of rights issues demise (Burton et al. 2005; Armitage, 2007)

Table 7.6: Multinomial Logit Regression Estimates of Choice of Equity Issuance Method											
PANEL A: Open Offers (1) vs Rights Issues (0)											
		Year 1		Year 2	Year 3						
Variables	(1)	(2)	(3)	(4)	(5)						
ROA	-15.706***	-14.138***	-15.739***	-11.558***	-8.817***						
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)						
MB	0.046	0.039	0.047	0.031	0.035						
	(0.29)	(0.19)	(0.28)	(0.45)	(0.38)						
AR	1.854**	1.222*	1.794**	1.137	1.235						
	(0.02)	(0.05)	(0.02)	(0.18)	(0.11)						
SIZE	-0.115	-0.103	-0.118	-0.145	-0.126						
	(0.41)	(0.33)	(0.39)	(0.31)	(0.37)						
DISC	0.419***	0.358***	0.441	0.493***	0.414***						
	(0.00)	(0.01)	(0.00)	(0.00)	$(0.01)_{*}$						
ACC_QUAL	-3.007***		-2.949***	-2.489**	-2.100*						
DIGI	(0.01)	5.0 00	(0.01)	(0.02)	(0.07)						
RISK		-7.233	-3.610	-1.066	-1.646						
D) (I)	7 0 7 1 ***	(0.39)	(0.74)	(0.92)	(0.88)						
PMV	-7.851	-5.711	-7.915***	-7.113***	-7.187***						
CDIGIG	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)						
CRISIS	17.433***	18.283***	17.473	16.806***	16.592***						
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)						
Constant	-11.319***	-12.998***	-11.121	-10.431***	-11.265						
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)						

PANEL B: Private Placement (1) vs Rights Issues (0)					
		Year 1		Year 2	Year 3
Variables	(1)	(2)	(3)	(4)	(5)
ROA	-13.374***	-12.568***	-13.661***	-10.594***	-3.172
	(0.00)	(0.00)	(0.00)	(0.00)	(0.19)
MB	0.091^{**}	0.063^{**}	0.092**	0.080^{**}	0.085**
	(0.01)	(0.02)	(0.01)	(0.03)	(0.01)
AR	1.579**	1.162**	1.538**	1.189	1.127
	(0.04)	(0.05)	(0.04)	(0.13)	(0.13)
SIZE	0.029	0.072	0.006	0.014	-0.042
	(0.82)	(0.44)	(0.96)	(0.92)	(0.73)
DISC	-0.354**	-0.282**	-0.305**	-0.247	-0.288**
	(0.01)	(0.03)	(0.04)	(0.14)	(0.05)
ACC_QUAL	-1.543		-1.223	-1.346	-1.023
	(0.17)		(0.29)	(0.31)	(0.44)
RISK		-8.131	-12.658	-10.182	-6.941
		$(0.26)_{}$	$(0.19)_{}$	(0.33)	(0.47)
PMV	-2.686***	-2.183***	-2.680***	-2.758***	-2.402***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
CRISIS	4.044***	4.315***	4.248***	4.247***	4.011***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Constant	2.131	0.935	2.746^*	2.594	2.498
	(0.16)	(0.45)	(0.08)	(0.10)	(0.12)
Year Dummies	Yes	Yes	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes	Yes	Yes
N	601	772	601	548	555
pseudo R^2	0.391	0.355	0.394	0.392	0.377
chi2	4041.922	4794.190	4066.833	4723.022	5682.218
p	0.000	0.000	0.000	0.000	0.000

Panel B shows the results for the multinomial logit regression of the choice between rights issues and private placement. The results show that there is a high probability that firms' ROA subsequent to the issuance is higher when issuing equity through rights issues rather than private placements. In particular, the parameter coefficient of ROA is positive and statistically significant at the 1% level within most of the estimations, supporting the

economical interpretation of the information asymmetry hypothesis. When firms raise equity through rights issues, they tend to experience better operating performance after the issuance. MB is significantly negative indicating that firms are more likely to issue overvalued equity through private placement. Also, Prior AR increases the probability of private placement relative to rights issues. This result suggests that rights issuing firms are not likely to issue overpriced equity to the existing shareholders. In other words, issues that involve outside investors, such as private placement, are more likely to be timed to take advantage of significant overvaluation.

However, the parameter coefficient of ACC QUAL does not support the information asymmetry hypothesis in accounting terms, indicating that a conventional information asymmetry proxy does not capture the interpretation of asymmetric information. The positive coefficient of DISC implies rights issues are less likely to be associated with significant issue price discount. This price discount compensates investors for the costs of obtaining information about the true value of the issuing firm. However, RISK has a negative and significant coefficient at the 1% level indicating asymmetric information does not increase the probability of rights issues relative to private placement. Also, rights issues generate larger PMV and have lower stock returns prior to the issuance, indicating that managers are less prone to take advantage of the timing of the issue when selecting rights issues. The negative parameter coefficient for CRISIS indicates that there is a lower probability for firms to issue equity through rights issues during the financial crisis, supporting the evidence that the use of rights issues in the UK market has become less common in recent years (e.g., Armitage, 2010).

These findings are consistent with the hypothesis that expected operating performance determines the choice of equity issuance method. It further supports the extant evidence about market timing of equity issues through private placement. The study departs from most previous studies that look at the short-term returns associated with share issuance methods. Consistent with Capstaff and Fletcher (2011), the study finds that rights issuing firms outperform open offers and private placement in the long run. Unlike Slovin et

al. (2000) and Barnes and Walker (2006), rights issuing firms are associated with quality information signalling since it is unlikely that firms would issue overvalued equity to exploit existing shareholders. It appears rights issuing firms are not overvalued but significantly perform better than firms making open offers and private placement. In effect, the evidence implies that the hypothesis that high operating performance is associated with rights issues is accepted. Also, the evidence is consistent with the hypothesis that rights issues are less likely to be overvalued and hence less likely to be timed.

7.3.2. Long-term Stock returns

[Hypothesis H_8]

This section analyses the long run stock returns associated with rights issues (RI), open offers (OO), and private placement (PP). The hypothesis about stock returns associated with equity issues is re-stated below.

H₇: Equity issues through private placement generate more negative abnormal returns than rights issues

Test of hypothesis H_8 is carried out using the OLS regression model below, also discussed in Section 5.2.2.4 of Chapter 5.

$$BHAR_{i} = \beta_{0} + \beta_{1}ROA_{i} + \beta_{2}ISSUE_{i} + \beta_{3}ROA*ISSUE_{i} + \sum_{i=1}^{n} \beta_{i}X_{i} + \varepsilon_{i}.....(4)$$

As discussed in Section 5.3.2.1 of Chapter 5, long run returns is the buy and hold abnormal returns (BHAR) computed as the excess of buy and hold returns of the issuing firm over that of its control firm. Table 7.7 provides the results of the BHAR for RI, OO, and PP as well as for RI-OO and RI-PP indicating the difference in returns between rights issues and open offers and between rights issues and private placement, respectively. The results show that BHAR for rights issues are not significantly different from zero over the three years after the issue. For the open offer sample firms, the BHAR is -7.9% (significant at the 5% level), -20.5% (significant at the 1% level), and -21.8% (significant at the 1% level) over year 1, year 2, and year 3, respectively. Similarly, the BHAR for private placement are -6.9%, -20.1%, and -32.7% all significant at the 1% level for year 1, year 2, and year 3, respectively.

Table	7.7: Long Ru	n Buy and Hol	ld Abnormal l	Returns and Eq	uity Issuance	Method
Year	RI	00	PP	All issues	RI-OO	RI-PP
1	-0.008	-0.079**	-0.069***	-0.056***	0.071	0.061*
2	0.006	-0.205***	-0.201***	-0.148***	0.211***	0.207^{***}
3	-0.078	-0.218***	-0.327***	-0.234***	0.140	0.249***
N	327	321	608	1256		

This table shows the long run stock returns for each of the three years following the issuance of rights issues (RI), open offers (OO) and private placements (PP). Returns are buy-and-hold based on Barber and Lyon's (1997) methodology, and t-statistics are bootstrapped skewness-adjusted. *, ** and *** show significance at the 10%, 5% and 1% levels, respectively.

Together, long run post-issue stock returns are significantly negative for equity issues. Also, the mean difference in returns is significantly positive over year 2 between rights issues and open offers and significantly positive over year 2 and 3 between rights issues and private placement. These results support the findings by Capstaff and Fletcher (2011) that rights issuing firms outperform open offers and private placement and highlight the argument that rights issuing firms are not poor quality firms seeking to exploit investors by issuing overvalued equity. Stated differently, firms issuing equity through open offers and private placement are more likely to time their issues and subsequently report significant stock price underperformance.

Table 7.8 presents results that explore the significance of operating performance for post-issuance stock returns. In particular, the long-term stock returns on firms are partitioned into firms with positive and negative ROA at each year after issuance. These results show that over the three-year period, rights issuing firms do not report operating losses which strengthen the earlier evidence that these firms are associated with less negative stock returns. Instead, around half of the firms issuing equity through open offers and private placements report deteriorating long-term operating performance after issuance. ROA also seems to influence the magnitude of stock returns after issuance, since poor stock returns are more prominent within firms that had negative ROA. It appears operating performance have significant implications for long-term stock returns after equity issuance.

Table 7.8: Long Run Stock Returns and Operating Performance (Positive and Negative) Year RIPPTotal 00 Negative ROA -0.092 1 n/a -0.109-0.081 2 -0.259 -0.242n/a -0.213 3 -0.394n/a -0.559 -0.493 N 0 136 208 344 Positive ROA 1 -0.008-0.057-0.063 -0.0422 0.006 -0.200-0.119 -0.175 3 -0.078-0.075-0.170-0.118 N 400 912 327 185 T-test (Positive - Negative ROA) 1 0.052 0.018 0.050 n/a 2 n/a 0.013 0.084 0.123^* 0.375*** 0.389^{***} 0.319^{**} n/a

This table shows the returns for each of the three years following the issuance of rights issues (RI), open offers (OO) and private placements (PP). Panel A shows the returns of the full sample, Panel B separately for negative and positive ROA firms and Panel C separately for low and high ROA firms when the split is based on the median ROA. ROA is the return on assets. Returns are buy-and-hold based on Barber and Lyon's (1997) methodology, and t-statistics are bootstrapped skewness-adjusted. *, ** and *** show significance at the 10%, 5% and 1% levels, respectively.

Further analysis of stock returns partitions the sample in terms of the level of operating performance. Table 7.9 shows long-term stock returns when firms are split into high and low ROA based on their median ROA. This test intends to offer further robustness, since no firm that issued rights issues had operating losses. Once again, firms with low ROA are associated with lower long-term stock returns after issuance than counterpart firms with high ROA. For example, the mean difference in BHAR between high ROA and low ROA over three years after issuance are 67.4%, 42.2%, and 58.7% for rights issues, open offers and private placements, respectively. These stock returns also provide further evidence that long run stock returns are significantly determined by operating performance. Moreover, the results demonstrate that rights issuing firms outperform their counterpart open offers and private placement.

Table 7.9: Long Run Stock Returns and Operating Performance (High and						
Low)						
Year	RI	00	PP	Total		
		Low ROA				
1	-0.050	-0.094	-0.066	-0.069		
2	-0.125	-0.221	-0.238	-0.203		
3	-0.434	-0.345	-0.588	-0.485		
N	158	150	285	593		
		High ROA				
1	0.027	-0.016	-0.070	-0.030		
2	0.158	-0.005	-0.171	-0.038		
3	0.240	0.076	-0.001	0.085		
N	157	171	284	591		
Test of Mean Difference T-test (High - Low ROA)						
	R1	00	PP	Total		
1	0.077	0.078	-0.003	0.039		
2	0.283***	0.216**	0.067	0.165***		
3	0.674***	0.422***	0.587***	0.570***		

This table shows the returns for each of the three years following the issuance of rights issues (RI), open offers (OO) and private placements (PP). Panel A shows the returns of the full sample, Panel B separately for negative and positive ROA firms and Panel C separately for low and high ROA firms when the split is based on the median ROA. ROA is the return on assets. Returns are buy-and-hold based on Barber and Lyon's (1997) methodology, and t-statistics are bootstrapped skewness-adjusted. *, *** and **** show significance at the 10%, 5% and 1% levels, respectively.

Table 7.10 provides OLS regression results for long run abnormal stock returns. The dependent variable BHAR for Year 1, Year 2 and Year 3 after equity issuance and the main independent variable is ROA after issuance. The control variables are the same as those used in the multinomial logit regression estimation in Table 7.6. ROA*OO and ROA*PP are the interaction terms for returns on assets and open offers and returns on assets and private placement respectively. The interaction term for rights issues and returns on assets (ROA*RI) is the baseline. The OLS results show that ROA is positively related to long-term stock returns after equity issuance. The relation is significant at the 1% level for two years and three years following the issuance. The parameter coefficients for the interaction variables ROA*OO and ROA*PP also tend to be negative, especially during the second year of issuance, indicating that rights issues tend to be related to higher operating performance than counterpart open offers and private placements. Also, the coefficients for control variables tend to be statistically insignificant, indicating that ROA seems to have strong explanatory power for long-term stock returns after issuance. Overall, these results further show that operating performance

explains the long run stock returns associated with equity issues and more importantly rights issuing firms are associated with more positive operating performance and long run stock returns than open offers and private placement.

Table 7.10: Cross-sectional Regression Analysis of Long Run Stock Returns							
	Year 1			Year 2		Year 3	
Variables	(1)	(2)	(3)	(4)	(5)	(6)	
Dependen	t Variable i	s the Long-ri	un Buy and Ho		Return, BHAI	R	
ROA	1.805**	1.926	4.161***	4.943***	6.298***	8.619***	
	(0.04)	(0.17)	(0.00)	(0.01)	(0.00)	(0.00)	
OO	0.068	0.021	0.198	0.199	0.343***	0.121	
	(0.38)	(0.86)	(0.12)	(0.30)	(0.00)	(0.46)	
PP	0.066	0.099	0.105	0.204	0.053	-0.009	
	(0.37)	(0.37)	(0.39) -2.961**	(0.27)	(0.61)	(0.95)	
ROA*OO	-0.887	0.218	-2.961 ^{**}	-3.975 [*]	-3.513 [*]	-5.103 ^{**}	
	(0.41)	(0.89)	(0.03)	(0.05)	(0.06)	(0.02)	
ROA*PP	-1.758 [*]	-1.915	-3.620***	-4.441**	-2.026	-2.485	
	(0.07)	(0.18)	(0.00)	(0.02)	(0.27)	(0.21)	
MB		0.004		0.016**		0.007	
		(0.47)		(0.05)		(0.41)	
AR		0.006		0.020		-0.032	
		(0.95)		(0.90)		(0.86)	
SIZE		-0.046***		-0.074***		-0.076***	
		(0.00)		(0.00)		(0.01)	
DISC		-0.003		-0.011		-0.017	
		(0.88)		(0.75)		(0.69)	
ACC_QUAL		0.026		-0.327		-0.714	
		(0.92)		(0.40)		(0.17)	
RISK		0.252		0.041		2.526	
		(0.87) 0.002^{***}		(0.99)		(0.40)	
PMV		0.002^{***}		0.001		0.002	
		(0.01)		(0.32)		(0.29)	
CRISIS		-0.101		0.143		0.519^*	
		(0.39)		(0.49)		(0.06)	
Constant	-0.136**	0.253	-0.327***	0.044	-0.424***	-0.454	
	(0.05)	(0.27)	(0.00)	(0.91)	(0.00)	(0.31)	
Year Dummies	Yes	Yes	Yes	Yes	Yes	Yes	
Industry Dummies	Yes	Yes	Yes	Yes	Yes	Yes	
N	1184	662	1085	609	1086	618	
adj. R^2	0.003	0.047	0.021	0.055	0.087	0.131	

Notes: This table shows the cross sectional regression results of long run abnormal returns associated with equity issues. The dependent variable is the 12-month buy-and-hold abnormal returns BHAR (2, 12) for models 1 and 2, 24-month buy-and-hold abnormal returns BHAR (2, 36) for models 5 and 6. Variables included in the regressions are operating performance (ROA), dummy for open offers (OO), and private placement (PP), the interaction terms for operating performance and open offers (ROA*OO) and private placement (ROA*PP), market-to-book ratio (MB), prior stock returns (Prior AR), logarithm of book value of total assets (SIZE), issue price discount (DISC), accruals quality (ACC_QUAL), idiosyncratic risk (RISK), ratio of proceeds to market value (PMV), dummy for financial crisis (CRISIS), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. *,** and **** denote significance at the 10%, 5% and 1% level respectively.

7.3.3. Long-term Operating Performance

Table 7.11 gives the results for the post-issue operating performance associated with equity issuing firms on whether the issue is made through rights issues (RI), open offers (OO), and private placement (PP). If managers select the issuance method in line with their expectations of firm operating performance, as argued above, rights issuing firms are more likely to have higher operating performance than counterpart firms that issue open offers and private placements. As discussed in Section 5.3.2.2 of Chapter 5, operating performance is the returns on assets of the issuing firm less the returns on assets of its control firm. As reported in the table rights issuing firms report higher average ROA than counterpart firms using open offers and private placements. The mean difference in ROA between RI and OO is significantly positive at the 1% level over the three-year period. Similarly, the ROA difference between RI and PP is also significantly positive over the three-year period. These results are to a large extent in line with the pattern previously found in returns after the issuance, offering further credence that firm managers select the equity issuance method based on a firm's future operating performance, which is then linked with differential long-term stock returns.

Table 7.11: Post-issue Changes in Operating Performance						
Year	RI	OO	PP	Total	RI-OO	RI-PP
1	0.069***	0.013***	0.031***	0.036***	0.056***	0.038***
2	0.082***	0.014^{***}	0.031***	0.041^{***}	0.069***	0.051***
3	0.052***	-0.019***	0.018***	0.018^{***}	0.071***	0.034***
N	328	321	608	1257		

Notes: This table shows the average operating performance (ROA) for rights issues (RI), open offers (OO) and private placements (PP) after equity issuance.*** shows significance at the 1% level.

7.4. Robustness and Sensitivity Analysis

This section presents the results of a number of robustness and sensitivity analysis. These additional results provide further support for the main results and also ensure the results are robust to alternative estimation techniques. The section is divided into Subsection 7.4.1 which provides results for alternative measures of operating performance; Subsection 7.4.2 gives results for firms using multiple issuance method during the period and Subsection 7.4.3 shows the tests of tobit regression estimates.

7.4.1. Logit Regression Results for Abnormal Returns on Assets (aROA) and Returns on Equity (ROE)

Table 7.12 presents the results for choice of equity issuance method based on operating performance measured as abnormal returns on assets (aROA). aROA is measured as the difference between the returns on assets of equity issuing firm and the returns on assets of the control firms. Control firms are identified in line with Spiess and Affleck-Graves (1995), Loughran and Ritter (1995) and Capstaff and Fletcher (2011). Panel A provides the results for the abnormal returns on assets proxy for the choice between open offers and rights issues whereas Panel B gives the results for the choice between private placement and rights issues. Tables 7.12 and 7.13 show multinomial logit estimations with independent variables α ROA and ROE. Similar to prior results and consistent with the predictions, the parameter coefficients of *aROA* and ROE are significantly negative, indicating that firms issuing open offers and private placements tend to anticipate poor operating performance at the time of issuance. Therefore, rights issuing firms are more likely to be profitable and better quality firms than those conducting open offers and private placement.

Table 7.12:	Multinomial Logit	Regression	Estimates of					
Equity Issuance Method (abnormal ROA)								
	(Year 1)	(Year 2)	(Year 3)					
Variables	(1)	(2)	(3)					
PANE	PANEL A: Open Offers (1) vs Rights Issues (0)							
aROA	-12.770***	- 9.137***	-5.254*					
	(0.00)	(0.00)	(0.05)					
MB	0.018	-0.042	-0.024					
	(0.68)	(0.29)	(0.54)					
AR	2.221**	2.091^{*}	2.250^{**}					
	(0.01)	$(0.09)_{a}$	(0.04)					
SIZE	-0.183	-0.293*	-0.132					
	(0.23)	(0.09)	(0.43)					
DISC	0.492***	0.596***	0.533***					
	$(0.01)_{**}$	(0.01)	$(0.01)_{**}$					
ACC_QUAL	-3.232**	-3.980	-2.918**					
	(0.03)	(0.00)	(0.04)					
RISK	7.605	11.678	8.849					
	$(0.54)_{***}$	(0.40)	$(0.50)_{***}$					
PMV	-10.266***	-9.426***	-9 [.] 995 ^{***}					
	(0.00)	(0.00)	(0.00)					
CRISIS	18.321	17.143	15.967 ^{***}					
	$(0.00)_{***}$	(0.00)	(0.00)					
Constant	-13.337***	-11.542	-12.002					
	(0.00)	(0.00)	(0.00)					

PANEL B: Private Placement (1) vs Rights Issues (0)						
aROA	-9.123***	-9.106***	-2.012			
	(0.00)	(0.00)	(0.39)			
MB	0.071*	0.047	0.053			
	(0.09)	(0.20)	(0.13)			
AR	1.915**	2.027^{*}	1.928*			
	(0.02)	(0.07)	(0.06)			
SIZE	-0.008	-0.008	0.018			
	(0.95)	(0.95)	(0.89)			
DISC	-0.312*	-0.295	-0.274			
	(0.08)	(0.17)	(0.14)			
ACC_QUAL	-0.615	-1.713	-1.461			
	(0.62)	(0.27)	(0.35)			
RISK	-5.301	2.211	1.812			
	(0.64)	(0.86)	(0.88)			
PMV	-3.212***	-3.258***	-2.911 ^{***}			
	(0.00)	(0.00)	(0.00)			
CRISIS	4.182***	3.895***	3.645***			
	(0.00)	(0.00)	(0.00)			
Constant	1.224	1.094	0.961			
	(0.48)	(0.56)	(0.60)			
Year Dummies	Yes	Yes	Yes			
Industry Dummies	Yes	Yes	Yes			
N	467	393	403			
pseudo R^2	0.431	0.446	0.420			
chi2	3706.498	3328.775	4118.569			
p	0.000	0.000	0.000			

Notes: This table shows multinomial logit results equity issuance method during the sample period. The dependent variable is 1 open offers (OO) in Panel A and private placements (PP) in Panel B and 0 for rights issues. The main independent variable is aROA and the following control variables are included: MB is the market-to-book ratio, Prior AR is the cumulative returns, SIZE is the logarithm of book value of total assets, DISC is the issuing price discount, ACC_QUAL measures earnings management, RISK is idiosyncratic volatility of stock returns, PMV is the total proceeds relative to firm size and CRISIS is a dummy variable that takes the value of 1 during the years 2008-2010. Standard errors are heteroscedasticity consistent. Year and industry dummies are included. P-values are shown in parentheses. *, ** and *** show significance at the 10%, 5% and 1% levels, respectively.

These results imply that the earlier results are robust to alternative definitions of operating performance. It further supports the hypothesis about the effects of anticipated operating performance on the choice of equity issuance method. Therefore, the hypothesis is confirmed regardless of the measure of operating performance, firms making rights issues anticipate high post-issue operating performance. This evidence is consistent with the overall outperformance of long run stock returns compared to open offers and private placements. P-

values, like the main results, are also significant suggesting that the joint hypotheses about the coefficients are significantly different from zero. In other words, the variables significantly define the choice of equity issuance method.

Table 7.13: Multinomial Logit Regression Estimates of							
Equity Issuance	Method (ROE)						
	(Year 1)	(Year 2)	(Year 3)				
Variables	(1)	(2)	(3)				
PANEL	A: Open Offers (1)	vs Rights Issu	ues (0)				
ROE	-4.117***	-2.281**	-2.842**				
	(0.00)	(0.05)	(0.04)				
MB	0.033	0.034	0.016				
	(0.44)	(0.41)	(0.71)				
AR	1.665**	1.143	0.990				
	(0.03)	(0.14)	(0.26)				
SIZE	-0.065	-0.215	-0.183				
	(0.65)	(0.14)	(0.28)				
DISC	0.415***	0.442***	0.609^{***}				
	(0.01)	(0.01)	(0.00)				
ACC_QUAL	-2.495 ^{**}	-2.369**	-2.274				
	(0.03)	(0.04)	(0.15)				
RISK	-5.427	-5.386	-2.594				
	(0.60)	(0.64)	$(0.84)_{}$				
PMV	-7.715 ^{***}	-6.894 ^{***}	-7.526***				
	(0.00)	(0.00)	(0.00)				
CRISIS	18.817***	17.635***	31.812***				
	(0.00)	(0.00)	(0.00)				
Constant	-14.103***	-11.386***	-12.438***				
	(0.00)	(0.00)	(0.00)				

PANEL B: Privat	te Placement ((1) vs Rights Iss	sues (0)
ROE	-1.370	-1.294	-0.880
	(0.21)	(0.22)	(0.47)
MB	0.084^{**}	0.087**	0.077**
	(0.02)	(0.02)	(0.02)
AR	1.501**	0.992	0.750
	(0.04)	(0.16)	(0.33)
SIZE	0.011	-0.110	-0.103
	(0.93)	(0.40)	(0.49)
DISC	-0.275*	-0.268*	-0.232
	(0.05)	(0.08)	(0.20)
ACC_QUAL	-1.053	-1.453	-1.755
	(0.40)	(0.28)	(0.26)
RISK	-10.225	-9.426	-6.302
	(0.26)	(0.36)	(0.59)
PMV	-2.458***	-2.028***	-2.103***
	(0.00)	(0.00)	(0.00)
CRISIS	4.205***	3.979***	17.630***
	(0.00)	(0.00)	(0.00)
Constant	1.895	3.069^*	3.409 [*]
	(0.22)	(0.06)	(0.07)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	597	534	443
pseudo R^2	0.379	0.369	0.401
chi2	•	11235.336	
p	•	0.000	

Notes: This table shows multinomial logit results equity issuance method during the sample period. The dependent variable is 1 open offers (OO) in Panel A and private placements (PP) in Panel B and 0 for rights issues. The main independent variable is ROE and the following control variables are included: MB is the market-to-book ratio, Prior AR is the cumulative returns, SIZE is the logarithm of book value of total assets, DISC is the issuing price discount, ACC_QUAL measures earnings management, RISK is idiosyncratic volatility of stock returns, PMV is the total proceeds relative to firm size and CRISIS is a dummy variable that takes the value of 1 during the years 2008-2010. Standard errors are heteroscedasticity consistent. Year and industry dummies are included. P-values are shown in parentheses. *, ** and **** show significance at the 10%, 5% and 1% levels, respectively.

7.4.2. Multinomial Logit Regression Results for Multiple Equity Issuance Method during the Period

If anticipated operating performance or asymmetric information determines the choice of equity issuance method, it can also be argued that managers may be reluctant to change their issuing approach. Thus, some firms may have selected the same issuance method throughout the sample period irrespective of managers' expectations of forthcoming operating performance. For example, Iqbal et al. (2013) contend that long-term stock returns after issuance for rights issues is influenced by whether firms had previously used the same equity issuance method. They find higher long-term stock returns after issuance for firms that had previously issued rights issues at least twice.

Table 7.14 shows the results for the logit estimations using only firms that have changed their equity issuance method during the sample period. As noted in the reported results, the coefficient of ROA is significantly negative in Panels A and B indicating that anticipated operating performance actually influences the choice of rights issues method over open offers and private placement. The parameter coefficients of the control variables are similar to the main results for the full sample. These results offer support to the acceptance of hypothesis about the effects of anticipated operating performance on the choice of rights issuance method.

Table 7.14: Multin	nomial Logit	Regression	Estimates of
Changes in Firm's Is	suance Metho	d During the S	Sample Period
	(Year 1)	(Year 2)	(Year 3)
Variables	(1)	(2)	(3)
PANEL A: (Open Offers (1)		es (0)
ROA	-14.976 ^{***}	-10.283***	-5.157
	(0.00)	(0.00)	(0.14)
MB	0.015	0.005	0.006
	(0.71)	(0.91)	(0.87)
AR	1.579*	0.956	1.099
	(0.09)	(0.32)	(0.22)
SIZE	-0.201	-0.207	-0.210
	(0.26)	(0.24)	(0.22)
DISC	0.542***	0.534***	0.485***
	(0.00)	(0.01)	(0.01)
ACC_QUAL	-2.912**	-2.047	-1.630
	(0.04)	(0.15)	(0.29)
RISK	4.464	7.480	7.108
	(0.68)	(0.51)	(0.55)
PMV	-7.517***	-6.883***	-6.805***
	(0.00)	(0.00)	(0.00)
CRISIS	16.721	17.865***	17.356
	(0.00)	$(0.00)_{***}$	(0.00)
Constant	-11.460	-12.693***	-12.869
	(0.00)	(0.00)	(0.00)

PANEL B: Priva	PANEL B: Private Placement (1) vs Rights Issues (0)							
ROA	-13.675***	-10.237***	-0.923					
	(0.00)	(0.00)	(0.73)					
MB	0.065**	0.059^{*}	0.066^{**}					
	(0.05)	(0.07)	(0.03)					
AR	1.743**	1.395	1.385*					
	(0.04)	(0.11)	(0.10)					
SIZE	-0.040	-0.039	-0.084					
	(0.80)	(0.81)	(0.58)					
DISC	-0.151	-0.100	-0.149					
	(0.38)	(0.60)	(0.36)					
ACC_QUAL	-2.163	-2.437	-2.458					
	(0.14)	(0.13)	(0.13)					
RISK	-7.634	-5.014	0.489					
	(0.43)	(0.65)	(0.96)					
PMV	-2.096***	-2.091**	-1.811***					
	(0.01)	(0.02)	(0.01)					
CRISIS	3.494***	3.544***	3.438***					
	(0.01)	(0.01)	(0.01)					
Constant	2.692	2.531	2.241					
	(0.16)	(0.20)	(0.23)					
Year Dummies	Yes	Yes	Yes					
Industry Dummies	Yes	Yes	Yes					
N	441	406	407					
pseudo R^2	0.358	0.354	0.337					
chi2	4453.672	4609.836	5905.474					
p	0.000	0.000	0.000					

Notes: This table shows multinomial logit results for firms that used more than one equity issuance method during the sample period. Firms that only used one issuance method to raise equity are excluded. The dependent variable is 1 open offers (OO) in Panel A and private placements (PP) in Panel B and 0 for rights issues. The main independent variable is ROA and the following control variables are included: MB is the market-to-book ratio, Prior AR is the cumulative returns, SIZE is the logarithm of book value of total assets, DISC is the issuing price discount, ACC_QUAL measures earnings management, RISK is idiosyncratic volatility of stock returns, PMV is the total proceeds relative to firm size, and CRISIS is a dummy variable that takes the value of 1 during the years 2008-2010. Standard errors are heteroscedasticity consistent. Year and industry dummies are included. P-values are shown in parentheses. , and show significance at the 10%, 5% and 1% levels, respectively.

7.4.3. Tobit Regression Estimation of Main Results

Table 7.15 provides the tobit regression results for equity issues given the level of operating performance. The dependent variable is the dollar value of equity issues divided by the market value of equity. This variable explains the effects of operating performance on the amount of shares issued. Model 1, 2, and 3 give the results of the effects of operating performance on equity issues for year 1, 2, and 3 respectively. As can be seen from the table, the coefficient of ROA is positive and statistically significant at the 1% level in Models 1 and 2. Overall, the results indicate that high operating performing firms are more likely to increase the amount of equity issues.

The coefficient of ROA*OO and ROA*PP are negative and significant at the 1% level in Models 1 and 2. This result suggests high operating performing firms increase the amount generated from equity issues when they make rights issues rather than open offers and private placement. Thus, firms are more likely to generate greater value from rights issues when they have better operating performance. This result suggests that the level of operating performance not only determine the probability of equity issues but also the amount that can potentially be generated from such issues. Overall, the above results provide further evidence about the effects of operating performance on equity issuance decisions.

Table 7.15: Tobit Regression Analysis of Equity
Issuance Method

	Year 1	Year 2	Year 3
Variables	(1)	(2)	(3)
ROA	9.659***	2.438***	1.342
	(0.00)	(0.00)	(0.11)
OO	1.478**	2.197***	-2.007***
	(0.02)	(0.00)	(0.00)
PP	0.827	1 235*	-2.911 ^{***}
	(0.23)	(0.07)	(0.00)
ROA*OO	-3.066 ^{***}	-3.091***	-2.847
	(0.00)	(0.00)	(0.11)
ROA*PP	-3.960***	-7.825 ^{***}	-4.938
	(0.00)	(0.00)	(0.17)
MB	0.064**	0.082**	0.042
	(0.04)	(0.02)	(0.41)
AR	0.198	0.556	0.279
	(0.74)	(0.44)	(0.70)
SIZE	1.099***	1.266***	1.248***
	(0.00)	(0.00)	(0.00)
DISC	-0.016	-0.055	-0.075
	(0.93)	(0.77)	(0.69)
ACC_QUAL	1.095	0.432	0.444
	(0.50)	(0.82)	(0.82)
RISK	3.378	8.618	3.881
	(0.70)	(0.42)	(0.73)
CRISIS	2.476***	2.048^{*}	2.072^{**}
	(0.00)	(0.05)	(0.04)
Constant	-12.835***	-14.718***	-10.052*
	(0.01)	(0.00)	(0.05)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	662	609	618
pseudo R^2	0.019	0.021	0.019
p	0.000	0.000	0.000

Notes: This table shows Tobit regression results for equity issuance method. The dependent variable is the total proceeds divided by market value of equity. The main independent variable is ROA, OO is 1 for open offers and 0 for rights issues, PP is 1 for private placement and 0 for rights issues, ROA*OO is the interaction between returns on assets and 1 for open offers and 0 for rights issues, and ROA*PP is the interaction between returns on assets and 1 for private placement and 0 for rights issues. MB is the market-tobook ratio, Prior AR is the cumulative returns, SIZE is the logarithm of book value of total assets, DISC is the issuing price discount, ACC_QUAL measures earnings management, RISK is idiosyncratic volatility of stock returns, PMV is the total proceeds relative to firm size, CRISIS is a dummy variable that takes the value of 1 during the years 2008-2010. Standard errors are heteroscedasticity consistent. Year and industry dummies are included. P-values are shown in parentheses. *, **and *** show significance at the 10%, 5% and 1% levels, respectively.

7.5. Conclusion

This chapter presented empirical results to explore the information asymmetry effects on the choice of equity issuance methods of UK listed firms. It specifically investigated how anticipated or future operating performance determines firm's selection of the equity issuance methods. The main hypothesis tested in this chapter was that firms with anticipated high operating performance are more likely to choose rights issues in issuing equity. In other words, firms with high information asymmetry in the form of anticipated improved operating performance prefer to issue equity to the existing shareholders in order to transfer part of the future benefits to them. Conversely, firms with overvalued equity are more likely to issue equity through open offers and private placement that involve outside or new investors. The implication of this hypothesis was that better performing firms prefer to transfer the benefits to existing shareholders whereas firms which are likely to record poor performance are motivated to exploit outside investors who would have insufficient information about the performance of the firms. In addition to the traditional proxies for information asymmetry such as accruals quality, issue price discount, idiosyncratic risks, and market to book ratio (which all measure the quality of the issuing firms, operating performance is used as the main descriptor of information asymmetry.

Consistent with prior studies, the results of this chapter find strong support for the preference for rights issues when operating performance is anticipated to improve. Thus, better performing firms are more likely to issue equity through rights issues than open offers and private placement. This evidence indicates that the firms do not exploit existing shareholders and that rights issuing firms are of better quality than open offers and private placement. Also, the chapter 7 finds that rights issuing firms report more positive post-issue operating performance which is consistent with stock price outperformance over open offers and private placement. Moreover, the chapter addresses issues with estimation assumptions and also presented results using alternative measures of operating performance. Finally, the chapter presented robust results for alternative estimation techniques.

These findings are consistent with Kabir and Roosenboom (2003), Armitage and Snell (2004), Chi and Gupta (2009), and Capstaff and Fletcher (2011). However, the results in this chapter do not support the findings by Slovin et al. (2000) and Barnes and Walker (2006). Unlike previous studies, this chapter used operating performance to demonstrate that future anticipated changes in operating performance by which firms determine their choice of equity issuance method. Thus, holding the measures of firm and issue quality constant- such as mispricing, the decision to utilise a particular issuance method is influenced by operating performance. However, the results presented in this chapter demonstrate that firms choose equity issuance method based on specific firm and issue characteristics. This evidence complements the extant literature on the equity issuance method in the UK. It shows that operating performance does not only influence the post-issue stock price performance but also and more importantly determine the choice of the issuance method.

Chapter 8

LEVERED SHARE REPURCHASE AND INTERNAL FINANCIAL CONSTRAINTS

8.1. Introduction

This section discusses the empirical results for levered share repurchase of internally financially unconstrained firms. During the last decades, share repurchases have become very popular as a means to return cash to shareholders. In 2007, share repurchases peaked at more than \$700 billion near the market top.³⁷ A large number of these repurchases were funded with debt issues. For instance, in 2009 alone 37 companies announced plans to spend \$39 billion on these levered repurchases.³⁸ It may be understandable for internally financially constrained firms to repurchase their shares by borrowing due to limited financial resources available, but it seems puzzling for internally financially unconstrained firms to borrow money in order to repurchase shares. The empirical analysis in this chapter investigates why internally financially unconstrained firms borrow money to repurchase shares. It further explores the stock returns associated with share repurchase given the level of internal financial constraints and sources of repurchase financing.

The extant empirical literature finds strong support for positive announcement period abnormal stock returns associated with share repurchase programs (see, for example, Vermaelen, 1981; Ikenberry et al., 1995). Theories such as the signalling hypothesis, the undervaluation hypothesis (see Vermaelen, 1981; Comment and Jarrell, 1991; Dittmar, 2000; Chan et al., 2007) and the free cash flow hypothesis of Jensen (1986) all explain the positive price reactions subsequent to share repurchase announcements. Since internally financially unconstrained firms already have sufficient excess cash flow to finance share repurchases, managers may borrow at the time of share repurchase in order to fund wealth-maximizing projects. Also, stock returns of internally financially

³⁷ See Michael Milken: "Why capital structure matters", *The Wall Street Journal*, April 21, 2009.

³⁸ Source: research by JP Morgan cited by Herb Greenbert in: "Debt to buy back stock", CNBC, November 8, 2011. See: http://www.cnbc.com/id/45209702 (retrieved on March 30, 2015).

unconstrained firms with a high level of investment should increase on the announcement of share repurchases; otherwise the increase in their stock returns will appear in the long-run stock returns when the benefits of the investments materialize, especially in an efficient market.

It can be argued that internally financially unconstrained firms conduct levered share repurchases to support investment expenditures. A number of studies have shown that share repurchases involve a significant cost that may limit the availability of internal cash to fund investments. Grullon and Michaely (2004), for example, show that firms may reduce their investments after share repurchase activities due to limited available cash flows. However, according to the precautionary motive for holding cash, future uncertainties in the capital market motivate firms to build cash buffers that allow them to undertake future investments (See, Opler, et al., 1999; Bates et al., 2006). Hahn and Lee (2009) further argue that firms with a high debt capacity are able to borrow more, which in turn allows for more investments in assets that can serve as collateral for further borrowing. Since internally financially unconstrained firms both have the borrowing capacity and the access to attractive interest rates, it is hypothesised that they borrow externally to retain adequate capital for funding investments. Thus, the hypothesis explained in Section 4.5 of Chapter 4 explores the share repurchase financing of internally financially unconstrained firms. Specifically, the study predicts that internally financially unconstrained firms are likely to conduct levered share repurchase when they also invest more. This behaviour should also have significant stock price implications for the levered share repurchasing firms, both in the short run and the long run. The hypotheses about the levered share repurchase are tested using both univariate and multivariate analyses.

Recent evidence by Chen and Wang (2012) highlights the share repurchase behaviour of constrained firms and the subsequent post-repurchase stock price underperformance. The inference from their studies is consistent with debt-financed repurchase of financially constrained firms. This study explores the levered share repurchase behaviour of internally financially unconstrained firms and tests the factors that drive cash-rich firms to finance their share

repurchases through borrowings. Minnick and Zhao (2007) find that firms borrow at the time of repurchase whereas De Jong et al. (2011) also assert that convertible bond issuers repurchase their stock in order to facilitate short selling. Both evidence shed some light on firm share repurchase financing. However this study contributes to the literature by focusing on unconstrained firms that undertake levered share repurchases.

The chapter consists of four parts. Section 8.1 describes the sample used to perform the empirical analysis. Section 8.2 presents the empirical findings about why internally financially unconstrained firms borrow to conduct share repurchase. Section 8.3 discusses additional results and performs robustness checks using alternative definitions of internal financial constraints and estimation techniques. Section 8.4 concludes the chapter.

8.2. Sample Statistics

This section discusses the descriptive statistics of the variables selected to undertake the empirical analysis. The variables presented here are used in prior studies to explain share repurchase transactions of firms. Thus, the variables allow the study to test the reasons why internally financial unconstrained firms conduct levered or debt-financed share repurchase. Section 8.2.1 presents the descriptive statistics for the final sample.

8.2.1. Descriptive Statistics of Independent and Control Variables

This section presents the descriptive statistics of independent and control variables used to undertake the empirical analysis. It focuses on the statistics of the firm characteristics of share repurchasing firms. These repurchase firms are partitioned into constrained and unconstrained. The unconstrained firm subsample is further divided into levered and unlevered to draw meaningful comparisons about the characteristics of repurchase firms based on the source of financing.

8.2.1.1. Share Repurchase during the Sample Period

Table 8.1 presents the number of yearly share repurchase announcements during the sample period, the anticipated proceeds from the repurchase, and the

average ratio of the proceeds to market value. As can be seen from the table, the annual issues are generally stable during the early part of the sample period from 1990 to 1993. During years 1994 and 2000, the number of share repurchases announcements increases significantly and drops to the levels in the 1990s between years 2001 and 2006 with a marginal increase in 2007 and 2008. Finally, the number drops significantly after year 2008; perhaps the financial crisis accounted for such drop as more firms faced significant cash constraints to even distribute to shareholders. The results on the proposed proceeds show a marked different picture from the volume of repurchase. Specifically, the proceeds are significantly lower during the first part of the sample period from year 1990 to 1999.

Table 8.1: Yearly Average Proceeds and							
Rati	Ratio of Share Repurchase Proceeds						
Year	N	Proceeds	PMV				
1990	233	336.82	0.13				
1991	113	58.31	0.08				
1992	194	81.82	0.07				
1993	170	92.44	0.05				
1994	253	106.19	0.07				
1995	308	110.89	0.08				
1996	442	185.46	0.06				
1997	416	153.68	0.22				
1998	642	145.78	0.08				
1999	468	122.69	0.09				
2000	256	251.05	0.09				
2001	155	289.08	0.08				
2002	123	268.01	0.09				
2003	100	243.81	0.07				
2004	153	570.20	0.08				
2005	229	553.56	0.08				
2006	193	980.25	0.08				
2007	298	720.96	0.11				
2008	394	418.06	0.15				
2009	135	545.49	0.08				
2010	188	758.96	0.09				
2011	264	648.55	0.11				
2012	147	1429.43	0.10				
Total	5,874	345.84	0.10				

This table shows the annual share repurchase of sample firms. It also indicates the average yearly proceeds from share repurchase announcements (PROCEEDS) and the ratio of proceeds to market value of firms (PMV).

However, between 2000 and 2012 the proceeds increase markedly. This amount of proceeds does not seem to depend on the number of repurchase announcements. In fact, during some periods when fewer repurchase announcements are made, the amount of proceeds is even larger than some periods of significantly higher number of repurchase announcements. Finally, the ratios indicate a stable value of proceeds per unit of firm value. It appears large firms announced more share repurchases than small firms. This evidence is consistent with free cash hypothesis that hold that large and mature firms (which have limited growth opportunities) are more likely to have significant free cash flows to distribute to shareholders (Jensen, 1986; Grullon and Michaely, 2004; Liang et al. 2013).

8.2.1.2. Descriptive Statistics of Variables

Table 8.2 provides descriptive statistics of some firm characteristics that affect share repurchase by internally financially constrained and unconstrained firms. ³⁹ It also performs independent sample *t-test* of the mean difference between constrained and unconstrained repurchasing firms. This mean difference test provides preliminary differences in firm characteristics that explain the share repurchase behaviour given the level of internal financial constraints. Panel A shows the descriptive statistics for the unconstrained firms whereas Panel B displays the descriptive statistics for constrained firms. The mean difference tests are reported in Panel C.

As can be seen from Table 8.2, the mean MB for unconstrained firms is 2.77 compared to 2.24 for constrained firms. AR is -4% and -6% for unconstrained and constrained firms, respectively. The mean difference for MB is 0.53 and statistically significant at the 1% level, where as that of AR is 1% and significant at the 1% level. These results indicate that constrained firms are more undervalued than unconstrained firms. Perhaps, the level of undervaluation motivates constrained firms to conduct share repurchase.

_

³⁹ The definition of financial constraints is given in Section Chapter 5. It further describes how repurchasing firms are classified as constrained or unconstrained.

Table 8.2: Summary Statistics: Comparison of Unconstrained and Constrained Firms

	Firms						
Panel A: Unconstrain	Panel A: Unconstrained						
Variables	N	Mean	Median	S.D	Min	Max	
MB	4700	2.77	2.11	2.12	0.44	10.26	
Prior AR	4700	-0.04	-0.03	0.17	-1.43	1.03	
INV	4700	0.06	0.05	0.06	0.00	0.68	
SIZE	4700	2.79	2.76	0.95	-0.06	5.80	
PROCEEDS (\$'m)	4623	388.61	30.00	2184.47	0.12	90000.00	
PMV	4611	0.08	0.06	0.16	0.00	8.46	
Panel B: Constrained							
Variables	N	Mean	Median	S.D	Min	Max	
MB	1174	2.24	1.66	2.02	0.44	10.26	
Prior AR	1174	-0.06	-0.04	0.17	-1.00	0.50	
INV	1174	0.09	0.06	0.12	-0.02	1.09	
SIZE	1174	2.61	2.64	0.80	0.00	5.57	
PROCEEDS (\$'m)	1162	175.69	25.00	803.40	0.11	15000.00	
PMV	1160	0.16	0.07	1.91	0.00	64.94	

Panel C: Test of Mean Difference of unconstrained and constrained firms

.,	.,,,		J
Variables	N	Mean difference	t-stat
MB	5874	0.53***	(7.73)
Prior AR	5874	0.01***	(2.60)
INV	5874	-0.03***	(-12.59)
SIZE	5874	0.18***	(5.92)
PROCEEDS	5785	212.92***	(3.27)
PMV	5771	-0.07**	(-2.54)

Notes: This table shows the descriptive statistics of the variables used in the empirical analysis for testing levered share repurchase behaviour of firms. Panel A provides results for constrained firms, Panel B for unconstrained firms and Panel C tests the mean difference of variables. Variables include market to book ratio (MB), pre-announcement abnormal returns (Prior AR), ratio of capital expenditure and research and development expenditure to total assets (INV), logarithm of book value of total assets (SIZE), value of repurchase proceeds (PROCEEDS), and ratio of repurchase value to market value (PMV). T-statistics are shown in parenthesis in Panel C. ** and *** denote significance at the 5% and 1% level respectively.

Unexpectedly, the mean difference in INV between unconstrained and constrained is negative and significant at the 1% level indicating that constrained firms invest more than unconstrained firms. The table also shows mean difference of 0.18 for SIZE at 1% significance level. The difference in firm size is consistent with the theory that large firms are more likely to be unconstrained. Moreover, the proceeds from share repurchase are larger for the unconstrained firms than the constrained firms. Overall, these results show that unconstrained firms have more free cash flows and thus are more likely to disburse cash flows to shareholders. It appears the evidence is consistent with prior literature about share repurchase of firms.

Table 8.3: Summary Statistics: Comparison of Unlevered and Levered Repurchase

Panel A: Unlevered	l					
Variables	N	Mean	Median	S.D	Min	Max
MB	3359	2.49	1.92	1.93	0.44	10.26
Prior AR	3359	-0.05	-0.03	0.19	-1.43	1.03
INV	3359	0.06	0.04	0.06	0.00	0.68
SIZE	3359	2.46	2.43	0.84	-0.06	5.80
PROCEEDS	3302	170.77	15.00	1750.08	0.12	90000.00
PMV	3290	0.08	0.06	0.10	0.00	1.95
Panel B: Levered						_
Variables	N	Mean	Median	S.D	Min	Max
MB	1341	3.47	2.70	2.41	0.44	10.26
Prior AR	1341	-0.03	-0.02	0.12	-0.60	0.51
INV	1341	0.07	0.05	0.07	0.00	0.67

3.57

224.40

0.06

0.70

2938.40

0.25

1.65

0.35

0.00

5.59

65000.00

8.46

Panel C: Test of Mean Difference of levered and unlevered repurchase

3.60

933.14

0.09

1341

1321 1321

SIZE

PROCEEDS

.,	.,,,	1	
Variables	N	Mean difference	t-stat
MB	1923	-2.28***	(-10.32)
Prior AR	2500	-0.06***	(-5.15)
INV	1658	-0.76***	(-7.74)
SIZE	2401	-0.84***	(-13.97)
PROCEEDS	2327	0.28***	(11.34)
PMV	2154	0.04***	(5.55)

Notes: This table shows the descriptive statistics of the variables used in the empirical analysis for testing levered share repurchase behaviour of unconstrained firms. Panel A provides results for unlevered, Panel B for levered repurchase and Panel C tests the mean difference of variables. Variables include market to book ratio (MB), pre-announcement abnormal returns (Prior AR), ratio of capital expenditure and research and development expenditure to total assets (INV), logarithm of book value of total assets (SIZE), value of repurchase proceeds (PROCEEDS), and ratio of repurchase value to market value (PMV). T-statistics are in parenthesis in Panel C and denotes significance at the 1% level.

Table 8.3 provides the summary statistics of firm and repurchase characteristics of unconstrained firm subsample. The sample is further divided into unlevered and levered depending on whether share repurchase financing is from internal cash flows or external debt financing. Panel A presents the results for unlevered repurchase whereas Panel B gives the results for levered repurchase. The tests of the mean difference in the variables between unlevered and levered repurchase is provided in Panel C.

It is found that the mean MB is 2.49 and 3.47 for unlevered and levered repurchase, respectively with a negative and significant mean difference at the

1% level. The average prior repurchase abnormal returns are -5.0% and -3.0% for unlevered and levered repurchase respectively with mean difference significant at the 1% level. These two results indicate that unlevered repurchase firms are more undervalued than levered repurchase firms. The table also reports that the mean difference in investment between unlevered and levered is negative and significant at the 1% level. This result implies levered repurchase firms expend more on investments than unlevered repurchase firms. Levered repurchase firms are bigger than unlevered repurchase firms but the ratio of repurchase amount is not statistically different between unlevered and levered repurchase firms. Generally, large firms have adequate debt capacity and collateral to enable them obtain external debt financing. Overall, evidence shows the first sign that levered repurchase firms are likely borrow to invest.

8.2.1.3. Pearson's Correlation Matrix of Variables

Logistic regression is used to estimate the model that tests the effects of internal financial constraints on share repurchase (H_8 and H_9) as explained in Section 5.3.3.4. However, OLS is used to test hypothesis H_{I0} and H_{II} about the stock returns following share repurchase. As discussed in Chapter 6 Section 6.2.1.3, a correlation coefficient close to or equal to +1 and -1 suggests high collinearity or perfect multicollinearity among the variables. Table 8.4 provides the Pearson's correlation matrixes of independent and control variables for the sample of share repurchase.

As the shown in Table 8.4, none of the variable indicates high collinearity with the highest being 0.507 for SIZE and LEVD. Most of the variables are correlated at either the 1% or 5% level, with most of the correlations lying close or below 0.20. This results implies low multicollinearity and thus tolerable for OLS estimation. The high correlation between SIZE and LEVD could explain that large firms possess collateral to obtain external borrowings. Overall, the results show that multicollinearity is not a problem for estimating the regression model using OLS.

		Ta	ble 8.4: Pea	rson's Corr	elation Matr	rix of Variab	les		
	LEVD_{dum}	KZ	MB	Prior AR	INV	SIZE	VALUE	PMV	$ m CRISIS_{dum}$
LEVD _{dum} KZ	1.000 0.110***	1.000							
MB	0.110	-0.265***	1.000						
Prior AR	0.033^{**}	-0.053***	0.021	1.000					
INV	0.107^{***}	0.136***	0.080***	-0.055***	1.000				
SIZE	0.507***	-0.164***	0.469^{***}	0.097^{***}	0.023^{*}	1.000			
VALUE	0.139^{***}	-0.087***	0.130^{***}	0.028^{**}	-0.012	0.314***	1.000		
PMV	-0.009	0.026^{**}	-0.021	-0.014	-0.011	-0.041***	0.067***	1.000	
CRISIS _{dum}	0.020	-0.034***	-0.025*	0.038^{***}	-0.128***	0.154***	0.080^{***}	0.008	1.000

Notes: Variables are defined as dummy for levered share repurchase ($LEVD_{dum}$), KZ index measure of internal financial constraints (KZ), market to book ratio (MB), pre-announcement abnormal returns ($Prior\ AR$), ratio of capital expenditure and research and development expenditure to total assets (INV), logarithm of book value of total assets (SIZE), financial crisis dummy ($CRISIS_{dum}$), amount of share repurchase (VALUE), ratio of repurchase value to market value (PMV). * , * and *** denote significance at the 10%, 5% and 1% level respectively.

8.3. Testing the Effects of Internal Financial Constraints on Levered Share Repurchase

This section tests the hypotheses specified in Section 4.4.1 and 4.4.2 of Chapter 4 using the logit regression model (Model 5) stated in Chapter 5. The hypotheses are re-stated below:

H₈: Financially constrained firms are more likely to conduct debtfinanced repurchase than unconstrained firms

H_9 : Financially unconstrained firms borrow to repurchase especially when they also undertake investments

Model 5 discussed in Section 5.2.3.4 tests the effects of internal financial constraints on levered share repurchase. The model is re-stated below:

$$\Pr(y_i = 1 | x_i) = \beta_0 + \beta_1 K Z_{it} + \beta_2 I N V_{it} + \beta_3 K Z * I N V_{it} + \sum_{i=1}^n \beta_i X_{it} + \varepsilon_{it} \dots (5)$$

The model tests the probability of levered share repurchases given internal financial constraints and a set of firm characteristics that explain share repurchase behaviour of firms. Since the model represents logit regression estimation, the dependent variable is 1 for levered repurchase and 0 for unlevered repurchase firm.

8.3.1. Logistic Regression Results for Levered Share Repurchase of Unconstrained Firms

[Hypothesis H_8 and H_9]

Table 8.5 presents the logit regression results about the probability that a firm conducts levered share repurchase given the level of internal financial constraints and a set of firm characteristics. The results for Models 1 and 2 use the full sample of share repurchase firms. Regression Model 1 shows the results for the effects of internal financial constraints on levered share repurchase holding constant the firm level variables including investment, INV. Model 2 gives the results of the logit regression when internal financial constraints and investments are interacted (KZ*INV). Both Models also contain control variables and year and industry dummies to control for year and

industry specific effects. Model 3 restricts the sample to only internally financially unconstrained firms that conducted share repurchase. The rationale is to detach the effects of investments on repurchase from the internal financial constraints variables that includes the constrained firm subsample. It also allows the study to investigate clearly how INV affects the levered share repurchase of internally financially unconstrained firms. In other words, Model 3 answers the question about why unconstrained firms borrow to conduct share repurchases.

As hypothesised, KZ has a positive and significant coefficient in Models 1 and 2 indicating that internal financial constraints increase the probability of levered share repurchase. Also, the coefficient of INV is positive and statistically significant at the 1% level in both Models 1 and 2. It shows that high investing firms are more likely to borrow to conduct share repurchase. The evidence about investment is consistent with building cash buffers at the time of share repurchase since repurchase reduces subsequent cash flows and hence stifle investment expenditure. Importantly, the coefficient of the interaction term KZ*INV is negative and statistically significant at the 1% level as shown in Model 2.

Table 8.6 is similar to Table 8.5 with SIZE variable excluded due to high correlation with LEVD and MB variables. The coefficients are similar to what is reported in Table 8.5. The results about the interaction term may produce two conclusions: They could imply that internally financially unconstrained and high investing firms are more likely to conduct levered share repurchase or that internally financially constrained firms which invest less are likely to conduct levered repurchase. The former conclusion is consistent with the developed hypothesis about share repurchase as explained in Sections 4.5.1 and 4.5.2 of Chapter 4.

Table 8.5: Logit Regression Estimates (Raw Coefficient Estimates) of Levered Repurchase and Internal Financial Constraints

Variables	(1)	(2)	(3)
KZ	1.162***	0.873***	
	(0.00)	(0.00)	
INV	0.466***	0.424***	0.384***
	(0.00)	(0.00)	(0.00)
KZ^*INV		-2.673***	
		(0.00)	
MB	-0.101***	-0.085***	-0.119 ^{***}
	(0.00)	(0.00)	(0.00)
Prior AR	-0.651***	-0.633***	-0.408
	(0.00) -1.018***	(0.01)	(0.13)
CRISIS	-1.018***	-1.082***	-1.589 ^{***}
	(0.00)	(0.00)	(0.00)
SIZE	1.995***	2.021****	2.106****
	(0.00)	(0.00)	(0.00)
PMV	0.030	0.033^{*}	1.072**
	$(0.11)_{}$	(0.08)	(0.02)
Constant	-6.496***	-6.470***	-6.548***
	(0.00)	(0.00)	(0.00)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	5771	5771	4611
pseudo R^2	0.319	0.324	0.346
chi-squared	1302.556	1306.272	1043.698
Notare This table shows	0.000	0.000	0.000

Notes: This table shows the coefficient estimates of the logit regression for share repurchase. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Models 1 and 2 provide results for the full sample whereas model 3 uses only the subsample of unconstrained firms. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), investment expenditure (INV), the interaction term between internal financial constraints and investment (KZ*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), the ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.*, ** and **** denote significance at the 10%, 5% and 1% level respectively.

Table 8.6: Logit Regression Estimates (Raw Coefficient Estimates) of Levered Repurchase and Internal Financial Constraints

Variables	(1)	(2)	(3)
KZ	0.692***	0.565***	_
	(0.00)	(0.00)	
INV	0.361***	0.346***	0.249^{***}
	(0.00)	(0.00)	(0.01)
KZ^*INV		-1.139***	
		(0.00)	
MB	0.188***	0.196***	0.221***
	(0.00)	(0.00)	(0.00)
Prior AR	0.189	0.202	0.530***
	(0.27)	(0.24)	(0.01)
CRISIS	0.659**	0.652**	0.230
	(0.02)	(0.02)	(0.47)
PMV	-0.035	-0.035	0.469***
	(0.18)	(0.19)	(0.01)
Constant	-2.338***	-2.313***	-2.367***
	(0.00)	(0.00)	(0.00)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	5771	5771	4611
pseudo R^2	0.107	0.108	0.116
chi-squared	638.615	638.383	528.748
<u>p</u>	0.000	0.000	0.000

Notes: This table shows the coefficient estimates of the logit regression for share repurchase. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Models 1 and 2 provide results for the full sample whereas model 3 uses only the subsample of unconstrained firms. Variables included in the regressions include KZ-index measure of internal financial constraints (KZ), investment expenditure (INV), the interaction term between internal financial constraints and investment (KZ*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), the ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. *** and ***** denote significance at the 5% and 1% level respectively.

Tables 8.7 and 8.8 provide the odds ratios of the logit regression estimations with SIZE and without SIZE respectively. From these results, the odds ratio for KZ*INV indicates that constrained firms which invest more are 0.48 times likely to conduct levered repurchase. The inverse produces higher odds ratio for unconstrained high investing firms indicating that these firms would more likely conduct levered repurchase.

Table 8.7: Logit Regression Estimates (Odds Ratio) of Levered Repurchase and Internal Financial Constraints

Variables	(1)	(2)	(3)
KZ	3.197***	2.393***	
	(0.00)	(0.00)	
INV	1.594***	1.528***	1.468***
	(0.00)	(0.00)	(0.00)
KZ^*INV		0.480***	
		(0.00)	
MB	0.904^{***}	0.919***	0.888^{***}
	(0.00)	(0.00)	(0.00)
Prior AR	0.522***	0.531***	0.665
	(0.00)	(0.01)	(0.13)
CRISIS	0.361***	0.339***	0.204***
	(0.00)	(0.00)	(0.00)
SIZE	7.355***	7.544***	8.213***
	(0.00)	(0.00)	(0.00)
PMV	1.030	1.033*	2.921**
	(0.11)	(0.08)	(0.02)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	5771	5771	4611
pseudo R^2	0.319	0.324	0.346
chi-squared	1302.556	1306.272	1043.698
p	0.000	0.000	0.000
Notes This table shows t	1	as officient estima	4 (- 11 4: - \ - -

Notes: This table shows the exponentiated coefficient estimates (odds ratio) of the logit regression for share repurchase. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Models 1 and 2 provide results for the full sample whereas model 3 uses only the subsample of unconstrained firms. Variables included in the regressions are KZ-index measure of internal financial constraints (KZ), investment expenditure (INV), the interaction term between internal financial constraints and investment (KZ*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), the ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.*, *** and **** denote significance at the 10%, 5% and 1% level respectively.

Levered Repurchase and Internal Financial Constraints Variables (2)(3) 1.998 1.760 ΚZ (0.00)(0.00)1.414*** 1.283*** **INV** 1.435* (0.00)(0.00)(0.01)KZ*INV 0.124* (0.00)1.248*** 1.206*** 1.217*** MB (0.00)(0.00)(0.00)1.208 1.224 1.698* Prior AR (0.24)(0.27)(0.01)**CRISIS** 1.933 1.920 1.259 (0.02)(0.02)(0.47)1.599* **SIZE** 0.965 0.966 (0.19)(0.01)(0.18)**PMV** Yes Yes Yes

Yes

5771

0.107

638.615

0.000

1.998

(0.00)

Year Dummies

pseudo R^2

chi-squared

Industry Dummies

Yes

5771

0.108

638.383

0.000

 1.760^{**} (0.00)

Yes

4611

0.116

528.748

0.000

Table 8.8: Logit Regression Estimates (Odds Ratio) of

Notes: This table shows the exponentiated coefficient estimates (odds ratio) of the logit regression for share repurchase. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Models 1 and 2 provide results for the full sample whereas model 3 uses only the subsample of unconstrained firms. Variables included in the regressions are KZ-index measure of internal financial constraints (KZ), investment expenditure (INV), the interaction term between internal financial constraints and investment (KZ*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), the ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. *, ** and **** denote significance at the 10%, 5% and 1% level respectively.

The other variables give the expected signs for the effects on the probability of a levered share repurchase. For example, the coefficients of MB and AR are negative and statistically significant at the 1% level in Models 1 and 2. These evidence support the undervaluation hypothesis of share repurchase transactions (Stephens and Weisbach, 1998; D'Mello and Shroff, 2000). More undervalued firms are likely to conduct levered repurchase. It seems firms take advantage of undervalued equity to borrow to enable them buy back undervalued shares in order to gain from subsequent price appreciations. The

CRISIS coefficient is also negative and statistically significant at the 1% level in all the outputs indicating that firms are likely to repurchase during periods of favourable economic conditions. Moreover, during crisis periods, firms are likely to be internally financially constrained internally which will imply less available free cash flows to distribute to shareholders. It further means that debt capital would be expensive such that borrowings become suboptimal. SIZE is also positive and significant at the 1% level in all three models consistent with the free cash flow hypothesis that explain that large firms with limited growth opportunities are more likely to have significant amount of free cash flows to distribute to shareholders to avoid investing in value-destroying projects. Repurchase ratio is positive and significant at 10% and 5% levels for Models 2 and 3 respectively.

These results are consistent with hypotheses about the effects of internal financial constraints on levered share repurchase and I argue that firms do not only utilize internal cash flows to finance share repurchase but also a significant number of firms borrow to finance share repurchase (Chen and Wang, 2012; Lei and Zhang, 2015; Farre-Mensa et al. 2015). These findings extend the evidence by Chen and Wang (2012) that some internally financially constrained firms repurchase their shares. Moreover, this study highlights the effects of investments in the decisions for internally financially unconstrained firms to finance their share repurchase programs through borrowings. Whereas unconstrained firms have internal cash flows to distribute through share repurchase programs, levered share repurchase by these firms are accompanied by significant investments expenditure. Thus, by obtaining external debt finance unconstrained firms build cash buffer sufficient to finance both share repurchase and investments. The rationale for unconstrained firms to borrow is that the external debt financing is more likely to be cheaper for these firms than it will be for constrained firms that might not even have debt ratings.

It further suggests that for unconstrained firms, there is no motivation to borrow to finance share repurchase since that will encourage excess perquisites and hence the agency costs of free cash flows (Jensen, 1986). The increased corporate liquidity is an advantage to obtain cheap debt financing which allows

that there is positive stock price reaction following announcements of share repurchase, levered repurchase of unconstrained firms should be combined with investments. Constrained firms cannot borrow cheap and this constraint limits the ability to combine share repurchase with investments. The stock reactions associated with levered repurchase of constrained firms is that the cash constraint is the only reason these firms will borrow. Share repurchase limits subsequent investment in capital projects since more internal cash flows are disbursed to shareholders (Grullon and Michaely, 2004). In other words, by taking advantage of potential cheap debt financing, unconstrained firms generate sufficient funds to also undertake investments. Thus, levered share repurchase allows the firm to simultaneously distribute cash flows to shareholders and also undertake investments.

Overall, evidence indicates that the decision to conduct levered share repurchase extends to the possibility of firm not foregoing investment projects at the time of share repurchase announcements. Share repurchase generally utilises internal cash flows and reduces available funds for subsequent investments (Denis and Sibilkov, 2010). Therefore, these findings about the effects of internal financial constraints on borrowings and investments are consistent with the effects of repurchase on investments. It also indicates that the effects of investments on the decision to finance repurchase through borrowings is driven by the unconstrained firms. It is only when firms are unconstrained will they be able to obtain sufficient amount of debt financing to undertake both share repurchase transactions and investments. Unconstrained firms have higher debt capacity that allows them to borrow more and increase investments expenditure (Hahn and Lee, 2009).

8.3.2. Share Repurchase and Stock Returns

This section discusses the post repurchase stock returns associated with repurchasing firms. It analyses both the announcement date abnormal returns and the long run abnormal returns. If the market underreacts to initial announcement of repurchase, then the long run returns should be more positive for these repurchasing firms. The analyses further consider the returns based on

whether the repurchase is unlevered or levered for unconstrained firms. The hypotheses about stock returns are re-stated below:

 H_{11} : Cash-financed repurchase is associated with more positive abnormal returns than debt-financed repurchase, especially when the firm is internally financially unconstrained.

 H_{12} : All else equal, unconstrained firms which conduct levered share repurchase should be associated with more positive returns for high investing firms than low investing firms.

As discussed in Section 5.2.3.4 of Chapter 5, the above hypotheses about stock returns are tested using the following cross-sectional regression model.

$$RET_i = \beta_0 + \beta_1 LEVD_i + \beta_2 INV_i + \beta_3 LEVD * INV_i + \sum_{i=1}^n \beta_i X_i + \varepsilon_i \dots (6)$$

From the model above, the dependent variable is the 3-day cumulative abnormal returns for announcement date abnormal returns and 36-month buy and hold abnormal. KZ and MB are the main independent variables with set of control variables defined in Section 5.2.1.3 of Chapter 5.

8.3.2.1. Share Repurchase and Announcement Date Abnormal Returns

[Hypothesis H_{11} and H_{12}]

The extant literature finds significant support for positive announcement date abnormal stock returns associated with share repurchase programs. These stock returns are explained by undervaluation hypothesis (Vermaelen, 1981; Comment and Jarrell, 1991; Stephens and Weisbach, 1998), information signalling hypothesis (Vermaelen, 1981 and 1984; Grullon and Michaely; 2004; Liang et al. 2013) as well as the free cash flow hypothesis (Jensen, 1986; Grullon and Michaely, 2004; Chan et al. 2004; Bozanic, 2010). Thus, each of the theories emphasises that repurchase constitutes positive information to the market about the future prospects of the repurchasing firms. This section explores whether the selection of unconstrained firms to borrow to repurchase shares influence their announcement date returns. Borrowing to repurchase may signal future investment prospects because firms are able to build cash

buffers during the share repurchase announcements without depleting the internal cash flows.

Table 8.9: Univariate Analysis of 3-day Announcement Date Abnormal Returns $(CAR_{(-1,1)})$ following Share Repurchase

Panel A	Panel A: Full sample - Investments and Levered Share Repurchase						
		Unlevered	Levered	Total	Mean diff.		
	N	1	2	3	t-stat		
High investment	1700	0.030	0.014	0.025	0.015		
		$(18.29)^{***}$	$(9.53)^{***}$	$(20.42)^{***}$	$(5.78)^{***}$		
Low investment	4700	0.017	0.015	0.016	0.002		
		(4.88)***	$(4.24)^{***}$	$(6.40)^{***}$	(0.44)		
Total	5874	0.027	0.014	0.023	0.013		
		$(18.58)^{***}$	$(10.12)^{***}$	$(21.06)^{***}$	$(5.48)^{***}$		
Mean diff		0.012	-0.000	0.008			
t-stat		$(3.16)^{***}$	(-0.29)	$(3.07)^{***}$			

Panel B: Unconstrained Firms - Investments and Levered Share Repurchase

		Unlevered	Levered	Total	Mean diff.
	N	1	2	3	t-stat
High investment	835	0.030	0.014	0.026	0.016
_		$(17.33)^{***}$	$(9.57)^{***}$	$(19.36)^{***}$	$(5.36)^{***}$
Low investment	3865	0.014	0.014	0.014	-0.000
		$(3.28)^{***}$	$(4.14)^{***}$	$(4.61)^{***}$	(-0.08)
Total	4700	0.027	0.014	0.024	0.013
		$(17.04)^{***}$	(10.34)***	$(19.41)^{***}$	$(4.91)^{***}$
Mean diff		0.017	0.000	0.012	. ,
t-stat		$(3.84)^{***}$	(0.06)	$(3.79)^{***}$	

Panel C: Constrained Firms - Investments and Levered Share Repurchase

		Unlevered	Levered	Total	Mean diff.
	N	1	2	3	t-stat
High investment	339	0.026	0.014	0.021	0.012
		$(6.03)^{***}$	$(3.44)^{***}$	$(6.93)^{***}$	$(1.97)^{**}$
Low investment	835	0.031	0.017	0.023	0.014
		$(4.52)^{***}$	$(2.25)^{**}$	$(4.60)^{***}$	(1.38)
Total	1174	0.027	0.015	0.022	0.012
		$(7.42)^{***}$	$(4.04)^{***}$	(8.32)***	$(2.35)^{**}$
Mean diff		-0.005	-0.003	-0.002	
t-stat		(-0.54)	(-0.38)	(-0.37)	

Notes: This table shows the univariate analysis of initial abnormal returns associated with share repurchase. The abnormal return measure is the 3-day cumulative abnormal returns, CAR(-1,1).

Panel A of Table 8.9 shows the 3-day abnormal returns associated with share repurchase based on investment and levered repurchase. Levered repurchase firms are associated with more positive stock returns (2.7%) compared to levered repurchase firms (1.4%), both significant at the 1% level. The mean difference is 1.3% and significant at the 1.0% level. Firms investing more show

abnormal returns of that are 1.5% significant at the 0.8% level more than those investing less. For the unconstrained firm subsample, as reported in Panel B, unlevered firms report more positive stock returns than levered repurchase firms. These results are consistent with the hypothesis (H_{II}) that cash-financed share repurchase is associated with more positive stock returns than debt-financed share repurchase. The reason is that the additional debt increases leverage and risks of financial distress and bankruptcy (Fama and French, 1992; Daniel and Titman, 1997).

Table 8.10 presents the cross-sectional regression analysis of the initial abnormal returns following share repurchase announcements. The dependent variable is the 3-day cumulative abnormal stock returns during the repurchase announcement period. LEVD is the dummy variable that equals 1 for levered share repurchase and 0 otherwise whereas LEVD*INV is the interaction term for levered repurchase and investments. INV is continuous variable for investment expenditure. MB is the market to book ratio, AR is the preannouncement stock returns, CRISIS is dummy variable for financial crisis, SIZE is the firm size and REP_RATIO is the repurchase value relative to firm size. In addition to the control variables, we also include one-digit SIC and year dummies in the regressions to control for the potential effects of industry-specific and time-specific differences.

The coefficient of LEVD is significantly negative in Models 1 and 3 indicating that levered repurchase of internally financially unconstrained firms is associated with less positive stock returns. However, the coefficient is positive in Models 4 (significant) and 5 (insignificant). INV is significantly negative in all outputs suggesting that investment is associated with less positive stock returns. This evidence is consistent with market perception about investment in value-destroying projects of internally financially unconstrained firms. The free cash flow hypothesis also describes this phenomenon about the stock reactions of investing firms, especially when they are internally financially unconstrained. The coefficient of LEVD*INV is positive and statistically significant at the 5% level in Model 3 and insignificant in Model 5. This result indicates that high investing of levered share repurchase firms is associated

with more positive stock returns. AR is significantly negative which is in line with the extant evidence. Large firms are associated with lower stock returns as shown by the negative coefficient of the SIZE variable. This result is consistent with the firm size effects and stock returns.

Table 8.10: Cross-sectional Regression Analysis of Announcement Date							
Abnormal Returns							
** . 11	(1)	(2)	(2)	(4)	(5)		

AU	noi mai k ei	lul IIS		
(1)	(2)	(3)	(4)	(5)
	riable is the			
-0.013***		-0.018***	0.006**	0.001
(0.00)		(0.00)	(0.03)	(0.75)
	-0.063***	-0.086***	-0.048**	-0.071**
	(0.00)	(0.01)	(0.04)	(0.02)
		0.088^{**}		0.065
		(0.03)		(0.11)
			0.000	0.000
			(0.84)	(0.83)
			-0.023**	-0.023**
			(0.04)	(0.04)
			0.013	0.013
			(0.14)	(0.15)
			-0.015***	-0.015***
			(0.00)	(0.00)
			0.008	0.008
			(0.36)	(0.38)
0.027^{***}	0.028^{***}	0.032^{***}	0.062^{***}	0.064***
(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Yes	Yes	Yes	Yes	Yes
Yes	Yes	Yes	Yes	Yes
4700	4700	4700	4611	4611
0.005	0.002	0.007	0.029	0.029
0.000	0.004	0.000	0.000	0.000
	(1) pendent Van -0.013*** (0.00) 0.027*** (0.00) Yes Yes 4700 0.005	(1) (2) pendent Variable is the -0.013*** (0.00) -0.063*** (0.00) (0.00) Yes Yes Yes Yes Yes Yes Yes 4700 4700 0.005 0.002	(1) (2) (3) Pendent Variable is the 3-day CAR(-0.013****	October Color Co

Notes: This table shows the cross sectional regression results of initial abnormal returns associated with repurchase of unconstrained firms. The dependent variable is the 3-day cumulative abnormal returns, CAR(-1,1). Variables included in the regressions dummy for levered repurchase (LEVD), investment expenditure (INV), the interaction term between levered repurchase and investment (LEVD*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. ** and **** denote significance at the 5% and 1% level respectively.

These findings support the hypothesis that levered repurchase of internally financially unconstrained firms affects the announcement period abnormal returns. Unconstrained firms should finance share repurchase with internal finance and not borrow from external debt market. The additional leverage and the effects on firm risk mean these firms will be associated with less positive

announcement date abnormal returns. Share repurchase of these firms will be associated with poorer stock performance when they finance the repurchase through borrowings, unless there is a simultaneous expenditure in more capital investments. The evidence provided in the regression output implies the hypotheses about post-repurchase stock returns are accepted.

8.3.2.2. Share Repurchase and Long Run Abnormal Returns

The long-run returns are given as the buy and hold abnormal returns 36-months post share repurchase announcements. Generally, unlevered share repurchase is associated with more positive long-term stock price performance than levered repurchase. This evidence is consistent with the adverse effects of increased leveraged on firm financial distress risk. In essence, the level of investment at the time of levered share repurchase has long-run impact on stock price reactions. For levered repurchase firms, it appears investment motivates firms to borrow in order to undertake investment projects that have long run benefits to the firms. This finding also implies investors initially underreact to levered share repurchase and the level of investment at the repurchase announcement. For example, Grullon and Michaely (2004) report that investors underreact to repurchase announcements because they initially underestimate reduction in the costs of capital. The increased investments of levered repurchase firms also indicate no or limited agency costs of debt that results in underinvestment (Myers, 1977). These findings suggest that investors react positively in the long run to levered share repurchase of unconstrained firms especially when these firms invest.

Table 8.11 reports the univariate analysis of long run abnormal returns associated with share repurchase announcements. Panel A analyses the stock returns for the full sample and examines whether or not the returns differ based on investment expenditure and levered share repurchase. The stock returns are 13.0% and 2.6% for unlevered and levered repurchase firms, respectively (H_{II}). The mean difference between the returns is 10.4% and significant at the 10% level. High investing firms are associated with 18.0% returns more than low investing firms. Panel B provides the stock returns for unconstrained firms based on investment expenditure and levered share repurchase. Levered

repurchase firms that undertake more investment projects are associated with more positive abnormal returns of 25.9% significant at the 10% level than low investing firms. Again, high investing firms are associated with more positive abnormal returns of 21.5% significant at the 5% level, more than low investing firms. These results are consistent with the hypothesis (H_{12}) that unconstrained firms that conduct levered share repurchase should be associated with more positive returns for high investing firms than low investing firms.

Table 8.11: Univariate Analysis of 36-month Buy and Hold Abnormal Returns $(BHAR_{36-month})$ following Share Repurchase

Panel A	Panel A: Full sample - Investments and Levered Share Repurchase							
		Unlevered	Levered	Total	Mean diff.			
	N	1	2	3	t-stat			
High investment	1089	0.253	0.221	0.240	0.032			
		$(2.95)^{***}$	$(2.48)^{**}$	$(3.84)^{***}$	(0.25)			
Low investment	4361	0.103	-0.039	0.069	0.142			
		$(2.93)^{***}$	(-0.53)	(0.51)	$(1.97)^{**}$			
Total	5450	0.130	0.026	0.096	0.104			
		$(3.96)^{***}$	(0.44)	$(3.31)^{***}$	$(1.66)^*$			
Mean diff		0.150	0.259	0.180				
t-stat		$(1.75)^*$	$(1.90)^*$	$(2.47)^{**}$				

Panel B: Unconstrained Firms - Investments and Levered Share Repurchase

		Unlevered	Levered	Total	Mean diff.
	N	1	2	3	t-stat
High investment	775	0.252	0.270	0.258	-0.018
		$(2.51)^{**}$	$(2.63)^{***}$	(3.45)***	(-0.11)
Low investment	3581	0.103	-0.107	0.043	0.210
		$(2.69)^{***}$	(-1.23)	(1.18)	$(2.57)^{**}$
Total	4356	0.128	-0.029	0.082	0.157
		$(3.55)^{***}$	$(-0.40)^{***}$	$(2.46)^{**}$	$(2.16)^{**}$
Mean diff		0.149	0.377	0.215	
t-stat		(1.54)	$(2.12)^{**}$	$(2.48)^{**}$	

Panel C: Constrained Firms - Investments and Levered Share Repurchase

		Unlevered	Levered	Total	Mean diff.
	N	1	2	3	t-stat
High investment	314	0.257	0.146	0.196	-0.111
J		(1.60)	(0.90)	$(1.71)^*$	(-0.48)
Low investment	780	0.103	0.181	0.135	0.078
		(1.15)	(1.44)	$(1.83)^*$	(0.52)
Total	1094	0.139	0.169	0.152	0.030
		$(1.78)^*$	$(1.69)^*$	$(2.45)^{**}$	(0.24)
Mean diff		0.154	-0.035	0.061	. ,
t-stat		(0.83)	(-0.17)	(0.45)	

Notes: This table shows the univariate analysis of initial abnormal returns associated with share repurchase. The abnormal return measure is the 36-month buy and hold abnormal returns, BHAR(2,36).

 Table 8.12: Cross-sectional Regression Analysis of Long Run Buy and Hold Abnormal Returns

 Variables
 (1)
 (2)
 (3)
 (4)
 (5)

 Dependent Variable is the 3-year BHAR(2,36)

 LEVD
 -0.157*
 -0.133
 -0.149
 -0.141

 (0.05)
 (0.25)
 (0.13)
 (0.29)

Береп	ueni variai	ne is the 5	year biiAi	A(2,30)	
LEVD	-0.157*		-0.133	-0.149	-0.141
	(0.05)		(0.25)	(0.13)	(0.29)
INV	, ,	1.332***	1.652***	1.664***	1.706***
		(0.01)	(0.01)	(0.00)	(0.01)
LEVD*INV		,	-0.590	,	-0.113
			(0.57)		(0.91)
MB			,	-0.045**	-0.045**
				(0.03)	(0.03)
Prior AR				-0.010	-0.010
				(0.95)	(0.96)
CRISIS				-0.386	-0.388
000				(0.26)	(0.26)
SIZE				0.054	0.054
				(0.23)	(0.23)
PMV				0.060	0.060
1 1/1 /				(0.79)	(0.78)
Constant	0.128***	-0.001	0.030	0.282	0.281
Constant	(0.00)	(0.98)	(0.54)	(0.37)	(0.37)
Year Dummies	Yes	Yes	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes	Yes	Yes
N	4356	4356	4356	4280	4280
adj. R^2	0.001	0.001	0.002	0.008	0.008
p	0.052	0.007	0.011	0.000	0.000
NI 4 771 1 1 1	41			C 1	7 7

Notes: This table shows the cross sectional regression results of long run abnormal returns associated with repurchase of unconstrained firms. The dependent variable is the 3-year buy and hold abnormal returns. Variables included in the regressions dummy for levered repurchase (LEVD), investment expenditure (INV), the interaction term between levered repurchase and investment (LEVD*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.*, ** and *** denote significance at the 10%, 5% and 1% level respectively.

Table 8.12 above reports the cross-sectional regression analyses of the long run stock price reactions to share repurchase announcements. The dependent variable is the BHAR over 36-month period post-repurchase announcements and independent/control variables are the same as in earlier regressions. ⁴⁰ Also year and industry dummies are controlled for to account for time and industry specific effects. Model 1 tests the effects on stock returns of levered repurchase. LEVD is marginally significant at the 10% level indicating that

⁴⁰ Similar cross-sectional regression analyses are provided for buy and hold returns over 12-

month and 24-month.

233

levered repurchase is associated with less positive stock returns. INV is significantly positive at the 1% level showing that investment is associated with positive long run stock returns. However, LEVD is not significant in Models 3, 4, and 5. Importantly, LEVD*INV is not significant in Model 3 and 5.

These findings are consistent with the adverse effects of leverage on firm value and the associated stock price reactions. Studies such as Ikenberry et al. (1995, 2000) and Peyer and Vermaelen (2009) and Chen and Wang (2012) all find strong support for the long-term abnormal returns associated with share repurchase. For example, Peyer and Vermaelen (2009) assert that the long run abnormal returns are a correction of prior overreaction to bad news about the firms at the time of repurchase. However, as hypothesised, the source of financing the repurchase of financially unconstrained could affect the magnitude of the long run stock returns.

8.4. Robustness and Sensitivity Analysis

This section presents the results of a number of robustness and sensitivity analysis. These additional results provide further support for the main results and also ensure the results are robust to alternative estimation techniques. The section is divided into subsection 8.4.1 which provides results for alternative cut-off points for KZ, 8.4.2 which provides results for WW index measure of internal financial constraints, subsection 8.4.3 gives results for the effects of bond ratings on share repurchase; subsection 8.4.4 shows results for the effects of financial distress on share repurchase, Subsection 8.4.5 estimates tobit regression of the main model, and Subsection 8.4.6 provides results for random and fixed effects logit regression estimation techniques.

8.4.1. Unconstrained firms defined using alternative cut-off point of KZ

This section provides results for alternative cut-off points for classifying share repurchase firms as constrained or unconstrained. Results from previous sections define share repurchase firms within the bottom 80% as unconstrained and the top 20% as constrained. Here firms are classified as internally

financially unconstrained when they fall within the bottom 90%, 75%, and 50% of KZ. In other words, internally financially constrained firms are those in the top 10%, 25% and 50% of KZ. Table 8.13 reports the results of the logit regressions for these alternative KZ cut-off points. In all three models, the coefficient of INV is significantly positive indicating that investment increases the probability that unconstrained firms will conduct levered share repurchase. MB is significantly negative in line with the undervaluation hypothesis of share repurchase. The coefficient of CRISIS and SIZE are significantly negative and positive, respectively. Overall, these results provide further evidence that the classification of internal financial constraints produce consistent evidence about the probability of levered share repurchase programs.

Table 8.13: Logi	Table 8.13: Logit Regression Estimates of Main Results					
(a	lternative cu	t-off points)				
	90%	75%	50%			
Variables	(1)	(2)	(3)			
INV	0.417***	0.371***	0.509***			
	(0.00)	(0.00)	(0.00)			
MB	-0.132***	-0.105***	-0.096***			
	(0.00)	(0.00)	(0.00)			
Prior AR	-0.598**	-0.395	-0.508			
	(0.02)	(0.17)	(0.21)			
CRISIS	-1.336***	-1.486 ^{***}	-2.075 ^{***}			
	(0.00)	(0.00)	(0.00)			
SIZE	2.073***	2.142***	2.289***			
	(0.00)	(0.00)	(0.00)			
PMV	0.954**	1.057**	0.550			
	(0.02)	(0.03)	(0.42)			
Constant	-6.465 ^{***}	-6.774***	-6.844***			
	(0.00)	(0.00)	(0.00)			
Year Dummies	Yes	Yes	Yes			
Industry Dummies	Yes	Yes	Yes			
N	5193	4323	2882			
pseudo R^2	0.328	0.356	0.385			
chi-squared	1142.077	972.443	662.240			
Notes: This table sho	0.000	0.000	0.000			

Notes: This table shows the results of the logit regression for share repurchase using 95% (model 1), 75% (model 2) and 50% (model 3) cut-off points of KZ to classify firms as unconstrained. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Variables included in the regressions are investment expenditure (INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), ratio of repurchase proceeds to market value (PMV) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. ** and *** denote significance at the 5% and 1% level respectively.

8.4.2. Levered Share Repurchase and WW index Internal Financial Constraints

As a robustness check, this section uses Whited and Wu (2006) index measure of internal financial constraints to analyse the effects of internal financial constraints on levered share repurchase. The logit regression results are shown in Table 8.14. Consistent with the results given in Table 8.2 using the KZ index measure of internal financial constraints, the coefficient of WW is positive and significant at 1% level in Model 1 and at 5% level in Model 2 for full sample. However, as shown in Model 2, the coefficient of the interaction term WW*INV is not significant. Investment INV is positive and significant in all the Models indicating that investment is drives levered share repurchase. In other words, firms that conduct debt-financed share repurchase are more likely to invest more. Overall, the findings are similar when KZ index and WW index are used as measure of internal financial constraints.

Table 8.14: Logit Regression Estimates of Levered			
Repurchase and Internal Financial Constraints (WW index)			
Variables	(1)	(2)	(3)
WW	0.491***	0.403**	
	(0.00)	(0.01)	
INV	0.586***	0.602***	0.558***
	(0.00)	(0.00)	(0.00)
WW^*INV		-2.604	
		(0.20)	
MB	-0.120***	- 0.119***	-0.124***
	(0.00)	(0.00)	(0.00)
Prior AR	-0.726***	-0.730***	-0.817 ^{***}
	(0.00)	(0.00)	(0.00)
CRISIS	-0.771 ^{**}	-0.775**	-0.708**
	(0.02)	(0.01)	(0.03)
SIZE	1.813***	1.803****	1.804***
	(0.00)	(0.00)	(0.00)
PMV	0.046^{**}	0.046^{**}	0.770^{***}
	(0.01)	(0.02)	(0.01)
Constant	-5.737***	-5.710	-5.749 ^{***}
	(0.00)	(0.00)	(0.00)
Year Dummies	Yes	Yes	Yes
Industry Dummies	Yes	Yes	Yes
N	5678	5678	4547
pseudo R^2	0.297	0.298	0.240
chi-squared	1224.464	1229.421	909.225
p	0.000	0.000	0.000

Notes: This table shows the coefficient estimates of the logit regression for share repurchase. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Models 1 and 2 provide results for the full sample whereas model 3 uses only the subsample of unconstrained firms. Variables included in the regressions include WW-index measure of internal financial constraints (KZ), investment expenditure (INV), the interaction term between internal financial constraints and investment (WW*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), the ratio of repurchase proceeds to market value (PMV), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are in parenthesis.*, ** and * denote significance at the 10%, 5% and 1% level respectively.

8.4.3. Levered Share Repurchase and Bond Ratings

Prior studies assert that share repurchase have significant post-repurchase effects of bond ratings. Specifically, the increased leverage from repurchase results in credit ratings downgrades (Minnick and Zhao, 2007; Chen and Wang, 2012). This implies that levered share repurchase should be associated with significant credit rating downgrades. It also follows that investment grade rating firms will be more likely to borrow more to finance share repurchase programs. Firms with bond ratings and investment grade ratings would be internally financially unconstrained and therefore will be more likely to conduct levered share repurchase to allow them to also invest. Firms are classed as internally financially constrained when they have positive debt but without S&P rating, whereas internally financially unconstrained firms have positive debt and an S&P bond rating. 41 Bond ratings represent the Standard and Poor's ratings assigned to bonds one month prior to the share repurchase announcement. In addition to this definition, firms with ratings are further classified into investment and non-investment grade ratings where investment grade rating firms are unconstrained and non-investment grade firms are constrained. Rated and investment grade rated firms are able to borrow large amounts and at low interests than non-rated and non-investment grade rated firms.

Table 8.15 presents the results for the probability of levered repurchase using bond ratings as internal financial constraints measure. Model 1 provides the

⁴¹ Whited (1992), and Hahn and Lee (2009) use the presence of bond ratings as a proxy for access to debt financing and financial constraints.

results for rated firms with INV significantly positive. Also, in Model 2 for investment grade firms INV is significantly positive. MB and AR are both negative and marginally significantly in Models 1 and 2 whereas SIZE is significantly positive in both models. These results are consistent with the hypothesis that investment increases the probability of levered share repurchase for internally financially unconstrained firms. This borrowing ability increases the probability that rated firms are likely to conduct levered share repurchase. This result suggests that rated firms that are unconstrained are likely to borrow to finance repurchase when they also invest more. Results are robust when investment grade and non-investment grade ratings definition is used. To the extent that financially constrained or non-rated firms will suffer downgrades and increase risk of financial distress, financially unconstrained (rated firms) are expected to borrow to finance share repurchase.

Table 8.15: Logit Regression Estimates of						
	Levered Repurchase and Bond Ratings					
Variables	(1) (2)					
INV	0.615***	0.514***				
	(0.00)	(0.00)				
MB	-0.045*	-0.047				
	(0.10)	(0.15)				
Prior AR	-0.691 [*]	-0.425				
	(0.07)	(0.41)				
CRISIS	-0.685	-1.037*				
	(0.16)	(0.09)				
SIZE	1.496**** 1.607****					
	(0.00)	(0.00)				
PMV	0.015 -0.155					
	(0.98)	(0.83)				
Constant	-4.506 ^{***} -4.498 ^{***}					
	(0.00)	(0.00)				
Year Dummies	Yes	Yes				
Industry Dummies	Yes	Yes				
N	1978	1398				
pseudo R^2	0.224	0.249				
chi-squared	415.344	312.294				
p No. This state is	0.000	0.000				

Notes: This table shows the results of the logit regression for share repurchase using bond ratings to classify firms as unconstrained. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Variables included in the regressions are investment expenditure (INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), ratio of repurchase proceeds to market

value (PMV) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.*, and **** denote significance at the 10% and 1% level respectively.

8.4.4. Levered Share Repurchase and Financial Distress

Financially distressed firms are highly leveraged with high costs of bankruptcy. These firms have limited debt capacity and as such more expensive to obtain external debt financing. The combined effects of additional leverage and lower cash levels as a result of share repurchase increase firm risk of financial distress (Stephens and Weisbach, 1998; Hovakimian, 2004; Chen and Wang, 2012). All else equal, levered repurchase should increase financial distress risk. Stated differently, prior financial distress risk should limit debt financed share repurchase. Altman's Z score is used to proxy for financial distress as discussed in Section 5.2.3.2 of Chapter 5.

The results in Table 8.16 provide tests of the effects of prior financial distress risk on the probability of levered share repurchase. Models 1 and 2 give the results for the full sample whereas Model 3 shows the results for less financially distressed firm subsample. The coefficient of Z-score is significantly negative for both Models 1 and 2. This evidence is consistent with the hypothesis that financially constrained firms are less likely to conduct debt financed share repurchase due to the significant increase in distress risk associated with debt increasing share repurchase program. INV is positive and significant at the 1% level indicating that investment is associated with levered repurchase. The coefficient of Z*INV is significantly negative which implies that less financially distressed firms will conduct levered repurchase when they invest more. This result also suggests the debt capacity associated with less distressed firms allows them to borrow more and provide cash buffer for investment expenditure at the time of distributing cash to shareholders. MB, AR, CRISIS, and SIZE all have the expected signs about the effects on share repurchase. INV in Model 3 is significantly positive indicating that for less financially distressed firms' investment determines the probability of levered share repurchase.

Overall, the above results provide further evidence about the levered share repurchase of financially unconstrained firms. It highlights the role of investment in conducting debt financed share repurchase. Thus, this evidence implies that the hypothesis that less financially distressed firms are more likely to conduct levered repurchase when they invest more is accepted. Less financially constrained firms have debt capacity and therefore are more likely to borrow to finance their share repurchase, especially when they have more investment expenditure. Hahn and Lee (2009) assert that higher debt capacity, a feature of non-distressed firms-allow firms to borrow and undertake more investments. Also as Chen and Wang (2012) point out, share repurchase depletes corporate liquidity and results in high leverage such that the firm is exposed to greater financial risk. Stated differently, unconstrained firms are less likely to suffer greater financial distress since the cash levels do not fall substantially following share repurchase.

Repurchase and Financial Distress Variables (1) (2) (3) Z score -0.211*** -0.244*** (0.00) (0.00) INV 0.615*** 0.422*** 0.446*** 0.446*** (0.00) (0.00) (0.00) Z*INV -0.361** (0.00) MB 0.004 -0.001 -0.125*** (0.84) (0.96) (0.00) Prior AR -0.807*** -0.784*** -0.798*** (0.00) (0.00) (0.00) CRISIS -0.967*** -0.945*** -0.769** (0.00) (0.00) (0.03) SIZE 1.861*** 1.869*** 2.030*** (0.00) (0.00) (0.00) PMV 0.031 0.031* 0.045** (0.10) (0.10) (0.02) Constant -5.311*** -5.250*** -6.536** (0.00) (0.00) (0.00) Year Dummies Year Year Year Year	Table 8.16: Logit Regression Estimates of Levered				
Z score $ \begin{array}{ccccccccccccccccccccccccccccccccccc$	Repurcha	se and Finar	icial Distres	S	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Variables	(1)	(2)	(3)	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Z score	-0.211***	-0.244***		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.00)	(0.00)		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	INV	0.615^{***}	0.422^{***}	0.446^{***}	
$\begin{array}{c} \text{MB} & \begin{array}{c} (0.00) \\ 0.004 \\ (0.84) \\ (0.96) \\ (0.00) \\ \end{array} & \begin{array}{c} (0.00) \\ (0.00) \\ \end{array} \\ \begin{array}{c} (0.00) \\ (0.00) \\ \end{array} & \begin{array}{c} (0.00) \\ \end{array} $		(0.00)		(0.00)	
MB	Z^*INV		-0.361***		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$			(0.00)		
Prior AR -0.807*** -0.784*** -0.798*** (0.00)	MB	0.004	-0.001	-0.125***	
Prior AR -0.80/ -0.784 -0.798 (0.00) (0.00) (0.00) (0.00) -0.945*** -0.769** (0.00) (0.00) (0.03) SIZE 1.861*** 1.869*** 2.030*** (0.00) (0.00) (0.00) PMV 0.031 0.031 0.045** (0.10) 0.10) (0.10) Constant -5.311*** -5.250*** -6.536*** (0.00) (0.00)		(0.84)	(0.96)	(0.00)	
CRISIS -0.967*** -0.945*** -0.769** (0.00) (0.00) (0.03) SIZE 1.861*** 1.869*** 2.030*** (0.00) (0.00) (0.00) PMV 0.031 0.031* 0.045** (0.10) (0.10) (0.10) (0.02) Constant -5.311*** -5.250** -6.536*** (0.00) (0.00)	Prior AR	-0.807***	-0.784***	-0.798***	
SIZE		(0.00)	(0.00)	(0.00)	
SIZE	CRISIS	-0.967***	-0.945***	-0.769**	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.00)	(0.00)	(0.03)	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	SIZE	1.861***	1.869***	2.030^{***}	
Constant		(0.00)			
Constant -5.311*** -5.250*** -6.536*** (0.00) (0.00)	PMV	0.031	0.031^*	0.045^{**}	
Constant -5.311 -5.250 -6.536 (0.00) (0.00)		(0.10)	(0.10)	(0.02)	
	Constant	-5.311***	-5.250 ^{***}	-6.536 ^{***}	
Year Dummies Yes Yes Yes		(0.00)	(0.00)	(0.00)	
100 100	Year Dummies	Yes	Yes	Yes	
Industry Dummies Yes Yes Yes	Industry Dummies	Yes	Yes	Yes	
N 5771 5771 5088	N	5771	5771	5088	
pseudo R^2 0.334 0.335 0.311	pseudo R^2	0.334	0.335	0.311	
chi-squared 1245.789 1245.597 1134.734	chi-squared	1245.789	1245.597	1134.734	
p 0.000 0.000 0.000	p	0.000	0.000	0.000	

Notes: This table shows the results of the logit regression for share repurchase using financial distress (Altman's Z score) to classify

firms as unconstrained. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Variables included in the regressions are investment expenditure (INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), ratio of repurchase proceeds to market value (PMV) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis. *, ** and *** denote significance at the 10%, 5% and 1% level respectively.

8.4.5. Tobit Regression Estimation of Main Results

Table 8.17 provides the tobit regression results for share repurchase by internally financially unconstrained firms. The dependent variable is the dollar value of share repurchases divided by the market value of equity, indicating the effects of investment and borrowings on the value of shares firms are likely to repurchase in the market. Model 1 gives the results of the effects of levered (LEVD) and investment (INV) on the value of share repurchase whereas Model 2 examines the simultaneously effects of levered and investment (LEVD*INV) on the value of share repurchase. As can be seen from the table, the coefficient of LEVD is positive and statistically significant at the 5% level. This evidence is consistent with the hypothesis that internally financially constrained firms are less likely to conduct debt financed share repurchase due to the significant increase in distress risk associated with debt increasing share repurchase program.

The coefficient of LEVD*INV is positive and significant at the 5% level indicating that borrowings and increase the value of share repurchase even for high investing firms. This result suggests the increased cash flows from debt financing allow firms to repurchase greater amount of shares even when they also engage in investment projects. As argued earlier, the borrowings provide cash buffer for investment expenditure at the time of distributing cash to shareholders. The negative coefficients of MB and prior AR are consistent with the undervaluation and information signaling theories by which firms repurchase their shares when their shares are cheap. Overall, the above results provide further evidence about the levered share repurchase of internally financially unconstrained firms. It underlines the effects of debt financing on share repurchase decisions of firms. Thus, this evidence implies the hypothesis about debt financed share repurchase is accepted.

Table 8.17: Tobit Regression Analysis of Levered Share Repurchase

Variables	(1)	(2)			
LEVD	0.016**	0.004			
	(0.01)	(0.65)			
INV	-0.008	-0.071			
	(0.83)	(0.14)			
LEVD*INV		0.180^{**}			
		(0.03)			
MB	-0.006***	-0.006***			
	(0.00)	(0.00)			
Prior AR	-0.025*	-0.025*			
	(0.08)	(0.07)			
CRISIS	0.026***	0.026***			
	(0.00)	(0.00)			
SIZE	-0.006*	-0.005			
	(0.10)	(0.11)			
Constant	0.107***	0.110^{***}			
	(0.00)	(0.00)			
Year Dummies	Yes	Yes			
Industry Dummies	Yes	Yes			
N	4611	4611			
pseudo R^2	-0.017	-0.018			
chi-squared	64.026	68.886			
р	0.000	0.000			
Notes: This table shows the Tobit regression results for share					

Notes: This table shows the Tobit regression results for share repurchase of unconstrained firms. The dependent variable is the ratio of repurchase proceeds to market value of equity. Variables included in the regressions include dummy which is 1 for levered repurchase and 0 otherwise (LEVD), investment expenditure (INV), the interaction term between levered repurchase and investment (LEVD*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are in parenthesis. *, ** and *** denote significance at the 10%, 5% and 1% level respectively.

8.4.6. Random and Fixed Effects Logit Regression Estimation of Main Results

The results presented in Table 8.18 are the random and fixed effects estimation of the main results. From the table, Models 1 and 2 are for random effects whereas fixed effects estimation results are given in Models 3 and 4. For a start the fixed effects eliminate time invariant variables (CRISIS variable has no output) and no constants are given. The chi-square and the probability statistics indicate the model is significant and that the coefficients are jointly significantly different from zero. Both the random effects and fixed effects results confirm the previous findings reported in the chapter about the share repurchase behaviour of internally financially unconstrained firms. Here again, the coefficients of KZ and INV are significantly positive for all the Models. Also, the coefficient of the interaction term KZ*INV is significantly negative in Models 2 and 4 for random and fixed effects, respectively. These results provide further evidence that internally financially unconstrained firms are more likely to conduct levered share repurchase when they also invest more. The parameter coefficient of MB, AR, CRISIS, and SIZE all give the expected signs. For example, MB and AR show that undervalued shares are more likely to be repurchased when firms also borrow. Overall, these results after using alternative estimation techniques still support the hypotheses about the levered share repurchase of internally financially unconstrained driven by investment expenditure.

The results of the hausman test are shown in Appendix 8. The starting point for the hausman test is that the two models do not produce significantly different coefficients. Thus, the null hypothesis is that the coefficient are not different between random and fixed effects in which case both estimations are appropriate otherwise the fixed effects is chosen because it gives consistent coefficients. As can be seen from the hausman test output in Appendix 8, the chi-square χ^2 is 11.77 and probability is 0.1083 indicating that the null hypothesis cannot be rejected in favour of the alternative hypothesis. Therefore, the difference in the coefficients between random and fixed effects is not systematic. This implies the random effects output is efficient and fixed effects

give consistent results for the coefficient. The coefficients of the two estimation techniques are comparable and not significantly different.

Table 8.18: Random and Fixed Effects Logit Regression of Levered
Repurchase and Internal Financial Constraints

Repurchase and Internal Financial Constraints						
	Random Ef	fects Model	Fixed Effects Model			
Variables	(1)	(2)	(3)	(4)		
Depende	ent Variable is	Levered (1) vs	Unlevered (0)			
KZ	1.162***	0.873****	1.156***	0.868***		
	(0.00)	(0.00)	(0.00)	(0.00)		
INV	(0.00) 0.466***	(0.00) 0.424***	$(0.00) \\ 0.464^{***}$	(0.00) 0.422^{***}		
	(0.00)	(0.00)	(0.00)	(0.00)		
KZ*INV	, ,	-2.673***	, ,	-2.660 ^{***}		
		(0.00)		(0.00)		
MB	-0.101***	(0.00) -0.085***	-0.101***	(0.00) -0.084***		
	(0.00)		(0.00)			
Prior AR	-0.651***	(0.00) -0.633***	-0.647***	(0.00) -0.630***		
	(0.01)	(0.01)	(0.01)	(0.01)		
CRISIS	-1.018 ^{***}	-1.082***	,	, ,		
		(0.00)				
SIZE	(0.00) 1.995***	(0.00) 2.021***	1.985***	2.010***		
	(0.00)	(0.00)	(0.00)	(0.00)		
PMV	0.030	0.033	0.030	0.032		
	(0.50)	(0.46)	(0.50)	(0.47)		
Constant	-6.496***	-6.470***	,	, ,		
	(0.00)	(0.00)				
Year Dummies	Yes	Yes	Yes	Yes		
Industry Dummies	Yes	Yes	Yes	Yes		
\overline{N}	5771	5771	5771	5771		
pseudo R^2			0.302	0.307		
chi-squared	1272.563	1284.977	2071.254	2103.090		
p No. This table is	0.000	0.000	0.000	0.000		
			CC . 1			

Notes: This table shows the results of the random and fixed effects logit regression for share repurchase. The dependent variable is 1 for levered repurchase and 0 for unlevered repurchase. Variables included in the regressions are KZ index measure of internal financial constraints (KZ), investment expenditure (INV), interaction term between internal financial constraints and investment (KZ*INV), market-to-book ratio (MB), prior stock returns (Prior AR), dummy for financial crisis (CRISIS), logarithm of book value of total assets (SIZE), ratio of repurchase proceeds to market value (PMV) and year and industry dummies. Standard errors are heteroscedasticity consistent. P-values are shown in parenthesis.*** denote significance at the 1% level.

Summarily, the results of the random and fixed effects as well as the main analysis using the logit regression provide strong support for the hypothesis about the levered share repurchase of internally financially unconstrained firms. The findings indicate that financially constrained firms are more likely to

conduct levered repurchase. Also, firms with significant investment expenditure at the time of share repurchase are also more likely to borrow to finance their share repurchase programs. And more importantly, unconstrained firms that do not require external financing to finance repurchase programs only borrow when they also invest more. The results seem to assert that the investment expenditure drives unconstrained firms to seek external financing as cash buffer during the repurchase period. Thus, the additional financing from borrowings enable to firm to invest without forgoing investment opportunities. This evidence reinforces the post-repurchase reduced cash flow and limited liquidity hypothesis.

8.5. Conclusion

This chapter presented empirical results to explore the share repurchase financing of firms. It specifically, investigated the levered share repurchase behaviour of internally financially unconstrained firms. The main hypothesis tested in this chapter was that internally financially unconstrained firms are likely to conduct levered share repurchase when they also invest more. This hypothesis emphasised that without investment expenditure, internally financially unconstrained firms will be less likely to borrow at the time of share repurchase because these firms would have sufficient internal cash flow to support their share repurchase programs. The implication of this hypothesis was that internally financially constrained firms would need external financing in order to conduct share repurchase. The results used the Kaplan and Zingales (1997) index to proxy for internal financial constraints. In addition, bond ratings and Altman's Z score measure of financial distress are additional proxies that were used to measure financial constraints. Change in leverage was used to indicate whether a firm has financed share repurchase using external borrowings.

Consistent with prior studies, the results of this chapter find strong support for the levered share repurchase of unconstrained firms when investment expenditure is high. Thus, unconstrained firms borrow at the time of share repurchase when they also invest more. This evidence indicates that the additional external financing provides cash buffer to preserve the postrepurchase cash and liquidity levels of the firm to support investment expenditure. Also, the chapter finds that unlevered repurchase firms outperform levered repurchase firms. However, the levered repurchase firms' investment seems to increase returns at least in the short run. Moreover, the chapter addresses issues with estimation assumptions and also presented results using alternative proxies for internal financial constraints. Finally, the chapter presented robust results for alternative estimation techniques.

The findings in this chapter of the study are consistent with Minnick and Zhao (2007), De Jong et al. (2011), and Chen and Wang (2012). However, the results in this chapter differ markedly from prior studies on several fronts. For example, Minnick and Zhao (2007) focus on the wealth effect of levered share repurchases whereas De Jong et al. (2011) assert that convertible issuers repurchase their stock in order to facilitate short selling. Both studies examine the market as whole without differentiating their results for internally financially constrained and unconstrained firms. Also, Chen and Wang (2012) instead focus on share repurchases of internally financially constrained firms and report that managerial hubris drives constrained firms to repurchase their shares. However, the results presented in this chapter demonstrate the share repurchase financing of internally financially unconstrained firms. This evidence has not previously been explored in the literature and therefore the results presented in this chapter contribute to the extant literature on share repurchase.

Chapter 9

CONCLUSION

9.1. Introduction

This chapter provides the conclusions in this thesis. It presents the main findings and the research implications for corporate financing decisions and policy changes. It also discusses the limitations of the current study, suggests areas of improvement including suggestions for future research. The remainder of the chapter is organised into four sections. Section 9.2 presents the main findings of the three empirical chapters. Section 9.3 discusses the research implications for corporate financing decisions and policy makers. Section 9.4 outlines the limitations of the study. Finally, Section 9.5 proposes suggestions for future research.

9.2. Main Findings of the Thesis

9.2.1. Share Issuance and Financial Constraints

The first empirical chapter (Chapter 6) explored the simultaneous effects of internal and external financial constraints on share issuance decisions of a sample of UK listed firms. Following the predictions of market timing-where firms are more likely to issue overvalued equity- this thesis tested the extent to which financially constrained firms are restricted in their attempts to issue overvalued equity. In this regard, the research hypothesised that internally financially constrained firms are less likely to issue overvalued equity compared to counterpart financially unconstrained firms. Financial constraints limit firms' ability to issue equity due to frictions associated with the external equity financing market. This effect arises because equity financing is associated with significant transaction costs in terms of commissions and fees as well as the adverse selection costs occasioned by such transactions. In essence, internally financially unconstrained firms should be able to afford such costs and will be likely to issue overvalued equity. However, given that internally financially constrained firms would more likely require external equity financing to finance investment projects, these firms would be equally

motivated to issue overvalued equity. Thus, the hypothesis (H_I) tested in Chapter 6 (defined in Subsection 2.5 in Chapter 2) states that financially constrained firms are more likely to issue overvalued equity. This study contributes to the market timing and information asymmetry theories as explained in Section 2.2 of Chapter 2.

In addition to internal financial constraints, the study also tested the share issuance behaviour of firms during periods of external financial constraints. As defined in the thesis, external financial constraints relate to the conditions of the external equity market including the general economic conditions and market liquidity. Consistent with hypothesis (H_2) , I find that firms that are internally financially unconstrained issue overvalued equity even when external financial constraints are high. Conversely, internally financially constrained firms are less likely to issue overvalued equity during periods of financial crisis or when there is low market liquidity. During periods of favourable market conditions, such as high market liquidity at times of no financial crisis, costs of equity issues are relatively low. This is because investor confidence in the market increases resulting in increases in the supply of financing in the market. Therefore, firms that intend to issue overvalued equity would be motivated during periods of favourable market conditions, even when these firms are internally financially constrained. The above effects of financial constraints on market timing of equity issues translate into the stock price reactions subsequent to the equity issues.

A sample of UK listed firms is used to test the developed hypotheses about the effects of financial constraints on share issuance decisions. The study finds that both internal and external financial constraints are important factors that influence the market timing of equity issues. It also shows that post-issue stock price reactions are related to the level of financial constraints and mispricing. The findings of the study support the developed hypotheses indicating that internal financial constraints determine whether or not firm issues overvalued equity. It further shows that during periods of economic downturn, only internally financially unconstrained firms are likely to issue overvalued equity. The results are robust to alternative definitions of financial constraints and

estimation techniques. The study finds strong support for the simultaneous effects of internal and external financial constraints on market timing of equity issues.

Overall, the findings in this study are consistent with prior studies about market timing of equity issues. Studies such as Stein (1992), Baker and Wurgler (2002) and Dong et al. (2012), all find evidence of equity market timing by which firms issue overvalued equity. For example, Baker and Wurgler (2002) stress that long-term changes in capital structure reflect the effects of persistent market timing of equity issues. Also, Korajczyk and Levy (2003), Dong et al. (2012) and (McLean and Zhao, 2014) assert that internal financial constraints restricts external equity financing. Favourable market conditions produce investor confidence and boost stock prices. Thus, all else equal, favourable market conditions motivate the issuance of overvalued equity even when the firm is financially constrained. The results, therefore, highlight the importance of external financing constraints on the equity issuance decisions of financially constrained firms. By this, the study synthesises the effects of both internal and external financial constraints and demonstrates how these two conditions determine the issuance of overvalued equity.

9.2.2. Share Issuance Methods and Operating Performance

The second empirical chapter (Chapter 7) focuses on the impact of operating performance on the choice of equity issuance methods among a sample of UK listed firms. Expectations of future operating performance have significant effects on the choice of equity issuance method. This is because information about future operating performance is unlikely to be available to outside investors who might have to incur substantial costs to obtain information about the firm. The choice of equity issuance method depends on both firm and issue characteristics. It is therefore argued that contemporaneous long-term operating performance might determine whether a firm chooses rights issues, open offers, and private placement. Firms would be more likely to choose rights offerings over open offers and private placement when operating performance is anticipated to be good, consistent with the objective of shareholder wealth maximisation. Stated differently, for anticipated poor operating performance,

issuance which involves outside investors will be preferred in order to exploit less informed investors, all other things being equal. Thus, it is hypothesised that rights issuing firms anticipate high operating performance than open offers and private placement (H_6). Rights issuing firms are likely to be firms of better quality and are less likely to issue equity in order to exploit equity mispricing. In other words, open offers and private placement are likely to be undertaken by firms seeking to time the issue of overvalued equity. It further implies that rights issuing firms are more likely to outperform open offers and private placement subsequent to the equity issues (H_7).

The study finds that operating performance is an important determinant of the choice of equity issuance method. In effect, the findings provide strong support for the developed hypothesis about the effects of long term operating performance on the choice of equity issuance method. This study asserts that operating performance not only determines the choice of issuance methods but it also explains the post-issue long run stock returns associated with each equity issuance method. Operating performance explains better the firm quality theory over the long term. Subsequent to equity issues, firms conducting rights issues report better operating performance than open offers and private placement. Thus, firms associated with better operating performance seem more likely to conduct rights issues rather than open offers and private placements. Consistent with Capstaff and Fletcher (2011), the study further finds that rights issuing firms outperform open offers and private placement in the long run.

Firms that make rights issues are associated with anticipated high operating performance than open offers and private placement. Through rights issues, firms return part of the potential benefit of better operating performance to existing shareholders. When the operating performance is eventually realised, stock prices rise accordingly causing shareholders value to increase in turn. For open offers and private placement, firms are motivated by the attempt to profit from temporary mispricing. These firms do not anticipate better operating performance but the pre-issue stock price run-up provides benefits to allow them to make the issue. The effects of long-term operating performance on the

choice of equity issuance method also explain the long run stock price performance. Rights issues outperform both open offers and private placement. Stated another way, open offers and private placement are associated with more negative stock returns than rights issues.

These findings are consistent Capstaff and Fletcher (2011) about the relation between post-issue stock returns and operating performance associated with equity issuing firms. Both the operating performance and stock performance indicate that high quality firms undertake rights issues. However, unlike Capstaff and Fletcher (2011), the current study focuses on how operating performance determines the probability of equity issues through rights issues, open offers, and private placement as well as explaining the long run stock price performance. For instance, Loughran and Ritter (1997) find that long run operating performance explains stock returns subsequent to US equity issues that are dominated by firm commitments. Holding the measures of firm and issue quality constant, the decision to utilise a particular issuance method is influenced by operating performance. Thus, the current study offers insights into the choice of UK issuance methods explained by the differential long run operating performance.

9.2.3. Share Repurchase and Internal Financial Constraints

The third empirical chapter (Chapter 8) focuses on the effects of internal financial constraints on share repurchase financing among a sample of US listed firms. It is argued that firms borrow to finance share repurchase when they are financially constrained (H_8). However, internally financially unconstrained conduct levered repurchase when they also undertake significant investment projects (H_9). While internally financially constrained would obtain debt financing to provide sufficient cash flow to finance share repurchase programs, it is unlikely that internally financially unconstrained that have excess cash flows would seek debt financing when they repurchase their shares. The hypothesis is consistent with the prediction that additional cash flows from borrowings allow the firm to build cash buffers in order to finance investment projects.

As argued in prior studies, share repurchase transactions reduce subsequent cash flows for investment (Grullon and Michaely, 2004; Denis and Sibilkov, 2010). Therefore, by borrowing the firm would be in better financial position to both disburse cash to shareholders without foregoing investment projects. In essence, the study explored the levered share repurchase behaviour of financially unconstrained firms and tests the factors that drive cash-rich firms to finance their share repurchases through borrowings. Consistent with hypotheses (H_8) and (H_9) , I find that internally financially unconstrained firms borrow when they conduct share repurchase and undertake significant capital expenditure. Firms do not only utilise internal cash flows to finance share repurchase but also a significant number of firms borrow to finance share repurchase (Chen and Wang, 2012; Lei and Zhang, 2015; Farre-Mensa et al., 2015). Whereas Chen and Wang (2012) assert that internally financially constrained firms conduct share repurchases due to managerial overconfidence and hubris, this study highlights that internally financially unconstrained firms also borrow during share repurchase programs. Similar to Farre-Mensa et al. (2015), firms undertaking share repurchases obtain external debt-financing. In particular, these unconstrained firms borrow during share repurchase announcements and potentially utilize the additional cash buffers to finance investment expenditure. The study further argued that the financing of share repurchase determines the post-repurchase stock returns associated the firm (H_{10}) consistent with the undervaluation, information signaling, and the free cash flow hypotheses.

A sample of share repurchase by US listed firms is used to test the hypothesis about levered share repurchase and financial constraints. The choice of US data is based on the evidence that a significant number of share repurchase programs involve debt financing (Milken, 2009). 42 In general, the results provide strong support for the hypothesis about why internally financially unconstrained firms conduct levered share repurchase. It further demonstrates the stock price reactions associated with share repurchase firms based on both the degree of financial constraints and the source of financing. Overall, the current research shed more light on share repurchase and notes that investment

_

 $^{^{\}rm 42}$ See Section 1.3 of Chapter 3 for explanation of the choice of US data.

expenditure has explanatory power as to whether firms borrow during share repurchase programs, given the level of internal financial constraints. This is because unless firms are willing to undertake investment expenditures, it is unlikely that unconstrained firms would borrow during share repurchase programs. It also appears unconstrained firms have debt capacity and are also able to obtain cheap debt financing due to favourable credit ratings. Thus, the presence of debt capacity and credit ratings are both symptoms of unconstrained firms.

Firms conducting share repurchases are undervalued and associated with low stock prices prior to the repurchase announcements. Thus, repurchase firms attempt to benefits from subsequent stock price appreciations. The study also finds that internally financially constrained firms are likely to borrow to finance share repurchase programs, all else equal. However, internally financially unconstrained firms are likely to conduct levered share repurchase when they also invest more. The results imply that the effects of share repurchase on reducing cash flow and stifling investments motivate firms to borrow to add to their cash buffers. Unconstrained firms also appear to be firms with investment grade bond ratings that allow them to borrow cheap in the external capital market. Internally financially constrained firms face significant costs of additional borrowings to the extent that would be sufficient to conduct share repurchase and also invest. Excessive borrowings would increase leverage ratios as to result in probability of financial distress. Unlike Chen and Wang (2012), this study not only show evidence of share repurchase by internally financially constrained firms, but it also provides evidence that debt financing affect the relation between share repurchase and internal financial constraints.

The three empirical chapters provide new insights into share issuance and repurchase decisions of firms. Financial constraints- both internal and external-affect share issuance decisions. Firms systematically time their equity issues to coincide with periods of mispricing. Financially constrained firms issue overvalued equity in order to profit from the temporary overpricing. However, additional analysis indicates that issue overvalued equity by financially

constrained only occurs during periods of favourable economic conditions. Also, in regards to share issuance methods, the study show that expected long run operating performance influence the choice of equity issuance methods. Better quality firms are likely to conduct rights issues rather than open offers and private placement, and are associated with significant stock price outperformance. For share repurchase, financially unconstrained firms borrow to finance share repurchase programs insofar as they also invest more. In effect, the costs of investment expenditure motivate financially unconstrained to obtain additional cash flows from the debt market while also disbursing cash flows to existing shareholders.

9.3. Research Implications and Recommendations

This section outlines and discusses implications of the findings for theoretical underpinnings of corporate decisions and for policy directions. Subsection 9.3.1 discusses the implications of the study for academics whereas Subsection 9.3.2 presents the implications for practitioners and policy makers.

9.3.1. Implications for Academics

The study tested the market timing theory in relation to share issuance and choice of issuance methods. It also explored the information asymmetry impact on share issuance decisions and the choice of issuance methods. The study on share issuance and issuance methods provides implications for the market timing theory. Market timing implies firms are likely to issue overvalued equity. Whereas this phenomenon is extensively explored in the literature, the current research indicates that financial constraints are an important determinant of the extent to which firms can systematically time their equity issues. The study also shows that financial constraints are related to the degree of information asymmetry. Overall, the study finds that financial constraint is an important variable in determining equity issuance decisions. Financial constraints have strong explanatory power after controlling for firm characteristics that have been found in the empirical literature to affect share issuance decisions. Thus, the findings provide insights for academics in the

study of share issuance behaviour of firms and note that future research should consider the effects of financial constraints on market timing of equity issues.

As regards share issuance methods, the study notes that operating performance reflects the degree of information asymmetry associated with equity issuing firms. After controlling for both firm and issue characteristics, the research finds that future expected operating performance is an important variable in choosing equity issuance methods. Operating performance measures information asymmetry that would not be captured by the traditional measures of information asymmetry such as accruals quality or idiosyncratic risk. Studies on the equity issuance methods should, therefore, consider the effects of operating performance. It should also examine how long run stock price reacts for each issuance methods given the level of anticipated long run operating performance.

The study further asserts that academic research on share repurchase should consider the source of financing. By undervaluation hypothesis, firms are likely to repurchase cheap shares whereas excess free cash flows motivate firms to repurchase under the free cash flow hypothesis. Information signalling hypothesis states that firms choose repurchase programs in order to convey significant information about the prospects of the firm to the market. The level of financial constraints is an important variable that explains share repurchase behaviour beyond the predictions of the undervaluation, free cash, and information signalling theories. As emphasised in the study, financially constrained firms also undertake repurchase by engaging in debt financing. Thus, the ability of firms to borrow might motivate firms to conduct share repurchase contrary to predictions of the free cash flow hypothesis. Therefore, academics should account for the effects of firms financial conditions in future research on share repurchases.

9.3.2. Implications for Practitioners and Policymakers

The implications of this study for practitioners and policymakers are particularly important given the impact of the recent financial crisis on corporate decisions. Regulations and policies for capital structure should ensure that firms are not significantly exposed to high leverage to the extent that they would be unable to finance important corporate events. The effects of the financial crisis meant that several firms could not undertake investment projects or employ people leading to overall downturn in economic activities. Most firms could only resort to the sale of assets in order to provide some liquidity to sustain the day-to-day running of their businesses. The study highlights that given the huge costs associated with equity issues; it might be suboptimal to issue them during periods of financial crisis. Therefore, practitioners need to appropriately time their equity issues to coincide with periods when costs of issuance are relatively low, especially during periods of favourable economic conditions with increased market liquidity.

It is important that business strategies are aligned with financial strategies to obtain optimum operational and financial benefits. Firms should also consider the appropriate financial strategy that is consistent with the corporate life cycle. It is important that practitioners institute a borrowing behaviour that is in line with the level of internal financial constraints, share repurchase decisions, and corporate investment strategy. Aggressive borrowings by internally financially constrained firms could significantly distort capital structure, especially when such debt financing is used to finance share repurchase programs. Furthermore, corporate governance issues determine financial policy changes of firms. Therefore, corporate governance policies should be formulated to assess the impact of share issuance and repurchase decisions in obtaining the overall optimum capital structure. In carrying out their legally mandated responsibilities, boards influence the financial performance of firms. Therefore, the findings of this study would guide board decisions to achieve optimal financial performance for their firms.

Share repurchase regulation in the US allows firms to exploit undervaluation by repurchasing cheap shares in the market. This phenomenon has given firms the latitude to systematically disburse excess cash to shareholders with the aim of buying them out of the ownership interest in the company. Also, the repurchase regulations make it possible for firms to engage in levered share repurchase even when they are significantly financially constrained. Increased

leverage adversely affects financial distress costs that could plunge the firm into bankruptcy. Regulations and policy changes could stifle the persistent attempts by firms to repurchase when it is suboptimal for the long-term sustainability of the firm.

9.4. Limitations of the Study

The empirical analysis of share issuance considers the probability of issuing and not issuing equity given the simultaneous effects of internal and external financial constraints. However, not issuing equity could imply issuing debt or even repurchasing equity. Thus, the dataset could consist of equity on one side and debt/repurchase on the other side of the dependent variable. For instance UK regulations restrict share repurchase announcements (Rau and Vermaelen, 2002) whereas bank loans that are not normally traded dominate corporate lending. 43 The unique features of UK share repurchase and debt issues make it difficult to model a choice analysis between equity issues on one hand and share repurchase and debt issues on the other hand. However, the study captures the effects of potential share repurchase and debt issues by selecting matching firms where the non-issue firm belongs to the same industry group and have comparable market to book ratio as the issuing firms.

Other factors such as the level of corporate governance affect corporate decisions. Also, external factors such as regulation, institutional arrangements, and corporate governance have significant impact on corporate decisions (Rau and Vermaelen, 2002; Bhagat and Bolton, 2008; Cornett et al. 2009). For example, regulatory and institutional changes take time to reflect on corporate decisions and market conditions such that the timing of the impact is difficult to capture. Furthermore, corporate governance variables need to be hand collected since most of such variables- board size, committees, board independence etc.-are not reported in databases. But given that the sample size of equity issues used in this study, it is impossible to collect data on governance. This study excludes these variables due to the difficulty in collecting them and also in line with prior studies. It is also impossible to even

⁴³ Bank of England (2009)

accurate measure human characteristics that define these variables. Therefore, their exclusion amounts to potential omitted variables bias in the models.

9.5. Suggestions for Future Research

Like all other studies, there are still avenues to extend and improve the predictions of both share issuance and repurchase decisions.

Following Dong et al. (2012), this current study has investigated the effects of financial constraints on share issuance decisions. However, like previous studies, it does not look at how corporate governance issues can potentially affect the relationship between internal financial constraints and market timing of equity issues. Corporate governance issues have become more important in this era of financial crisis and the effects of corporate decisions cannot be overemphasised. As such, future research should focus on the moderating impacts of governance issues in determining how financial constraints relate to governance and the subsequent effects on equity market timing. Furthermore, corporate governance impact on the choice of equity issuance method could be analysed in future research.

Given the global impact of the financial crisis, a cross-country study involving major European countries, Asian countries, Canada, and the US would provide an improved understanding of how external financing constraints moderate the relation between internal financing constraints and equity market timing. Since countries were affected and responded differently to the financial crisis, it can be argued that market timing of equity issues would be pursued in different fashions. For instance, this present study finds different predictions for the effects of financial constraints on market timing from the findings by Dong et al. (2012) who used Canadian data. Perhaps, regulatory differences and institutional arrangements impact differently on equity issuance decisions. Future research would help delineate these potential differences.

Similarly, share repurchase decisions, like other corporate events, should relate to how the firm is governed. Thus, whether or not firms repurchase based on undervaluation, free cash flow, or information signalling would be explained by the quality of corporate governance. Moreover, since levered share repurchase has significant impact on debt ratios with increased financial distress, corporate governance issues should determine the decision to borrow to finance share repurchase. It is even more important when the firm anticipating levered share repurchase program is internally financially constrained or not. Therefore, future research should consider how better governed firms behave in their share repurchase decisions. Thus, future research could empirically test the share repurchase behaviour of financially constrained given the level of corporate governance.

The objective of the third empirical chapter was to investigate the levered share repurchase behaviour of a sample of financially unconstrained US listed firms. Share repurchase transactions have increased significantly over the last two decades among US firms. However, several European countries that previously did not permit share repurchase have recently allowed firms to repurchase their shares, albeit in a relatively controlled fashion. Thus, it is more likely firms in these countries pursue the notion of levered share repurchase. Therefore, future research could conduct a cross-country empirical analyse of levered share repurchase to present a holistic view of this corporate policy. Moreover, this research could focus on the effects of financial constraints on levered share repurchase for these countries.

Given the importance of financial constraints on corporate decisions, including share issuance and repurchase, the potential effects on dividend policy could be explored. Generally, dividend payments are made from excess profits or cash flows that imply firms with high levels of profitability should be paying more dividends than low profitability firms. Also, given the adverse effects of decreasing dividends, firms that are cash-strapped would be in a difficult situation in retaining current dividends. It follows, therefore, that the degree of internal financial constraints would significantly affect dividend payments. Dividend payments relate the degree of internal financial constraints such that dividend-paying firms are classed as internally financially unconstrained. However, future research could explore how internally financially constrained firms finance subsequent dividend payments. Thus, the idea of levered

dividend payments could be investigated to provide empirical understanding of dividend payment financing.

Future research could also explore the choice between dividend payment and share repurchase given the level of internal financial constraints. Since both payout policies are considered substitutes in the empirical literature, it would be ideal to investigate the effects of internal financial constraints on the decision to pay dividends or repurchase shares. Moreover, the idea of leverage-financed corporate decisions could be extended to the choice between dividend payments and share repurchase. Thus, the research could examine the corporate events that are more likely to be financed using debt issues. For instance, it can be argued that since repurchase is less binding than dividend payments, internally financially constrained firms might be motivated to borrow to conduct share repurchase than they would for dividend payments unlikely to be sustained in the future.

APPRENDICES

Appendix 6A					
Hausman Test for Random versus Fixed Ef	fects				

	b	В	b-B	sqrt(diag(V_b-V_B))
Variables	Random	Fixed	Diff. (b-B)	S.E
KZ	-1.928	-1.645	0.283	0.288
MB	0.433	0.147	-0.286	0.040
KZ*MB	1.658	2.123	0.465	0.328
SIZE	0.289	0.400	0.111	0.024
ED	1.188	0.427	-0.762	0.135
PROF	-4.203	-1.379	2.824	0.496
CAPX	-8.727	-9.635	-0.908	

b = consistent

under Ho and Ha

B = inconsistent under Ha, efficient under Ho

Test: Ho: difference in coefficients not systematic

chi-squared = $(b-B)'[(V_b-V_B)^{-1}](b-B)$

= 185.02

Prob>chi-squared = 0.0000

(V_b-V_B is not positive definite)

Notes: This table displays the results for the hausman test for random versus fixed effects logit regression estimation given in Table 6.12

Appendix 6B
Hausman Test for Random versus Fixed Effects (Crisis Period)

	Hausman Test for Kandom versus Fixed Effects (Crisis Feriod)				
	b	В	b-B	sqrt(diag(V_b-V_B))	
Variables	Random	Fixed	Difference	S.E	
KZ	-3.241	-3.561	0.319	0.696	
MB	0.248	-0.027	0.275	0.077	
KZ*MB	-3.033	-3.946	0.914	0.831	
SIZE	0.329	0.304	0.025	0.061	
ED	2.269	1.184	1.085	0.339	
PROF	-7.102	-2.886	-4.216	1.344	
CAPX	-10.393	-9.035	-1.357		

b = consistent under Ho and Ha

B = inconsistent under Ha, efficient under Ho

Test: Ho: difference in coefficients not systematic

chi-squared = $(b-B)'[(V_b-V_B)^{-1}](b-B)$

= 21.66

Prob>chi-squared = 0.1170

(V b-V B is not positive definite)

Notes: This table displays the results for the hausman test for random versus fixed effects logit regression estimation given in Table 6.13.

Appendix 6C Hausman Test for Random versus Fixed Effects (Non-Crisis Period)

	b	В	b-B	sqrt(diag(V b-V B))
Variables	Random	Fixed	Difference	S.E
KZ	-1.246	-1.240	-0.006	0.192
MB	0.458	0.308	0.151	0.042
KZ*MB	1.334	1.429	-0.095	0.218
SIZE	0.318	0.359	-0.041	0.018
ED	0.813	0.744	0.069	0.096
PROF	-3.360	-1.797	-1.563	0.484
CAPX	-8.557	-8.839	0.282	0.159

b = consistent under Ho and Ha

B = inconsistent under Ha, efficient under Ho

Test: Ho: difference in coefficients not systematic

chi-squared = $(b-B)'[(V_b-V_B)^{-1}](b-B)$

= 16.06

Prob>chi-squared = 0.3783

(V b-V B is not positive definite)

Notes: This table displays the results for the hausman test for random versus fixed effects logit regression estimation given in Table 6.13

Appendix 8A
Hausman Test for Random versus Fixed Effects

Hausman Test for Random versus Fixed Effects				
	b	В	b-B	sqrt(diag(V_b-V_B))
Variables	Random	Fixed	Diff. (b-B)	S.E
KZ	0.936	0.935	-0.001	
INV	0.342	0.33	-0.012	0.004
KZ*INV	-2.684	-2.707	-0.023	0.032
MB	-0.1	-0.098	0.001	0.001
Prior AR	-0.413	-0.424	-0.01	0.015
SIZE	1.986	1.994	0.008	0.006
PMV	0.035	0.034	0	

b = consistent under Ho and Ha

B = inconsistent under Ha, efficient under Ho

Test: Ho: difference in coefficients not systematic

chi-squared = $(b-B)'[(V_b-V_B)^{-1}](b-B)$

= 11.77

Prob>chi-squared = 0.1083

(V b-V B is not positive definite)

Notes: This table displays the results for the hausman test for random versus fixed effects logit regression estimation.

REFERENCES

Abhyankar, A. & Ho, K. (2007), 'Long-Horizon Event Studies and Event Firm Portfolio Weights: Evidence from UK Rights Issues Re-Visited', *International Review of Financial Analysis*, Vol. 16, No. 1, pp. 61-80

Akerlof, G. A. (1970), 'The Market for "Lemons": Quality Uncertainty and the Market Mechanism', *The Quarterly Journal of Economics*, Vol. 84, No. 3, pp. 488-500.

Aldrich, J. H. & Nelson, F. D. (1984), 'Linear Probability, Logit, and Probit Models', Vol. 45. SAGE.

Alexander, G. J., Stoner, R. D., & Kuhnau, D. B. (1979), 'Market Timing Strategies In Convertible Debt Financing', *The Journal of Finance*, Vol. 34, No. 1, pp. 143-155.

Allen, D. E. (1993), 'The Pecking Order Hypothesis: Australian Evidence', *Applied Financial Economics*, Vol. 3, No. 2, pp. 101-112.

Almeida, H. & Campello, M. (2007), 'Financial Constraints, Asset Tangibility, and Corporate Investment', *Review of Financial Studies*, Vol. 20, No. 5, pp. 1429-1460.

Almeida, H., Campello, M., & Weisbach, M. S. (2004), 'The Cash Flow Sensitivity of Cash', *The Journal of Finance*, Vol. 59, No. 4, pp. 1777-1804.

Alti, A. (2006), 'How Persistent Is the Impact of Market Timing on Capital Structure', *The Journal of Finance*, Vol. 61, No. 4, pp. 1681-1710

Alti, A. & Sulaeman, J. (2012), 'When Do High Stock Returns Trigger Equity Issues?' *Journal of Financial Economics*, Vol. 103, No. 1, pp. 61-87.

Altman, E. I. (1968), 'Financial Ratios, Discriminant Analysis and the Prediction of Corporate Bankruptcy', *The Journal of Finance*, Vol. 23, No. 4, pp. 589-609.

Andrikopoulos, P. 2009, "Seasoned Equity Offerings, Operating Performance and Overconfidence: Evidence from the UK", *Journal of Economics and Business*, Vol. 61, No. 3, pp. 189-215

Andriosopoulos, D., & Lasfer, M. (2015), 'The Market Valuation of Share Repurchases in Europe', *Journal of Banking and Finance*, Vol. 55, pp. 327-339.

Antoniou, A., Guney, Y., & Paudyal, K. (2006), 'The Determinants of Debt Maturity Structure: Evidence from France, Germany and the UK', *European Financial Management*, Vol. 12, No. 2, pp. 161-194.

Apple Inc., (2012), 'Apple Announces Plans to Initiate Dividend and Share Repurchase Program (Press Release)'. Accessed at: http://www.apple.com/pr/library/2012/03/19apple-announces-plans-to-initiate dividend-and-share-repurchase-program.html.

Armitage, S. (2007), 'Discounts in Placing Pre-Renounced Shares in Rights Issues', *Journal of Business Finance and Accounting*, Vol. 34, No. 7-8, pp. 1345-69.

Armitage, S. (1998), 'Seasoned Equity Offers and Rights Issues: A Review of the Evidence', *European Journal of Finance*, Vol. 4, No. 1, pp. 29-59.

Armitage, S. (2000), 'The Direct Costs of UK Rights Issues and Open Offers', *European Financial Management*, Vol. 6, No. 1, pp. 57-68.

Armitage, S. (2002), 'Do Underwriters Certify Value? Evidence from UK Rights Issues and Open Offers', *Journal of Business Finance and Accounting*, Vol. 29, No. 9&10, pp. 1239-1273.

Armitage, S. (2010), 'Block Buying and Choice of Issue Method in UK Seasoned Equity Offerings', *Journal of Business Finance and Accounting*, Vol. 37, No. 3-4, pp. 422-448.

Armitage, S. (2012), 'The Calculation of Returns During Seasoned Equity Offers', *European Journal of Finance*, Vol. 18, No. 5, pp. 393-417.

Armitage, S. & Snell, A. (2004), 'Rights Issues Versus Private Placement: Theory and UK Evidence', Working Paper, University of Edinburgh.

Armitage, S., Dionysiou, D., & Gonzalez, A. (2014), 'Are the Discounts in UK Open Offers and Placings Due to Inelastic Demand?' *Journal of Business Finance and Accounting*, Vol. 41, No. 5-6, pp. 743-772.

Asquith, P. & Mullins, D. W. (1986), 'Equity Issues and Offering Dilution', *Journal of Financial Economics*, Vol. 15, No. 1-2, pp. 61-89.

Bagnoli, M., Gordon, R., & Lipman, B. L. (1989), 'Stock Repurchase As a Takeover Defense', *Review of Financial Studies*, Vol. 2, No. 3, pp. 423-443.

Baker, H. K., Powell, G. E., & Veit, E. T. (2003), 'Why Companies Use Open-Market Repurchases: A Managerial Perspective', *The Quarterly Review of Economics and Finance*, Vol. 43, No. 3, pp. 483-504.

Baker, M. & Wurgler, J. (2000), 'The Equity Share in New Issues and Aggregate Stock Returns', *The Journal of Finance*, Vol. 55, No. 5, pp. 2219-2257.

Baker, M. & Wurgler, J. (2002), 'Market Timing and Capital Structure', *The Journal of Finance*, Vol. 57, No. 1, pp. 1-32.

Baker, M. & Wurgler, J. (2004), 'A Catering Theory of Dividends', *The Journal of Finance*, Vol. 59, No. 3, pp. 1125-1166.

Baker, M., Stein, J.C., & Wurgler, J. (2003), 'When Does the Market Matter? Stock Prices and the Investment of Equity-Dependent Firms', *The Quarterly Journal of Economics*, Vol. 118, No. 3, pp. 969-1005.

Balachandran, B., Faff, R., & Theobald, M. (2008), 'Rights Offerings, Takeup, Renounceability, and Underwriting Status', *Journal of Financial Economics*, Vol. 89, No. 2, pp. 328-346.

Balachandran, B., Faff, R., Theobald, M., Velayutham, E., & Verwijmeren, P. (2013), 'Does Quality Signalling and Mispricing Explain the Choice and Long-Term Impact of Seasoned Equity Offering Methods?' *European Financial*

Management Association Meeting, Available At: http://www.Efmaefm.Org/0EFMAMEETINGS/EFMA%20ANNUAL%20 MEETINGS/2013-Reading/Papers/EFMA2013 0415 Fullpaper.Pdf.

Bancel, F. & Mittoo, U. R. (2004), 'Cross-Country Determinants of Capital Structure Choice: a Survey of European Firms', *Financial Management*, Vol. 33, No. 4, pp. 103-132.

Bank of England. (2009), 'Trends in Lending', Accessed online at: http://www.bankofengland.co.uk/publications/Documents/other/monetary/TrendsNovember09.pdf.

Baran, L. C. & King, T. D. (2010), 'Going Private Transactions, Bondholder Returns, and Wealth Transfer Effects', *Journal of Banking and Finance*, Vol. 34, No. 8, pp. 1856-1872.

Barber, B. M. & Lyon, J. D. (1996), 'Detecting Abnormal Operating Performance: The Empirical Power and Specification of Test-Statistics', *Journal of Financial Economics*, Vol. 41, No. 3, pp. 359-399.

Barber, B. M. & Lyon, J. D. (1997), 'Detecting Long-Run Abnormal Stock Returns: The Empirical Power and Specification of Test-Statistics', *Journal of Financial Economics*, Vol. 43, No. 3, pp. 341-372.

Barclay, M. J., Holderness, G. C., & Sheehan, D. P. (2007), 'Private Placements and Managerial Entrenchment', *Journal of Corporate Finance*, Vol. 13, No. 4, pp. 461-484.

Barclay, M. J., Smith, C. W. (1988), 'Corporate Payment Policy: Cash Dividends Versus Open Market Repurchase', *Journal of Financial Economics*, Vol. 22, No. 1, pp. 61-82.

Barclay, M.J., Smith, C.W., & Watts, R. L. (1995), 'The Determinants of Corporate Leverage and Dividends Policies', *Journal of Applied Corporate Finance*, Vol. 7, No. 4, pp. 4-19.

Bargeron, L., Kulchania, M., & Thomas, S. (2011), 'Accelerated Share Repurchase', *Journal of Financial Economics*, Vol. 101, No. 1, pp. 69-89.

Barnes, E. & Walker, M. (2006), 'The Seasoned Equity Issues of UK Firms: Market Reaction and Issuance Method Choice', *Journal of Business Finance and Accounting*, Vol. 33, No. 1-2, pp. 45-78.

Baskin, J. (1989), 'An Empirical Investigation of the Pecking Order Hypothesis," *Journal of Financial Management and Analysis*, Vol. 18, No. 1, pp. 26-35.

Bates, T. W., Kahle, K. M., & Stulz, R. M. (2009), 'Why Do US Firms Hold So Much More Cash Than They Used to?' *The Journal of Finance*, Vol. 64, No. 5, pp. 1985-2021.

Bayless, M. & Chaplinsky, S. (1991), 'Expectations of Security Type and the Information Content of Debt and Equity Offers', *Journal of Financial Intermediation*, Vol. 1, No. 3, pp. 195-214.

Bayless, M. & Chaplinsky, S. (1996), 'Is There a Window of Opportunity for Seasoned Equity Issuance? *The Journal of Finance*, Vol. 51, No. 1, pp. 253-278.

Bayless, M. E. & Diltz, J. D. (1994), "Securities Offering and Capital Structure Theory', *Journal of Business Finance and Accounting*, Vol. 21, No. 1, pp. 77-91.

Beattie, V., Goodacre, A., & Thomson, S. J. (2006), 'Corporate Financing Decisions: UK Survey Evidence', *Journal of Business Finance and Accounting*, Vol. 33, No. 9-10, pp. 1402-1434.

Benartzi, S., Michaely, R., & Thaler, R. (1997), 'Do Changes in Dividends Signal the Future Or the Past?' *The Journal of Finance*, Vol. 52, No. 3, pp. 1007-1034.

Bens, D. A., Nagar, V., Skinner, D. J., & Wong, M. H. F. (2003), 'Employee Stock Options, EPS Dilution, and Stock Repurchases', *Journal of Accounting and Economics*, Vol. 36, No. 1-3, pp. 51-90.

Bhagat, S. & Bolton, B. (2008), 'Corporate Governance and Firm Performance', *Journal of Corporate Finance*, Vol. 14, No. 3, pp. 257-273.

Bhargava, A. (2010), 'An econometric analysis of dividends and share repurchases by US firms', *Journal of the Royal Statistical Society: Series A (Statistics in Society)*, Vol. 173, No. 3, pp. 631-656.

Bhattacharya, U. & Dittmar, A. K. (2004), 'Costless Versus Costly Signaling in Capital Markets: Theory and Evidence', In *FIRS Conference on Banking, Insurance and Intermediation In Capri In May* 2004.

Bilinski, P., Liu, W., & Strong, N. (2012), 'Does Liquidity Risk Explain Low Firm Performance Following Seasoned Equity Offerings?' *Journal of Banking and Finance*, Vol. 36, No. 10, pp. 2770-2785.

Billet, M. T., Flannery, M. J., & Garfinkle, J. A. (2011), 'Frequent Issuers' Influence on Long-Run Post-Issuance Returns', *Journal of Financial Economics*, Vol. 99, No. 2, pp. 349-364.

Billet, M. T., Garfinkel, J. A., & Yu, M. (2015), 'The Effect of Asymmetric Information on Product Market Outcomes', Working Paper Accessed at SSRN: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1929008.

Billet, M. T., Mauer, D. C., & Zhang, Y. (2010), 'Stockholder and Bondholder Wealth Effects of CEO Incentive Grants', *Financial Management*, Vol. 39, No. 2, pp. 463-487.

Bøhren, Ø., Eckbo, B. E., & Michalsen, D. (1997), 'Why Underwrite Rights Offerings? Some New Evidence', *Journal of Financial Economics*, Vol. 46, No. 2, pp. 223-261.

Bollinger, C. & Chandra, A. (2005), 'Iatrogenic Specification Error: A Cautionary Tale of Cleaning Data ', *Journal of Labor Economics*, Vol. 23, No. 2, pp. 235-257.

Bolton, P., Chen, H., & Wang, N. (2011), 'A Unified Theory of Tobin's Q, Corporate Investment, Financing, and Risk Management', *The Journal of Finance*, Vol. 66, No. 5, pp. 1545-1578.

Bolton, P., Chen, H., & Wang, N. (2013), 'Market Timing, Investment, and Risk Management', *Journal of Financial Economics*, Vol. 109, No. 1, pp. 40-62.

Bonaime, A. A. & Ryngaert, M. D. (2013), 'Insider Trading and Share Repurchases: Do Insiders and Firms Trade in the Same Direction?', *Journal of Corporate Finance*, Vol. 22, pp. 35-53.

Boudry, W. I., Kallberg, J. G., & Liu, C. H. (2013), 'Investment Opportunities and Share Repurchases', *Journal of Corporate Finance*, Vol. 23, pp. 23-38.

Bozanic, Z. (2010), 'Managerial Motivation and Timing Of Open Market Share Repurchases', *Review Of Quantitative Finance And Accounting*, Vol. 34, No. 4, pp. 517-531.

Bozec, Y. & Laurin, C. (2008), 'Large Shareholder Entrenchment and Performance: Empirical Evidence from Canada', *Journal of Business Finance and Accounting*, Vol. 35, No. 1-2, pp. 25-49.

Bradley, M., Wakeman, L. M. (1983), 'The Wealth Effects Of Targeted Share Repurchase', *Journal Of Financial Economics*, Vol. 11, No. 1-4, Pp. 301-328.

Brav, A., Geczy, C., & Gompers, P. A. (2000), "Is the abnormal return following equity issuances anomalous?" *Journal of Financial Economics*, Vol. 56, No.2, pp. 209-249.

Brav, A., Graham, J. R., Harvey, C. R., & Michaely, R. (2005), 'Payout Policy in the 21st Century', *Journal of Financial Economics*, Vol. 77, No. 3, pp. 483-527.

Brennan, M. J. & Schwartz, E. S. (1980), 'Analysing Convertible Bonds', *Journal of Financial and Quantitative Analysis*, Vol. 15, No. 4, pp. 907-929.

Brennan, M. J. & Schwartz, E. S. (1988), 'The Case for Convertibles', *Journal of Applied Corporate Finance*, Vol. 1, No. 2, pp. 55-64.

Brockman, P. & Chung, D. Y. (2001), 'Managerial Timing and Corporate Liquidity: Evidence from Actual Share Repurchases', *Journal of Financial Economics*, Vol. 61, No. 3, pp. 417-448.

Brockman, P., Howe, J. S., & Mortal, S. (2008), 'Stock Market Liquidity and the Decision to Repurchase', *Journal of Corporate Finance*, Vol. 14, pp. 446-459.

Brockman, P., Khurana, I. K., & Martin, X. (2008), 'Voluntary Disclosures Around Share Repurchase', *Journal of Financial Economics*, Vol. 89, No. 1, pp. 175-191.

Brounen, D., De Jong, A., & K. Koedijk, K. (2006), 'Capital Structure Policies in Europe: Survey Evidence', *Journal of Banking And Finance*, Vol. 30, No. 5, pp. 1409-1442.

Brown, R. L. & Chan H. (2004), 'Rights Issues versus Placements in Australia: Regulation or Choice?' *Company and Securities Law Journal*, Vol. 22, No. 5, pp. 301-312.

Brown, S. J. & Warner, J. B. (1985), 'Using Daily Stock Returns: The Case of Event Studies. *Journal of Financial Economics*, Vol. 14, No. 1, pp. 3-31.

Bruinshoofd, A. W. & De Haan, L. (2012), 'Market Timing and Corporate Capital Structure: A Transatlantic Comparison', *Applied Economics*, Vol. 44, No. 28, pp. 3691-3703.

Burch, T. R., Christie, W. G., & Nanda, V. (2004), 'Do Firms Time Equity Offering? Evidence from the 1930s and 1940s', *Financial Management*, Vol. 33, No. 1, pp. 5-23.

Burton, B. M., Lonie, A. A., & Power, D. M. (1999), 'Does the Issue Method Influence the Market Reaction to Seasoned Equity Offer Announcements?' *Applied Economics Letters*, Vol. 6, No. 7, 459-462.

Butler, A. W., Grullon, G., & Weston, J. P. (2005), 'Stock Market Liquidity and the Cost of Issuing Equity', *Journal of Financial and Quantitative Analysis*, Vol. 40, No. 2, pp. 331-348.

Byoun, S. (2008), 'How and When Do Firms Adjust Their Capital Structures Toward Targets?' *The Journal of Finance*, Vol. 63, No. 6, pp. 3069-3096.

Byoun, S. (2011), 'Financial Flexibility and Capital Structure Decision', Available at SSRN: http://ssrn.com/abstract=1108850.

Cameron, A. C. & Trivedi, P. K. (2005), 'Microeconometrics: Methods and Applications', Cambridge University Press.

Campello, M., Giambona, E., Graham, J. R., & Harvey, C. R. (2011), 'Liquidity Management and Corporate Investment During a Financial Crisis', *Review of Financial Studies*, Vol. 24, No. 6, pp. 1944-1979.

Campello, M., Graham J.R., & Harvey, C. R. (2010), 'The Real Effects of Financial Constraints: Evidence from a Financial Crisis', *Journal of Financial Economics*, Vol. 97, No. 3, pp. 470-487.

Capstaff, J. & Fletcher, J. (2011), 'Long Term Performance and Choice of SEO Method by UK Firms', *Journal of Business Finance and Accounting*, Vol. 38, No. 9-10, pp. 1262-1289.

Cassell, C. A., Huang, S. X., Sanchez, J. M., & Stuart, M. D. (2012), 'Seeking Safety: The Relation Between CEO Inside Debt Holdings and the Riskiness of Firm Investment and Financial Policies', *Journal of Financial Economics*, Vol. 103, No. 3, pp. 588-610.

Chae, J. (2005), 'Trading Volume, Information Asymmetry and Timing information', *The Journal of Finance*, Vol. 60, No. 1, pp. 413-442.

Chan, K., Ikenberry, D. L., & Lee, I. (2004), 'Economic Sources of Gain in Stock Repurchase', *Journal of Financial and Quantitative Analysis*, Vol. 39, No. 3, pp. 461-479.

Chan, K., Ikenberry, D. L., & Lee, I. (2007), 'Do Managers Time the Market? Evidence from Open-Market Share Repurchases', *Journal of Banking and Finance*, Vol. 31, No. 9, pp. 2673-2694.

Chan, K., Ikenberry, D. L., Lee, I., & Wang, Y. (2010), 'Share Repurchases As A Potential Tool to Mislead Investors', *Journal of Corporate Finance*, Vol. 16, No. 2, pp. 137-158.

Chang, X., Hilary, G., Shih, C. M. & Tam, L.H.K. (2010), 'Conglomerate Structure and Capital Market Timing', *Financial Management*, Vol. 39, No. 4, pp. 1307-1338.

Chang, X., Tam, L. Tan, T. J., & Wong, G. (2007), 'The Real Impact of Stock Market Mispricing: Evidence from Australia', *Pacific-Basin Finance Journal*, Vol. 15, No. 4, pp. 388-408.

Chen, A. S., Cheng, L. Y., Cheng, K. F., & Chih, S. W. (2010), 'Earnings Management, Market Discounts and the Performance of Private Equity Placements. *Journal of Banking and Finance*, Vol. 34, No. 8, pp. 1922-1932.

Chen, H. J. & Chen, S. (2012), 'Investment-Cash Flow Sensitivity Cannot Be a Good Measure of Financial Constraints: Evidence from the Time Series', *Journal of Financial Economics*, Vol. 103, No. 2, pp. 393-410.

Chen, S. & Wang, Y. (2012), 'Financial Constraints and Share Repurchases', *Journal of Financial Economics*, Vol. 105, No. 2, pp. 311-331.

Chi, J. D. & Gupta, M. (2009), 'Overvaluation and Earnings Management', *Journal of Banking and Finance*, Vol. 33, No. 9, pp. 1652-1663.

Choe, H., Masulis, R. W., & Nanda, V. K. (1993), 'Common Stock Offerings across the Business Cycle: Theory and Evidence', *Journal of Empirical Finance*, Vol. 1, No. 1, pp. 3-31.

Chou, D. W., Gombola, M., & Liu, F. Y. (2009), 'Long-Run Underperformance Following Private Equity Placements: The Role of Growth Opportunities', *The Quarterly Review of Economics and Finance*, Vol. 49, No. 3, pp. 1113-1128.

Comment, R. & Jarrell, G. A. (1991), 'The Relative Signalling Power of Dutch-Auction and Fixed-Price Self-Tender Offers and Open-Market Share Repurchases', *The Journal of Finance*, Vol. 46, No. 4, pp. 1243-1271.

Constantindes, G. & Grundy, B. (1989), 'Optimal Investment with Stock Repurchase and Financing As Signals', *Review of Financial Studies*, Vol. 2, No. 4, pp. 445-465.

Cook, D. O. & Tang, T. (2010), 'Macroeconomic Conditions and Capital Structure Adjustment Speed', *Journal of Corporate Finance*, Vol. 16, No. 1, pp. 73-87.

Core, J. E. & Guay, W. R. (2001), 'Stock Plans For Non-Executive Employees', *Journal of Financial Economics*, Vol. 61, No. 2, pp. 253-287.

Cornett, M. M., McNutt, J. J., & Tehranian, H. (2009), 'Corporate Governance and Earnings Management at Large US Bank Holding Companies', *Journal of Corporate Finance*, Vol. 15, No. 4, pp. 412-430.

Corwin, S. A. (2003), 'The Determinants of Underpricing for Seasoned Equity Offers', *The Journal of Finance*, Vol. 58, No. 5, pp. 2249-2279.

Cowan, A.R., Nayar, N., & Singh, A.K. (1990), 'Stock Returns Before and After Calls of Convertible Bonds', *Journal of Financial and Quantitative Analysis*, Vol. 25, No. 4, pp. 549-554.

Cox, D. R. (1970), 'The Analysis of Binary Data', London: Methuen.

Cronqvist, H. & Nilsson, M. (2005), 'The Choice between Rights Offerings and Private Equity Placements', *Journal of Financial Economics*, Vol. 78, No. 2, pp. 375-407.

D'Mello, R. & Shroff, P. K. (2000), 'Equity Undervaluation and Decisions Related to Repurchase Tender Offers: An Empirical Investigation', *The Journal of Finance*, Vol. 55, No. 5, pp. 2399-2424.

Daniel, K. D. & Titman, S. (1997), 'Evidence on the Characteristics of Cross-sectional Variation in Common Stock Returns', *The Journal of Finance*, Vol. 52, No. 1, pp. 1-33.

Daniel, K. D., Hirshleifer, D., & Subrahmanyam, A. (2001), 'Overconfidence, Arbitrage, And Equilibrium Asset Pricing', *The Journal of Finance*, Vol. 56, No. 3, pp. 921-965.

Dann, L. Y. & DeAngelo, H. (1988), 'Corporate Financial Policy and Corporate Control: A Study of Defensive Adjustments in Asset and Ownership Structure', *Journal of Financial Economics*, Vol. 20, pp. 87-127.

Dann, L.Y. & Mikkelson, W.H. (1984), 'Convertible Debt Issuance, Capital Structure Change and Financing-Related Information: Some New Evidence', *Journal of Financial Economics*, Vol. 13, No. 2, pp. 157-186.

De Jong, A., Dutordoir, M., & Verwijmeren, P. (2011), 'Why Do Convertible Issuers Simultaneously Repurchase Stock? An Arbitrage-Based Explanation', *Journal of Financial Economics*, Vol. 100, No. 1, pp. 113-129.

De Jong, A., van Dijk, R., & Veld, C. (2003), 'The Dividend and Share Repurchase Policies of Canadian Firms: Empirical Evidence Based on An Alternative Research Design', *International Review of Financial Analysis*, Vol. 12, No. 4, pp. 349-377.

De Jong, A., Verbeek, M., & Verwijmeren, P. (2010), 'The Impact of Financial Surpluses and Large Financing Deficits on Tests of the Pecking Order Theory', *Financial Management*, Vol. 39, No. 2, pp. 733-756.

DeAngelo, H., DeAngelo, L., & R. Stulz, R. (2010), 'Seasoned Equity Offerings, Market Timing, and the Corporate Lifecycle', *Journal of Financial Economics*, Vol. 95, No. 3, pp. 275-295.

Dechow, P. M. & Dichev, I. D. (2002), 'The Quality of Accruals and Earnings: The Role of Accrual Estimation Errors', *The Accounting Review: Supplement* 2002, Vol. 77, No. s-1, pp. 35-59.

Denis, D. J. (1994), 'Investment Opportunities and the Market Reaction to Equity Offerings', *Journal of Financial and Quantitative Analysis*, Vol. 29, No. 2, pp. 159-177.

Denis, D. J. & McKeon, S. B. (2012), 'Debt Financing and Financial Flexibility Evidence from Proactive Leverage Increases', *The Review of Financial Studies*, Vol. 25, No. 6, pp. 1897-1929.

Denis, D. J. & Osobov, I. (2008), 'Why Do Firms Pay Dividends? International Evidence on the Determinants of Dividend Policy', *Journal of Financial Economics*, Vol. 89, No. 1, pp. 62-82.

Denis, D. J. & Sibilkov, V. (2010), 'Financial Constraints, Investment, and the Value of Cash Holdings', *Review of Financial Studies*, Vol. 23, No. 1, pp. 247-269.

Dierkens, N. (1991), 'Information Asymmetry and Equity Issues', *Journal of Financia* Dudley 2012

Dionysiou, D. (2015), 'Timing, Earnings Management and Overreaction Around Pure Placings', *European Journal of Finance*, Vol. 21, No. 8, pp. 646-671.

Dissanaike, G., Faasse, J., & Jayasekera, R. (2014), 'What Do Equity Issuances Signal? A Study of Equity Issuances in the UK Before and During the Financial Crisis', *Journal of International Money and Finance*, Vol. 30, Part B, pp. 1-28.

Dittmar, A. & Thakor, A. (2007), 'Why Do Firms Issue Equity?' *The Journal of Finance*, Vol. 52, No. 1, pp. 1-54.

Dittmar, A. (2000), 'Why Do Firms Repurchase Stock?' *The Journal of Business*, 73, No. 3, pp. 331-355.

Donaldson, G. (1961), 'Corporate Debt Capacity: A Study of Corporate Debt Policy and the Determination of Corporate Debt Capacity', Harvard Business School, Division of Research, Harvard University.

Dong, M., Loncarski, I., ter Horst, J., & Veld, C. (2012), 'What Drives Security Issuance Decisions: Market Timing, Pecking Order, Or Both?' *Financial Management*, Vol. 41, No. 3, pp. 637-663.

Duca, E., Dutordoir, M., Veld C., & Verwijmeren, P. (2010), 'Why Are Convertible Bond Announcement Associated with Increasing Negative Abnormal Stock Returns? An Arbitrage Based Explanation', *Journal of Banking and Finance*, Vol. 36, No. 11, pp. 2884-2899.

Duchin, R., Ozbas, O., & Sensoy, B. A. (2010), 'Costly External Finance, Corporate Investment and The Subprime Mortgage Credit Crisis', *Journal of Financial Economics*, Vol. 97, No. 3, pp. 418-435.

Dudley, E. (2012), 'Capital Structure and Large Investment Projects', *Journal of Corporate Finance*, Vol. 18, No. 5, pp. 1168-1192.

Dutordoir, M. & De Gucht, L.V. (2009), 'Why Do Western European Firms Issue Convertibles Instead of Straight Debt or Equity?' *European Financial Management*, Vol. 15, No. 3, pp. 563-583.

Eckbo, B. E. & Masulis, R.W. (1992), 'Adverse Selection and the Rights Offer Paradox', *Journal of Financial Economics*, Vol. 32, No. 3, pp. 293-332.

Eckbo, B. E., Masulis, R.W., & Norli, Ø. (2000), 'Seasoned Public Offerings: Resolution 0f the 'New Issues Puzzle', *Journal of Financial Economics*, Vol. 56, No. 2, pp. 251-291.

Elliot, W. B., Prevost, A. K., & Rao, R. P. (2009), 'The Announcement Impact of Seasoned Equity Offerings on Bondholder Wealth', *Journal of Banking and Finance*, Vol. 33, No. 8, pp. 1472-1480.

Elliott, W.B., Koëter-Kant, J., & Warr, R.S. (2008), 'Market Timing and the Debt-Equity Choice', *Journal of Financial Intermediation*, Vol. 17, No. 2, pp. 175-197.

Erickson, T. & Whited, T. M. (2000), 'Measurement Error and The Relationship between Investment and q', *Journal of Political Economy*, Vol. 108, No. 5, pp. 1027-1057.

Espenlaub, S., Iqbal, A., & Strong, N. (2009), 'Datastream Returns and UK Open Offers', *The European Journal of Finance*, Vol. 15, No. 1, pp. 61-69.

Fama, E. F. (1998), 'Market Efficiency, Long-Term Returns and Behavioral Finance', *Journal of Financial Economics*, Vol. 49, No. 3, pp. 283-306.

Fama, E. F. & French, K. R. (1992), 'The Cross-Section of Expected Stock Returns', *The Journal of Finance*, Vol. 47, No. 2, pp. 427-465.

Fama, E. F. & French, K. R. (1993), 'Common Risk Factors in Returns on Stocks and Bonds', *Journal of Financial Economics*, Vol. 33, No. 1, pp. 3-56.

Fama, E. F. & French, K. R. (2001), 'Disappearing Firm Dividends, Changing Firm Characteristics Or Lower Propensity To Pay?', *Journal of Financial Economics*, Vol. 60, No. 1, pp. 3-43.

Fama, E. F. & French, K. R. (2004), 'The Capital Asset Pricing Model: Theory and Evidence', *Journal of Economic Perspectives*, Vol. 18, No. 3, pp. 25-46.

Fama, E. F. & French, R. K. (1995), 'Size and Book-to-Market Factors in Earnings and Returns', *The Journal of Finance*, Vol. 50, No. 1, pp. 131-155.

Fama, E. F. & Miller, M. H. (1972), 'The Theory of Finance (Vol. 3), Hinsdale, IL: Dryden Press.

Fama, E. F., & French, K. R. (2002), 'Testing Trade-Off and Pecking Order Predictions About Dividends And Debt', *Review of Financial Studies*, Vol. 15, No. 1, pp. 1-33.

Fama, E. F., Fisher, L., Jensen, M. C., & Roll, R. (1969), 'The Adjustment of

Stock Prices to New Information', *International Economic Review*, Vol. 10, No. 1, pp. 1-21.

Fama, E.F. & French, K. R. (2005), 'Financing Decisions: Who Issues Stock?' *Journal of Financial Economics*, Vol. 76, No. 3, pp. 549-582.

Farre-Mensa, J., Michaely, R., & Schmalz, M.C. (2015), 'Financing Payout', Ross School of Business Paper No. 1263. Available at SSRN: http://ssrn.com/abstract=2535675 or http://dx.doi.org/10.2139/ssrn.2535675.

Faulkender, M., Flannery, M. J. K., Hankins, W., & Smith, J. M. (2012), 'Cash Flows and Leverage Adjustments', *Journal of Financial Economics*, Vol. 103, No. 3, pp. 632-646.

Fazzari, S. M., Hubbard, G. R., & Petersen, B. C. (1988), 'Financing Constraints and Corporate Investment', *Brookings Papers on Economic Activity*, Vol. 1, pp. 141-206.

Fazzari, S. M., Hubbard, G. R., & Petersen, B. C. (2000), 'Investment-Cash Flow Sensitivities Are Useful: A Comment on Kaplan and Zingales', *The Quarterly Journal of Economics*, Vol. 115, No.2, pp. 695-705.

Fenn, G. W. & Liang, N. (1998), 'Good News and Bad News about Share Repurchases', FEDS Paper No. 98-4, Available at SSRN: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=113268

Fenn, G. W. & Liang, N. (2001), 'Corporate Payout Policy and Managerial Stock Incentives', *Journal of Financial Economics*, Vol. 60, No. 1, pp. 45-72.

Foley, C. F., Hartzell, J. C., Titman, S., & Twite, G. (2007), 'Why Do Firms Hold So Much Cash? A Tax-Based Explanation', *Journal of Financial Economics*, Vol. 86, No. 3, pp. 579-607.

Frank, M.Z. & Goyal, V. K. (2003), 'Testing the Pecking Order Theory of Capital Structure', *Journal of Financial Economics*, Vol. 67, No. 2, pp. 217-248.

Fried, J. M. (2004), 'Open Market Repurchases: Signalling Or Managerial Opportunism?' *Theoretical Inquiries in Law*, Vol. 2, No. 2, pp. 865-894.

Gajewski, J. F. & Ginglinger, E. (2002), 'Seasoned Equity Issues in a Closely Held Market: Evidence from France', *European Finance Review*, Vol. 6, No. 3, pp. 291-319.

Gara, A. (2013), 'Goldman Sachs and Deutsche Bank Lead A 6-Part Debt Offering', Accessed at: http://www.forbes.com/sites/thestreet/2013/04/30/goldman-deutsche-to-lead-apple-bond-offering-for-dividend-boost On April 30, 2013.

Gay, G. D., Kale, J. R., & Noe, T. H. (1996), '(Dutch) Auction Share Repurchases', *Economica*, *New Series*, Vol. 63, No. 249, pp. 57-80.

Ginglinger, E., Matsoukis, L., & Riva, F. (2013), 'Seasoned Equity Offerings: Stock Market Liquidity and the Rights Offer Paradox', *Journal of Business Finance and Accounting*, Vol. 40, No. 1-2, pp. 215-238.

Giot, P. & Petitjean, M. (2009), 'Short-Term Market Timing Using the Bond-Equity Yield Ratio', *European Journal of Finance*, Vol. 15, No. 4, pp. 365-384.

Gomes, A. & Phillips, G. (2007), 'Why Do Public Firms Issue Private and Public Securities?' AFA 2008 New Orleans Meeting Paper. Available at SSRN: http://ssrn.com/abstract=972723.

Goyal, V. K., & Park, C. W. (2002), 'Board Leadership Structure and CEO Turnover', *Journal of Corporate Finance*, Vol. 8, No. 1, pp. 49-66.

Graham, J. & Harvey, C. (2001), 'The Theory and Practice of Corporate Finance: Evidence from the Field', *Journal of Financial Economics*, Vol. 60, No. 2-3, pp. 187-243.

Green, R. C. (1984), 'Investment Incentives, Debt and Warrants', *Journal of Financial Economics*, Vol. 13, No. 1, pp. 115-136.

Greenbert, H. (2011, November 8), 'Debt to buy back stock', *CNBC*. Retrieved from: http://www.cnbc.com/id/45209702.

Grullon, G. & Ikenberry, D. (2000), 'What Do We Know About Stock Repurchases?', *Journal of Applied Corporate Finance*, Vol. 13, No. 1, pp. 31-51.

Grullon, G. & Michaely, R. (2002), 'Dividends, Share Repurchases, and the Substitution Hypothesis', *The Journal of Finance*, Vol. 57, No. 4, pp. 1649-1684.

Grullon, G. & Michaely, R. (2004), 'The Information Content of Share Repurchases Programs', *The Journal of Finance*, Vol. 59, No. 2, pp. 651-680.

Grullon, G., Michaely, R., & Swaminathan, B. (2002), 'Are Dividend Changes a Sign of Firm Maturity?', *The Journal of Business*, Vol. 75, No. 3, pp. 387-424.

Guay, W. & Harford, J. (2000), 'The Cash Flow Permanence and Information Content of Dividend Increases Versus Repurchases', *Journal of Financial Economics*, Vol. 57, No. 3, pp. 385-415.

Hadlock, C.J. & Pierce, J.R. (2010), 'New Evidence on Measuring Financial Constraints: Moving beyond the KZ index', *The Review of Financial Studies*, Vol. 23, No. 5, pp. 1909-1940.

Hahn, J. & Lee, H. (2009), 'Financial Constraints, Debt Capacity, and the Cross Section of Stock Returns', *The Journal of Finance*, Vol. 64, No. 2, pp. 891-921.

Hansen, R.S. (1988), 'The Demise of the Rights Issue', *Review of Financial Studies*, Vol. 1, No. 3, pp. 289-309.

Harford, J. (2005), 'What Drives Merger Waves?' *Journal of Financial Economics*, Vol. 77, No. 3, pp. 529-560.

Harris, M. & Raviv, A. (1988), 'Corporate Control Contests and Capital Structure', *Journal of Financial Economics*, Vol. 20, pp. 55-86.

Healy, P. M. & Palepu, K. G. (1990), 'Effectiveness of Accounting-Based Dividend Covenants', *Journal of Accounting and Economics*, Vol. 12, No. 1, pp. 97-123.

Heaton, J. B. (2002), 'Managerial Optimism and Corporate Finance', *Financial Management*, Vol. 31, No. 2, pp. 33-45.

Helwege, J. & Liang, N. (1996), 'Is There A Pecking Order? Evidence from a Panel of IPO Firms', *Journal of Financial Economics*, Vol. 40, No. 3, pp. 429-458.

Hertzel, M. & Smith, L. R. (1993), 'Market Discounts and Shareholder Gains for Placing Equity Privately', *The Journal of Finance*, Vol. 48, No. 2, pp. 459-485.

Hertzel, M. G. & Li, Z. (2010), 'Behavioral and Rational Explanations of Stock Price Performance Around SEOs: Evidence from A Decomposition of Market-to-Book Ratios', *Journal of Financial and Quantitative Analysis*, Vol. 45, No. 4, pp. 935-958.

Hertzel, M., Lemmon, M., Linck, S. J., & Rees, L. (2002), 'Long-Run Performance Following Private Placements of Equity', *The Journal of Finance*, Vol. 57, No. 6, pp. 2595-2617.

Ho, K. (2005), 'Long-Horizon Abnormal Performance Following Rights Issues and Placings: Additional Evidence from the UK Market', *Review of Financial Economics*, Vol. 14, No. 1, pp. 25-45.

Hovakimian, A. (2006), 'Are Observed Capital Structure Determined by Equity Market Timing?' *Journal of Financial and Quantitative Analysis*, Vol. 41, No. 1, pp. 221-243.

Hovakimian, A., Hovakimian, G., & Tehranian, H. (2004), 'Determinants of Target Capital Structure: The Case of Dual Debt and Equity Issues', *Journal of Financial Economics*, Vol. 71, No. 3, pp. 517-540.

Hovakimian, A., Opler, T., & Titman, S. (2001), 'The Debt-Equity Choice', *Journal of Financial and Quantitative Analysis*, Vol. 36, No. 1, pp. 1-21.

Howcroft, B., Kara, A., & Marques-Ibanez, D. (2014), 'Determinants of Syndicated Lending in European Banks and the Impact of the Financial Crisis', *Journal of International Financial Markets, Institutions and Money*, Vol. 32, pp. 473-490.

Huang, R. & Ritter, J. R. (2009), 'Testing Theories of Capital Structure and Estimating the Speed of Adjustment', *Journal of Financial and Quantitative Analysis*, Vol. 44, No. 2, pp. 237-271.

Ibbotson, R. & Jaffe, J. (1975), 'Hot Issues Market', University of Illinois at Urbana-Champaign's Academy for Entrepreneurial Leadership Historical Research Reference in Entrepreneurship. Available at SSRN: http://ssrn.com/abstract=1505241.

Ikenberry, D. & Vermaelen, T. (1996), 'The Option to Repurchase Stock', *Financial Management*, Vol. 25, No. 4, pp. 9-24.

Ikenberry, D., Lakonishok, J., & Vermaelen, T. (1995), Market Underreaction to Open Market Share Repurchase', *Journal of Financial Economics*, Vol. 39, No. 2-3, pp. 181-208.

Ikenberry, D., Lakonishok, J., & Vermaelen, T. (2000), 'Stock Repurchases in Canada: Performance and Strategic Trading', *The Journal of Finance*, Vol. 55, No. 5, pp. 2373-2397.

Iqbal, A. (2008), 'The Importance of the Sequence in UK Rights Issues', *Journal of Business Finance and Accounting*, Vol. 35, No. 1, pp. 150-176.

Iqbal, A., Akbar, S., & Shiwakoti, R. K. (2013), 'The Long-Run Performance of UK Firms Making Multiple Rights Issues', *International Review of Financial Analysis*, Vol. 28, No. 2, pp. 156-165.

Iqbal, A., Espenlaub, S., & Strong, N. (2006), 'The Long-Run Performance of UK Rights Issuers', *Frontiers in Finance and Economics*, Vol. 3, No. 2, pp. 19-54.

Iqbal, A., Espenlaub, S., & Strong, N. (2009), 'Datastream Returns and UK Open Offers', *European Journal of Finance*, Vol. 15, No. 1, pp. 61-69.

Iqbal, A., Espenlaub, S., & Strong, N. (2009), 'Earnings Management Around UK Open Offers', *European Journal of Finance*, Vol. 15, No. 1, pp. 29-51.

Isagawa, N. (2000), 'Open-Market Stock Repurchases and Stock Price Behaviour When Management Values Real Investment', *Financial Review*, Vol. 35, No. 4, pp. 95-108.

Isagawa, N. (2002), 'Open-Market Repurchases Announcements and Stock Price Behaviour in Inefficient Markets', *Financial Management*, Vol. 31, No. 3, pp. 5-20.

Ivashina, V. & Scharfstein, D. (2010), 'Bank Lending During the Financial Crisis of 2008', *Journal of Financial Economics*, Vol. 97, No. 3, pp. 391-338.

Jagannathan, M. & Stephens, C. P. (2003), 'Motives for Multiple Open Market Repurchase Programs', *Financial Management*, Vol. 32, No. 2, pp. 71-91.

Jagannathan, M., Stephens, C. P., & Weisbach, M. S. (2000), 'Financial Flexibility and the Choice between Dividends and Stock Repurchases', *Journal of Financial Economics*, Vol. 57, No. 3, pp. 355-384.

Jalan, P. & Barone-Adesi, G. (1995), 'Equity Financing and Corporate Convertible Bond Policy', *Journal of Banking and Finance*, Vol. 19, No. 2, pp. 187-206.

Jalilvand, A. & Harris, R.S. (1984), 'Corporate Behavior in Adjusting to Capital Structure and Dividend Targets: An Econometric Study', *The Journal of Finance*, Vol. 39, No. 1, pp. 127-45.

Jensen, M. C. (1986), 'Agency Costs of Free-Cash-Flow, Corporate Finance

and Takeovers', American Economic Review, Vol. 76, No. 2, pp. 323-329.

Jensen, M. C. (2005), 'Agency Costs of Overvalued Equity', *Financial Management*, Vol. 34, No. 1, pp. 5-19.

Jensen, M. C. & Meckling, W. H. (1976), 'Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure', *Journal of Financial Economics*, Vol. 3, No. 4, pp. 305-360.

Jiang, Z., Kim, K. A., Lie, E., & Yang, S. (2013), 'Share Repurchases, Catering, and Dividend Substitution', *Journal of Corporate Finance*, Vol. 21, pp. 36-50.

Jolls, C. (1998), 'Stock Repurchases and Incentive Compensation', Working Paper No. 6467, Cambridge, Mass.: NBER.

Jun, S., Jung, W., & Walkling, R. A. (2009), 'Share Repurchase, Executive Options and Wealth Changes to Stockholders and Bondholders', *Journal of Corporate Finance*, Vol. 15, No. 2, pp. 212-229.

Jung, K., Kim, Y. C., & Stulz, R. (1996), 'Timing, Investment Opportunities, Managerial Discretion, and the Security Issue Decision', *Journal of Financial Economics*, Vol. 42, No. 2, pp. 159-186.

Jung, M. & Sullivan, M. J. (2009), 'The Signalling Effects Associated with Convertible Debt Design', *Journal of Business Research*, Vol. 62, No. 12, pp. 1358-1363.

Kabir, R. & Roosenboom, P. (2003), 'Can the stock market anticipate future operating performance? Evidence from equity rights issues', *Journal of Corporate Finance*, Vol. 9, No. 1, pp. 93-113.

Kahle, K. M. (2002), 'When A Buyback Isn't A Buyback: Open Market Repurchases and Employee Options', *Journal of Financial Economics*, Vol. 63, No. 2, pp. 235-261.

Kahle, K. M., Dyl, E. A., & Banyi, M. (2005), 'Measuring Share Repurchases', Working Paper, University of Arizona', Accessed at: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=726284.

Kaplan, S.N. & Zingales, L. (1997), 'Do Investment-Cash Flow Sensitivities Provide Useful Measures of Financing Constraints?' *The Quarterly Journal of Economics*, Vol. 112, No. 1, pp. 169-215.

Kashyap, A., Lamont, O., & Stein, J. (1994), 'Credit Conditions and the Cyclical Behavior of Inventories: A Case Study of the 1981-82 Recession', *Quarterly Journal of Economics*, Vol. 109, No. 3, pp. 565-592.

Katy, B. & Mike C. (2013, April 30), 'Apple's Record Plunge into Debt Pool', *The Wall Street Journal*. Accessed online at: http://www.wsj.com/articles/SB10001424127887324482504578454691936382 274.

Kayhan, A. & Titman, S. (2007), 'Firms' Histories and Their Capital Structures', *Journal of Financial Economics*, Vol. 83, No. 1, pp. 1-32.

Kedia, S. & Mozumdar, A. (2002), 'Performance Impact of Employee Stock Options', AFA 2003 Washington, DC Meetings. Accessed at: http://ssrn.com/abstract=304188.

Kim, J. & Varaiya, N. (2004), 'Disclosure on Open Market Repurchases Transactions in the U.S.: Does It Create A Conflict of Interest?', Working Paper, San Diego State University, CA.

Kim, W. & Weisbach, M. (2008), 'Motivations for Public Equity Offering: An International Perspective', *Journal of Financial Economics*, Vol. 87, No. 2, pp. 281-307.

Kim, Y. O. & Kallberg, J. (1998), 'Convertible Calls and Corporate Taxes Under Asymmetric Information', *Journal of Banking And Finance*, Vol. 22, No. 1, pp. 19-40.

Klassen, K. J. & Sivakumar, R. (2001), 'Measuring Managerial Stock Option Incentives: The Case of Stock Repurchases', Working Paper, University of Waterloo, Waterloo, ON.

Klein, A. & Rosenfeld, J. (1988), 'Targeted Share Repurchases and Top Management Changes', *Journal of Financial Economics*, Vol. 20, pp. 493-506.

Korajczyk, R. A. & Levy, A. (2003), 'Capital Structure Choice: Macroeconomic Conditions and Financial Constraints', *Journal Of Financial Economics*, Vol. 68, No. 1, pp. 75-109.

Korajczyk, R. A., Lucas, D. J., & McDonald, R. L. (1991), 'The Effect of Information Releases on the Price and Timing of Equity Issues', *Review of Financial Studies*, Vol. 4, No. 4, pp. 685-708.

Korajczyk, R. A., Lucas, D. J., & McDonald, R. L. (1992), 'Equity Issues with Time-Varying Asymmetric Information', *Journal of Financial and Quantitative Analysis*, Vol. 27, No. 3, pp. 397-417.

Korteweg, A. & Renneboog, L. (2003), 'The Choice between Rights-Preserving Issue Methods: Regulatory and Financial Aspects Issuing Seasoned Equity in the UK', ECGI - Finance Working Paper No. 15/2003. Available at SSRN: http://ssrn.com/abstract=389001.

Kothare, M. (1997), 'The Effects of Equity Issues on Ownership Structure and Stock Liquidity: A Comparison of Rights and Public Offerings', *Journal of Financial Economics*, Vol. 43, No. 1, pp. 131-48.

Kothari, S. P. & Warner, J. B. (1997), 'Measuring Long-horizon Security Price Performance', *Journal of Financial Economics*, Vol. 43, No. 3, pp. 301-339.

Kothari, S. P. & Warner, J. B. (2007), 'Econometrics of Event Studies', In *Handbook of Corporate Finance: Empirical Corporate Finance*, Vol. 1, Edited By B. E. Eckbo, pp. 3-32. North-Holland: Elsevier.

Krishnaswami, S. & Subramaniam, V. (1999), 'Information Asymmetry, Valuation, and the Corporate Spin-off Decision', *Journal of Financial*

Economics, Vol. 53, No. 1, pp. 73-112.

Lamont, O., Polk, C., & Saa-Requejo, J. (2001), 'Financial Constraints and Stock Returns', *The Review of Financial Studies*, Vol. 14, No. 2, pp. 529-554.

Lang, L. H. P. & Litzenberger, R. H. (1989), 'Dividend Announcements: Cash Flow Signalling Versus Free Cash Flow Hypothesis', *Journal of Financial Economics*, Vol. 24, No. 1, pp. 181-191.

Leary, M., & Roberts, M. (2010), 'The Pecking Order, Debt Capacity, and Information Asymmetry', *Journal of Financial Economics*, Vol. 95, No. 3, pp. 332-355.

Lee, D. S., Mikkelson, W. H. & Partch, M. M. (1992), 'Managers' Trading Around Stock Repurchases', *The Journal of Finance*, Vol. 47, No. 5, pp. 1947-1961.

Lee, G. & Masulis, R. W. (2009), 'Seasoned Equity Offerings: Quality of Accounting Information and Expected Flotation Costs', *Journal of Financial Economics*, Vol. 92, No. 3, pp. 443-469.

Lee, I. (1997), 'Do Firms Knowingly Sell Overvalued Equity?' *The Journal of Finance*, Vol. 52, No. 4, pp. 1439-1466.

Lee, I., Lochhead, S., Ritter, J., & Zhao, Q. (1996), 'The Cost of Raising Capital', *Journal of Financial Research*, Vol. 19, No. 1, pp. 59-74.

Lei, Z. & Zhang, C. (2015), 'Leveraged Buybacks', WBS Finance Group Research Paper. Available at SSRN: http://ssrn.com/abstract=2528478 or http://dx.doi.org/10.2139/ssrn.2528478.

Lemmon, M.L. & Zender, J.F. (2010), 'Debt Capacity and Tests of Capital Structure Theories', *Journal of Financial Quantitative Analysis*, Vol. 45, No. 5, pp. 1161-1187.

Levis, M. (1995), 'Seasoned Equity Offerings and the Short- and Long-Run Performance of Initial Public Offerings in the UK', *European Financial Management*, Vol. 1, No. 2, pp. 125-146.

Li, K. & McNally, W. (2003), 'The Decision to Repurchase, Announcement Returns and Insider Holdings: A Conditional Event Study', The ICFAI Journal of Applied Finance, Vol. 9, No. 6, pp. 55-70.

Liang, W., Chan, K., Lai, W., & Wang, Y. (2013) 'Motivation for Repurchases: A Life Cycle Explanation', *Journal of Financial Services Research*, Vol. 43, No. 2, pp. 221-242.

Lintner, J. (1956), 'Distribution of Incomes of Corporations Among Dividends, Retained Earnings, and Taxes', *The American Economic Review*, Vol. 46, No. 2, pp. 97-113.

Livingston, M., Naranjo, A., & Zhou, L. (2007), 'Asset Opaqueness and Split Bond Ratings', *Financial Management*, Vol. 36, No. 3, pp. 49-62.

Loncarski, I., ter Horst, J., & Veld, C. (2008), 'Why Do Companies Issue Convertible Bond Loans? An Empirical Analysis for the Canadian Market', *Canadian Journal of Administrative Sciences*, Vol. 25, No. 3, pp. 214-236.

Loughran, T. & Ritter, J. R. (1995), 'The New Issues Puzzle', *The Journal of Finance*, Vol. 50, No. 1, pp. 23-51.

Loughran, T. & Ritter, J. R. (1997), 'The Operating Performance of Firms Conducting Seasoned Equity Offerings', *The Journal of Finance*, Vol. 52, No. 5, pp. 1823-1850.

Louis, H. & White, H. (2007), 'Do Managers Intentionally Use Repurchase Tender Offers to Signal Private Information? Evidence from Firm Financial Reporting Behavior', *Journal of Financial Economics*, Vol. 85, No. 1, pp. 205-233.

Lyandres, E., Sun, L., & Zhang, L. (2008), 'The New Issues Puzzle: Testing the Investment-Based Explanation', *The Review of Financial Studies*, Vol. 21, No. 6, pp. 2825-2855.

Lyon, J. D., Barber, B. M., & Tsai, C. L. (1999), 'Improved Methods for Tests of Long-Run Abnormal Stock Returns', *The Journal of Finance*, Vol. 54, No. 1, pp. 165-201.

Mahajan, A. & Tartaroglu, S. (2008), 'Equity Market Timing and Capital Structure: International Evidence', *Journal of Banking and Finance*, Vol. 32, No. 5, pp. 754-766.

Marciukaityte, D., Szewczyk, S. H., & Varma, R. (2005), 'Investor Overoptimism and Private Equity Placements', *Journal of Financial Research*, Vol. 28, No. 4, pp. 591-608.

Markides, C. C. & Oyon, D. (1994), 'Stealing from Thy Neighbour: Leveraged Recapitalizations and Wealth Redistribution', *British Journal of Management*, Vol. 5, No. 2, pp. 139-152.

Marsh, P. (1982), 'The Choice between Equity and Debt: An Empirical Study', *The Journal of Finance*, Vol. 37, No. 1, pp. 121-144.

Martin, D. (1977). Early Warning of Bank Failure: A Logit Regression Approach', *Journal of Banking and Finance*, Vol. 1, No. 3, pp. 249-276.

Martin, K. J. (1996), 'The Method of Payment in Corporate Acquisitions, Investment Opportunities, and Management Ownership', *The Journal of Finance*, Vol. 51, No. 4, pp. 1227-1246.

Massa, M., Rehman, Z., & Vermaelen, T. (2007), 'Mimicking Repurchases', *Journal of Financial Economics*, Vol. 84, No. 3, pp. 624-666.

Maxwell, W. & Stephens, C. (2003), 'The Wealth Effects Of Repurchases On Bondholders', *The Journal of Finance*, Vol. 58, No. 2, pp. 895-920.

Mayers, D. (1998), 'Why Firms Issue Convertible Bonds: The Matching of Financial and Real Investment Options', *Journal of Financial Economics*, Vol. 47, No. 1, pp. 83-102.

McLaughlin, R., Safieddine, A., & Vasudevan, G. K. (1996), 'The Operating Performance of Seasoned Equity Issuers: Free Cash Flow and Post-Issue

Performance', Financial Management, Vol. 25, No. 4, pp. 41-53.

McLean, D. R. (2011), 'Share Issuance and Cash Savings', *Journal of Financial Economics*, Vol. 99, No. 3, pp. 693-715.

McLean, D. R., & Zhao, M. (2014), 'The Business Cycle, Investor Sentiment, and Costly External Finance', *The Journal of Finance*, Vol. 69, No. 3, pp. 1377-1409.

McNally, W. J. (1999), 'Multi-Dimensional Signaling with Fixed-Price Repurchases Offers', *Managerial and Decision Economics*, Vol. 20, No. 3, pp. 131-150.

McNally, W. J. (1999), 'Open Market Stock Repurchases Signalling', *Financial Management*, Vol. 28, No. 2, pp. 55-67.

McNichols, M. F. (2002), 'Discussion of the Quality of Accrual and Earnings: The Role of Accrual Estimation Errors', *The Accounting Review*, Vol. 77, No. s-1, pp. 61-69.

Mikkelson, W. H. & Ruback, R. S. (1991), 'Targeted Repurchases and Common Stock Returns', *The Rand Journal of Economics*, Vol. 22, No. 4, pp. 544-561.

Mikkelson, W.H. & Partch, M. M. (1986), 'Valuation Effects of Security Offerings and the Issuance Process', *Journal of Financial Economics*, Vol. 15, No. 1-2, pp. 31-60.

Milken, M. (2009), 'Why Capital Structure Matters: Companies That Repurchased Stock Two Years Ago Are in a World of Hurt', *The Wall Street Journal*, Accessed online at: http://www.wsj.com/articles/SB124027187331937083.

Minnick, C. & Zhao, M. (2007), 'Levered Repurchases: Who Benefits?', FMA Annual Meeting, Orlando, FL.

Modigliani, F. & Miller, M. H. (1958), 'The Cost of Capital, Corporation Finance, and the Theory of Investment," *The American Economic Review*, Vol. 48, No. 3, pp. 261-297.

Myers, S. C. (1984), 'The Capital Structure Puzzle', *The Journal of Finance*, Vol. 39, No. 3, pp. 574-592.

Myers, S. C. (2001), 'Capital Structure', *The Journal of Economic Perspectives*, Vol. 15, No. 2, pp. 81-102.

Myers, S. C. & Majluf, N. S. (1984), 'Corporate Financing and Investment Decisions When Firms Have Information That Investors Do Not Have', *Journal of Financial Economics*, Vol. 13, No. 2, pp. 187-221.

Naeem, S., Danbolt, J., & Siganos, A. (2014), 'Credit Ratings and Capital Structure: A Non-Linear Relation, Unpublished Results.

Naes, R., Skjeltorp, J. A., & Odegaard, B. A. (2011), 'Stock Market Liquidity and the Business Cycle', *The Journal of Finance*, Vol. 66, No. 1, pp. 139-176.

Ngatuni, P., Capstaff, J., & Marshall, A. (2007), 'Long-Term Performance Following Rights Issues and Open Offers in the UK', *Journal of Business Finance and Accounting*, Vol. 34, No. 1-2, pp. 33-64.

Nohel, T. & Tarhan, V. (1998), 'Share Repurchases and Firm Performance: New Evidence on the Agency Costs of Free Cash Flow', *Journal of Financial Economics*, Vol. 49, No. 2, pp. 187-222.

Oded, J. (2011), 'Stock Repurchases: How Firms Choose between A Self-Tender Offer and an Open Market Program', *Journal of Banking and Finance*, Vol. 35, No. 12, pp. 3174-3187.

Office for National Statistics, Ownership of UK Quoted Shares In 2010: Statistical Bulletin 2012; Accessed at: http://www.ons.gov.uk/ons/dcp171778_257476.pdf.

Opler, T. C. & Titman, S. (1993), 'The Determinants of Leveraged Buyout Activity: Free Cash Flows Vs. Financial Distress Costs', *The Journal of Finance*, Vol. 48, No. 5, pp. 1985-1999.

Opler, T. C. & Titman, S. (1994), 'Financial Distress and Corporate Performance', *The Journal of Finance*, Vol. 49, No. 3, pp. 1015-1040.

Opler, T., Pinkowitz, L., Stulz, R., & Williamson, R. (1999), 'The Determinants and Implications of Corporate Cash Holdings', *Journal of Financial Economics*, Vol. 52, No. 1, pp. 3-46.

Oswald, D. & Young, S. (2004), 'What Role Taxes and Regulation? A Second Look at Open Market Share Buyback Activity in the UK', *Journal of Business Finance and Accounting*, Vol. 31, No. 1-2, pp. 257-292.

Oswald, D. & Young, S. (2008), 'Share Reacquisitions, Surplus Cash and Agency Problems', *Journal of Banking and Finance*, Vol. 32, No. 5, pp. 795-806.

Oyer, P. & Schaefer, S. (2005), 'Why Do Some Firms Give Stock Options to All Employees? An Empirical Examination of Alternative Theories', *Journal of Financial Economics*, Vol. 76, No. 1, pp. 99-133.

Ozkan, A. (2001), 'Determinants of Capital Structure and Adjustments to Long Run Target: Evidence from UK Company Panel Data', *Journal of Business Finance and Accounting*, Vol. 28, No. 1-23, pp. 175-198.

Pastor, L. & Stambaugh, R.F. (2003), 'Liquidity Risk and Expected Stock Returns', *Journal of Political Economy*, Vol. 111, No. 3, pp. 642-685.

Pastro-Llorca, M. J. & Martin-Ugedo, J. F. (2004), 'Long-Run Performance of Spanish Seasoned Equity Issues with Rights', *International Review of Financial Analysis*, Vol. 13, No. 2, pp. 191-215.

Perez-Quiros, G. & Timmermann, A. (2000), 'Firm Size and Cyclical Variations in Stock Returns', *The Journal of Finance*, Vol. 53, No. 3, pp. 1229-1262.

Persons, J. C. (2007), 'Heterogeneous Shareholders and Signaling with Share Repurchases', *Journal of Corporate Finance*, Vol. 3, No. 3, pp. 221-249.

Peyer, U. C. & Vermaelen, T. (2005), 'The Many Facets of Privately Negotiated Stock Repurchase', *Journal of Financial Economics*, Vol. 75, No. 2, pp. 361-395.

Pilotte, E. (1992), 'Growth Opportunities and the Stock Price Response to New Financing', *Journal of Business*, Vol. 65, No. 3, pp. 371-394.

Polk, C. & Sapienza, P. (2004), 'The Real Effects of Investor Sentiment', Working Paper No. 10563, National Bureau of Economic Research.

Pontiff, J. (2006), 'Costly Arbitrage and the Myth of Idiosyncratic Risk', *Journal of Accounting and Economics*, Vol. 42, No. 1&2, pp. 35-52.

Pontiff, J. & Irvine, P. (2009), 'Idiosyncratic Return Volatility, Cash Flows, and Product Market Competition', *Review of Financial Studies*, Vol. 22, No. 3, pp. 1149-1177.

Purnanandam, A. K. (2008), 'Financial Distress and Corporate Risk Management: Theory & Evidence', *Journal of Financial Economics*, Vol. 87, No. 3, pp. 706-739.

Rajan, R. & Servaes, H. (1997), 'Analyst Following of Initial Public Offerings', *The Journal of Finance*, Vol. 52, No. 2, pp. 507-529.

Rajan, R. G. & Zingales, L. (1995), 'What Do We Know About Capital Structure? Some Evidence from International Data', *Journal of Finance*, Vol. 50, No. 5, pp. 1421-1460.

Rajan, R. G. & Zingales, L. (1998), 'Financial Dependence and Growth', *American Economic Review*, Vol. 88, No. 3, pp. 559-586.

Rangan, S. (1998), 'Earnings Management and the Performance of Seasoned Equity Offerings', *Journal of Financial Economics*, Vol. 50, No. 1, pp. 101-122.

Rau, R. P. & Vermaelen, T. (2002), 'Regulation, Taxes and Share Repurchase in the United Kingdom', *The Journal of Business*, Vol. 75, No. 2, pp. 245-282.

Rhodes-Kropf, M., Robinson, D., & Viswanathan, S. (2005), 'Merger Waves and Merger Activity: The Empirical Evidence', *Journal of Financial Economics*, Vol. 77, No. 3, pp. 561-603.

Ritter, J. R. (1984), 'The "Hot Issue" Market of 1980', *The Journal of Business*, Vol. 57, No. 2, pp. 215-240.

Ritter, J. R. (1991), 'The Long-Run Performance of Initial Public Offerings', *The Journal of Finance*, Vol. 46, No. 1, pp. 3-27.

Sadka, R. & Scherbina, A. (2007), 'Analyst Disagreement, Mispricing, and Liquidity', *The Journal of Finance*, Vol. 62, No. 5, pp. 2367-2404.

Scholes, M. S. (1972), 'The Market for Securities: Substitution versus Price Pressure and the Effects of Information on Share Prices', *The Journal of Business*, Vol. 45, No. 2, pp. 179-211.

Schultz, P. (2003), 'Pseudo Market Timing and the Long-Run Underperformance Of IPOs', *The Journal of Finance*, Vol. 58, No. 2, pp. 483-517.

Scott A. J. & Wild, C. J. (1991), 'Fitting Logistic Models in Stratified Case-Control Studies', *Biometrics*, Vol. 47, pp. 497-510.

Scott, A. J. & Wild, C. J. (1986), 'Fitting Logistic Models Under Case-Control or Choice Based Sampling', *Journal of the Royal Statistical Society, Series B* (*Methodological*), Vol. 48, No. 2, pp. 170-182.

Share buy-backs: The Repurchase Revolution (September 13, 2014). *The Economist*. Retrieved from: http://www.economist.com/news/business/21616968-companies-have-been-gobbling-up-their-own-shares-exceptional-rate-there-are-good-reasons.

Shivakumar, L. (2000), 'Do Firms Mislead Investors by Overstating Earnings Before Seasoned Equity Offerings?' *Journal of Accounting and Economics*,

Vol. 29, No. 3, pp. 339-371.

Shyam-Sunder, L. & Myers, S. C. (1999), 'Testing Static Tradeoff Against Pecking Order Models of Capital Structure', *Journal of Financial Economics*, Vol. 51, No. 2, pp. 219-244.

Siddiqi, M. A. (2009), 'Investigating The Effectiveness of Convertible Bonds in Reducing Agency Costs: A Monte-Carlo Approach', *The Quarterly Review of Economics and Finance*, Vol. 49, No. 4, pp. 1360-1370.

Silvapulle, M. J. (1981), 'On the Existence of Maximum Likelihood Estimators for the Binomial Response Models', *Journal of the Royal Statistical Society*, *Series B (Methodological)*, Vol. 43, No. 3, pp. 310-313.

Sinha, S. (1991), 'Share Repurchase as A Takeover Defense', *Journal Of Financial And Quantitative Analysis*, Vol. 26, No. 2, Pp. 233-244.

Skinner, D. J. & Sloan, R. G. (2002), 'Earnings Surprises, Growth Expectations, and Stock Returns or Don't Let an Earnings Torpedo Sink Your Portfolio', *Review of Accounting Studies*, Vol. 7, No. 2&3, pp. 289-312.

Slovin, M. B., Sushka, M. E., & Lai, K. W. L. (2000), 'Alternative Flotation Methods, Adverse Selection, and Ownership Structure: Evidence from Seasoned Equity Issuance in the U.K.', *Journal of Financial Economics*, Vol. 57, No. 2, pp. 157-190.

Smith, C.W. (1977), 'Alternative Methods for Raising Capital - Rights Versus Underwritten Offerings', *Journal of Financial Economics*, Vol. 5, No. 3, pp. 273-307.

Spiess, K.D. & Affleck-Graves, J. (1995), 'Underperformance in Long Run Stock Returns Following Seasoned Equity Offerings', *Journal of Financial Economics*, Vol. 38, No. 3, pp. 243-67.

Stein, J. (1992), 'Convertible Bonds As Back Door Equity Financing', *Journal of Financial Economics*, Vol. 32, No. 2, pp. 3-21.

Stein, J. (1996), 'Rational Capital Budgeting in An Irrational World', *The Journal of Business*, Vol. 69, No. 4, pp. 429-455.

Stephens, C. P. & Weisbach, S. M. (1998), 'Actual Share Reacquisitions in Open Market Repurchase Operations', *The Journal of Finance*, Vol. 53, No. 1, pp. 313-333.

Stulz, R. M. (1988), 'Managerial Control of Voting Rights: Financing Policies and the Market for Corporate Control', *Journal of Financial Economics*, Vol. 20, pp. 25-54.

Taffler, R. J. (1982), 'Forecasting Company Failure in the UK Using Discriminant Analysis and Financial Ratio Data', *Journal of the Royal Statistical Society*, Vol. 145, No. 3, pp. 342-358.

Taffler, R. J. (1983), 'The Assessment of Company Solvency and Performance Using a Statistical Model: A Comparative UK-Based Study', *Accounting and Business Research*, Vol. 15, No. 52, pp. 295-307.

Taggart, R. A. (1977), 'A Model of Corporate Financing Decisions', *The Journal of Finance*, Vol. 32, No. 5, pp. 1467-1484.

Tennant, C. (1977), 'The General Health Questionnaire: A Valid Index of Psychological Impairment in Australian Populations', *Medical Journal of Australia*, Vol. 2, No. 12, pp. 392-394.

Teoh, S. H., Welch, I. & Wong, T. J. (1998), 'Earnings Management and Long-Run Market Performance of Initial Public Offerings', *The Journal of Finance*, Vol. 53, No. 6, pp. 1935-1974.

Teoh, S. H., Welch, I., & Wong, J. T. (1998), 'Earnings Management and the Underperformance of Seasoned Equity Offerings', *Journal of Financial Economics*, Vol. 50, No. 1, pp. 63-99.

Thorbecke, W. (1997), 'On Stock Market Returns and Monetary Policy', *The Journal of Finance*, Vol. 52, No. 2, pp. 635-654.

Titman, S., Li, K. J., & Xie, F. (2004), 'Capital Investments and Stock Returns', *Journal of Financial and Quantitative Analysis*, Vol. 39, No. 4, pp. 677-700.

Tsangarakis N. V. (1996), 'Equity Rights Issues: Signalling vs Issue Price Irrelevance Hypothesis', *European Financial Management*, Vol. 2, No. 3, pp. 299-310.

U.S. Securities and Exchange Commission (2004), 'Division of Market Regulation: Answers to Frequently Asked Questions Concerning Rule 10b-18("Safe Harbor" For Issuer Repurchases)', Accessed at: http://www.sec.gov/divisions/marketreg/r10b18faq0504.htm.

Ursel, N. (2006), 'Rights Offerings and Corporate Financial Condition', *Financial Management*, Vol. 35, No. 1, pp. 31-52.

Veld, C. & Veld-Merkoulova, Y. (2004), 'Do Spin-Offs Really Create Value? The European Case', *Journal of Banking and Finance*, Vol. 28, No. 5, pp. 1111-1135.

Vermaelen, T. (1981), 'Common Stock Repurchases and Market Signalling: An Empirical Study', *Journal of Financial Economics*, Vol. 9, No. 2, pp. 139-183.

Vermaelen, T. (1984), 'Repurchase Tender Offers, Signaling, And Managerial Incentives', *Journal of Financial and Quantitative Analysis*, Vol. 19, No. 2, pp. 163-181.

Vermaelen, T. (2005), 'Share Repurchases', Foundations and Trends in Finance, Vol.1, No. 3, pp. 171-268.

Voss, J. (2012), 'Why Do Firms Repurchase Stock?', *Major Themes in Economics*, pp. 55-75.

Wang, C., Strong, N., Tung, S., & Lin, S. (2009), 'Share Repurchases, the Clustering Problem, and the Free Cash Flow Hypothesis', *Financial Management*, Vol. 38, No. 3, pp. 487-505.

Weisbenner, S. J. (2000), 'Corporate Share Repurchase in the 1990s: What Role Do Stock Options Play?', Working Paper, University of Illinois At Urbana Champaign.

Weston, J. F. & Siu, J. A. (2003), 'Changing Motives for Share Repurchases', Working Paper, Anderson Graduate School of Management, UCLA.

White, H. (1980), 'A Heteroskedasticity-Consistent Covariance Matrix Estimator and a Direct Test for Heteroskedasticity', *Econometrica*, Vol. 48, No. 4, pp. 817-838.

Whited, T. & Wu, G. (2006), 'Financial Constraints Risk', *Review of Financial Studies*, Vol. 19, No.2, pp. 531-559.

Womack, K. L. (1996). Do brokerage analysts' recommendations have investment value? *The Journal of Finance*, Vol. 51, No. 1, pp. 137-167.

Wooldridge, J. M. (2009), 'Introductory Econometrics: A Modern Approach', 4th ed. South-Western College Publishing.

Wooldridge, J. M. (2010), 'Econometric Analysis of Cross Section and Panel Data', Cambridge, Massachusetts The MIT Press.

Wruck, K. H. (1989), 'Equity Ownership Concentration and Firm Value: Evidence from Private Equity Financings', *Journal of Financial Economics*, Vol. 23, No. 1, pp. 3-28.

Wruck, K. H. & Wu, Y. (2009), 'Relationships, Corporate Governance and Performance: Evidence from Private Placements of Common Stock', *Journal of Corporate Finance*, Vol. 15, No. 1, pp. 30-47.

Wu, Y. L., Wang, X., & Yao, J. (2005), 'Understanding the Positive Announcement Effects of Private Equity Placements: New Insights from Hong Kong Data', *Review of Finance*, Vol. 9, No. 4, pp. 385-414.

Zeidler, F., Mietzner, M., & Schiereck, D. (2012), 'Risk Dynamics Surrounding the Issuance of Convertible Bonds', *Journal of Corporate Finance*, Vol. 18, No. 2, pp. 273-290.

Zwiebel, J. (1996), 'Dynamic Capital Structure under Managerial Entrenchment', *American Economic Review*, Vol. 86, No. 5, pp. 1197-1215.